Hesser G. McBride, Jr. Associate General Regulatory Counsel Law Department **PSEG Services Corporation**

80 Park Plaza - T5, Newark, New Jersey 07102-4194

973-430-5333 fax: 973-430-5983

email: hesser.mcbride@pseg.com



VIA ELECTRONIC MAIL & OVERNIGHT MAIL

Fart July 2, 2018

RECEIVED CASE MANAGEMENT

June 29, 2018

JUL 0 2 2018

In the Matter of the Provision of Basic Generation Service for the Period Beginning June 1, 2019

BOARD OF PUBLIC UTILITIES TRENTON, NJ

Docket No. ER18040356

BOARD OF PUBLIC UTILITIES TRENTON, NJ

JUL 02 2018

Aida Camacho-Welch, Secretary New Jersey Board of Public Utilities 44 South Clinton Avenue, 3rd Floor, Suite 314 Trenton, New Jersey 08625-0350

Dear Secretary Camacho-Welch:

Enclosed is an original and ten copies for submission to the Board of Public Utilities (Board) of the generic Proposal for the Basic Generation Service (BGS) of the New Jersey Electric Distribution Companies (EDCs): Public Service Electric and Gas Company (PSE&G); Jersey Central Power & Light Company (JCP&L); Atlantic City Electric Company (ACE); and, Rockland Electric Company (RECO). This filing is made in compliance with the Procedural Schedule in the above-captioned docket approved in a Board Order dated April 25, 2018.

This generic Proposal for Basic Generation Service Effective June 1, 2019 briefly describes the actual auction process, including the pre-qualification of bidders, setting of starting prices, BGS pricing and rate design methodology, and roles of the EDCs and the Board in the process. Also included are proposed Supplier Master Agreements, which define the duties and responsibilities of BGS suppliers and provide assurance that the full requirements load will be reliably delivered to the EDCs' distribution systems.

In addition, each EDC will submit under separate cover a Company Specific Addendum that will supply specific company proposals for committed supply, contingency plans, BGS accounting methodology, cost recovery, details on BGS pricing and tariff details, and other matters. PSE&G's Addendum is enclosed herein.

Respectfully submitted,

Hose D. Misself.

Encl.

C Service List (Electronic)

a Mercon

Chinings, ; Bolin Tourniche (Lorie

Section of the

A HERBERT, COLOUR, ROCKERS, A AND WOTWERF

Public Service Electric and Gas Company BGS 2019 ER18040356

ACE

Susan DeVito
Pepco Holdings, LLC - 92DC56
500 N, Wakefield Drive
P.O. Box 6066
Newark DE 19714-6066
(302) 451-5324
susan.devito@pepcoholdings.com

ACE

Philip J. Passanante Esq.
Atlantic City Electric Company - 92DC42
500 N. Wakefield Drive
P.O. Box 6066
Newark DE 19714-6066
(609) 909-7034
philip.passanante@pepcoholdings.com

<u>BPU</u>

Mark Beyer
Board of Public Utilities
44 South Clinton Avenue,
3rd Floor, Suite 314
P.O. Box 350
Trenton NJ 08625-0350
(609) 292-2637
mark.beyer@bpu.ni.gov

<u>BPU</u>

Stacy Peterson
Board of Public Utilities
44 South Clinton Avenue
3rd Floor, Suite 314
P.O. Box 350
Trenton NJ 08625-0350
(609) 292-4517
stacy.peterson@bpu.nj.goy

BPU Consultants

Craig R. Roach
Bates White, LLC
1300 Eye St NW, Suite 600
Washington DC 20005
craig.roach@bateswhite.com

JCP&L

Kevin Connelly
Jersey Central Power & Light Co.
300 Madison Avenue
P.O. Box 1911
Morristown NJ 07962
(973) 401-8708
kconnelly@firstenergycorp.com

ACE

Thomas M. Hahn
Pepco Holdings, LLC - 63ML38
5100 Harding Highway
Mays Landing NJ 08330-2239
(609) 625-5853
thomas.hahn@pepcoholdings.com

ACE

Alison L. Regan
Pepco Holdings, LLC - 92DC56
500 N. Wakefield Drive
P.O. Box 6066
Newark DE 19714-6066
(302) 451-5280
alison.regan@pepcoholdings.com

BPU

Aida Camacho-Welch Board of Public Utilities 44 South Clinton Avenue, 3rd Floor, Suite 314 Post Office Box 350 Trenton NJ 08625-0350 aida.camacho@bpu.ni.gov

BPU

Bethany Rocque-Romaine Esq. Board of Public Utilities 44 South Clinton Avenue 3rd Floor, Suite 314 P.O. Box 350 Trenton NJ 08625-0350 (609) 292-1496 bethany.romaine@bpu.nj.goy

DAG

Andrew Kuntz Esq.
NJ Department of Law & Public Safety
124 Halsey Street
Post Office Box 45029
Newark NJ 07101
Andrew.kuntz@lps.state.nj.us

JCP&L

Gregory Eisenstark Esq.
Windels Marx Lane & Mittendorf, LLP
120 Albany Street Plaza
New Brunswick NJ 08901
732.448.2537
geisenstark@windelsmarx.com

ACE

Joseph F. Janocha
PEPCO Holdings, LLC - 92DC56
500 N. Wakefield Drive
P.O. Box 6066
Newark DE 19714-6066
(302) 451-5326
joseph.janocha@pepcoholdings.com

ACE

Daniel A. Tudor
Pepco Holdings, LLC - EP6412
701 Ninth Street NW
Washington DC 20001
(202) 872-2090
datudor@pepco.com

BPU

Richard DeRose
Board of Public Utilities
Post Office Box 350
44 South Clinton Ave., 3rd Floor
Trenton NJ 08625-0350
Richard.derose@bpu.nj.gov

BPU Consultants

Frank Mossburg
Bates White, LLC
1300 Eye St NW, Suite 600
Washington DC 20005
(202) 652-2194
frank.mossburg@bateswhite.com

JCP&L

Sally J. Cheong
Jersey Central Power & Light Co.
300 Madison Avenue
P.O. Box 1911
Morristown NJ 07962-1911
(973) 401-8699
scheong@firstenergycorp.com

NERA

Paul Cardona
NERA Economic Consulting
777 S. Figueroa, Suite 1950
Los Angeles CA 90017
Paul.Cardona@NERA.com

			ť	1
		,		

Public Service Electric and Gas Company BGS 2019 ER18040356

NERA

Chantale LaCasse NERA 1166 Avenue of the Americas New York NY 10036 (212) 345-9313

chantale.lacasse@nera.com

NJLEUC

Steven S. Goldenberg Esq. Fox Rothschild LLP 997 Lenox Drive, Bldg. 3 Lawrenceville NJ 08648-2311 (609) 896-4586 sgoldenberg@foxrothschild.com

Other Parties

Sean Gallagher
Solar Energy Industries Assoc.
575 7th Street, NW
Suite 400
Washington DC 20005
sgallagher@seja.org

Other Parties

Holly Minogue Gabel Associates 417 Denison Street Highland Park NJ 08904 holly.minogue@gabelassociates.com

PSE&G

Hesser G. McBride, Jr. Esq. PSEG Service Corporation 80 Park Plaza, T5G P.O. Box 570 Newark NJ 07102 (973) 430-5333 hesser.mcbride@pseg.com

Rate Counsel

James Glassen
Division of Rate Counsel
140 East Front Street, 4th Flr.
P.O. Box 003
Trenton NJ 08625
(609) 984-1460
jglassen@rpa.state.nj.us

NERA

Eugene Meehan NERA 1255 23rd Street Suite 600 Washington DC 20037 (202) 466-9287 eugene.meehan@nera.com

Other Parties

Sara Bluhm NJBIA 102 West State Street Trenton NJ 08608-1199 (609) 393-7707 sbluhm@njbia.org

Other Parties

John Holub NJ Retail Merchants Assoc. 332 West State Street Trenton NJ 08618 john@njrma.org

Other Parties

Lyle Rawlings
Mid-Atlantic Solar Energy Industries Assoc.
Rutgers EcoComplex
Suite 208-B
1200 Florence-Columbus Rd.
Bordentown NJ 08505
lyle@adyancedsolarproducts.com

PSE&G

Terrence J. Moran
Public Service Electric & Gas Co.
80 Park Plaza, T-13
Newark NJ 07101
terrence.moran@pseg.com

Rate Counsel

Lisa Gurkas
Division of Rate Counsel
140 East Front Street, 4th Flr.
P.O. Box 003
Trenton NJ 08625
(609) 984-1460
lgurkas@rpa.state.nj.us

NJLEUC

Paul F. Forshay Esq.
Eversheds-Sutherland, LLP
700 Sixth Street, NW, Suite 700
Washington DC 20001-3980
(202) 383-0100
paul.forshay@eyersheds-sutherland.com

Other Parties

Bruce H. Burcat Esq.
Mid-Atlantic Renewable Energy Coalition
208 Stonegate Way
Camden DE 19934
bburcat@marec.us

Other Parties

Robert Macksoud CEP
EnergySolve
One Executive Drive
Suite 401
Somerset NJ 08873
rmacksoud@energysolve.com

Other Parties

Larry Spielvogel PE L. G. Spielvogel, Inc. 190 Presidential Blvd#310 Bala Cynwyd PA 19004-1151 spielvogel@comcast.net

Rate Counsel

Stefanie A. Brand Division of Rate Counsel 140 East Front Street, 4th Flr. P.O. Box 003 Trenton NJ 08625 (609) 984-1460 sbrand@rpa.state.nj.us

Rate Counsel

Brian O. Lipman
Division of Rate Counsel
140 East Front Street, 4th Flr.
P.O. Box 003
Trenton NJ 08625
(609) 984-1460
blipman@rpa.nj.goy

. f

Public Service Electric and Gas Company BGS 2019 ER18040356

Rate Counsel

Ami Morita
Division of Rate Counsel
140 East Front Street, 4th Flr.
P.O. Box 003
Trenton NJ 08625
(609) 984-1460
amorita@rpa.state.nj.us

Rate Counsel Consultant

Robert Fagan Synapse Energy Economics, Inc. 485 Massachusetts Ave., Suite 2 Cambridge MA 02139 (617) 661-3248 rfagan@synapse-energy.com

RECO

Margaret Comes
Consolidated Edison Co. of NY
Law Dept
4 Irving Place
New York NY 10003
(212) 460-3013
comesm@coned.com

Suppliers

Matthew Davies
TransCanada Power Marketing Ltd.
110 Tumpike Road, Suite 300
Westborough MA 01581
matthew_davies@transcanada.com

Suppliers

Divesh Gupta Esq.
Exelon Business Services Corp.
111 Market Place
Suite 1200C
Baltimore MD 21202
divesh.gupta@constellation.com

Suppliers

Thomas Hoatson LS Power Development, LLC 2 Tower Center East Brunswick NJ 08816 (732) 867-5911 thoatson@lspower.com Rate Counsel

Diane Schulze Esq.
Division of Rate Counsel
140 East Front Street, 4th Flr.
P.O. Box 003
Trenton NJ 08625
(609) 984-1460
dschulze@rpa.state.nj.us

RECO

William A. Atzl Jr.
Rockland Electric Company
4 Irving Place
Room 515-S
New York NY 10003
(212) 460-3308
atzlw@coned.com

RECO

James C. Meyer Esq.
Riker, Danzig, Scherer, Hyland & Perretti
Headquarters Plaza
One Speedwell Avenue
Morristown NJ 07962
(973) 538-0800
jmeyer@riker.com

Suppliers

Raymond Depillo
PSEG Energy Resources & Trade
80 Park Plaza
P.O. Box 570
Newark NJ 07101
(973) 430-8866
raymond.depillo@pseg.com

Suppliers

Mark Haskell Cadwalader, Wickersham & Taft LLP 700 Sixth Street, N.W. Washington DC 20001 mark.haskell@cwt.com

Suppliers

Don Hubschman American Electric Power 155 W. Nationwide Blvd. Columbus OH 43215 (614) 583-7019 dmhubschman@aepes.com Rate Counsel Consultant

Max Chang
Synapse Energy Economics, Inc.
485 Massachusetts Ave., Suite 2
Cambrige MA 02139
(617) 661-3248
mchang@synapse-energy.com

RECO

John L. Carley Esq.
Consolidated Edison Co. of NY
Law Dept.
4 Irving Place, Room 1815-S
New York NY 10003
(212) 460-2097
carleyi@coned.com

Suppliers

Craig S. Blume
UGI Energy Services
One Meridian Boulevard
Suite 2C01
Wyomissing PA 19610
cblume@ugies.com

Suppliers

Steve Gabel - IEPNJ
Gabel Associates
417 Denison Street
Highland Park NJ 08904
steven@gabelassociates.com

Suppliers

Marcia Hissong
DTE Energy Trading, Inc.
414 South Main Street
Suite 200
Ann Arbor MI 48104
(734) 887-2042
hissongm@dteenergy.com

Suppliers

Adam Kaufman
Independent Energy Producers of NJ
Five Vaughn Drive
Suite 101
Princeton NJ 08540
akaufman@kzgrp.com

		• ,
	•	

Public Service Electric and Gas Company BGS 2019 ER18040356

Suppliers

James Laskey - BGS Esq. Norris McLaughlin & Marcus 721 Route 202-206, Suite 200 Bridgewater NJ 08807 (908) 252-4221 jlaskey@nmmlaw.com

Suppliers

Ira G. Megdal-BGS Esq. Cozen O'Connor 457 Haddonfield Rd. Suite 300 Cherry Hill NJ 08002 (856) 910-5019 IMEGDAL@COZEN.COM

Suppliers

Anthony Pietranico
ConEdison Solutions Inc.
(732) 741-5822 X204
pietranicoa@conedsolutions.com

Suppliers

Glen Thomas
The P3 Group
GT Power Group LLC
1060 First Avenue
Suite 400
King of Prussia PA 19406
Gthomas@gtpowergroup.com

Suppliers

Aundrea Williams
NextEra Power Marketing LLC
700 Universe Boulevard
Juno Beach FL 33408
Aundrea.williams@nexteraenergyservices.com

Third Party Suppliers

David Gill
NextEra Energy Resources, LLC
700 Universe Boulevard
Juno Beach FL 33408
david.gill@extraenergy.com

Third Party Suppliers

Stacey Rantala
National Energy Marketers Association
3333 K Street, N.W.
Suite 110
Washington DC 20007
srantala@energymarketers.com

Suppliers

Shawn P. Leyden (BGS/CB) PSEG Services Corporation 80 Park Plaza P. O. Box 570 Newark NJ 07101 (973) 430-7698 shawn.leyden@pseg.com

Suppliers

Becky Merola Noble Americas Energy Solutions, LLC 5325 Sheffield Avenue Powell OH 43065 bmerola@noblesolutions.com

Suppliers

David K. Richter Esq.
PSEG Services Corporation
80 Park Plaza, T5
P.O. Box 570
Newark NJ 07102
(973) 430-6451
david.richter@pseg.com

Suppliers

Howard O. Thompson - BGS Esq. Russo Tumulty Nester Thompson & Kelly, LLP 240 Cedar Knolls Road Suite 306 Cedar Knolls NJ 07927 (973) 993-4477 https://doi.org/10.1001/pdf.

Third Party Suppliers

David B. Applebaum
NextEra Energy Resources, LLC
21 Pardee Place
Ewing NJ 08628
david.applebaum@nexteraenergy.com

Third Party Suppliers

Marc A. Hanks
Direct Energy Services LLC
Government & Regulatory Affairs
marc,hanks@directenergy.com

Third Party Suppliers

Dana Swieson
EPEX
717 Constitutional Drive
Suite 110
Exton PA 19341
dana.swieson@epex.com

Suppliers

Christine McGarvey AEP Energy Partners, Inc. 155 W Nationwide Blvd. Suite 500 Columbus OH 43215

clmcgarvey@aepes.com

Suppliers

Christi L. Nicolay
Macquarie Energy LLC
500 Dallas St., Level 31
Houston TX 77002
christi.nicolay@macquarie.com

Suppliers

Glenn Riepl
AEP Energy Services
1 Riverside Plaza
14th Floor
Columbus OH 43215-2373
(614) 324-4502
gfriepl@aep.com

Suppliers

Sharon Weber
PPL Energy Plus
2 North 9th Street TW 20
Allentown PA 18101
sjweber@pplweb.com

Third Party Suppliers

Murray E. Bevan Esq.
Bevan, Mosca, Giuditta & Zarillo, P.C.
222 Mount Airy Road, Suite 200
Basking Ridge NJ 07920
(908) 753-8300
mbevan@bmgzlaw.com

Third Party Suppliers

Kathleen Maher
Constellation NewEnergy
810 Seventh Avenue
New York NY 10019-5818
kathleen.maher@constellation.com

				w 21 3	
		•			
			·		

CASE MANAGEMENT

JUL 02 2018

BOARD OF PUBLIC UTILITIES TRENTON, NJ

IN THE MATTER OF THE PROVISION OF BASIC GENERATION SERVICE FOR THE PERIOD BEGINNING JUNE 1, 2019

: Docket No. ER18040356

Public Service Electric and Gas Company, Jersey Central Power & Light Company, Atlantic City Electric Company, and Rockland Electric Company

PROPOSAL FOR BASIC GENERATION SERVICE REQUIREMENTS TO BE PROCURED EFFECTIVE JUNE 1, 2019

June 29, 2018

TABLE of CONTENTS

I.	INTRODUCTION	1
II.	GOALS OF THE BGS PROCUREMENT PROCESS	2
III.	THE ESTABLISHED BGS AUCTION PROCESS IS UNIQUELY SUITED TO THE GOALS O	
	I. A. Product Definition	
	I. B. Auction Format	
	I. C. Competitive Safeguards	
	I. D. Qualification Procedures	
II	I. E. Rate Design	
II	I.F. Roles	8
IV.	OVERVIEW OF THE EDCs' PROPOSAL FOR THE 2019 AUCTIONS	10
	V. A. BGS – Residential Small Commercial Pricing ("BGS-RSCP")	
	V. B. BGS – Commercial and Industrial Energy Pricing ("BGS-CIEP")	
	V. C. BGS-RSCP and BGS-CIEP	
V.	CONDUCT OF THE AUCTIONS	16
	. A. The Roles of the Board, the Board Advisor, the EDCs and the Auction	10
•	Manager	16
	V. A. 1. The Role of the Board and the Board Advisor	16
	V. A. 2. The Role of the EDCs	
	V. A. 3. The Role of the Auction Manager	
V	. B. Confidentiality of Auction Information	
V	. C. Regulatory Milestones and Tentative Auction Timeline	
VI.	BOARD APPROVAL AND EXECUTION OF BGS SUPPLIER MASTER AGREEMENTS	23
VII.	LIST OF CHANGES	24
VIII.	APPENDIX A	26
IX.	APPENDIX B	103
X.	APPENDIX C	184
VI	A DDENINIY D	281

I. INTRODUCTION

The New Jersey electric distribution companies ("EDCs") – Public Service Electric and Gas Company ("PSE&G"), Jersey Central Power & Light Company ("JCP&L"), Atlantic City Electric Company ("ACE"), and Rockland Electric Company ("RECO") – first filed a joint proposal for the procurement of electric power for basic generation service ("BGS") customers on July 1, 2001. On December 10, 2001, after allowing various stakeholders to comment and holding public hearings, the New Jersey Board of Public Utilities ("BPU" or "Board") approved the EDCs' joint proposal to implement a statewide simultaneous descending clock auction ("clock auction"). The first BGS clock auction was held in February 2002. In the ensuing years, the EDCs have jointly filed an annual auction proposal to procure BGS Supply, which has been consistently approved by the Board and successfully implemented through the combined efforts of the Board, its Staff, the Auction Manager, the Board's Advisor, and the EDCs. Since 2002, the annual energy auctions created and administered by the New Jersey BPU have been used to efficiently meet the demand of BGS customers.

After the first five years of BGS Auction experience, the Board conducted a review of the BGS Auction process, which led to some further improvements in the process, including the ability for bidders to comment on the alternate guaranty process and application forms. The Board conducted another comprehensive review following the 2012 BGS Auctions in Docket No. ER12020150. The Board solicited comments from all parties with a focus on the procurement process and policy issues directly related thereto. The Board Order in Docket No. ER18040356 instructed the EDCs to submit a proposal to procure BGS Supply beginning June 1, 2019. Accordingly, the EDCs hereby submit the instant proposal, which proposes to continue to conduct a statewide clock auction to procure power for BGS customers.

II. GOALS OF THE BGS PROCUREMENT PROCESS

In the following sections of this filing, the EDCs review why the BGS Auction Process continues to best meet the interests of New Jersey customers. The starting point of the explanation is a review of the goals of the BGS procurement process. In the Electric Discount and Energy Competition Act of 1999 ("EDECA"), the Legislature found and declared that it is the policy of the State, consistent with other important policy objectives, to rely upon competitive markets, where such markets exist, to deliver energy services to consumers (N.J.S.A. 48:3-50a(2)). To accomplish these policy objectives, EDECA directed the Board to implement "retail choice." Retail choice allows retail electric customers to be given the choice of shopping directly for their electric service or opting not to shop and to receive BGS.

EDECA is quite clear on the subject of power procurement for BGS: "Power procured for basic generation service by an electric power supplier shall be purchased at prices consistent with market conditions." (N.J.S.A. 48:3-57d). The same section goes on to note that "charges assessed to customers for basic generation service shall be regulated by the Board, and shall be based on the reasonable and prudent cost to the supplier of providing such service, including the cost of power purchased at prices consistent with market conditions, by the supplier in the competitive wholesale marketplace." (Emphasis added.) BGS is a regulated market-priced service.

The goal of having BGS reflect market conditions and competitive power market prices not only is consistent with EDECA, but also is consistent with several other important objectives. Retail choice can only develop and proliferate efficiently if the alternative to retail choice – BGS – is efficiently priced.¹ As currently structured by the BPU, the BGS Auction Process ensures that BGS is efficiently priced. The BGS product is a full requirements product. Suppliers bear and include in their price a wide variety of risks including load variability, market volatility, fuel price increases, migration, and changes in the PJM marketplace. These risks are not managed by regulation; rather, these risks are managed by competitive entities that can employ creative risk

¹ "Efficient proliferation of retail choice" should not be confused with artificially stimulating retail choice or maximizing switching. Rather, efficient proliferation of retail choice occurs when Third Party Suppliers ("TPSs") are able to offer added value that can induce customers to switch away from a market-priced BGS offering.

management strategies. Hence, the price that retailers compete against is the risk-adjusted price of competitive BGS suppliers providing full requirements service. A market price means the best or lowest market price that can be reasonably obtained, a price that will vary with market conditions including conditions in world energy markets. This is exactly what EDECA intended.

Intrinsic characteristics of the customer classes must also be considered in an examination of the appropriate BGS procurement process. In implementing the BGS Auction Process, the Board recognizes that some customer classes are able to understand the risks of price movements in competitive power markets and will be able to absorb risks or contract for the management of those risks. On the other hand, other customer classes may neither understand nor be able to manage these price risks in a way that yields a more economical result than the BGS offering.

In this regard, the BGS-RSCP product is designed to provide residential customers and smaller commercial and industrial customers with BGS at a stable price that still reflects competitive, market-based costs for a price-risk managed service. While it is entirely appropriate that such customers receive a stable market price, larger, more sophisticated, commercial and industrial customers can react to real-time electricity price signals to facilitate demand response and can contract for price-risk management in the competitive market. The BGS-CIEP product, which reflects real-time hourly energy spot prices, has been implemented for such larger, more sophisticated customers.

These broad policy goals of the BGS procurement process are reflected in and expanded upon by a number of specific goals that apply to the BGS procurement process:

- To obtain reliable supply on behalf of BGS customers, at prices consistent with market conditions.
- To establish a fair and transparent competitive process that will maximize participation. The process should be transparent in terms of the requirements for participation, the supply contract, the retail rates that will result from the Auction, and the manner in which final auction prices are determined and in which winners

² The terms "real-time hourly energy spot price" or "real-time hourly spot price" refer to PJM's Residual Metered Load aggregate real-time Locational Marginal Price.

emerge at the Auction. The process should be fair and transparent in terms of providing timely and equal access to information for all bidders.

- To allocate supply responsibility efficiently over the loads of the multiple EDCs. An efficient allocation of supply helps to ensure that prices best reflect the market, so that any market perceptions regarding differences in serving various EDCs are reflected in the prices.
- To have competitive entities take, manage, and price BGS risks. BGS is a pricerisk management service where competitive entities assemble supply components
 in the competitive power market, and assess and price these risks. This ensures
 that customers obtain the full benefits of competition by opening the price-risk
 management function to competitive discipline.
- To implement BGS at market rates that reflect customer class, seasonal and timeof-day market differences in order to encourage efficient consumption and conservation decisions, and in order to encourage the development of efficient retail competition. BGS rates should also minimize customer switching in response to rate design inefficiencies.
- To minimize customer confusion by presenting customers who stay on BGS with the appropriate type of retail rate structure and design.
- To design a BGS product consistent with the ability of various customer classes to react to price and manage energy price risks.
- To preserve the financial integrity of the EDCs. BGS costs and revenues exceed 60% of total EDC cost and revenues. BGS costs are an order of magnitude greater than EDC earnings. It is imperative that the BGS process protects the financial integrity of the EDCs.

III. THE ESTABLISHED BGS AUCTION PROCESS IS UNIQUELY SUITED TO THE GOALS OF BGS PROCUREMENT

The Board's BGS Auction Process fully specifies and provides structure to all necessary aspects of a procurement process. In addition, all of the elements of the BGS Auction Process work together to achieve the goals identified in and flowing from EDECA's mandate. The remainder of this section discusses briefly how the principal elements of the BGS Auction Process work together to achieve these goals.

III. A. Product Definition

The BGS Auction Process defines the product as a full requirements product. BGS suppliers are responsible for serving a percentage of an EDC's default service load, whatever the load may be at any given point in time. Bidders compete to serve BGS customers by striving to be the best at assembling supply components (energy, capacity, renewable attributes, etc.) in the competitive power market, and at assessing and pricing the risks associated with serving a percentage of BGS Load. Having a full requirements product places the portfolio acquisition and price-risk management function in the hands of the competitive entities that can most efficiently carry out these tasks. The full requirements product is designed so that all aspects of BGS Supply that can be provided through the competitive market, including risk assessment and management, are provided and priced through the competitive market. The full requirements product is fully consistent with EDECA's preference for reliance on competitive forces. The full requirements product also obtains a full service price for BGS that serves as an efficient competitive benchmark for efficient retail choice and enables potential BGS suppliers to bid with knowledge as to the retail rates that will result from the Auction. The full requirements product thus also encourages the development and efficient working of competitive retail markets.

Currently, the full requirements product is procured separately for residential and smaller commercial and industrial customers (the "BGS-RSCP product") and for larger commercial and industrial customers (the "BGS-CIEP product"). The BGS-RSCP product is procured on a three-year rolling portfolio basis and the BGS-CIEP product is procured on an annual basis. The use of an appropriate term structure enables smaller commercial and residential customers to benefit from a stable yet market-based rate that is appropriate for these customers, consistent with

EDECA, and helps to minimize customer confusion. Providing larger customers price certainty for capacity, RPS compliance, and ancillary services also helps to minimize customer confusion and provides for an environment where the retail offerings can develop efficiently. This dual structure provides appropriate stability and a hedge against volatility for each customer type, and achieves the goal of designing a product that is consistent with the ability of various customer classes to react to price and to manage energy price risks.

III. B. Auction Format

The BGS Auction Process solicits bids through a clock auction: a multiple round process with dynamic information feedback. Bidders submit bids each round as prices tick down, and each round bidders get information about how the market views the auction opportunity. On the basis of that information, bidders have an opportunity to revise their bids, and switch their bids from one EDC to another. The information that bidders receive during the BGS Auction reduces the uncertainty that bidders face and leads to more aggressive bidding. In this way, the BGS clock auction format encourages competitive bidding and efficient market prices consistent with EDECA. The fact that bidders can switch from one EDC to another means that any price differences among the EDCs reflect the market's view of differences in the cost to serve each EDC's BGS Load. Hence, the BGS Auction achieves efficient relative prices and an efficient allocation of supply responsibility among the EDCs. As explained later in greater detail, the BGS Auction also provides a large degree of transparency as all bidders understand how prices are determined and how winners emerge. This transparency encourages participation and further helps to obtain reliable supply at prices consistent with market conditions.

III. C. Competitive Safeguards

In addition to the fact that a clock auction format promotes competitive bidding, several competitive safeguards are part and parcel with the BGS Auction Process: (i) the Association and Confidential Information Rules; (ii) the setting of load caps; and (iii) the ability to reduce the volume at the Auction.

The Association and Confidential Information Rules ensure that bidders independently and vigorously compete against each other at the Auction, resulting in competitive bidding, and leading to the procurement of reliable supply at an efficient market price. Additionally, the

Association and Confidential Information Rules ensure that the bidders' confidential information is properly kept confidential so that participation in the BGS procurement process does not damage or hinder any other market activities that the bidder undertakes. Through its Auction Orders, the Board has consistently upheld the confidentiality of bidders' information as an exception to OPRA. Bidders' certainty that their competitively-sensitive information will be protected as confidential encourages participation. Robust participation in turn leads to efficient market prices.

The load cap limits the number of tranches that a bidder can bid and win at the Auction. This competitive safeguard not only controls the ability of any entity to unduly influence the auction-closing price but also provides for needed diversity in the pool of BGS suppliers. This diversity limits the exposure of the EDCs and their customers to the credit risk of any one entity, providing reliable supply sources to BGS customers.

III. D. Qualification Procedures

The BGS Auction Process provides for a standardized qualification process. As part of the qualification process, all prospective bidders must accept in advance the terms of a form of master contract between the BGS supplier and the EDC acting as agent for its customers, and the terms of the auction rules under which BGS is procured. All prospective bidders also must meet standard credit provisions, which ensure that customers receive the benefit of the BGS bargain in terms of reliable service for the term of the contract. The standard contract terms, qualification process, credit terms, and the price-only basis for the evaluation of bids, directly contribute to the transparency of the process as bidders understand the terms under which they participate in the BGS process. These features also contribute to the fairness of the process. Once a prospective supplier has been qualified, and registered as a bidder, its bids are evaluated on a price-only basis, which ensures that all bidders compete on a "level playing field." The transparency and fairness of the process encourage maximum participation, and maximum participation leads to efficient market prices.

III. E. Rate Design

The BGS Auction Process features a method for translating the auction closing prices into retail rates. For residential and smaller commercial and industrial customers, the rate design

methodology properly reflects time-of-use pricing differentials. For larger commercial and industrial customers, the rate design methodology reflects energy prices at the hourly market. In both cases, the methodologies employed provide the proper benchmarks to advance the goal of efficient retail pricing and development of competitive retail markets. The EDCs specify and communicate the rate design methodology to bidders in advance of the Auction. Thus, bidders can properly assess the risk in serving BGS Load, reducing bidders' uncertainty and encouraging participation in the process.

III. F. Roles

The EDCs, the BGS Auction Manager, Board Staff, and the Board's Advisor all have clearly defined roles that allow them to contribute to the management of the BGS procurement process. The EDCs file with the Board their procurement proposal each year, provide bidders with data and documents needed to prepare their bids, assess the financial and creditworthiness qualifications of suppliers, support the promotion of the auction opportunity, and manage the contracts with BGS suppliers on behalf of their customers. These activities maximize the participation in the process and ensure efficient market prices. The Board considers the procurement proposal as well as accounting, contingency plans, and cost recovery. Approval of these items helps to further the goal of protecting the financial integrity of the EDCs.

The BGS Auction Manager serves as a single point of contact for bidder questions and concerns, maintains a website through which bidders are kept informed about the process, ensuring the fairness of the process by providing equal access to information for all bidders. Additionally, the BGS Auction Manager manages the qualification procedure and the bid process. Board Staff and the Board Advisor monitor the entire process and monitor the bids round by round. This oversight further enhances the fairness and the transparency of the process, promoting participation.

Further, the BGS Auction Manager and the Board's Advisor each submit a report to the Board promptly at the close of the Auction so that the Board can be in a position to evaluate whether the process was competitive and whether the process was conducted as approved. The Board has in the past committed to make this assessment within two business days. This prompt Board review of auction results enables suppliers to give their best bids and contributes to the

goal of obtaining reliable supply at prices consistent with market conditions. The way in which the BGS Auction Process is managed and the oversight provided by the Board and its Advisor are important factors in its success and in the ability of the process to meet its goals.

IV. OVERVIEW OF THE EDCS' PROPOSAL FOR THE 2019 AUCTIONS

The EDCs have again worked together to develop a detailed proposal for the competitive bidding to procure BGS Supply that builds on the experience of prior BGS Auctions. The EDCs' proposal for the BGS competitive bidding process for the supply period beginning June 1, 2019 ("2019 Auctions") is summarized below.

IV. A. BGS – Residential Small Commercial Pricing ("BGS-RSCP")

- BGS-RSCP procurement offers will be solicited through a statewide Auction Process that simultaneously seeks offers for all BGS-RSCP Load in the State. Appendix B, the BGS-RSCP Auction Rules, further describe this Auction Process.
- 2. The BGS-RSCP Auction will seek offers for the supply of full requirements tranches of each EDC's BGS-RSCP Load for a three-year period. For each EDC, tranches in the 2019 BGS-RSCP Auction will be identical and uniform and will represent a fixed percentage of that EDC's total BGS-RSCP Load. Approximately two-thirds of the EDCs' BGS-RSCP Load for the period from June 1, 2019 through May 31, 2020 was secured through the 2017 and 2018 Auctions. Therefore, approximately one-third of the EDCs' BGS-RSCP Load will be procured for the BGS Supply period beginning June 1, 2019. Following a successful Auction Process, the EDCs will have under contract one-third of their total BGS-RSCP Load with a remaining contract term of one year, one-third of their total BGS-RSCP Load with a remaining contract term of two years, and one-third of their total BGS-RSCP Load for a term of three years.
- 3. The EDCs will use a multiple round descending clock auction to procure BGS-RSCP Supply. In a round, bidders will state how many tranches they wish to serve of an EDC's BGS-RSCP Load at the price in that round. The going price will decrease each round in which there is excess supply and the BGS-RSCP Auction will end when the amount proposed to be supplied is equal to the amount the EDCs wish to procure. There will be a single clearing price for each EDC's BGS-RSCP Load that will apply to all tranches for that EDC procured in this Auction. Payments to bidders from June through September will be shaped by the use of a summer multiplicative factor on the Auction price, and

payments for the remaining months will be shaped by the use of a winter multiplicative factor.

- 4. The EDCs propose changing the shape of the decrement function from a linear function to a stepwise function for PSE&G, JCP&L, and ACE (RECO already has a stepwise function for its decrement). This is the same change that was proposed and approved for the BGS-CIEP Auction in 2018 and thus this proposal would harmonize the approach across the two Auctions. The stepwise functions, which are described in detail in the BGS-RSCP Auction Rules, closely approximate the final decrement formulas used for the 2018 BGS Auctions. These decrement functions are expected to serve their main purpose, namely to ensure that the Auction is paced appropriately, with price decreases sufficiently large when excess supply is abundant and price levels are above those consistent with the market, and with smaller price decreases as the Auction nears its end and prices approach bidders' valuations. These functions are easier for bidders to understand. Furthermore, the use of the stepwise functions for the larger EDCs, which was done for the first time in the 2018 BGS-CIEP Auction, has shown that stepwise functions minimize departures from efficient relative prices over the course of the Auction. This occurs because the prices for different EDCs are more likely to decrease by exactly the same percentage from one round to the next, keeping relative prices more stable and closer to their efficient levels. Finally, these decrement functions provide less information to bidders about the excess supply of a given EDC. Less information of this type means that bidders are more likely to switch and generally change their bids considering only their valuations rather than also taking into consideration their impact on the excess supply and the price of any particular EDC. While this last benefit is more important in the BGS-CIEP Auction where a bidder controls a larger number of tranches because of the 45% statewide load cap, it is still expected to promote competition in the BGS-RSCP Auction and favor auction results that yield efficient market prices, to the benefit of ratepayers.
- 5. A rate design methodology that accounts for supply procured in prior Auctions will be used to translate final Auction prices into BGS-RSCP customer rates for one year beginning June 1, 2019. The EDCs' rate design proposals, including the timing of setting

these rates, are detailed in each EDC's Company Specific Addendum. Bidders will be provided with a spreadsheet that converts final Auction prices into customer rates. This will enable bidders to assess migration risk at various price levels. BGS-RSCP rates will reflect market-influenced seasonality and time-of-day use, where appropriate and feasible, in order to provide efficient price signals.

- 6. The rate design methodology yields for each EDC the summer multiplicative factor and the winter multiplicative factor used for supplier payments. The EDCs propose that the summer and winter payment factors both be set to 1 for any EDC whose rate design methodology results in a summer payment factor less than 1 and a winter payment factor over 1. The EDCs will update the seasonal payment factors at the time of the compliance filing to the Board, to reflect updates of the inputs. Further, the EDCs will update inputs to the rate design methodology one final time approximately ten days before the BGS-RSCP Auction. All such updates will be communicated to bidders. The last update will be early enough to provide bidders certainty before the Auction while allowing the EDCs to update the PJM transmission obligations to their 2019 values and potentially to integrate any decisions by the Board regarding transmission rates issued at its January 2019 agenda meeting.
- 7. To ensure supplier diversity, each EDC will have a load cap for its BGS-RSCP Load ("EDC load cap"). An EDC load cap is a maximum number of tranches that a bidder can win in the Auction for that EDC. In addition, there will be a statewide load cap that limits the aggregate amount of BGS-RSCP Load for all EDCs that can be won by any bidder.

IV. B. BGS – Commercial and Industrial Energy Pricing ("BGS-CIEP")

The EDCs proposal for the 2019 BGS-CIEP Auction can be summarized as follows:

1. The EDCs propose a multiple round descending clock auction to procure BGS-CIEP Supply. BGS procurement offers will be solicited through a statewide Auction Process that simultaneously seeks offers for all BGS-CIEP Load in the State. Appendix A, the BGS-CIEP Auction Rules, further describe this Auction Process.

- 2. The BGS-CIEP Auction will seek offers for the supply of full requirements tranches of each EDC's BGS-CIEP Load. For each EDC, tranches will be identical and uniform and will represent a fixed percentage of that EDC's total BGS-CIEP Load based on a tranche size of approximately 75 MW on an eligible basis. The procurement term for BGS-CIEP load will be one year with the BGS-CIEP Supply period beginning on June 1, 2019.
- 3. The BGS-CIEP Auction will determine the BGS-CIEP Price, which will be paid to bidders on the basis of the capacity obligation of customers. The EDCs propose, consistent with previous BGS-CIEP Auctions, a pre-specified ancillary service component of \$6.00/MWh.
- 4. For BGS-CIEP tranches, rate schedules will specify the BGS-CIEP charge resulting from the final BGS-CIEP Auction Price as a per kW or per kWh rate, a per kW or per kWh rate for the transmission obligation, a pre-specified per kWh rate for ancillary services, and a provision to pass through the real-time hourly energy spot price. All CIEP customers will be charged the CIEP Standby Fee, which the EDCs propose be set at 0.015¢/kWh. BGS-CIEP suppliers will receive a proportional share of revenue, comprised of five elements: (i) the pre-specified CIEP Standby Fee for sales made to all CIEP-eligible customers at the customer meter, (ii) the daily BGS-CIEP capacity obligation times the BGS-CIEP Price determined at the BGS-CIEP Auction, (iii) the daily BGS-CIEP transmission obligation times the transmission rate, (iv) the hourly BGS-CIEP Load at the EDC's PJM zone bus times the real-time hourly spot price, and (v) the prespecified ancillary services rate times BGS-CIEP sales adjusted for losses.

IV. C. BGS-RSCP and BGS-CIEP

1. Each BGS supplier will be required to assume PJM Load Serving Entity ("LSE") responsibility for the portion of BGS Load (whether BGS-CIEP or BGS-RSCP) served by that supplier. In accordance with PJM Agreements, BGS suppliers will be physically and financially responsible for the day-to-day provision of electricity to BGS customers. This full requirements service includes the provision of capacity, energy, transmission, ancillary services and any other services as may be required by PJM. The LSE obligations of each BGS tranche are subject to uncertainty and bidders will be required to manage

- this uncertainty. All BGS customers are free to come and go from BGS, provided that they give notice at least 13 days before their next scheduled meter reading.
- 2. The detailed commercial terms and conditions under which the BGS supplier will operate will be set forth in the BGS Supplier Master Agreements. The BGS Supplier Master Agreements are designed to govern the interaction of each EDC and the BGS suppliers during the period when the BGS supplier is serving BGS Load. The BGS Supplier Master Agreements for BGS-CIEP and BGS-RSCP, as proposed by the EDCs and subject to Board approval, are attached to this filing as Appendices C and D.
- 3. The EDCs propose that the CIEP line remain unchanged at 500 kW for the 2019 BGS Auctions. BGS customers with a Peak Load Contribution ("PLC") of 500 kW or more will be required to take service under a BGS-CIEP tariff or rate.
- 4. The EDCs propose to continue to allow potential bidders to propose modifications to the standard form of the Post-Auction Letter of Credit as well as to the standard form of the Pre-Auction Letter of Credit. The EDCs propose to continue to make available an alternate guaranty process for bidders that have corporate policies that preclude them from using the Standard Form of Guaranty appended to the BGS Supplier Master Agreements. The EDCs propose that the specific deadlines associated with the multiple steps of such a process be published separately in the guidelines to the alternate guaranty process (rather than those deadlines being included in the general Auction timeline as they are not generally relevant to a majority of bidders).
- 5. The final bidder information webcast is an opportunity to communicate to bidders a number of parameters including the level of the tranche fees and the final decrement formulas. The EDCs propose to release to bidders the transmission rates at this time also. This allows the EDCs to highlight the release of this information for bidders and potentially to integrate any decision from the Board taken in this regard at the January 2019 agenda meeting.
- 6. The EDCs have developed contingency plans, tariff sheets, and accounting and cost recovery proposals that are detailed in their respective Company Specific Addenda.

These are essential elements of the EDCs' Auction proposal, and the EDCs request that the Board review and approve these elements of the proposal.

- 7. The Board will render a decision on the Auction Process and render a decision on the Auction results. Under the proposed Auction Process, the Board will approve or reject in their entirety the results of the BGS-RSCP Auction and, separately, the results of the BGS-CIEP Auction, by the end of the second business day following the day on which the last Auction closes. The Board, at its discretion, has the option of rendering a decision on the results of one Auction and of rendering a decision on the results of the other Auction at different points in time. For example, the Board may render a decision on the first Auction that closes while the second Auction is still in progress.
- 8. The bids at the Auctions will represent binding commitments on behalf of bidders and full acceptance of all contract terms. Upon Board approval, Auction results will be a binding commitment on the EDCs and the winning BGS suppliers.

V. CONDUCT OF THE AUCTIONS

In this section, the EDCs explain particular aspects of the conduct of the 2019 BGS Auctions. The roles of the Board, the Board Advisor, the EDCs and the Auction Manager are explained. In addition, the EDCs' proposal for continued confidentiality of sensitive Auction information is presented. Finally, a preliminary timeline for the Auction Process is provided.

V. A. The Roles of the Board, the Board Advisor, the EDCs and the Auction Manager

V. A. 1. The Role of the Board and the Board Advisor

Past Auction Processes have proven successful in achieving the benefit of market-based prices for BGS Supply. The EDCs believe that it is appropriate for the Auction Process to be similar to that approved in the previous Auctions.

The EDCs believe that the Board should again play a substantial role in the 2019 BGS Auctions. Specifically, the EDCs respectfully recommend that the Board and the Board Advisor be responsible for the following activities:

- The Board will approve the Auction Process set forth in this Proposal, the Auction Rules, and the EDCs' Company Specific Addenda;
- The Board will approve the BGS-CIEP Supplier Master Agreement and the BGS-RSCP Supplier Master Agreement;
- The Board Advisor will oversee the conduct of the Auctions and brief the Board during the Auction Process; and
- The Board will render a decision on final Auction results by the end of the second business day following the day on which the last Auction closes. The Board, at its discretion, has the option of rendering a decision on the results of one Auction and on the results of the other Auction at different points in time. For example, the Board may render a decision on the first Auction that closes while the second Auction is still in progress.

V. A. 2. The Role of the EDCs

It is important for the EDCs to work with the Board and with other parties to design a process that assures that supply for BGS customers is procured at a cost consistent with market conditions, that there is a smooth and seamless transfer of responsibility for BGS Supply from the prior year's BGS suppliers to BGS suppliers for the supply period beginning June 1, 2019, and that adequate protections are in place to assure that the BGS suppliers are physically and financially reliable.

In that regard, the EDCs believe that it is appropriate for the EDCs to continue to assume the previously-approved logistical responsibilities that include:

- Retention of NERA as Auction Manager to administer the Auction;
- Development of the Auction Process, which is presented to the Board in this filing;
- Promotion of the Auctions in conjunction with the Auction Manager;
- Supply of the data and other key information that the suppliers would use to prepare their bids and that will be made available through an Auction website maintained by the Auction Manager;
- Provision of follow-up technical support to the Auction Manager in response to specific
 questions received from bidders and potential bidders with respect to the data and Auction
 Process information and pre-Auction information packages;
- Development of the BGS Supplier Master Agreements and instruments for financial guarantees;
- Review and approval of financial qualifications, including review of alternate guaranty forms; and
- Execution of the BGS Supplier Master Agreements on behalf of their customers.

In addition, the EDCs propose to continue to fund the Board's retention of an independent Board Advisor to oversee the Auctions under the Board's supervision and to advise the Board with respect to interim and final approvals. As in years past, the cost of the independent Board Advisor will be recovered through the tranche fees paid by winning bidders.

V. A. 3. The Role of the Auction Manager

The EDCs will retain NERA as Auction Manager to administer the Auctions. The Auction Manager will be responsible for day-to-day administration, for dissemination of information about the Auction Process. Further, if necessary, the Auction Manager will be responsible for the development of software that will implement the rules of the Auction.

The Auction Manager would be primarily responsible for the following tasks:

- Setting up and maintaining a website for the dissemination of Auction information to stakeholders. This Auction information includes application deadlines and information webcast dates, as well as the information packages prepared by the EDCs and the Auction Manager;
- Receiving queries from interested parties, directing the questions to EDC representatives
 if necessary, and returning the answers to the inquiring parties. To maintain fairness and
 to ensure that all parties have the same information, the Auction Manager will also
 maintain a database of all questions and answers on the website;
- Receiving applications for qualification and notifying interested parties of the results of the qualification procedure;
- Managing the Alternate Guaranty process;
- Receiving indicative offers and letters of credit, ensuring that these are in accordance with the rules and notifying registered bidders of their initial eligibility;
- Developing and testing bidding procedures that implement the Auction Rules;
- Providing technical help to bidders with respect to the Auction Rules and the bidding procedures;
- Managing the interface during the Auctions, to ensure that Auction parameters such as length of rounds and decrements are set appropriately;
- Developing information packages that will be made available to bidders at the time of the bidder information webcasts:
- Drafting manuals for the Auctions;
- Reviewing other information required of bidders before and after qualification and resolving issues over associations with the Board Advisor;

- Training potential bidders in the bidding procedures;
- Providing to the Board and the EDCs a full factual report on the Auctions and on the final results; and
- Preparing the BGS Supplier Master Agreements for the EDCs upon the completion of the Auctions, but before the Board renders its decision on the Auction results, in order to streamline and expedite the contract execution process.

In addition, the Auction Manager would support the EDCs and the Board by providing assistance, when appropriate, with the following tasks:

- Promoting the Auctions to potential participants; and
- Coordinating between the Board with its Advisor, and the EDCs.

The Auction Manager is important to a well-run process, from the promotion of the Auctions to the certification of the results, and performs an essential task in developing and testing the bidding procedures for the Auctions. As in years past, the cost of the Auction Manager will be recovered through the tranche fees paid by winning bidders.

V. B. Confidentiality of Auction Information

On October 22, 2004, the Board issued an Order (Docket No. EO04040288) wherein a list of information was found exempt from the requirements of the Open Public Records Act ("OPRA"), N.J.S.A. 47:1A-1 et seq., and the rules promulgated by the Board at N.J.A.C. 14:1-12 et seq. The Board found the following information, filed as part of the Auction Process, resulting from the BGS-RSCP or the BGS-CIEP Auctions, or provided by market participants for the purpose of participating in the Auctions, to be information that would provide an advantage to competitors or bidders, and deemed it confidential and not included as a government record pursuant to OPRA:

- 1. EDC-specific starting prices that are in effect for the first round of bidding;
- 2. Logic processes and algorithms used by the Auction Manager to determine the starting prices, and volume adjustments during the Auction rounds;

- 3. Indicative offers consisting of the number of tranches a qualified bidder is willing to supply at the maximum and minimum starting prices;
- 4. Auction round prices and individual bids in each round;
- 5. Bidder information supplied to qualify for the Auctions from the Part 1 Application;
 - The identities of the bidders except for the identities of the winners as released by the Board;
 - Information on Bidding Agreements;
 - Financial and Credit Requirements;
 - Guarantors' Information;
 - Justification for Omissions.
- 6. Bidder information supplied to register for the Auctions from the Part 2 Application:
 - The identities of the bidders except for the identities of the winners as released by the Board;
 - Qualified Bidders' Indicative Offers and Calculations of Required Bid Bond;
 - Qualified Bidders' Preliminary Maximum Interest in Each Product;
 - Additional Financial and Credit Requirements;
 - Associations and Confidential Information Certifications:
 - Justification for Omissions.

The EDCs request that the Board find and conclude that the foregoing information be deemed non-public proprietary commercial and financial information that would provide an advantage to competitors or bidders and not included as a government record pursuant to OPRA.

V. C. Regulatory Milestones and Tentative Auction Timeline

The proposed Auction timeline is as follows. The EDCs request that the Board approve this tentative timeline as a guideline for the conduct of the Auctions with the understanding that the time lapses between the various steps be materially adhered to (e.g., the time between the Part 2 application and the Auction not be unreasonably shortened). However, the specific dates would be subject to adjustment to accommodate holiday schedules, the schedules of other procurements in the region, and external events.

Event	Date
EDC proposal filed	Friday, June 29, 2018
Discovery request deadline	Monday, July 23, 2018
Discovery response deadline	Monday, August 06, 2018
First FAQ is posted	Tuesday, August 14, 2018
Release of Preliminary Draft of RSCP Pricing Spreadsheet	Tuesday, August 21, 2018
Deadline for Initial Comments on all proposals	Wednesday, September 05, 2018
Announce Alternate Guaranty Process Available	Tuesday, September 11, 2018
Legislative-type Board Hearing	September 2018
Public Hearings	September 2018
Deadline for Final Comments	Wednesday, October 03, 2018
Information Webcast for Potential Bidders	Thursday, October 04, 2018
Illustrative Part 1 and Part 2 Application Forms are posted	Thursday, October 04, 2018
Comment Process for Letters of Credit is posted	Thursday, October 04, 2018
Deadline for Expression of Interest in Alternate Guaranty Process	Tuesday, October 23, 2018
Deadline to propose modifications to the standard form of the Pre- Auction and Post-Auction Letters of Credit by NOON	Tuesday, October 23, 2018
Auction Manager provides individual responses to parties proposing modifications to the Letters of Credit	Thursday, November 01, 2018
All modifications to the standard form of the Letters of Credit that are acceptable on an optional basis are posted	Monday, November 05, 2018
Statewide minimum and maximum starting prices announced	Wednesday, November 14, 2018
Load caps announced	Wednesday, November 14, 2018
Tranche sizes announced	Wednesday, November 14, 2018
Board decision on Auction proposal	November 2018
EDC Compliance Filing	November/December 2018
Expected Board Decision on Compliance Filing	December 2018

Event	Date
Information Webcast for Potential Bidders	Monday, December 3, 2018 (tentative)
Online Application Portal made available to Bidders	Tuesday, December 4, 2018
Deadline to submit Part 1 Application due by NOON	Tuesday, December 18, 2018
Part 1 Applications are reviewed	December 18-21, 2018
Applicants are notified of Part 1 Application Results	Friday, December 21, 2018
Deadline to submit Part 2 Application due by NOON	Thursday, January 10, 2019
Part 2 Applications are reviewed	January 10-17, 2019
Applicants are notified of Part 2 Application results	Thursday, January 17, 2019
Mark-to-Market Information Release	Tuesday, January 22, 2019
Tranche Fee is announced	Tuesday, January 22, 2019
Transmission Rates announced	Thursday, January 22, 2019
Information Webcast for Registered Bidders	Tuesday, January 22, 2019 (tentative)
Auction Manager informs Registered Bidders of changes to decrement formulas or ranges of total excess supply (if necessary)	Wednesday, January 23, 2019
Final rate spreadsheets are posted	Thursday, January 24, 2019
Final seasonal factors announced	Thursday, January 24, 2019
Trial Auctions for Registered Bidders	Thursday, January 24, 2019
Auction Manager informs Bidders registered in BGS-CIEP Auction of starting prices	Tuesday, January 29, 2019
Auction Manager informs Bidders registered in BGS-RSCP Auction of starting prices	Wednesday, January 30, 2019
BGS-CIEP Auction Starts	Friday, February 01, 2019
BGS-RSCP Auction Starts	Monday, February 04, 2019
Board decision on Auction results	Within 2 business days of close of the BGS-RSCP Auction or BGS-CIEP Auction, whichever comes later
Winning suppliers execute BGS Supplier Master Agreements	Within 3 business days of Board decision
Documents provided to bidders by Auction Manager containing confidential information must be destroyed	Within 5 business days of Board decision
BGS-RSCP rates filed with Board	No later than 30 days prior to becoming effective
Power Flows	Saturday, June 01, 2019

VI. BOARD APPROVAL AND EXECUTION OF BGS SUPPLIER MASTER AGREEMENTS

The EDCs propose that the Board decide formally, within two business days of the day on which the last Auction closes, whether or not to accept the Auction results. The Board, at its discretion, has the option of rendering a decision on the results of one Auction and of rendering a decision on the results of the other Auction at different points in time. For example, the Board may render a decision on the first Auction that closes while the second Auction is still in progress. Since the Auction Process would have been previously approved by the Board, accepted bids resulting from the Auction would be deemed reasonable and prudent.

The EDCs recommend that each winning bidder be immediately notified by the Auction Manager of the Board's approval of the Auction results, and that each winning bidder and each EDC be given a period of three business days from receipt of the notification to formally execute the BGS-CIEP and BGS-RSCP Supplier Master Agreements. The obligations outlined in those Agreements will be part of an irrevocable offer that will become a binding, contractual obligation upon the award of the bid and contract execution will memorialize this commitment.

In other words, the purpose of the review and approval process recommended by the EDCs is for the Board to take the necessary time at the start of the Auction Process to resolve potentially contentious issues, to provide a mechanism for an expeditious decision from the Board in response to the Auction results, and to provide assurance to potential bidders that, once they are notified by the Board that their offer to serve one or more tranches has been accepted, they will indeed be serving that BGS Load.

VII. LIST OF CHANGES

Below is provided a listing of changes to the proposal for the provision of BGS for the period beginning June 1, 2019 compared to the proposal for the provision of BGS for the period beginning June 1, 2018. The EDCs do not consider any of these changes to be material and are providing this list for the convenience of the parties in this proceeding.

1. Overview of Changes to Proposal

The EDCs maintain all material aspects of the prior year's proposal with respect to the product, standard contract, auction format, rate design, bidder interface, and competitive safeguards. The EDCs propose minor changes as enhancements to the prior year's process, namely:

- As reflected in the BGS-RSCP Auction Rules (Appendix B to this filing), the EDCs propose to use stepwise functions for the decrement formulas.
- As reflected in the tentative Auction timeline: (i) intermediate deadlines in the
 alternative guaranty process would be included in a separate document rather than be
 part of the Auction timeline; (ii) transmission rates would be announced at the time
 of the final bidder information webcast; and (iii) inputs to the rate design methodology
 would be updated one final time approximately ten days before the BGS-RSCP
 Auction.
- As reflected in each Company Specific Addendum, each EDC has eliminated the
 input in its rate design spreadsheet for the incremental RPM cost results from PJM's
 Incremental Auctions for Capacity Performance. This reflects the fact that the last
 BGS Supplier Master Agreements to include supplier compensation for incremental
 RPM costs have expired.

2. Additional Minor Changes to documents

The following additional changes to the documents are minor and primarily administrative in nature.

BGS Supplier Master Agreements

- · Dates are updated to the current year; and
- The docket number is updated.

BGS Auction Rules

- Dates are updated to the current year;
- The tranche targets are updated with the most recent PJM data and the examples are modified as needed to reflect the change in the tranche targets; and
- Decrement formulas reflect the final decrement formulas from the prior year for the BGS-CIEP Auction (the decrement formulas reflect the new proposal for the BGS-RSCP Auction).

Company-Specific Addenda

- Dates are updated to the current year; and
- The docket number is updated.

VIII. APPENDIX A

Provisional BGS-CIEP Auction Rules

VIII. APPENDIX A: PROVISIONAL BGS-CIEP AUCTION RULES

Table of Contents

/III. Apper	ndix A: Provisional BGS-CIEP Auction Rules	1
VIII.A.	Introduction	
VIII.A.1.	Overview	3
VIII.A.2.	Basic Generation Service and Load to Be Procured	4
VIII.A.3.	Payment to BGS-CIEP Suppliers	6
VIII.A.4.	Reliance on Product Definition, Payment Information, Rates, and Customer	
	Switching in These Rules	9
VIII.B.	Before the Auction	9
VIII.B.1.	Information Provided to Bidders	9
VIII.B.2.	Qualification Process	10
VIII.B.3.	Starting Prices	15
VIII.B.4.	Extraordinary Events	15
VIII.C.	Bidding Rules	16
VIII.C.1.	Overview of Auction Format	16
VIII.C.2.	Round Phases and Bidding Day	19
VIII.C.3.	Round 1 of the Auction	20
VIII.C.4.	Bidding in Round 2	
VIII.C.5.	Calculating and Reporting Phases in Round 2	30
VIII.C.6.	Round 3 and All Subsequent Rounds	
VIII.C.7.	Reporting of Total Excess Supply	
VIII.C.8.	Calculation of Next Round Prices	
VIII.C.9.	Pauses in the Auction	
VIII.C.10.	Failure to Submit a Bid	
VIII.C.11.	End of Auction	
VIII.D.	Post Auction	
VIII.E.	Association and Confidential Information Rules	47
VIII.E.1.	Process for Reporting Associations, Identifying Concerns and Remedies	47
VIII.E.2.	Association Criteria	
VIII.E.3.	Definitions of Confidential Information	
VIII.E.4.	Certifications and Disclosures to Be Made	51
VIII.E.5.	Actions to Be Taken If Certifications Cannot Be Made	55
VIII.F.	Glossary of Auction Terms	57
VIII.F.1.	Associated With	57
VIII.F.2.	Auction Volume	57
VIII.F.3.	BGS	57
VIII.F.4.	BGS Load	57
VIII.F.5.	BGS-CIEP Customer	
VIII.F.6.	BGS-CIEP Load	
VIII.F.7.	BGS-CIEP Peak Load Share	
VIII.F.8.	BGS-RSCP Load	
VIII.F.9.	Bid	
VIII.F.10.	Bidding Phase	
VIII.F.11.	Calculating Phase	
VIII.F.12.	CIEP Customer	
VIII.F.13.	CIEP Load	
VIII F 14	CIED Deak I god Chara	50

VIII.F.15.	CIEP Standby Fee	. 59
VIII.F.16.	Contingency Plan	. 59
VIII.F.17.	Decrement	. 59
VIII.F.18.	Denied Switches	. 60
VIII.F.19.	Eligibility and Initial Eligibility	. 60
VIII.F.20.	Eligibility Ratio	
VIII.F.21.	End of Auction	. 60
VIII.F.22.	Excess Supply	. 60
VIII.F.23.	Exit Price	
VIII.F.24.	Extension	.61
VIII.F.25.	Financial Guarantee	.61
VIII.F.26.	Free Eligibility	. 61
VIII.F.27.	Full-Requirements Service	. 61
VIII.F.28.	Going Price	. 61
VIII.F.29.	Indicative Offer	. 61
VIII.F.30.	Initial Eligibility	. 62
VIII.F.31.	Minimum and Maximum Starting Prices	. 62
VIII.F.32.	MW-Measure	
VIII.F.33.	Outbid (Denied Switches)	. 62
VIII.F.34.	Oversupply Ratio	. 62
VIII.F.35.	Part 1 Application	. 62
VIII.F.36.	Part 1 Application Date	. 63
VIII.F.37.	Part 2 Application	. 63
VIII.F.38.	Part 2 Application Date	. 63
VIII.F.39.	Product in the Auction	. 63
VIII.F.40.	Qualified Bidder	. 63
VIII.F.41.	Recess	
VIII.F.42.	Reductions	. 64
VIII.F.43.	Registered Bidder	
VIII.F.44.	Reporting Phase	. 64
VIII.F.45.	Round	. 65
VIII.F.46.	Rounding	. 65
VIII.F.47.	RSCP Load	. 65
VIII.F.48.	Session	. 65
VIII.F.49.	Starting Prices	. 65
VIII.F.50.	Statewide Load Cap	. 65
VIII.F.51.	Switches and Switching	. 65
VIII.F.52.	Switching Priority	. 65
VIII.F.53.	Target Eligibility Ratio	
VIII.F.54.	Time-Out	
VIII.F.55.	Total Excess Supply	
VIII.F.56.	Tranche	. 66
VIII.F.57.	Tranche Size	
VIII.F.58.	Tranche Target	
VIII.F.59.	Withdrawal	
/III.G.	Appendix	
VIII.G.1.	Tranches	
VIII G 2	Decrements	67

VIII.A. Introduction

VIII.A.1. Overview

The four New Jersey Electric Distribution Companies ("EDCs") are Atlantic City Electric Company ("ACE"), Jersey Central Power & Light Company ("JCP&L"), Public Service Electric and Gas Company ("PSE&G"), and Rockland Electric Company ("RECO"). The EDCs will procure full-requirements service for their Commercial and Industrial Energy Pricing (BGS-CIEP) loads in a single, statewide Auction for a one-year term from June 1, 2019 to May 31, 2020. The Auction will determine a CIEP Price measured in \$/MW-day. Each EDC's CIEP Load includes the load of all customers served at transmission voltage and other large customers, as described in Table VIII-1. The BGS-CIEP Load for each EDC will be divided into a number of tranches of equal size, each accounting for the same percentage of the EDC's BGS-CIEP Load.

BGS-CIEP is not a term-price service, but a service that is priced to the hourly market. Suppliers will be bidding for the right to serve a portion of BGS-CIEP Load for one or more EDCs. Customers on BGS-CIEP (BGS-CIEP customers) will pay, and winning suppliers will receive, the PJM "Hourly Real-Time Spot Price" for the zone, which refers to PJM's loadweighted average Residual Metered Load aggregate Real-Time Locational Marginal Price, as well as the pre-specified ancillary services rate and transmission costs based on the PJM Open Access Transmission Tariffs ("OATT") in effect on January 1, 2019. Energy charges and ancillary service charges to customers will be adjusted for losses on each EDC's system as appropriate. CIEP customers (i.e., all customers eligible for BGS-CIEP) will pay, and winning suppliers will receive, the CIEP Standby Fee, which is a fee for the option available to each CIEP customer of taking BGS on this tariff. The EDCs propose that this fee be set at 0.015¢/kWh of the energy used by CIEP customers measured at the customer meter. The Auction will determine for each EDC a CIEP Price ("price") measured in \$/MW-day, that will be reflected in the rates charged to customers on BGS-CIEP, and that will be paid to each winning supplier on the basis of the share of the tranches for an EDC that each supplier has won at the Auction. The capacity obligation is the unforced capacity requirement for the aggregate group of BGS-CIEP customers determined in accordance with the EDC and PJM practices on a daily basis.

The EDCs will procure their BGS-CIEP Loads through a simultaneous descending clock auction ("clock auction"). The clock auction proceeds in a series of rounds. During the bidding phase of each round, each bidder must indicate how many tranches of each EDC's BGS-CIEP Load the bidder wishes to serve given the prices announced by the Auction Manager. The prices will be in dollars per megawatt-day. A bid is the number of tranches of each EDC's BGS-CIEP Load that a bidder wants to serve. After the bidding phase of a round, the Auction Manager reduces the price for the tranches of an EDC by a decrement if the number of tranches bid by all bidders is greater than the number of tranches that are needed for that EDC. The Auction Manager then announces new prices for each EDC before bidding in the next round opens. The Auction continues and the prices tick down until, for each EDC, the total number of tranches subscribed falls to the point where it equals the number of tranches needed. When the Auction ends, the bidders holding tranches in the final round are the winners. Following the end of the Auction, the Board of Public Utilities ("Board") will decide whether to approve the Auction results and if it does, winners will become authorized BGS-CIEP suppliers. All winners for an EDC's tranches receive the same price, as explained in section VIII.C.11.

VIII.A.2. Basic Generation Service and Load to Be Procured

Basic Generation Service, or BGS, is the electric supply for those retail customers who are not served by a third party supplier. BGS Load is the load associated with these customers, and it is obtained by subtracting the load of third party suppliers from the retail load in an EDC zone. For each EDC, BGS Load is divided into two portions, BGS Commercial and Industrial Energy Pricing Load ("BGS-CIEP Load") and BGS Residential Small Commercial Pricing Load ("BGS-RSCP Load"). An EDC's BGS-CIEP Load is defined to include the sum of the hourly load of all BGS-CIEP customers, times the EDC loss expansion factor associated with each CIEP rate class. CIEP customers refer to all customers that are eligible for BGS-CIEP, whether or not they choose BGS-CIEP. That is, CIEP customers are retail customers who are taking BGS-CIEP, or are retail customers served by a third party supplier who, were they to take BGS, would be served under a CIEP (rather than an RSCP) tariff or rate. Table VIII-1 provides each EDC's list of rate classes for which either all customers, or only customers with a peak load contribution of 500 kW or greater, must currently take BGS on a CIEP tariff or

rate. BGS-RSCP Load is the load associated with customers who take BGS but do so on an RSCP (rather than a CIEP) tariff or rate.

Table VIII-1. Customers Who Must Take BGS on a CIEP Tariff

EDC	Rate Class	Customers Included	Description	Voltage Level
	High Tension Service ("HTS-HV")	All	General purposes at high voltages	138,000V to 230,000V
DSEAC	High Tension Service ("HTS-Sub")	All	General purposes at subtransmission voltages	26,400V to 69,000V
PSE&G	Large Power and Lighting, Primary Service ("LPL-P")	All	General purposes at primary distribution voltages.	2,400V to 13,200V
	Large Power and Lighting, Secondary Service ("LPL-S")	500 kW or greater	General purposes at secondary distribution voltages	208V to 480V
	General Service Primary ("GP")	All	General purposes at primary distribution voltages	2,400V to 34,500YV
JCP&L	General Service Transmission ("GT")	All	General purposes at transmission voltages	34,500ΔV
		All	High Tension Service	230,000V
	General Service Secondary ("GS")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	General Service Secondary Time-of-Day ("GST")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	Transmission General Service ("TGS")	All	General purposes at high voltages	23,000V or higher
	Annual General Service – Primary ("AGS-Primary")	500 kW or greater	General purposes at primary distribution voltages	4,000 & 12,000V
ACE	Annual General Service – Secondary ("AGS-Secondary")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	Monthly General Service Primary ("MGS – Primary")	500 kW or greater	General purposes at primary distribution voltages	4,000 & 12,000V
	Monthly General Service Secondary ("MGS – Secondary")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
RECO	Service Classification No. 7 – Primary TOU Service and Separately Metered Space Heating Service	All	Primary service customers with demands exceeding 500 kW	2,400V or higher
	Service Classification No. 2 – General Service	500 kW or greater	General primary and secondary service	All primary and secondary voltages

For purposes of the Auction, the BGS-CIEP Load for an EDC is called a *product in the Auction*: the BGS-CIEP Load for PSE&G is a product in the Auction, and the BGS-CIEP Load for ACE is another product in the Auction, etc. Each product in the Auction is divided into units called *tranches*, each representing the same percentage of that EDC's BGS-CIEP load. The number of tranches that each EDC will procure in the Auction and the size of each tranche (i.e., the percentage of the BGS-CIEP Load that each tranche represents for an EDC) are provided below in Table VIII-2 also provides a MW-measure for each of an EDC's tranches. The MW-measure of each tranche is determined on the basis of the obligation attributable to all CIEP customers (those who receive BGS and those who do not) during 2018 (called the *CIEP Peak Load Share*). The data in Table VIII-2 are provisional. Following the October 2018 update of this obligation by PJM, this table will be updated. The Auction Manager will provide the update no later than 10 days before suppliers must first apply to participate in the Auction.

Table VIII-2. Provisional Number of Tranches and MW-Measures of Tranches per EDC

EDC	CIEP Peak Load Share (MW)	Number of tranches	Size of tranche (%)	MW-measure (MW)
PSE&G	1,797.77	24	4.17%	74.91
JCP&L	805.89	11	9.09%	73.26
ACE	318.40	4	25.00%	79.60
RECO	56.00	1	100.00%	56.00
Total	2,978.06	40		

The EDCs do not represent that each tranche will have the loads shown above or any particular value. The actual BGS-CIEP Load will depend upon many factors including customer migration to third party suppliers and weather conditions. Bidders are responsible for evaluating the uncertainties associated with BGS-CIEP Load.

VIII.A.3. Payment to BGS-CIEP Suppliers

Winners at the Auction are authorized to become BGS-CIEP suppliers by the Board. A BGS-CIEP supplier provides *full-requirements service* for the percentage of the EDC's BGS-CIEP Load for a given supply period corresponding to the number of tranches won by the BGS-CIEP supplier for that supply period. Full-requirements service means that the BGS-CIEP supplier is responsible for fulfilling all the requirements of a PJM Load Serving Entity

("LSE"), including capacity, energy, ancillary services and transmission, and any other service as PJM may require. A winning supplier may win one or more tranches for one or more EDCs.

Each BGS-CIEP supplier receives an EDC-specific network transmission rate applied to the supplier's share (the number of tranches won times the size of the tranche) of the BGS-CIEP transmission obligation. This transmission rate will be based on the OATT rate in effect on January 1, 2019 and will be announced prior to the Auction. To the extent that the transmission rates change and that the Board approves a corresponding change to the BGS-CIEP customers' rates; (1) if such a change results in an increase in BGS-CIEP customers' rates, then any additional revenues collected as a result of such an increase, once approved by the FERC in a final order and no longer subject to a refund, would be to the benefit of BGS-CIEP suppliers; or (2) if such a change results in a decrease in BGS-CIEP customers' rates, then any decrease in revenues would result in a reduction of payments to BGS-CIEP suppliers. The parameters of what constitutes a change in transmission rates are specified in Section 15.9 of the BGS-CIEP Supplier Master Agreement. Each BGS-CIEP supplier receives an ancillary service payment rate, pre-specified for each EDC, that includes PJM-administrative costs, and that is applied to the supplier's share of BGS-CIEP Load (energy).

Each BGS-CIEP supplier receives the "Hourly Real-Time Spot Price", which refers to the PJM load-weighted average Residual Metered Load aggregate Real-Time Locational Marginal Price for the zone, applied to the supplier's share of BGS-CIEP Load (energy), and the price determined in the Auction applied to the supplier's share of the BGS-CIEP capacity obligation.

With respect to the rates that will be paid by BGS-CIEP customers, the transmission rate is based on either a demand charge or transmission obligation charge, depending on the EDC. The energy rate is the Hourly Real-Time Spot Price, which will be adjusted each hour by the relevant EDC loss expansion factor. The ancillary service rate will be a pre-determined value based on the ancillary service payment rate to suppliers adjusted for losses. The price for an EDC is the final Auction price for that EDC and will be assessed as a specific capacity obligation charge, demand charge, or energy charge. All of these charges are levied on actual BGS-CIEP customers only. Details for each particular EDC are in the Company Specific Addenda.

Each BGS-CIEP supplier also receives the CIEP Standby Fee. All CIEP customers pay this standby fee, whether the customers are currently taking BGS or not. The EDCs propose that this standby fee be set at $0.015 \phi/kWh$ of the energy used by CIEP customers measured at the customer meter. This option fee is the charge for providing the option to each CIEP customer of taking BGS on this tariff.

Example 1. 1

Suppose, for purposes of this example, that the tranche target for PSE&G is 25 tranches. A BGS-CIEP supplier wins 5 of PSE&G's 25 tranches. The BGS-CIEP supplier serves 20% of the BGS-CIEP Load and would be paid as follows.

For capacity, assume an Auction clearing price of \$135/MW-day and a BGS-CIEP capacity obligation totaling 30,000 MW-days for the month. The BGS Supplier is paid 20% of 30,000 MW-days times \$135/MW-day, or \$810,000.

For transmission, assume a BGS-CIEP transmission obligation totaling 36,000 MW-days for the month and assume an OATT of \$50/MW-day. The BGS-CIEP supplier is paid 20% of 36,000 MW-days times \$50/MW-day, or \$360,000.

For ancillary services, assume an ancillary service payment rate of \$6/MWh and monthly energy of 500,000 MWh at the BGS-CIEP customer meter; assuming a loss expansion factor of 1.046759 and adjusting for distribution and transmission losses as de-rated for PJM marginal losses of 1.601%, monthly BGS-CIEP Load is 515,000 MWh. The BGS-CIEP supplier is paid 20% of 515,000 MWh times \$6/MWh, or \$618,000.

For energy, assume 800 MWh of energy measured at the BGS-CIEP customer meter for hour 1; assuming a loss expansion factor of 1.046759 and adjusting for distribution and transmission losses as de-rated for PJM marginal losses of 1.601%, hourly BGS-CIEP Load is approximately 824 MWh. Assume also an Hourly Real-Time Spot Price of \$50/MWh. The BGS-CIEP supplier is paid 20% of 824 MWh times \$50/MWh, or \$8,240. For the remaining hours in the month an energy payment is determined in the same way but with the actual hourly BGS-CIEP Load and the actual Hourly Real-Time Spot Price for the zone.

Assuming that energy used by all CIEP customers was 1,000,000 MWh measured at the customers' meter (double the corresponding value in the example for BGS-CIEP customers), the BGS-CIEP supplier would receive 20% of 1,000,000 MWh times 0.015¢/kWh, or \$30,000 for the month for the CIEP Standby Fee. Note that for this payment, energy at the customer meter is not increased for losses.

¹ All examples are for illustrative purposes. The prices, statewide load cap, and bids are illustrative only.

VIII.A.4. Reliance on Product Definition, Payment Information, Rates, and Customer Switching in These Rules

Information regarding the definition of a product in the Auction and information regarding the payment bases given in these Auction Rules are solely for the convenience of bidders and are not to be relied upon by bidders. The BGS-CIEP Supplier Master Agreement posted on the BGS Auction website is the document that provides the official definitions of products in the Auction and payment terms. Information regarding rates paid by customers and information on rules regarding the ability of customers to switch from or back to BGS-CIEP given in these Auction Rules are solely for the convenience of bidders and are not to be relied upon by bidders. Board Orders as well as the tariffs of each EDC are the documents that provide official information in this regard.

VIII.B. Before the Auction

VIII.B.1. Information Provided to Bidders

The EDCs make data available to potential bidders in advance of qualification. The data series are posted to the BGS Auction website, http://www.bgs-auction.com/bgs.dataroom.asp.

The EDCs provide historical data, consisting of hourly load and daily capacity and transmission peak load allocations, for the following load categories: total retail, BGS, CIEP, BGS-CIEP, RSCP, and BGS-RSCP. The data include associated zonal losses.

The EDCs provide generally three years of historical data. All data series at a given point in time use the contemporaneous definition of CIEP customers. The BGS-CIEP data for a given month includes customers who take BGS on a CIEP tariff in force in that month. The CIEP data includes customers who take BGS on a CIEP tariff in force in that month as well as CIEP customers (as defined for that month) who are served by a third party supplier. BGS-RSCP and RSCP data are derived as residuals; for example, RSCP Load is equal to total retail load less CIEP Load, and BGS-RSCP Load is equal to the BGS Load less BGS-CIEP Load. The EDCs provide monthly customer switching data (number of customers and estimated load) as currently provided to the Board. Hourly load data series, peak load allocations, and

switching statistics are updated monthly. Some historical data series older than three years remain posted to the BGS Auction website strictly for the convenience of bidders.

To the extent practicable, the EDCs will also provide additional data to assist bidders. The EDCs will provide one-time customer counts and historical aggregate energy usage for several groupings of customers who in the past were eligible to take BGS on an RSCP tariff but who are or will be required to take BGS on a CIEP tariff. These groupings may include: customers 500 to 749 kW, 750 to 999 kW, 1,000 to 1,249 kW, 1,250 to 1,499 kW, and 1,500 kW or greater. The EDCs may provide limited series of historical hourly load for specific customer classes or customer groupings for which customers have interval meters. The EDCs may provide load profiles for specific customer classes or customer groupings as well as additional historical customer switching data.

No later than 10 days² before interested parties first apply to participate in the Auction, the Auction Manager will announce a *statewide load cap*, a statewide *maximum starting price*, and a statewide *minimum starting price*. At the same time, the Auction Manager will announce the size of the CIEP Peak Load Share for each EDC, the number of tranches available to bid for each EDC in the Auction, and the MW-measure of each tranche for each EDC based on the percentage of the CIEP Peak Load Share that a tranche represents. The statewide load cap is a maximum number of tranches of BGS-CIEP Load that any one bidder can bid in the Auction and serve statewide. The statewide load cap limits the impact that any one bidder may have on the Auction. (There are no EDC-specific load caps.) The minimum and maximum starting prices establish the range of possible starting prices for the Auction: each EDC will choose a starting level for its price for round 1 of the Auction that is between the minimum and the maximum starting prices. The EDCs will agree on the statewide load cap, and on the statewide minimum and maximum starting prices. Board Staff and the Board Consultant will review these decisions.

VIII.B.2. Qualification Process

The application process is in two parts. All interested parties that have no impediments to meeting the PJM LSE requirements can submit a *Part 1 Application*. There is no state

² Unless otherwise specified, "days" refers to business days.

licensing requirement. Interested parties will be asked to submit financial information so that the EDCs can assess their creditworthiness. In addition, each interested party will be asked to comply with other qualification criteria that will have been agreed upon by all EDCs, including agreeing to comply with the BGS-CIEP Auction Rules and agreeing to the terms of the BGS-CIEP Supplier Master Agreement. Each interested party will also be asked to agree that if the interested party is successful in its Part 1 Application it will keep confidential the list of other successful applicants and it will not assign its rights or substitute another entity in its place. This is to ensure that the entity that agrees to the BGS-CIEP Auction Rules in the Part 1 Application is also the entity submitting bids in the BGS-CIEP Auction, and to ensure that the entity that agrees to the terms of the BGS-CIEP Supplier Master Agreement is the entity that will execute the BGS-CIEP Supplier Master Agreement should the interested party become an Auction winner. In accordance with these Auction Rules, execution of the BGS-CIEP Supplier Master Agreement must occur within three days of Board certification of the Auction results and within that period the Auction winner will demonstrate compliance with the creditworthiness requirements set forth in the BGS-CIEP Supplier Master Agreement. Such creditworthiness requirements will take into consideration all BGS obligations held by the Auction winner, including those from past BGS Auctions.

Applications must be submitted no later than noon³ on the *Part 1 Application Date*, which will be no earlier than ten (10) days after the maximum and minimum starting prices have been announced. All interested parties will have at least ten (10) business days to complete the Part 1 Application. Interested parties will be notified as to whether they succeeded in qualifying to participate in the Auction no later than three days after the Part 1 Application Date. An interested party that has qualified becomes a *qualified bidder*. The Auction Manager will send simultaneously to each qualified bidder a list of all qualified bidders, but the list of qualified bidders will not be publicly disclosed. Interested parties, in their Part 1 Applications, will have undertaken to maintain the confidentiality of the list of qualified bidders, and to destroy documents with this information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results, as explained further in this document in section VIII.E.4 in the "Association and Confidential Information Rules".

³ Unless otherwise specified, all times are Eastern Time Zone times.

Qualified bidders that wish to participate in the Auction must submit a Part 2 Application to the Auction Manager. Only qualified bidders may submit Part 2 Applications. Part 2 Applications must be submitted no later than noon on the Part 2 Application Date, which will be no later than 10 days before the start of the Auction. In the Part 2 Application, qualified bidders will make a number of certifications to ensure compliance with the association and confidential information portion of these rules. These certifications, provided in section VIII.E.4 below, ensure that each qualified bidder is bidding independently of other qualified bidders and ensure the confidentiality of information regarding the Auction. Each qualified bidder is also asked to agree to keep confidential the list of other successful applicants; to agree that the submission of any bid creates a binding and irrevocable offer to provide service under the terms of the BGS-CIEP Supplier Master Agreement; and not to assign its rights or substitute another entity in its place. On an exceptional basis, the Auction Manager may consider requests for substitution of another entity for a qualified bidder when the qualified bidder consolidates into, amalgamates into, or merges into another corporate entity. Such a request may only be considered if it is received prior to the Part 2 Application This is to ensure that the entity that agrees to comply with the Association and Date. Confidential Information Rules is also the entity submitting bids in the BGS-CIEP Auction, and to ensure that the entity that agrees that its bids create a binding and irrevocable offer to provide service under the terms of the BGS-CIEP Supplier Master Agreement is also the entity that will execute the BGS-CIEP Supplier Master Agreement should the qualified bidder become an Auction winner. With their Part 2 Application, qualified bidders will also be required to submit an indicative offer and to submit a financial guarantee in proportion to their indicative offer.

A qualified bidder is associated with another qualified bidder if the two bidders have ties that could allow them to act in concert or that could prevent them from competing actively against each other in the Auction. The competitiveness of the Auction and the ability of the Auction Process to deliver competitive prices may be harmed by the coordinated or collusive behavior that associations facilitate. The Auction Manager, who may rely, among other factors, on the number of independent competitors to set the Auction volume, would be using inaccurate information unless associations are duly disclosed in the Part 2 Application. The Auction volume is the number of tranches that the EDCs plan to purchase through the Auction.

Associations may be considered in setting the Auction volume and may be used in the application of the statewide load cap. See section VIII.E "Association and Confidential Information Rules" later in this document for precise criteria.

Sanctions can be imposed on a bidder for failing to disclose information relevant to determining associations, for coordinating with another bidder, or for failing to abide by any of the certifications that the bidder will have made in its Part 1 and Part 2 Applications. Such sanctions can include, but are not limited to, loss of all rights to serve any BGS-CIEP Load won in the Auction by such bidder, forfeiture of financial guarantees and other fees posted or paid, prosecution under applicable state and federal laws, debarment from participation in future BGS Auctions, and other sanctions that the Board may consider appropriate. For any failure to disclose information or any violation of the certifications, the Auction Manager will make a recommendation to the Board on a possible sanction and the Board will make the final determination.

An indicative offer specifies two numbers of tranches. The first number represents the amount that the qualified bidder is willing to serve at the maximum starting price on a statewide basis (i.e., for all EDCs combined). The second number represents the amount that the qualified bidder is willing to supply at the minimum starting price on a statewide basis. At each of the maximum and the minimum starting prices, the number of tranches indicated by the qualified bidder cannot exceed the statewide load cap. At the maximum starting price, the number of tranches indicated by a qualified bidder cannot be lower than two.

Indicative offers are important in two respects. First, the EDCs may use the indicative offers to inform their decision in setting the round 1 prices. Second, the number of tranches indicated by the qualified bidder at the maximum starting price determines the qualified bidder's *initial eligibility*. As explained in the bidding rules in section VIII.C.3, a bidder will never be able to bid in the Auction on a number of tranches greater than the bidder's initial eligibility. Thus, the qualified bidder is encouraged to state the maximum possible number of tranches that it would be willing to serve.

Each qualified bidder must post a financial guarantee, in the form of a letter of credit (or bid bond), proportional to its initial eligibility. A financial guarantee of \$375,000 per tranche is required. Letters of credit (or bid bonds) must be in a form acceptable to the EDCs. Sample

letters of credit and sample bid bonds that are acceptable to the EDCs will be posted to the Auction website.

Example 2.

Suppose that the maximum starting price is \$550/MW-day and the minimum starting price is \$425/MW-day. Suppose that the statewide load cap is 18 tranches. A qualified bidder submits an indicative offer of 8 tranches at the maximum starting price and 4 tranches at the minimum starting price. The qualified bidder is required to submit a financial guarantee for \$3,000,000 (8 x \$375,000) in the form of a letter of credit (or bid bond). The number of tranches of the indicative offer at the maximum starting price is not lower than the minimum of 2 tranches and does not exceed the maximum of the statewide load cap of 18 tranches. The number of tranches of the indicative offer at the minimum starting price does not exceed the maximum of the statewide load cap of 18 tranches.

For a Part 2 Application to be accepted, it must be complete, including its indicative offer and financial guarantee. The financial guarantee must be provided in a form acceptable to all EDCs and must be sufficient to cover the indicative offer at the maximum starting price. If its Part 2 Application is accepted, a qualified bidder becomes a *registered bidder*. The Auction Manager will send simultaneously to each registered bidder a list of all registered bidders and the total initial eligibility in the Auction. Neither the list of registered bidders nor the total initial eligibility in the Auction will be released publicly. Qualified bidders, in their Part 2 Applications, will have undertaken to maintain the confidentiality of the list of registered bidders and the total initial eligibility in the Auction, and to destroy documents with this information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results, as explained further in this document in section VIII.E.4.

Financial guarantees will remain in full force until the Board renders a decision on the Auction results and the bidder has won no tranches; or until the Board renders a decision on the Auction results and the bidder has won tranches, has signed the BGS-CIEP Supplier Master Agreement, and has complied with all creditworthiness requirements of that agreement. The EDCs can collect the financial guarantees if bidders fail to comply with their obligations.

BGS-CIEP suppliers must meet PJM LSE requirements by the start of the supply period.

VIII.B.3. Starting Prices

Three days before the Auction starts, the Auction Manager informs all registered bidders of each EDC's starting price, which will be the price in round 1 of the Auction. Each EDC's starting price will be no higher than the statewide maximum starting price and no lower than the statewide minimum starting price. Each EDC will set its own starting price in consultation with the Auction Manager, Board Staff and the Board Consultant.

VIII.B.4. Extraordinary Events

The EDCs, in consultation with the Auction Manager, Board Staff and the Board Consultant, may determine that, due to extraordinary events, the statewide maximum starting price and the statewide minimum starting price require revision. In this event, the schedule may also be revised. If the indicative offers have already been received, the Auction Manager would request that the registered bidders (or the qualified bidders if registration had not been completed) revise their indicative offers on the basis of the revised statewide maximum starting price and the revised statewide minimum starting price.

For such a revision to be necessary, an extraordinary event must occur between the time at which the statewide maximum starting price and the statewide minimum starting price are announced (no later than 10 days before the Part 1 Application is due) and the day on which the Auction starts. The EDCs, in consultation with the Auction Manager, Board Staff and the Board Consultant, will agree that an event constitutes an extraordinary event. Such events could include, for instance, the advent of war.

If an extraordinary event occurs during that time, the EDCs will determine a revised statewide maximum starting price, a revised statewide minimum starting price, a revised minimum indicative offer, and may also determine a revised schedule. Board Staff and the Board Consultant will review these decisions. New indicative offers will be required from bidders. The determination of new maximum and minimum starting prices, the revision to the minimum number of tranches in the indicative offer, the submission of new indicative offers, and if necessary the announcement of new starting prices, will be carried out so as to afford bidders sufficient time.

VIII.C. Bidding Rules

We first present an overview of the Auction format. We then proceed to explain the bidding and other rules in detail.

VIII.C.1. Overview of Auction Format

The Auction is a simultaneous, multiple round, descending clock auction. We can explain this format's features by simply "unpacking" this terminology.

The Auction is called simultaneous because tranches for all the EDCs are put on offer through the same auction. The Auction proceeds in rounds. In a round, the Auction Manager announces a price for each EDC. Bidders bid by providing the number of tranches that they are willing to serve for each EDC at the prices announced by the Auction Manager. If the number of tranches bid is greater than number of tranches needed for an EDC, the price for that EDC is reduced for the next round. In the next round, bidders are given an opportunity to bid again.

The Auction is called a descending clock auction because prices "tick down" throughout the Auction, starting high and being reduced gradually until the supply bid is just sufficient to meet the load to be procured. Prices that tick down in a round decrease by a decrement; a decrement is a given percentage of the previous price. Bidders holding the final bids when the Auction closes are the winners.

Example 3.

There are 11 bidders in the Auction. Suppose that the statewide load cap is 18 tranches. Consider the following sample round.

ROUND 1					
EDC	Price (\$/MW-day)	# Tranches bid	# Tranches desired	Excess supply	Oversupply ratio
PSE&G	475.00	52	24	28	0.700
JCP&L	475.00	11	11	0	0.000
ACE	475.00	6	4	2	0.050
RECO	475.00	3	1	2	0.200

The Auction Manager reduces the price for an EDC if the number of tranches bid is greater than the number of tranches desired. The amount of the price reduction depends on the oversupply ratio, which is the ratio of the excess supply on that EDC to an estimate of the maximum possible excess supply on that EDC, taking into account the total excess supply in the Auction. Roughly speaking, the larger the oversupply ratio for an EDC, the larger is the portion of maximum excess supply that is actually on that EDC, and the larger is the price decrease. A formula for the oversupply ratio is provided in section VIII.G.2.

In round 1, all bidders combined stand ready to supply 52 tranches of PSE&G at a price of \$475.00/MW-day. The number of tranches bid (52) exceeds the number of tranches desired (24) by 28 tranches. The price for PSE&G will tick down.

The actual excess supply on PSE&G is 28 tranches. The maximum possible excess supply is the total excess supply for all the EDCs in the Auction, or 32 (28 + 0 + 2 + 2). If all the excess bids in the Auction had been bid on PSE&G, PSE&G would have 32 tranches of excess supply. The estimate of the maximum excess supply used to calculate the oversupply ratio is the upper bound of the range of total excess supply at the Auction reported to bidders, which is 40. Thus, the oversupply ratio for PSE&G is 0.700 (28 / 40). Roughly speaking, 70% of the excess supply in the Auction is bid on PSE&G.

The calculation for RECO is similar. In round 1, all bidders combined stand ready to supply 3 tranches of RECO at a price of \$475.00/MW-day. The number of tranches bid (3) exceeds the number of tranches desired (1) by 2 tranches. The price for RECO will tick down.

The actual excess supply on RECO is 2 tranches. For RECO, the maximum possible excess supply is not the total excess supply for all the EDCs in the Auction (32, calculated above). Indeed, with only 11 bidders in the Auction and one tranche of RECO desired, the maximum possible number of tranches bid on RECO is 11 (11 x 1). The maximum possible excess supply on RECO is 10(11-1). For RECO, the oversupply ratio is 0.200(2/10).

The oversupply ratio is calculated in the same way for all other EDCs.

The Auction Manager will lower the price in round 2 for PSE&G, ACE, and RECO since for these EDCs the number of tranches bid exceeds the number of tranches needed. PSE&G, which has the largest oversupply ratio, will have the largest decrement.

In round 2 below, the price for PSE&G, which had the largest decrement from round 1, fell the most. Bidders submit new bids at these prices. The total excess supply range reported to bidders is 21-30 (so that 30 is used as the measure of total excess supply in calculating the oversupply ratio).

ROUND 2					
EDC	Price (\$/MW-day)	# Tranches bid	# Tranches desired	Excess supply	Oversupply ratio
PSE&G	451.25	33	24	9	0.300
JCP&L	475.00	18	11	7	0.233
ACE	460.75	12	4	8	0.267
RECO	460.75	2	1	1	0.100

Each EDC has more tranches bid than tranches desired. The Auction Manager will calculate the decrement for each EDC from that EDC's oversupply ratio and lower the price of each EDC for the next round accordingly.

Although we describe the main points of the Auction Rules in more detail below, four aspects should be briefly highlighted at the outset. These are as follows:

- 1. Winners for each EDC are not determined until the bidding has closed for all EDCs. When the number of tranches bid in a round for an EDC does not exceed the number of tranches desired, the price for that EDC will not tick down for the next round. However, as the Auction progresses and the prices for the other EDCs tick down, some bidders may re-assign tranches and increase the number of tranches bid on that EDC, which may cause the price for the EDC to tick down again. Hence, the winners cannot be determined for any one EDC until bidding stops for all EDCs. There is no matching of suppliers to customers during the course of the Auction or selection of winners by the Auction Manager. It is only at the end of the Auction that suppliers reveal themselves to be winners by not withdrawing from the Auction.
- 2. If a bidder bid on an EDC in the preceding round and the price for the EDC's tranches did not tick down for the current round, the bidder cannot reduce the number of tranches bid for that EDC in the current round, either through a withdrawal or a switch (these terms are defined in sections VIII.C.4.b and VIII.C.4.c). Any bid is a binding obligation to accept the BGS-CIEP supply responsibility at the price at which the bid was made for a one-year term. The bidder may be able to reduce the number of tranches bid on that EDC later in the Auction. If other bidders increase their number of tranches bid on the EDC (because its price has remained high relative to the prices for other EDCs) and cause the price for that EDC to tick down, the bidder will once again be able to reduce the number of tranches bid on that EDC.
- 3. Bidders can never increase the total number of tranches they bid during the Auction. If a bidder does not bid a tranche in the first round, that tranche cannot be bid later on. Once a tranche is withdrawn, it can never be bid again.
- 4. All bidders that win tranches for a particular EDC and are authorized as suppliers by the Board receive the same price per MW-day as well as all other payments specified in the BGS-CIEP Supplier Master Agreement.

VIII.C.2. Round Phases and Bidding Day

Each round of the Auction is divided into three phases: a bidding phase, a calculating phase, and a reporting phase.

In the bidding phase of the round, bidders place bids. To be valid, a bid must be submitted and verified in the bidding phase and processed by the Auction software. Bidders should allow time within the bidding phase of the round for submission, verification, and confirmation of the bid by the Auction software. The time-stamp of a bid is the time at which the bid is confirmed. A bidder that submits a bid in a round may change this bid as long as the bidding phase of the round is still open. The last valid bid by the bidder in the round becomes a firm offer to supply that cannot be rescinded.

In the calculating phase of the round, the Auction Manager tabulates the results of that round's bidding phase and calculates the prices for the next round. During this phase, bidders cannot submit bids and bidders do not yet have access to the results from that round's bidding phase.

In the reporting phase of the round, the Auction Manager informs bidders that are still in a position to win at the Auction of the results of that round's bidding phase. All Auction results are confidential as explained below in section VIII.E.3. Bidders are informed of the *going prices* for the next round's bidding phase and are provided with a range of total excess supply left in the Auction. The going prices in a round are the prices at which the Auction Manager solicits bids in the bidding phase of that round. (Total excess supply is defined more precisely in section VIII.C.5 and information provided concerning total excess supply is specified in more detail in section VIII.C.7 below.) A bidder receives no information regarding another bidder's bid. Each bidder privately receives the results of its own bid from that round, indicating to each bidder its obligation at this point in the Auction.

A typical schedule for a bidding day will have a number of rounds in a morning session, a lunch break, and then a similar number of rounds scheduled in an afternoon session. The round times will speed up over the course of the Auction as bidders become more familiar with the process and bidding becomes more routine.

VIII.C.3. Round 1 of the Auction

VIII.C.3.a. Definition of a Bid

A bidder selects how many tranches it wants to serve for each EDC at the round 1 prices. In round 1, a *bid* then consists of four numbers:

- a number of tranches that the bidder wants to supply for PSE&G at the PSE&G round 1 price;
- a number of tranches that the bidder wants to supply for JCP&L at the JCP&L round 1 price;
- a number of tranches that the bidder wants to supply for ACE at the ACE round 1 price; and
- a number of tranches that the bidder wants to supply for RECO at the RECO round 1 price.

The EDCs are always ranked in decreasing order of the *tranche targets*; a tranche target is the number of tranches needed for a given EDC.

The number of tranches that a bidder chooses for one EDC may or may not be the same as the number of tranches that the bidder chooses for another EDC. A number of tranches is an integer (0,1,2,...). A number of 0 (zero) for one EDC means that at the round 1 price for the EDC the bidder does not want to supply any of the BGS-CIEP Load for that EDC.

Example 4.

example 4.							
Suppose that the round 1 prices are:							
EDC PSE&G JCP&L ACE RECO							
Price (\$/MW-day) 460.00 475.00 440.00 445.00							
Then (5, 0, 3, 1) is Bidder A's round 1 bid and it indicates that the bidder stands ready to supply 5 PSE&G tranches, no JCP&L tranches, 3 ACE tranches, and 1 RECO tranche at the round 1 prices.							

1				
EDC	PSE&G	JCP&L	ACE	RECO
Price (\$/MW-day)	460.00	475.00	440.00	445.00
Bid (tranches)	5	0	3	1

VIII.C.3.b. Bidding Phase

The Auction Manager informs bidders of the starting prices for each EDC three days prior to the Auction; these starting prices are the prices in force, or the going prices, for round 1

of the Auction. The going prices in a round are the prices at which the Auction Manager solicits bids in the bidding phase of that round.

Any bid submitted in round 1 must satisfy the condition that the total number of tranches bid by a bidder cannot exceed the bidder's initial eligibility. As explained in section VIII.B.2, the bidder's initial eligibility is equal to the number of tranches in the bidder's indicative offer at the maximum starting price. As the bidder's indicative offer cannot exceed the statewide load cap, the bidder's bid automatically satisfies the statewide load cap.

Example 5.

Suppose that the statewide load cap is 18 tranches. Bidder A submits an indicative offer of 10 tranches at the maximum starting price. Bidder A's initial eligibility is 10 tranches.

Bidder A's bid in round 1 is (5, 0, 3, 1).

- Bidder A is bidding on 9 tranches, which does not exceed Bidder A's initial eligibility of 10 tranches.
- Bidder A is bidding on a number of tranches that does not exceed the statewide load cap of 18 tranches.

As explained further in section VIII.C.9, the bidding phase in round 1 is automatically extended for the convenience of bidders. If a bidder requests an extension in round 1, it will run concurrently with the automatic extension.

VIII.C.3.c. Calculating Phase, Reporting Phase and Potential Volume Cutback

The calculating phase of round 1 immediately follows the extended bidding phase. In the ordinary course of events, the Auction Manager reviews the results and sets the prices that will be in force in round 2 of the Auction. Round 1 moves to the reporting phase and the Auction Manager reports to bidders the results of bidding in round 1 as well as the round 2 prices. The price for an EDC is the same as the round 1 price when the number of tranches bid in round 1 is insufficient to fill the tranche target for the EDC, or when the number of tranches bid for the EDC is just equal to the tranche target for the EDC. The price for an EDC ticks down when the number of tranches bid on that EDC exceeds its tranche target. Details on the amount by which the price in each round is reduced are given in section VIII.G.2. The Auction Manager also provides to bidders an indication of the total excess supply in the Auction in

round 1, as explained in section VIII.C.7. A bidder is not provided any information regarding any other individual bidder's bids.

The Auction Manager may call a pause in the Auction during the calculating phase of round 1. This pause is called a *time-out* (see section VIII.C.9 for additional details on time-outs). The Auction Manager will call this time-out if the Auction Manager needs to consider whether to cut back the Auction volume to ensure the competitiveness of the Auction. It is not expected that the Auction Manager would revise the statewide load cap, but the Auction Manager retains the discretion to make such a revision based on the revised Auction volume, and will do so when the reduced Auction volume falls below the statewide load cap. As soon as practicable during the time-out, the Auction Manager will either announce that the Auction volume will not change, or will announce the revised Auction volume, the revised tranche target for each EDC, and the revised statewide load cap if applicable.

As soon as practicable during the time-out, the Auction Manager will start the reporting phase of round 1. The Auction Manager will report to the bidders the prices in force for round 2 as well as an indication of the total excess supply in the Auction in round 1. (Total excess supply is defined in section VIII.C.5.) The manner in which the tranche targets for the EDCs will be changed on the basis of the revised Auction volume will be announced no later than 10 days before the Part 2 Application is due. If the statewide load cap is reduced because it exceeds the reduced Auction volume, the Auction Manager reports to a bidder whose eligibility is greater than the reduced statewide load cap that the bidder's bid has been adjusted to conform to the reduced statewide load cap. The bidder withdraws any excess eligibility in round 2.

If the Auction volume is cut back, each EDC will implement a *Contingency Plan* for its tranches that have been removed from the Auction. Under their Contingency Plans, the EDCs will purchase necessary services to serve BGS-CIEP Load, including installed capacity, energy, transmission, and ancillary services through PJM-administered markets. Pursuant to these Contingency Plans, which are described in more detail in the Company Specific Addenda, the EDCs will not negotiate contracts with suppliers for BGS-CIEP Load after the Auction.

The Auction Manager will use a confidential set of guidelines approved by the Board to decide whether to cut back the Auction volume and to determine the magnitude of any necessary cutback. If the Auction volume is cut back, it will be cut back to the number of tranches bid in round 1 divided by a parameter called the *target eligibility ratio* (a desired ratio of tranches bid to the Auction volume). The precise value of this parameter depends on various factors, such as the number of bidders and characteristics of individual bids.

Example 6.

In the bidding phase of round 1, the Auction volume is 40 tranches (this is the sum of the tranche targets for all EDCs in the Auction: 24 + 11 + 4 + 1 = 40). Suppose that a total of 54 tranches are bid. Further, suppose that given the number of bidders and the characteristics of the bids, the Auction Manager sets the target eligibility ratio at 1.5.

The actual eligibility ratio is approximately 1.35 (54/40). The Auction Manager reduces the Auction volume to attain the target eligibility ratio of 1.5 by cutting back the volume by 4 tranches to 36 tranches (54/1.5 = 36).

After the volume cutback, there will be 1.5 tranches bid for every tranche to be purchased through the Auction (54/36 = 1.5).

The Auction Manager may further cut back the Auction volume on the basis of the bids as the Auction progresses, in accordance with the confidential guidelines approved by the Board. If such a revision is necessary to ensure a competitive bidding environment, the Auction Manager will call a time-out during the calculating phase of a round. As soon as practicable during the time-out, the Auction Manager will advise the bidders of the revised Auction volume, the revised tranche target for each EDC, and the revised statewide load cap (if applicable).

No later than three days before the start of the Auction, the Auction Manager may release further information regarding the possible values of the target eligibility ratio and the circumstances under which a second volume cutback may be undertaken.

VIII.C.4. Bidding in Round 2

VIII.C.4.a. Bid and Eligibility

A bidder selects how many tranches to serve from each EDC at the round 2 prices.

As stated in the introduction to the Auction format in section VIII.C.1, a bidder cannot increase its total number of tranches bid at the round 2 prices from its total number of tranches bid in round 1. This is more generally expressed by saying that the bidder cannot exceed its

eligibility. A bidder's eligibility in round 2 is the bidder's total number of tranches bid⁴ in round 1.

A bidder can always select the same number of tranches for each EDC in round 2 as in round 1. A bidder that wishes to change the number of tranches bid on a given EDC can reduce the number of tranches bid as long as the price for that EDC has ticked down. Such *reductions* in the number of tranches bid on an EDC for which the price has ticked down can be in the form of a *withdrawal* or a *switch*. A bidder reduces the number of tranches bid on an EDC through a withdrawal when the bidder is reducing the number of tranches bid on an EDC, not increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total. A bidder reduces the number of tranches bid on an EDC through a switch when the bidder is re-assigning the total number of tranches bid on other EDCs is increased by the same amount. If a bidder re-assigns tranches, the total number of tranches that the bidder bids on at least one EDC will increase, but the total number of tranches bid on all EDCs cannot increase. This is explained in more detail below in sections VIII.C.4.b and VIII.C.4.c.

In addition to specifying the number of tranches that a bidder is prepared to serve for each EDC, fully specifying a bid in round 2 may require the bidder to provide *switching priorities* or *exit prices* (defined below). A bidder is required to provide switching priorities when increasing the total number of tranches bid on two or more EDCs, as explained in section VIII.C.4.c. A bidder is required to provide exit prices if a bidder is reducing the number of tranches bid on an EDC through a withdrawal, as explained in section VIII.C.4.b.

In reporting to the bidder the result of its bid in the reporting phase, the Auction Manager will always report that it accepts a bid that is unchanged. The Auction Manager may disallow reductions that a bidder wants to make from an EDC. The Auction Manager retains withdrawn tranches if, by accepting all reductions from withdrawals and switches, the tranche target for that EDC would no longer be filled. The Auction Manager denies reductions from switches if, after retaining all withdrawn tranches from that EDC, accepting all reductions from

⁴ Given that the bidder's round 1 bid satisfied the statewide load cap, and given that the bidder cannot increase its total number of tranches bid, the bidder's bid will satisfy the statewide load cap in all subsequent rounds.

switches would prevent the tranche target for that EDC from being filled. To fill the tranche target of an EDC, the Auction Manager first takes tranches that are bid at the round 2 price, then retains tranches that bidders want to withdraw, and finally denies reductions from switches that bidders have requested, as necessary. As explained in more detail below, the Auction Manager retains tranches that a bidder has withdrawn at the exit price named by the bidder and keeps switches that must be denied at the last price at which the bidder freely bid these tranches.

VIII.C.4.b. Reductions through Withdrawals

A bidder can withdraw tranches from an EDC only if the price for that EDC has decreased from round 1 to round 2. If, instead, the price for an EDC has not changed from round 1 to round 2, then the bidder's offer in round 1 at that price is still standing, and the bidder cannot bid on fewer tranches for that EDC at the (unchanged) going price. As explained below, a bidder can always bid more tranches for an EDC whose price has not changed from round 1 by reducing the number of tranches from other EDCs whose prices have ticked down and switching them to the EDC whose price has not ticked down.

A bidder that withdraws one or more tranches from an EDC must name an exit price for those tranches. A bidder names an exit price when it bids one or more tranches at the previous going price but is unwilling to bid these tranches at the current going price. A bidder that withdraws several tranches previously bid at the round 1 price for a given EDC must specify the same exit price for all tranches from that EDC. An exit price must be less than or equal to the last price at which the tranches were freely bid (in round 2, this is the price in round 1) and must be higher than the EDC's going price (in round 2, this is the round 2 price, a price at which the bidder is no longer willing to bid the tranches being withdrawn). A bidder that withdraws tranches from more than one EDC can specify a different exit price for each EDC.

An exit price enables the Auction Manager to determine which bidder would have remained ready to serve an EDC had the price ticked down continuously rather than in lumpy, discrete decrements. The Auction Manager relies on exit prices when tranches are withdrawn from an EDC and when the number of tranches bid on that EDC at the round 2 price falls short of that EDC's tranche target due to reductions from withdrawals and possibly also due to

reductions from switches. The tranches with lower exit prices are retained first, and they are retained at the exit price that the bidder has named.

If, to fill the last tranches of the tranche target of an EDC, the Auction Manager must retain some but not all the tranches from two or more bidders that named the same exit price, then the Auction Manager, for each tranche to be retained, will choose at random the bidder whose tranche is retained. Thus, for the first tranche needed at the tied exit price, the probability that a particular bidder is chosen is the number of tranches that the bidder has bid at the exit price divided by the total number of tranches bid at the exit price. If a second tranche is needed at the exit price, the Auction Manager again will choose at random a bidder whose tranche will be retained. The probability that any one bidder is chosen is the number of tranches that the bidder has bid at the exit price and that have not yet been retained divided by the total number of tranches bid at the exit price and that have not yet been retained. The Auction Manager repeats this procedure until the tranche target for the EDC is filled.

A bidder that withdraws tranches from an EDC loses the eligibility associated with these tranches, and forfeits the right to bid these tranches for the remainder of the Auction on any EDC. Eligibility is lost even if the withdrawn tranches are retained. Eligibility is always lost as a result of withdrawing tranches.

Example 7.

Suppose that the round 1 prices and the round 1 bid for Bidder A are:						
EDC	PSE&G	JCP&L	ACE	RECO		
Round 1 Price (\$/MW-day)	460.00	475.00	440.00	445.00		
Bid (tranches)	5	0	4	1		
The round 2 prices are:						
EDC	PSE&G	JCP&L	ACE	RECO		
Round 2 Price (\$/MW-day)	460.00	460.75	426.80	445.00		

Bidder A cannot bid fewer tranches for PSE&G or RECO in round 2 because the prices for these EDCs have not changed from round 1.

Bidder A reduces the number of tranches bid for ACE from 4 to 2. The bidder enters an exit price of \$434.25/MW-day, which is between \$426.80/MW-day and \$440.00/MW-day. (The exit price could have been equal to \$440.00/MW-day, but not equal to \$426.80/MW-day.)

VIII.C.4.c. Reductions through Switches

When a bidder is switching and not withdrawing, a bidder is simultaneously decreasing the number of tranches bid for one or more EDCs and increasing the number of tranches for one or more EDCs while leaving the total number of tranches bid the same. As in the case when the bidder is reducing the number of tranches bid on an EDC because the bidder is withdrawing tranches, the bidder can reduce the number of tranches bid on an EDC by switching only if the price for that EDC has decreased in round 2. Again, if instead the price for an EDC has not ticked down, then the bidder cannot bid fewer tranches on that EDC (i.e., the bidder cannot withdraw tranches or switch tranches from that EDC).

Example 8.

Suppose that the following are the round 1 and round 2 prices and the bids for Bidder A.					
EDC PSE&G JCP&L ACE RECO					
Round 1 Price (\$/MW-day)	460.00	475.00	440.00	445.00	
Bid Round 1 (tranches) 5 0 4 1					
Round 2 Price (\$/MW-day)	460.00	460.75	426.80	445.00	
Bid Round 2 (tranches)	9	0	0	1	

In round 2, Bidder A is increasing by 4 the number of tranches bid on PSE&G. Also, Bidder A is reducing by 4 the number of tranches bid on ACE. Since the total number of tranches bid is the same (10), the bidder is switching. Bidder A can reduce the number of tranches bid on ACE since its price has decreased in round 2. Bidder A can bid for more tranches for PSE&G. If the price for an EDC does not change, a bidder cannot bid *fewer* tranches but can bid *more* tranches for that EDC.

When (and only when) a bidder is increasing the number of tranches bid at the going price on more than one EDC, the bidder must specify a *switching priority*. A switching priority is a rank assigned to each of the EDCs on which the bidder is increasing the number of tranches bid. The Auction Manager will use this switching priority only when accepting some, but not all, reductions involved in a switch that the bidder has requested. The Auction Manager will use this switching priority only when, to keep an EDC's tranche target filled, the Auction Manager must retain all tranches that were withdrawn out of that EDC and must deny some, but not all, reductions from that EDC that come from a bidder's switch. The Auction Manager first increases the number of tranches bid of the EDC to which the bidder has assigned the highest priority (1 is the highest priority). The switches that the Auction Manager denies to fill

the tranche target of an EDC are retained at the price at which they were last freely bid (in round 2, this is the round 1 price).

If a bidder intended to reduce the number of tranches bid on an EDC through a switch, some or all of these reductions may be denied. The Auction Manager will deny reductions from switches if the number of tranches bid at the going price, and the number of tranches that can be retained from withdrawals, are not together sufficient to fill the tranche target of an EDC. If the Auction Manager must deny one or several reductions from switches from two or more bidders, the Auction Manager chooses at random the bidder whose switch will be denied.

For each tranche of the target that must be filled by denying a reduction from a switch, the Auction Manager chooses at random the bidder whose switch is denied. Thus, for the first switch that must be denied, the probability that the Auction Manager chooses a tranche bid by a particular bidder requesting a switch is the number of tranches by which the bidder's bid on the EDC is reduced by the switch and that could be denied, divided by the total number of tranches by which the number of tranches bid on the EDC is reduced by switches from all bidders and that could be denied. If a second switch must be denied, the Auction Manager again chooses at random the bidder whose switch will be denied. The probability that the Auction Manager chooses a tranche bid by a bidder requesting a switch is the number of tranches by which the bidder's bid on the EDC is reduced by the switch and that could have, but have not yet been denied, divided by the total number of tranches by which the number of tranches bid on the EDC is reduced by all switches from bidders and that could have, but have not yet been denied. The Auction Manager repeats this procedure until the tranche target for the EDC is filled. The Auction Manager continues to report that some or all of these switches are being denied in subsequent rounds as long as they are still needed to fill the EDC's tranche target.

Example 9.

Bidder B submits the following bid in round 2 after this history in round 1:					
EDC	PSE&G	JCP&L	ACE	RECO	
Round 1 Price (\$/MW-day)	460.00	475.00	440.00	445.00	
Round 1 Bid (tranches)	2	7	2	1	
Round 2 Price (\$/MW-day)	460.00	460.75	426.80	445.00	
Round 2 Bid (tranches)	. 6	1	4	1	

In round 2, Bidder B is increasing the number of tranches bid on ACE and PSE&G while reducing the number of tranches bid on JCP&L. Since the total number of tranches bid is the same (12) in both rounds, the bidder is switching. Bidder B gives the switching priority to the increase on PSE&G.

If the Auction Manager will allow 2 of Bidder B's reductions from JCP&L but needs to deny the other 4, then the bidder will be allowed to increase its number of tranches bid for PSE&G by those 2 tranches because of the switching priority. The other 2 tranches by which Bidder B increased its number of tranches bid for PSE&G, and the 2 tranches by which Bidder B increased its number of tranches bid for ACE, will not be allowed. The 4 denied switches will remain JCP&L tranches with a price of \$475.00/MW-day, which is the last price at which they were freely bid.

EDC	PSE&G	JCP&L	ACE	RECO
Round 2 Price (\$/MWh)	460.00	460.75	426.80	445.00
Round 2 Accepted Bid (tranches)	4	1 @ 460.75	2	1
		4 @ 475.00		

VIII.C.4.d. Withdrawing and Switching Tranches Simultaneously

If a bidder is both switching and withdrawing, a bidder can reduce tranches from a particular EDC only if the price for that EDC has decreased in round 2. Additionally, if the bidder is switching and withdrawing, the bidder may be asked to specify which tranches are being withdrawn and which tranches are being switched. The bidder would be asked for this additional information, for instance, when the bidder is switching and withdrawing, and the bidder is reducing the number of tranches bid for more than one EDC. The tranche or tranches that the bidder specifies to be withdrawn are the tranche(s) for which the bidder will name an exit price.

Example 10.

Bidder C submits the following bid in round 2 after this history in round 1:					
EDC	PSE&G	JCP&L	ACE	RECO	
Round 1 Price (\$/MW-day)	460.00	475.00	440.00	445.00	
Round 1 Bid (tranches)	2	5	2	1	
Round 2 Price (\$/MW-day)	460.00	460.75	426.80	440.55	
Round 2 Bid (tranches)	4	2	3	0	

Bidder C bids a total of 10 tranches in round 1 and 9 tranches in round 2. Bidder C is withdrawing 1 tranche and will name an exit price for that tranche.

Since Bidder C reduces the number of tranches bid for both JCP&L and RECO, Bidder C's intentions are not clear unless the Auction Manager asks Bidder C for additional information. Indeed, it could be that Bidder C is withdrawing 1 tranche from JCP&L and switching the remaining tranches. Or it could be that Bidder C is withdrawing 1 tranche from RECO and switching the remaining tranches.

Bidder C is asked to select whether it is withdrawing a tranche from JCP&L or RECO. Bidder C selects that it is withdrawing a tranche from RECO and is asked to name an exit price for this tranche. Since Bidder C is increasing the number of tranches bid for more than one EDC (PSE&G and ACE), Bidder C is also asked for switching priorities. Bidder C assigns the first priority to ACE.

VIII.C.5. Calculating and Reporting Phases in Round 2

The calculating phase starts immediately after the bidding phase. Once the Auction Manager has tabulated and reviewed the results, the reporting phase begins. The Auction Manager informs all bidders of the round 3 price for each EDC. The Auction Manager provides to all bidders a range for the *total excess supply* in the Auction for round 2. The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus free eligibility, which is defined below in section VIII.C.6. The range of total excess supply reported to bidders will change as the Auction progresses. When total excess supply is high, this range will count as few as 5 integers; as total excess supply decreases, the range will become wider and count 10 or more integers. The exact ranges of total excess supply provided as the Auction progresses are specified in detail in section VIII.C.7.

In addition to what the Auction Manager tells all bidders about the general progress of the Auction, the Auction Manager reports privately to each bidder the outcome of its own bid. (The Auction Manager never reports the bid of one bidder to another bidder.) If the round 2 bid is unchanged from round 1, or if requested reductions (from switches or withdrawals) all have been granted, the Auction Manager reports the bid made in round 2.

If the bidder intended to reduce the number of tranches bid on an EDC by withdrawing tranches and some or all of these tranches are retained, the Auction Manager informs the bidder of the number of withdrawn tranches that are being retained and the price at which these tranches are retained. This can occur only if the EDC's number of tranches bid in round 1 exceeded its tranche target while the number of tranches bid at the round 2 price is not sufficient to fill its tranche target. The price at which the withdrawn tranches are retained is the exit price. The Auction Manager will continue to report that some or all of these tranches are being retained in subsequent rounds as long as they are needed to fill the EDC's tranche target.

If a bidder intended to reduce the number of tranches bid on an EDC through a switch, but if some or all of these reductions are denied, the Auction Manager informs the bidder of the number of tranches for which the switch is denied. The price at which a switch is denied is the last price at which the tranche was freely bid. In round 2, this price is the round 1 price. A switch can be denied only if, for the EDC from which the number of tranches bid is reduced, the number of tranches bid in round 1 exceeded its tranche target, and the number of tranches bid at the round 2 price plus the retention of all the withdrawals from the EDC are not sufficient to fill the tranche target.

Example 11.

Bidders A, B, and C are the only bidders bidding for tranches of JCP&L, for which the tranche target is 11. None of these bidders are bidding on RECO. Their bids in rounds 1 and 2 are:

EDC	PSE&G	JCP&L	ACE
Tranche Target	24	11	4
Round 1 Price (\$/MW-day)	460.00	475.00	440.00
Bidder A Bid (tranches)	0	4	0
Bidder B Bid (tranches)	0	4	0
Bidder C Bid (tranches)	0	4	0
Round 2 Price (\$/MW-day)	460.00	460.75	440.00
Bidder A Bid (tranches)	0	3	1
Bidder B Bid (tranches)	1	2	1
Bidder C Bid (tranches)	0	4	0

Bidder A switches, increasing the number of tranches bid for ACE to 1 and reducing the number of tranches bid on JCP&L. Bidder B is also switching. Bidder C's bid is unchanged from round 1 to round 2. In total, three fewer tranches are bid on JCP&L. The number of

tranches bid on PSE&G and ACE are both increased. Bidder B gives ACE the switching priority.

No other bidders bid on tranches for JCP&L in round 1 or in round 2. In round 1, JCP&L's tranche target is more than filled with 12 tranches bid at the round 1 price. In round 2, the 9 tranches bid at the round 2 price are insufficient to fill the tranche target. The Auction Manager must deny 2 of the intended switches.

Two of the three reductions from JCP&L must be denied. For the first tranche that must be filled by denying a switch, the probability that each bidder is chosen is 1/3 for Bidder A and 2/3 for Bidder B. Bidder B is chosen at random. The Auction Manager repeats the procedure for the second tranche that must be filled by denying a switch. The probability that each bidder is chosen is 1/2 for Bidder A and 1/2 for Bidder B. Bidder A is chosen at random. This means that the Auction Manager denies Bidder A its intended switch and that Bidder B is allowed to switch 1 of the 2 tranches requested. Bidder B gave ACE the switching priority. Adding the 2 denied switches to the 9 tranches bid at the round 2 price fills JCP&L's tranche target.

EDC	PSE&G	JCP&L	ACE
Round 1 Price (\$/MW-day)	460.00	475.00	440.00
Round 2 Price (\$/MW-day)	460.00	460.75	440.00
Report to Bidder A	0	3 @ \$460.75/MW-day 1 denied switch @ \$475.00/MW-day	0
Report to Bidder B	0	2 @ \$460.75/MW-day 1 denied switch @ \$475.00/MW-day	1 @ \$440.00/MW- day
Report to Bidder C	0	4 @ \$460.75/MW-day	0 -

The Auction Manager reports to Bidder A that its switch was denied. The Auction Manager also reports to Bidder A that it has 3 JCP&L tranches bid at the round 2 price of \$460.75/MW-day and 1 denied switch, a JCP&L tranche at the round 1 price of \$475.00/MW-day. The Auction Manager reports to Bidder B that Bidder B has 2 tranches of JCP&L bid at the round 2 price of \$460.75/MW-day and 1 denied switch, a JCP&L tranche bid at the round 1 price of \$475.00/MW-day. The Auction Manager also reports to Bidder B that Bidder B has 1 ACE tranche bid at \$440.00/MW-day (because of the switching priority). The Auction Manager reports to Bidder C that Bidder C has 4 tranches of JCP&L bid at the round 2 price.

VIII.C.6. Round 3 and All Subsequent Rounds

Rounds start with a bidding phase. When an EDC's price in a round has ticked down from the previous round, a bidder is free to maintain its number of tranches bid on the EDC or reduce its number of tranches bid on the EDC. Also, the bidder can increase its number of tranches bid on an EDC if the bidder has tranches bid on another EDC for which the price has ticked down. When an EDC's price has not ticked down from the previous round, a bidder must maintain its number of tranches bid on the EDC, or the bidder can increase its number of

tranches bid on the EDC (subject to the condition just noted). To fully specify a bid, in certain circumstances the bidder will be required to provide exit prices and switching priorities as explained in sections VIII.C.4.b and VIII.C.4.c.

At all times, the total number of tranches bid by a bidder cannot exceed the bidder's eligibility. The bidder's eligibility for a round is its eligibility at the start of the previous round minus the number of tranches that the bidder withdrew from the Auction in the bidding phase of the previous round. The total number of tranches bid by a bidder includes tranches bid at the going prices and denied switches that are kept at the last price at which they were freely bid.

In the reporting phase that follows the bidding phase and the calculating phase, the Auction Manager reports to all bidders a range for the total excess supply left in the Auction, as described in section VIII.C.7. The Auction Manager reports privately to a bidder on its bid. The Auction Manager reports any withdrawn tranches that are retained and any switches that are denied along with their associated prices, as described above in sections VIII.C.4.b and VIII.C.4.c. If a switch that had been denied in a previous round is freed to be bid once again, then the Auction Manager will report the number of tranches that are being freed.

To fill the tranche target for an EDC, the Auction Manager takes bids in increasing order of price. The Auction Manager first takes tranches bid at the current round price, then the Auction Manager retains withdrawn tranches, and finally the Auction Manager denies switches. Thus, when new tranches are bid at the current round price for an EDC from which switches had been denied, the new tranches replace the denied switches (partially or completely) in filling the EDC's tranche target, and therefore free up some or all of these denied switches. A denied switch that is being replaced by a new tranche at the going price is then said to be *outbid*. If switches from more than one bidder are retained, and if not all denied switches are outbid, the Auction Manager chooses at random, for each denied switch that will be outbid, the bidder whose switch will be outbid. Thus, for the first denied switch that is outbid, the probability that the Auction Manager chooses a particular bidder's denied switch is the bidder's number of denied switch must be outbid, the Auction Manager again will choose at random the bidder whose denied switch will be outbid. The probability that the Auction Manager chooses a bidder's denied switch is the bidder's number of denied switches that have not yet been

outbid divided by the total number of denied switches that have not yet been outbid. The Auction Manager repeats this procedure until the required number of denied switches has been outbid.

Once some denied switches are outbid for a bidder, the Auction Manager reports this fact to the bidder, along with the number of tranches that have been outbid. A denied switch that is outbid becomes *free eligibility* for the bidder in the next round. A tranche of free eligibility can be bid on any one of the EDCs in the next round, or it can be withdrawn in the next round, as explained further below.

The Auction Manager reports privately to a bidder if a withdrawn tranche that had been retained is now being released and thereby irrevocably removed from the Auction. Withdrawn tranches that had been retained at the highest exit prices are the first to be released and removed from the Auction. If withdrawn tranches from more than one bidder had been retained at the same exit price, and if not all retained tranches at that exit price are being released, the Auction Manager chooses at random the bidder or bidders whose tranches are released and thereby irrevocably removed from the Auction. Thus, for the first retained tranche that should be released, the probability that a particular bidder is chosen is the bidder's number of retained tranches at the tied exit price divided by the total number of retained tranches at that exit price. If a second retained tranche needs to be released, the Auction Manager again will choose at random the bidder whose retained tranche will be released, and the probability that a particular bidder is chosen is the bidder's number of retained tranches at the tied exit price that have not yet been released. The Auction Manager repeats this procedure until the required number of tranches has been released.

Two additional bidding rules come into play when a bidder has been denied a switch.

First, if a bidder holds tranches for an EDC from a switch that had been denied in a previous round, and if this bidder bids new tranches for this same EDC at the price for the current round, the bidder will be deemed to have bid any tranches from a previously denied switch at the current round price for that EDC. That is, tranches from the denied switch become tranches that are bid at the price for the current round. The Auction Manager, in filling the tranche target for the EDC, will take first tranches bid at the going price; in these tranches

at the going price, the Auction Manager will include any denied switches that have become tranches bid at the current round price (because the bidder has bid new tranches for this same EDC at the current round price).

Second, if a bidder's denied switches are partially or totally outbid in a round and become free eligibility, and if the bidder withdraws one or more of these tranches of free eligibility in the very next round, then the bidder does not name an exit price for these tranches. No exit price is needed to withdraw a tranche of free eligibility because such tranches are not associated with any one EDC. The Auction Manager always accepts a withdrawal of tranches of free eligibility, and these tranches are always irrevocably removed from the Auction.

Example 12.

Bidder A's bids in rounds 6 and 7 are given below (Bidder A does not bid for any other EDCs):				
EDC	JCP&L	ACE		
Round 6 Price (\$/MW-day)	443.33	407.89		
Bidder A Bid (tranches)	2	3		
Round 7 Price (\$/MW-day)	430.03	395.65		
Bidder A Bid (tranches)	5	0		

In round 7, Bidder A requests to switch 3 tranches. The Auction Manager denies part of the switch. Bidder A, in the reporting phase of round 7, is informed that its bid consists of 3 tranches of JCP&L at \$430.03/MW-day and 2 denied switches of ACE at \$407.89/MW-day (the price at which the tranches of ACE were last freely bid).

EDC	JCP&L	ACE
Report to Bidder A	3 @ \$430.03/MW-day	2 denied switches @ \$407.89/MW-day

In round 8, Bidder A reduces its number of tranches bid on JCP&L by 2 and increases its number of tranches bid on ACE. At the round 8 prices, Bidder A bids 1 tranche of JCP&L and 2 tranches of ACE at the going prices. The denied switches are kept on ACE and cannot be freely bid.

EDC	JCP&L	ACE
Round 8 Price (\$/MW-day)	422.50	395.65
Bidder A Bid (tranches)	1	2
Denied Switches		2 @ \$407.89/MW-day

Bidder A has bid new tranches of ACE at the round 8 price while having switches denied on the same EDC at a higher price. Bidder A is then deemed to have bid all 4 ACE tranches at the round 8 price of \$395.65/MW-day as shown below. All 4 tranches of ACE become tranches bid at the round 8 price.

EDC	JCP&L	ACE
Round 8 Price (\$/MW-day)	422.50	395.65
Bidder A Bid (tranches)	1	4 @ \$395.65/MW-day

VIII.C.7. Reporting of Total Excess Supply

In addition to the next round prices, the Auction Manager provides all bidders that could still be winners with a range for total excess supply in the Auction. The actual total excess supply in the Auction lies within the reported range. The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus all tranches of free eligibility.

The manner in which the total excess supply in the Auction is reported changes as the Auction progresses. In general, when total excess supply is high, the Auction Manager announces a range for the total excess supply that includes exactly 5 integers. The largest integer of each such range will be divisible by 5. When the total excess supply falls to 40 or fewer tranches, the Auction Manager announces a range for the total excess supply that includes at least 10 integers. The largest integer of each such range will be divisible by 5. When the total excess supply falls to 20 or fewer tranches, the Auction Manager will report to all bidders only that the total excess supply is between 0 and 20 tranches.

In summary, the ranges are expected to be the following:

Table VIII-3. Possible Ranges of Total Excess Supply

Total excess supply falls to 20 tranches or fewer: a single range remains	0-20
Total excess supply is between 21 and 40 tranches:	21 – 30
ranges count 10 integers	31 – 40
	41 – 45
Total excess supply is 41 tranches or more: ranges count 5 integers	46 – 50
	51 – 55 (etc.)

The Auction Manager may revise these ranges before the start of the Auction. The Auction Manager will inform the registered bidders no later than 3 days after registration of any changes in the possible ranges of total excess supply.

When the actual total excess supply is zero, the Auction ends, as explained in section VIII.C.11.

A bidder that has no remaining obligation loses its access to the Auction software and is no longer provided with the range of total excess supply and the next round prices. A bidder has no remaining obligation when the bidder has zero eligibility and has no retained withdrawals. A bidder that has no remaining obligation can no longer win at the Auction. Such a bidder loses its access to the Auction software and is no longer provided with the range of total excess supply and the next round prices no earlier than the round after the bidder has been first informed that it has no remaining obligation. The bidder will lose its access as soon as practicable, but in no event later than eight rounds after it has no remaining obligation.

Round results that are reported to all bidders will also be reported to a list of representatives from the EDCs, the Board and the Board Consultant. The bidders, the EDCs, the Auction Manager, the Board and the Board Consultant hold any Auction results to which they have access to be confidential. Before being registered to participate in the Auction, the bidders will agree to keep all Auction results, as well as the list of qualified bidders, the list of registered bidders, and the total initial eligibility in the Auction confidential. The bidders will agree not to disclose any such confidential information about the Auction Process, except for any aspects of the Auction results that the Board releases as part of rendering its decision on the Auction results, or that the Board explicitly authorizes can be released (see section VIII.E.3). Bidders will also agree to destroy documents with Auction information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results.

VIII.C.8. Calculation of Next Round Prices

An EDC's price in a round is calculated so that it ticks down more quickly the larger is the excess supply on the EDC. The excess supply on an EDC is measured against an estimate of the maximum possible excess supply for the EDC. This estimate takes into account the total excess supply in the Auction, the statewide load cap, the tranche target of each EDC, and the number of registered bidders. A decrease in price for an EDC that has a number of tranches bid greater than the tranche target is called the decrement; the decrement is a percentage of the previous going price. The decrements are calculated using formulas. The Auction Manager has the discretion to override the calculated decrement for any EDC in any round. The Auction Manager does not expect to use such discretion, but may do so. When the Auction Manager, in the reporting phase of a round, uses discretion to override the decrement formulas and

determine the going prices for the next round, the Auction Manager notifies bidders of that fact during the reporting phase of that round.

The decrement formulas allow the decrements to be larger at the start of the Auction than in later rounds. At the start of the Auction, when the number of tranches bid on an EDC exceeds the tranche target, the decrement is between 0.5% and 5% of the previous round price. Decrements for EDCs for which the number of tranches bid is greater than the tranche target continue to be between 0.5% and 5% of the previous round price until the going prices for round 4 are calculated. After that time, in the first round in which the upper bound of the total excess supply range reported to bidders is at least 15 tranches fewer than the upper bound of the total excess supply range reported to bidders in round 1 and the total excess supply is greater than 20 tranches, decrements (for EDCs for which the number of tranches bid is greater than the tranche target) will be between 0.375% and 3.75% of the previous round price. Decrements for EDCs for which the number of tranches bid is greater than the tranche target continue to be between 0.375% and 3.75% of the previous round price until total excess supply in the Auction is 20 or fewer tranches. At that point, decrements for EDCs for which the number of tranches bid is greater than the tranche target will be between 0.25% and 2.5% of the previous round price and will remain so until the end of the Auction.

The excess supply of an EDC is the number of tranches bid at the going price minus the EDC's tranche target, or it is zero, whichever is greater. Prices are rounded off to the nearest cent per MW-day.

Provisional decrement formulas are provided in section VIII.G.2. The formulas that will be used in the Auction, which depend on the statewide load cap and the number of registered bidders, will be released to all bidders no later than three days after bidder registration.

VIII.C.9. Pauses in the Auction

Any one bidder can pause the Auction by requesting an extension or a recess. A bidder requests an extension during the bidding phase of a round; such a request extends the bidding phase of the round by 15 minutes for all bidders. A bidder requests a recess during the

calculating or reporting phase of the round; such a request suspends the Auction for a period of no less than 20 minutes. A recess allows a bidder additional time to consider its bid.

The bidding phase of round 1 is automatically extended. The bidding phase of a round can be extended only once by 15 minutes. Each bidder is allowed two extensions during the Auction. A bidder with positive eligibility is automatically deemed to have requested an extension when — by the scheduled ending time of the bidding phase — the bidder has not submitted a bid and when the bidder has not already used the two allowable extensions. A bidder with positive eligibility that has already used the two allowable extensions and that does not submit a bid during a round will be assigned a default bid as explained in section VIII.C.10. Extension requests from all bidders are granted but all extensions run concurrently. All bidders that have requested an extension during the bidding phase of a round will see their available number of extensions reduced. The bidding phase in a round will be extended only once by 15 minutes. The Auction Manager reports to all bidders at the end of the planned bidding phase that the bidding phase has been extended.

A recess must be requested during the calculating phase or during the reporting phase. If the reporting phase of a round is scheduled to be shorter than 5 minutes, a recess must be requested during the scheduled calculating phase of the round. If the reporting phase is scheduled to be 5 minutes or longer, a recess must be requested before the scheduled last half of the reporting phase of a round starts, or before the last five minutes of the reporting phase of a round, whichever is earlier. (That is, if the reporting phase of a round is scheduled to be 8 minutes, then the recess must be requested before the last 5 minutes of the reporting phase; if the reporting phase of a round is scheduled to be 14 minutes, then the recess must be requested before the last 7 minutes of the reporting phase.) As soon as is feasible after the time at which a recess can be requested has passed, the Auction Manager reports to all bidders that a recess has been called. The Auction Manager retains the discretion to set the length of a recess but the Auction Manager will not set the recess time to be less than 20 minutes.

Each bidder is allowed to request at most one recess during the Auction. A bidder cannot request a recess in the first ten rounds of the Auction. Starting in the eleventh round, a bidder may request a recess in a round only if the total excess supply in the Auction reported in the previous round was 20 or fewer tranches. All recess requests are granted, but all requested

recesses run concurrently. All bidders making a request in a given calculating or reporting phase will be deemed to have used a recess request. All bidders that have requested a recess will see their available number of recesses reduced to zero and the recess will last a length of time determined by the Auction Manager for one recess and will be no less than 20 minutes.

Example 13.

The total number of tranches in the Auction is 40.

In round 5, total excess supply is reported to be in the 76-80 range. Bidders are not able to request a recess in round 5 since round 11 has not yet been reached.

In round 12, the total excess supply is reported to be in the 0-20 range for the first time. Bidders can request a recess in round 13, since the total excess supply reported in the previous round (round 12) was at 20 tranches or below, and round 11 had been reached.

The Auction Manager can call a time-out to the Auction at any time during a round. It is intended that a time-out suspend activity in the Auction for a period of no more than four hours; however, the Auction Manager retains the discretion to suspend activity for a longer period, if necessary. Whenever a time-out is called, the Auction Manager reports to all bidders how long the time-out is expected to last.

During any bidding phase, the Auction Manager may call a time-out and extend bidding in case of technical difficulties. Such a time-out would typically last fifteen minutes. The Auction Manager reports to all bidders as soon as practicable that the Auction Manager has extended the bidding phase.

During the calculating phase of round 1, the Auction Manager may call a time-out to the Auction to evaluate whether the Auction volume should be adjusted. The Auction Manager has the discretion to call additional time-outs during the Auction. Such discretion could be used, for example, in case of an extraordinary event occurring during the Auction. The Auction Manager expects to exercise this discretion rarely.

VIII.C.10. Failure to Submit a Bid

A bidder with positive eligibility must submit a bid in every round (even when the bidder's bid does not change or the bidder is bidding only on EDCs whose prices have not ticked down). As explained in section VIII.C.9, if a bidder with positive eligibility does not submit a bid during the bidding phase of a round, the bidder is granted an extension whenever

possible. If the bidder has previously used both extensions, or if the bidder does not submit a bid during the extension to the bidding phase, then the bidder has failed to submit a bid.

When a bidder with positive eligibility has failed to submit a bid in a round, the bidder is assigned a *default bid*. A default bid is the number of tranches on each EDC equal to the minimum number of tranches that the bidder could have bid on each EDC, as explained below. Any tranches of free eligibility that could have been bid in the round are deemed to be withdrawn and are irrevocably removed from the Auction. Any tranche that was bid on an EDC and that could be withdrawn is deemed to be withdrawn at the highest possible exit price for the round, which is equal to the going price in the previous round. A tranche that is withdrawn as a result of a default bid can be retained only if the number of tranches bid by other bidders at the going price, and the number of withdrawn tranches from bidders that have submitted a bid, together are not sufficient to fill the tranche target. In breaking any ties between tranches withdrawn at the same exit price, default bids will be disadvantaged compared to bids actually submitted by bidders. In general, in any circumstance where a tie must be broken, default bids will be disadvantaged compared to bids actually submitted by bidders, also as explained below.

As a consequence of a bidder being assigned a default bid, a bidder may lose the ability to submit bids for the remainder of the Auction. For example, if the prices of all the EDCs for which a bidder bid tranches tick down, then the default bid is zero on all EDCs. Indeed, the minimum number of tranches that can be bid on an EDC whose price has ticked down is zero. Any tranche that had previously been bid is withdrawn. After such a default bid, the bidder will be assigned zero eligibility in the next round, and the bidder will lose the ability to bid in the next and in all future rounds.

The default bid assigned to a bidder that has not submitted a bid during a round or during its extension, and the treatment of this default bid in any required tie-breaking procedure, are determined during the calculating phase of a round. The default bid for a bidder in round 1 is 0 tranches on each and every EDC. The default bid for a bidder in round 2 and all subsequent rounds is described in detail as follows.

If the bidder has some tranches of free eligibility in the current round, these tranches are deemed to be withdrawn and are irrevocably removed from the Auction.

If, in the previous round, a bidder did not bid any tranches on an EDC at the going price and in the reporting phase of that round the Auction Manager reported that the bidder did not have any retained withdrawals or denied switches for that EDC, then the bidder is assigned zero tranches for that EDC.

If, as of the reporting phase in the previous round, a bidder had tranches on a particular EDC at the going price for the previous round and if the EDC's price ticked down from the previous round to the current round, then the bidder is deemed to have withdrawn all tranches at the highest exit price, namely the price from the previous round. The bidder loses the eligibility associated with these tranches. Tranches with a lower exit price named by bidders that have submitted a bid in the current round are retained first. If needed, tranches with the same exit price named by bidders that have submitted a bid in the current round are retained next. If all the withdrawn tranches by the bidder and by other bidders that were assigned a default bid are needed to fill the tranche target, these tranches are retained. If some but not all of the tranches submitted by the bidder and other bidders that were assigned a default bid are needed, tranches are chosen at random to fill the tranche target, in a procedure analogous to that used for bidders that submitted a bid, as described in section VIII.C.4.

If, as of the reporting phase in the previous round, a bidder had some tranches on a particular EDC at the going price for the previous round, and/or retained withdrawals, and/or denied switches; if the EDC's price did not tick down from the previous round to the current round; and if there is excess supply for the EDC in the current round, so that the price will tick down from the current round to the next round, then:

- all withdrawals that were previously retained are released and the bidder has no remaining obligation from those tranches;
- all switches that had previously been denied are outbid and the bidder is assigned free eligibility for those tranches in the next round; and
- all tranches bid in the previous round at the going price are bid in the current round on the EDC at the going price. The bidder has eligibility associated with these tranches in the next round. If the bidder is assigned a default bid in the next round these tranches will be withdrawn and assigned the highest possible exit price.

If, as of the reporting phase in the previous round, a bidder had some tranches on a particular EDC bid at the going price, and/or retained withdrawals, and/or denied switches; if the EDC's price did not tick down from the previous round to the current round; and if there is no excess supply on the EDC in the current round so that the price will not tick down in the next round; then:

- any tranches bid at the going price in the previous round are deemed to be bid in the current round at the current going price;
- if any new tranches were bid by other bidders on the EDC at the going price in the current round, the denied switches (if any) of bidders that have been assigned default bids are outbid first, before the denied switches of bidders that have submitted a bid in the current round. If more than one bidder has been assigned a default bid, and if some but not all denied switches from such bidders are outbid, then for each denied switch that must be outbid, the Auction Manager chooses at random among the default bidders the bidder whose switch is outbid, in a procedure analogous to that used for bidders that submitted a bid, as described in section VIII.C.6; and
- if any new tranches were bid by other bidders on the EDC at the going price in the current round, and if all denied switches from default bidders and from bidders that submitted a bid are outbid, retained withdrawals are released, starting with the highest named exit price. For a given exit price, tranches from bidders that have been assigned default bids (if any) are released first, before the retained withdrawals of bidders that have submitted a bid in the current round. If more than one bidder has been assigned a default bid, and if some but not all of the retained withdrawals from such bidders must be released at a given exit price, then for each retained withdrawal that must be released, the Auction Manager chooses at random among the default bidders the bidder whose withdrawn tranche is released, in a procedure analogous to that used for bidders that submitted a bid, as described in section VIII.C.6.

The bidder can lose its ability to bid in all future rounds by failing to bid during the bidding phase of a round or during its extension. It is the responsibility of the bidder to ensure that bids are submitted on time.

Example 14.

Bidder A's bid in round 5 is given below:						
EDC	PSE&G	JCP&L	ACE	RECO		
Round 5 Price (\$/MW-day)	437.08	451.99	416.59	419.27		
Bidder A Bid (tranches)	0	4	2	0		
Bidder A's bid in round 6 is given below:						
EDC PSE&G JCP&L ACE RECO						
Round 6 Price (\$/MW-day)	429.43	444.08	404.09	406.69		
Bidder A Bid (tranches)	2	4	0	0		

This bid represents a switch, whereby Bidder A has increased the number of tranches on PSE&G while decreasing the number of tranches on ACE. The Auction Manager reports the following:

EDC	PSE&G	JCP&L	ACE	RECO
Round 7 Price (\$/MW-day)	429.43	430.76	404.09	394.49
Report to Bidder A (tranches)	0	4	2 denied switches @ 416.59	0

In round 7, the minimum number of tranches that Bidder A can bid on PSE&G and RECO is zero, because Bidder A did not have any tranches accepted for these EDCs in round 6. The minimum number of tranches that Bidder A can bid on JCP&L is zero, since the price for JCP&L has ticked down from \$444.08/MW-day to \$430.76/MW-day. The minimum number of tranches that Bidder A can bid on ACE is 2, since the price for ACE has not ticked down and Bidder A has two denied switches on ACE.

In round 7, Bidder A fails to submit a bid in the bidding phase of the round. Bidder A is granted an extension but does not submit a bid during the extension. Bidder A will be assigned the following bid, which is its default bid. Bidder A is the only bidder that is assigned a default bid.

EDC	PSE&G	JCP&L	ACE	RECO
Round 7 Price (\$/MW-day)	429.43	430.76	404.09	394.49
Default Bid for Bidder A (tranches)	0	0	2 denied switches @ 416.59	0

None of Bidder A's withdrawn tranches are retained; tranches from other bidders that actually submitted bids are sufficient to fill the tranche target for JCP&L. Three new tranches are bid on ACE by other bidders. Both of Bidder A's tranches become outbid denied switches. Bidder A's eligibility in round 8 is 2. Should Bidder A again fail to bid in round 8, this free eligibility will be lost. Bidder A would be assigned eligibility of zero in round 9 and would lose the ability to bid for the remainder of the Auction.

VIII.C.11. End of Auction

The Auction ends for all EDCs at the same time. The Auction ends in the reporting phase before the first round in which no bidder could change its bid from the previous round. For the Auction to end, the total excess supply in the Auction must be equal to zero and it must be that no price can tick down. At the end of the Auction, tranches are allocated to the winners and all winners for an EDC's tranches, should the Board authorize them as suppliers, will receive the same price for that EDC. The price given to the winners is determined as follows.

If, to fill the tranche target for an EDC in the final round, only tranches bid at the price from the final round are used, the winners are those that submitted bids at the price from the final round. The final price given to all winners is the price from the final round.

If, to fill the tranche target for an EDC in the last round, withdrawn tranches must be retained, then the winners are the bidders that submitted bids at the price from the last round and the bidders that submitted the lowest of the exit prices. If, to fill the last tranches of the tranche target of an EDC in the last round, the Auction Manager must use some but not all the tranches from two or more bidders tied at the same exit price, then the Auction Manager, for each tranche, will choose at random the bidder whose tranche is retained. For the first tranche needed at the tied exit price, the probability that a bidder is chosen is the number of tranches that the bidder has bid at the exit price divided by the total number of tranches bid at the tied exit price. If a second tranche is needed at the exit price, the Auction Manager again will choose at random the bidder whose tranche will be retained. The probability that any one bidder is chosen is the number of tranches that the bidder has bid at the exit price and that have not yet been retained divided by the total number of tranches bid at the exit price and that have not yet been retained. The Auction Manager repeats this procedure until the tranche target for the EDC is filled. The final price given to all winners is the last accepted price.

If, to fill the tranche target for an EDC in the last round the Auction Manager must disallow reductions from both withdrawals and switches, then the winners are the bidders that submitted bids at the last round price, the bidders that withdrew their tranches, and the bidders with reductions for that EDC from switches that were denied. The final price received by all winners is the price at which the denied switches were last freely bid.

Example 15.

The tranche target for PSE&G is 24.

In round 34, 26 tranches for PSE&G are bid at a price of \$223.66/MW-day. In round 35, 20 tranches for PSE&G are bid at a price of \$218.07/MW-day.

Bidder A bids 5 tranches for PSE&G in round 34 and 1 tranche in round 35. Bidder A enters an exit price equal to \$223.05/MW-day for the 4 tranches it is withdrawing.

Bidder B bids 3 tranches for PSE&G in round 34 and 1 tranche in round 35. Bidder B enters an exit price of \$221.56/MW-day for the 2 tranches it is withdrawing.

No other bidder changes its number of tranches bid on PSE&G. All other bidders collectively bid 18 tranches for PSE&G in both rounds 34 and 35. The total excess supply in round 35 is zero and the Auction ends in round 35. Twenty tranches for PSE&G are allocated to the bidders that bid at the going price of \$218.07/MW-day. Two additional tranches are allocated to Bidder B since it submitted a lower exit price. Finally, Bidder A wins two additional tranches so that the tranche target is filled. All winning bidders will receive a price of \$223.05/MW-day, which is the lowest price at which the tranche target is filled.

VIII.D. Post Auction

The Board will decide whether or not to approve the results of this Auction (the BGS-CIEP Auction) by the end of the second business day following the calendar day on which the BGS-RSCP or BGS-CIEP Auction closes, whichever comes later.

If the Board approves the Auction results and authorizes the winners to become BGS-CIEP suppliers, the Auction Manager will notify each winner of the tranches it has won and the final prices. The Auction Manager will notify each EDC of the following for its territory: the identity of the winners, the number of tranches won, and the final price.

Each winner and each EDC will have three days from the time at which the Board approves the Auction results to execute the BGS-CIEP Supplier Master Agreement. Each winner will have these three days to demonstrate compliance with the creditworthiness requirements set forth in the BGS-CIEP Supplier Master Agreement and to submit the executed BGS-CIEP Supplier Master Agreement to the Board for information. A winner's financial guarantee posted before the Auction may be forfeited if the winner does not execute the Agreement within three days, if the winner fails to demonstrate compliance with the creditworthiness requirements set forth in the BGS-CIEP Supplier Master Agreement, or if the winner fails to agree to any of the terms of the Agreement. Effective with the exercise by an EDC of its right to collect on the financial guarantees, any contractual rights or other

entitlements of the winners shall immediately terminate without further notice by the EDC. In addition, winners shall be liable for damages incurred by the EDCs, which damages shall be determined in accordance with the terms of the BGS-CIEP Supplier Master Agreement as if the winner were a defaulting party to that Agreement.

Each winner will have to pay a fee per tranche won. This fee will be set to recover all the costs associated with the Auction Process. The Auction Manager will announce the fee per winning tranche no later than 6 days before the Auction. The fee will be netted against the first payment made to the winner during the supply period.

VIII.E. Association and Confidential Information Rules

VIII.E.1. Process for Reporting Associations, Identifying Concerns and Remedies

Interested parties applying to qualify to bid in the Auction will be required to indicate in their Part 1 Applications whether they are part of a bidding agreement, a joint venture for purposes of participating in the Auction, a bidding consortium, or other arrangements pertaining to bidding in the Auction. Interested parties will also be required to certify in their Part 1 Application that, should they qualify to participate in the Auction, they will not disclose information regarding the list of qualified bidders. In addition to certifications regarding bidding agreements and confidential information, each interested party makes a number of other certifications signifying its agreement with the terms of the BGS-CIEP Auction Rules, the terms of the BGS-CIEP Supplier Master Agreement, as well as signifying its agreement that it will not assign its rights or substitute another entity in its place. These additional certifications are discussed in section VIII.B.2.

Once parties are qualified to bid in the Auction, each qualified bidder will be asked in its Part 2 Application to make a number of certifications, each detailed below, and each qualified bidder may be asked to provide additional information to the Auction Manager if these certifications cannot be made. In particular, each qualified bidder will be informed of the list of qualified bidders and will be asked to certify that it is not associated with any other qualified bidder. If a qualified bidder cannot make such a certification, it will be asked to identify associations it may have with other qualified bidders. The criteria that determine whether two bidders are associated with one another are given below. If two qualified bidders

are associated with one another, the Auction Manager will determine whether the two qualified bidders can both participate in the Auction, as well as the terms and conditions of such participation. The Auction Manager may require qualified bidders that are associated with one another to bid as one entity or to reorganize so as to no longer be associated with one another.

Qualified bidders will be asked to certify that they will undertake to appropriately restrict their disclosure of confidential information relative to their bidding strategy and confidential information regarding the Auction Process (both of which are defined in section VIII.E.3). Qualified bidders will also be asked to certify that they have not come and will not come to any agreement with another qualified bidder with respect to bidding at the Auction, except as disclosed and approved by the Auction Manager in their Part 1 Application. In addition to certifications regarding associations and confidential information, each qualified bidder makes a number of other certifications signifying its agreement that a bid is a binding offer to provide service under the terms of the BGS-CIEP Supplier Master Agreement, as well as signifying its agreement that it will not assign its rights or substitute another entity in its place. These additional certifications are discussed in section VIII.B.2.

Before obtaining final documentation necessary to participate in the Auction, registered bidders will be required to certify that they will continue to maintain the confidentiality of any information that they will have acquired through their participation in the Auction Process.

VIII.E.2. Association Criteria

1. Preliminary Definitions

- a. A party *controls* an entity directly if the party holds a majority of shares, majority voting power, a majority of common directors, can appoint a majority of directors, or if the party in fact controls the entity's affairs through some other means. A party controls an entity indirectly if the party controls another entity that controls the entity in question (or through a longer line of control; e.g., if the party controls another entity that controls an entity that controls the entity in question, etc.).
- b. A party *participates* directly in another entity Z if the party holds any class of listed shares, if it holds the right to acquire such shares, if it holds any option to purchase shares or if it has voting power. The participation is indirect if the party participates

in another entity that participates in Z (with potentially a longer line of "indirect participation"). When the participation is indirect, the percentage of participation of the party in the entity is obtained by multiplying the percentages of participation at each level.

- c. A party is concerned with the bid of a bidder if the party has confidential information relative to the bidders' bidding strategy (see definition in the next section), has agreed to provide assistance with financing or has agreed to provide assistance in another way.
- 2. Bidder A and Bidder B are associated with each other if Bidder A
 - a. Controls bidder B, directly or indirectly; or
 - b. Has at least a 10% participation in Bidder B and is concerned with Bidder B's bid; or
 - c. Controls an entity that has at least a 10% participation, direct or indirect, in Bidder B and that is concerned with Bidder B's bid; or
 - d. Is controlled by an entity that controls Bidder B directly or indirectly; or
 - e. Is controlled by an entity that has at least a 10% participation, direct or indirect, in Bidder B and that is concerned with Bidder B's bid.
- 3. Bidder A and Bidder B are associated if there is a party which
 - a. Controls Bidder A, directly or indirectly; or
 - b. Has at least a 10% participation in Bidder A, directly or indirectly, and is concerned with Bidder A's bid; or
 - c. Controls an entity that has at least a 10% participation in Bidder A, direct or indirect, and is concerned with Bidder A's bid; or
 - d. Has confidential information about Bidder A's bid and is controlled by Bidder A; or
 - e. Has confidential information about Bidder A's bid and is controlled by an entity or person that controls Bidder A directly or indirectly; or

f. Has confidential information about Bidder A's bid and is controlled, directly or indirectly, by an entity that has at least a 10% participation in Bidder A and is concerned with Bidder A's bid;

and if this same party has one of relationships a. to f. with Bidder B.

- 4. Bidder A and Bidder B are associated if there is a party that has at least a 20% participation, directly or indirectly, in both bidders.
- 5. Bidder A and Bidder B are associated if there is a party that has at least a 20% participation, directly or indirectly, in Bidder A and that:
 - a. Has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - b. Is controlled by Bidder B; or
 - c. Controls a person or entity that controls Bidder B; or
 - d. Controls a person or entity that: has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - e. Is controlled by a person or entity that controls Bidder B directly or indirectly; or
 - f. Is controlled by a person or entity that has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - g. Is controlled by a person or entity who controls a person has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid.

VIII.E.3. Definitions of Confidential Information

Confidential information relative to the bidding strategy means information relating to a bidder's bid, whether in writing or verbally, which if it were to be made public would be likely to have an effect on any of the bids that another bidder would be willing to submit at the Auction. Confidential information relative to the bidding strategy includes (but is not limited to) a bidder's Auction strategy; a bidder's indicative offer; the bidder's preference to bid for one EDC's system rather than another; the quantities that a bidder wishes to serve of one or

more EDCs' systems; the bidder's estimation of the value of a tranche of the various EDCs' systems; the bidder's estimation of the risks associated with serving BGS-CIEP Load or of serving a particular EDC; and a bidder's contractual arrangements for power with a party to serve the BGS-CIEP Load were the bidder to be a winner at the Auction.

Confidential information regarding the Auction Process means information that is not released publicly by the Board or the Auction Manager and that a bidder acquires as a result of participating in the Auction Process, whether in writing or verbally, which if it were to be made public could impair the integrity of current or future Auctions, impair the ability of the EDCs to hold future Auctions, harm consumers, or injure bidders or applicants. Confidential information regarding the Auction Process includes (but is not limited to) the list of qualified bidders, the list of registered bidders, the initial eligibility in the Auction, the status of a bidder's participation in the Auction, and all reports of results and announcements made by the Auction Manager to all or any one bidder during the Auction.

VIII.E.4. Certifications and Disclosures to Be Made

An interested party will be required in its Part 1 Application to disclose any bidding agreement or arrangement in which the interested party may have entered. In addition, the following certification will be required and will apply from the time that the Part 1 Application has been submitted.

1. An applicant must certify that if it qualifies to participate in the Auction, the applicant will not disclose at any time information regarding the list of qualified bidders, including the number of qualified bidders, the identity of any one or all entities that have been qualified, or the fact that an entity has not been qualified for participation in the Auction. The applicant must certify that it will destroy any document distributed by the Auction Manager that lists the qualified bidders within five days of the Board rendering a decision on the Auction results.

The following certifications will be required of each qualified bidder in its Part 2 Application and will apply from the time of qualification until the Board certification of the Auction results. Each qualified bidder must consult the list of all qualified bidders and attest to the following:

2. A qualified bidder must certify that it is not associated with another qualified bidder according to the criteria given above.

A qualified bidder unable to make certification 2 must identify the qualified bidders with which it is associated and the nature of the association.

3. A qualified bidder must certify that, other than qualified bidders that were explicitly named in its Part 1 Application as parties with whom the bidder has entered into a bidding agreement, joint venture for the purpose of bidding in the Auction, bidding consortium, or other arrangement pertaining to bidding in the Auction, the bidder has not entered into any agreement with another qualified bidder, directly or indirectly, regarding bids at the Auction, including, but not limited to, the amount to bid at certain prices, the system on which bids are placed, when or at what prices bids are withdrawn or switched, or the amount of exit prices.

An *Advisor* is an entity or person(s) that will be advising or assisting the qualified bidder with bidding strategy in the BGS-CIEP Auction, with estimation of the value of a system's tranches, or with the estimation of the risks associated with serving BGS-CIEP load.

4. A qualified bidder must certify to one of the following: (i) the qualified bidder has not retained an Advisor; or (ii) the qualified bidder has retained an Advisor, the Advisor will not provide any similar advice or assistance to any other qualified bidder, and the Advisor will not be privy to confidential information relative to another qualified bidder's bidding strategy; or (iii) the qualified bidder has retained an Advisor who will provide similar advice or assistance to another qualified bidder, or who will be privy to confidential information relative to any other qualified bidder's bidding strategy, but appropriate protections have been put into place to ensure that the Advisor does not serve as a conduit of information between, or as a coordinator of the bidding strategies of, multiple bidders.

A qualified bidder unable to make certification 4 must name the Advisor and the other bidder(s) concerned.

5. A qualified bidder must certify that the qualified bidder is not a party to any contract for the purchase of power that might be used as source of supply for BGS-CIEP, and that (i) would require the disclosure of any confidential information (confidential information relative to the bidding strategy or confidential information regarding the Auction Process) to the counterparty under such a contract; or (ii) that would require the disclosure of any confidential information (confidential information relative to the bidding strategy or confidential information regarding the Auction Process) to any other party; or (iii) that would provide instructions, direct financial incentives, or other inducements for the bidder to act in a way determined by the counterparty in the agreement and/or in concert with any other bidder in the Auction. Notwithstanding the above, a qualified bidder may, during negotiations prior to the Auction for contractual arrangements for power to serve BGS-CIEP Load were the bidder to be a winner at the Auction, discuss with the counterparty to such arrangements the nature of the products to be purchased, the volume, and the price at which it is willing to buy these products.

A qualified bidder unable to make certification 5 must disclose the contractual terms that prevent the qualified bidder from making the certification.

6. A qualified bidder must certify that it does not have any knowledge of confidential information relative to the bidding strategy of any other qualified bidder.

A qualified bidder unable to make certification 6 will be asked to name the other qualified bidder and the nature of the confidential information.

7. A qualified bidder must certify that it will not disclose confidential information relative to its own bidding strategy except to bidders that were explicitly named in its Part 1 Application as parties with whom the bidder has entered into a bidding agreement, joint venture for the purpose of bidding in the Auction, bidding consortium, or other arrangement pertaining to bidding in the Auction, bidders with which it is associated as disclosed through certification 2, to its Advisor, and to its financial institution.

A qualified bidder unable to make certification 7 will be asked to state all reasons.

8. A qualified bidder must certify that, other than entities affiliated with the bidder, and other than bidders with which the bidder has entered a bidding agreement, joint venture for purposes of the Auction, bidding consortium, or other arrangement pertaining to the Auction, no party has agreed to defray any of its costs of participating in the Auction, including the cost of preparing the bid, the cost of any financial guarantees, the cost to be paid upon winning a tranche, or any other participation cost.

A qualified bidder unable to make certification 8 must identify the party that has agreed to defray some or all of the qualified bidder's cost of participating in the Auction, and the nature of the participation costs that the party has agreed to defray.

Furthermore, in its Part 2 Application a qualified bidder will have to certify that it will not disclose any confidential information regarding the Auction Process that it has acquired or will acquire through its participation. These certifications will apply from the time of submission of the Part 2 Application.

- 9. A qualified bidder must certify that if it is registered to participate in the Auction, the qualified bidder will not disclose at any time information regarding the initial eligibility in the Auction or the list of registered bidders, including the number of registered bidders, the identity of any one or all entities that have been registered, or the fact that an entity has not been registered for participation in the Auction.
- 10. A qualified bidder must certify that it will not disclose any confidential information regarding the Auction Process to any party except to its Advisor and bidders with which it is associated.

A qualified bidder unable to make certification 9 or certification 10 will be asked to state all reasons.

Following a successful Part 2 Application, the registered bidder will be required to certify that it will continue to abide by its prior commitment to maintain the confidentiality of information regarding the Auction Process. The registered bidder will be required to do so before obtaining manuals and procedures essential to submit bids in the Auction. The

registered bidder also certifies that it will destroy all documents provided by the Auction Manager that contain confidential information within five days of the Board rendering a decision on the Auction results. Such certification will apply from the time at which it is made, no later than five days before the start of the Auction.

11. A registered bidder certifies that it continues to abide by its prior confidentiality certifications. The registered bidder will not disclose any confidential information regarding the Auction Process to any party except to its Advisor and bidders with which it is associated. Further, the registered bidder certifies that it will destroy all documents written or electronic provided by the Auction Manager that contain confidential information regarding the Auction Process within five days of the Board rendering a decision on the Auction results.

Once the Auction has been concluded, if the Board approves the Auction results, the Board may choose to release information regarding final BGS-CIEP prices and the names of the winners. At that point, a winner may itself release information only regarding the number of tranches it has won and the territories the winner will be serving, and a losing bidder may itself release information only regarding the fact that it participated in the Auction. The winners and losing bidders otherwise continue to be bound by their certifications as described previously.

VIII.E.5. Actions to Be Taken If Certifications Cannot Be Made

If a party cannot make all above certifications, the Auction Manager will decide within five days on a course of action on a case-by-case basis. To decide on this course of action, the Auction Manager may make additional inquiries to understand the reason for the inability of the bidder to make all the certifications.

In general, qualified bidders that are associated with one another, or that have entered into agreements regarding bidding at the Auction, are considered as one bidder for the purposes of application of the statewide load cap and for the administration of the Auction. Bidders can be allowed to bid independently or can be asked to end their association or agreement as a

condition of participation, as circumstances warrant. If qualified bidders are asked to end their associations they will be given five days to do so.

If qualified bidders do not comply with additional information requests by the Auction Manager regarding certifications required in the Part 2 Application, or do not comply with a request from the Auction Manager to end their associations, this may be sufficient grounds for the Auction Manager to reject the application.

Sanctions can be imposed on a qualified bidder for failing to disclose information relevant to determining associations, for coordinating with another bidder without disclosing this fact, for releasing confidential information except as provided in 3, 5, and 7 of section VIII.E.4, or for disclosing information during the Auction to a person other than those specified in 10 of section VIII.E.4. Such sanctions can include, but are not limited to, loss of all rights to serve any BGS-CIEP Load won in the Auction by such bidder, forfeiture of bid bonds and other fees posted or paid, liquidated damages of \$100,000, action under state or federal laws, attorneys' fees and court costs incurred in any litigation that arises out of the bidder's improper disclosure, debarment from participation in future BGS Auctions, prosecution under applicable state and federal laws, or other sanctions that the Board may consider appropriate. The Auction Manager will make a recommendation to the Board on a possible sanction and the Board will make the final determination.

VIII.F. Glossary of Auction Terms

VIII.F.1. Associated With

A bidder is associated with another if the two bidders have ties that could allow them to act in concert or that prevent them from competing actively against each other in the Auction. Specific criteria for associations are provided in section VIII.E.2.

VIII.F.2. Auction Volume

The Auction volume is a number of tranches that the EDCs taken together will purchase through the Auction. The Auction Manager, in accordance with confidential guidelines approved by the Board, may cut back the Auction volume.

VIII.F.3. BGS

Basic Generation Service is the electric supply for those retail customers who are not served by a third party supplier.

VIII.F.4. BGS Load

The BGS Load is the full electricity requirement (i.e., including energy, capacity, ancillary services, transmission, etc.) of retail customers who have not chosen a third party supplier, as measured and reported to PJM. It is the EDC aggregate zonal requirements less the wholesale requirements less the third party supplier requirements.

VIII.F.5. BGS-CIEP Customer

A BGS-CIEP Customer is a retail customer who takes BGS on a CIEP (as opposed to an RSCP) tariff.

VIII.F.6. BGS-CIEP Load

An EDC's BGS-CIEP Load includes the sum of the hourly load of all BGS-CIEP customers, adjusted for losses.

VIII.F.7. BGS-CIEP Peak Load Share

BGS-CIEP Peak Load Share is the portion of PJM-determined EDC zonal peak load on one proxy day attributable to customers in CIEP rate classes that are taking BGS on that proxy day. The BGS-CIEP Peak Load Share is a single value measured in MW.

VIII.F.8. BGS-RSCP Load

The BGS-RSCP Load is the BGS Load excluding the BGS-CIEP Load.

VIII.F.9. Bid

A bid consists of four numbers: a number of PSE&G tranches, a number of JCP&L tranches, a number of ACE tranches, and a number of RECO tranches. A bid represents the number of tranches that a bidder wishes to serve for each EDC at the going prices in force at that point in the Auction. In all rounds except the first, to fully specify a bid, a bidder may be asked for information in addition to the number of tranches that it wishes to bid for each EDC, such as switching priorities and exit prices. A bid is valid when it is submitted and verified in the bidding phase, and processed by the Auction software.

VIII.F.10. Bidding Phase

The bidding phase is the first phase of a round, during which bidders place bids. A bidder that submits a bid in a round may revise or change this bid as long as the bidding phase of the round is still open.

VIII.F.11. Calculating Phase

The calculating phase is the second phase of the round, during which the Auction Manager tabulates the results of that round's bidding phase and calculates the prices for the next round.

VIII.F.12. CIEP Customer

CIEP customers are retail customers who are taking BGS-CIEP, or are retail customers served by a third party supplier who, were they to take BGS, would be served under a CIEP (rather than an RSCP) tariff.

VIII.F.13. CIEP Load

CIEP Load for an EDC is defined as the sum of the hourly load of all BGS-CIEP customers, as well as the hourly load of retail customers served by a third party supplier who, were they to take BGS, would do so under a CIEP tariff, adjusted for losses.

VIII.F.14. CIEP Peak Load Share

CIEP Peak Load Share is the portion of PJM-determined EDC zonal peak load on one proxy day attributable to customers in CIEP rate classes. The CIEP Peak Load Share is a single value measured in MW, that consists of the BGS-CIEP Peak Load Share and the peak load attributable to customers served by third party suppliers on the proxy day and who are in CIEP rate classes.

VIII.F.15. CIEP Standby Fee

The CIEP Standby Fee is a fee for the option available to each CIEP customer of taking BGS on a CIEP tariff or rate. This fee is proposed to be set at 0.015 /e/kWh of the energy used by CIEP customers measured at the customer meter.

VIII.F.16. Contingency Plan

If the number of tranches procured through the Auction is less than the Auction volume set at the start of the Auction, each EDC whose tranche target is not filled at the end of the Auction will implement a Contingency Plan for the remaining tranches. Under their Contingency Plans, the EDCs will purchase necessary services including installed capacity, energy, and ancillary services through PJM-administered markets.

VIII.F.17. Decrement

If the number of tranches bid exceeds the number of tranches needed for an EDC, the price for the EDC falls by a decrement in the next round. The decrement varies in each round based on the excess supply on an EDC measured against a measure of maximum possible excess supply.

VIII.F.18. Denied Switches

The Auction Manager denies reductions in the number of tranches bid for an EDC from switches when the tranches bid at the going price and the withdrawn tranches that can be retained are not sufficient to fill that EDC's tranche target. Denied switches are retained at the last price at which they were freely bid.

VIII.F.19. Eligibility and Initial Eligibility

Initial eligibility is equal to the number of tranches that a bidder has financially guaranteed with a letter of credit (or bid bond) with its Part 2 Application. Initial eligibility can never exceed the statewide load cap.

A bidder's initial eligibility is the bidder's eligibility in round 1. A bidder's eligibility in round 2 is the total number of tranches bid for all EDCs at the going prices in round 1. For any subsequent round, a bidder's eligibility in a round is the bidder's eligibility at the start of the previous round minus the number of tranches that the bidder withdrew in the bidding phase of the previous round (whether or not the Auction Manager retains these withdrawn tranches).

VIII.F.20. Eligibility Ratio

An eligibility ratio is obtained by dividing the total eligibility in the Auction by the number of tranches available in the Auction.

VIII.F.21. End of Auction

The Auction ends in the reporting phase of the first round in which the total excess supply is zero.

VIII.F.22. Excess Supply

The excess supply on an EDC is the number of tranches bid at the going price for the EDC minus the EDC's tranche target, or it is zero, whichever is greater.

VIII.F.23. Exit Price

If a bidder is reducing eligibility in a round, in general the bidder names an exit price. The bidder names an exit price for the tranches of an EDC that the bidder is no longer willing to serve at the price for the current round. An exit price must be a price higher than the going price in the current round and less than or equal to the previous going price for the EDC. No exit price is named when withdrawing tranches of free eligibility.

VIII.F.24. Extension

An extension extends the bidding phase of a round by 15 minutes. Each bidder is allowed two extensions during the Auction. A bidder is automatically deemed to have requested an extension whenever no bid has been received from the bidder by the end of the scheduled bidding phase of a round and the bidder has not used the two allowable extensions. The bidding phase in round 1 is automatically extended for the convenience of bidders.

VIII.F.25. Financial Guarantee

Each qualified bidder must post a financial guarantee, in the form of a letter of credit (or bid bond), proportional to its indicative offer at the maximum starting price.

VIII.F.26. Free Eligibility

A denied switch that is replaced in filling the tranche target by a new tranche at the going price is said to be outbid. An outbid denied switch becomes a tranche of free eligibility. The bidder can assign a tranche of free eligibility to any EDC(s) and, if the bidder wishes to withdraw such a tranche, no exit price will be named and this tranche will not be retained.

VIII.F.27. Full-Requirements Service

Full-requirements service means that the supplier is responsible for fulfilling all the requirements of a PJM LSE including capacity, energy, ancillary services, transmission, and any other service as may be required by PJM.

VIII.F.28. Going Price

The going price for an EDC in a round is the price at which the Auction Manager is soliciting bids in that round. A bidder that submits a bid in a round declares itself ready to supply the number of tranches bid for each EDC at the going price for the EDC.

VIII.F.29. Indicative Offer

An indicative offer specifies two numbers of tranches. The first number represents the amount that the qualified bidder is willing to serve at the maximum starting price on a

statewide basis (i.e., for all EDCs combined). The second number represents the amount that the qualified bidder is willing to supply at the minimum starting price on a statewide basis. At each of the maximum and the minimum starting prices, the number of tranches indicated by the qualified bidder cannot exceed the statewide load cap. At the maximum starting price, the number of tranches indicated by a qualified bidder cannot be lower than two. Indicative offers are provided with the Part 2 Application.

VIII.F.30. Initial Eligibility

See Eligibility and Initial Eligibility.

VIII.F.31. Minimum and Maximum Starting Prices

The minimum and maximum starting prices establish the range of possible starting prices for the Auction: each EDC will choose a starting price for round 1 of the Auction that is between the minimum and the maximum starting prices. The EDCs will agree on the statewide minimum and maximum starting prices.

VIII.F.32. MW-Measure

The approximate measure in MW of a single tranche for an EDC, given the CIEP Peak Load Share for the EDC and the percentage of load represented by the tranche.

VIII.F.33. Outbid (Denied Switches)

A denied switch that is replaced in filling the tranche target by a new tranche at the going price is said to be outbid.

VIII.F.34. Oversupply Ratio

The oversupply ratio is the ratio of the excess supply on an EDC to a measure of the maximum possible excess supply on that EDC. The measure of the maximum possible excess supply on the EDC takes into account the total excess supply in the Auction, the statewide load cap, and the number of registered bidders.

VIII.F.35. Part 1 Application

In their Part 1 Applications, interested parties will be asked to submit financial information so that the EDCs can assess their creditworthiness. In addition, interested parties

will be asked to comply with other qualification criteria including agreeing to comply with the BGS-CIEP Auction Rules and agreeing to the terms of the BGS-CIEP Supplier Master Agreement. Each interested party will also be asked to agree that if the interested party is successful in its Part 1 Application it will keep confidential the list of other successful applicants and it will not assign its rights or substitute another entity in its place.

VIII.F.36. Part 1 Application Date

Date at which Part 1 Applications are due. This date will be set no earlier than 10 days after the maximum and minimum starting prices have been announced.

VIII.F.37. Part 2 Application

In the Part 2 Application, qualified bidders will make a number of certifications to ensure compliance with the association and confidential information portion of these rules. Each qualified bidder is also asked to agree to keep confidential the list of other successful applicants; to agree that the submission of any bid creates a binding and irrevocable offer to provide service under the terms of the BGS-CIEP Supplier Master Agreement; and not to assign its rights or substitute another entity in its place.

VIII.F.38. Part 2 Application Date

Date at which Part 2 Applications are due. This date will be set no later than 10 days before the start of the Auction.

VIII.F.39. Product in the Auction

A product in the (BGS-CIEP) Auction is the BGS-CIEP Load for an EDC for a one-year supply period.

VIII.F.40. Qualified Bidder

An interested party that has submitted a Part 1 Application and that has satisfied all conditions of the Part 1 Application becomes a qualified bidder. Interested parties will be notified that they have qualified no later than three days after the Part 1 Application Date.

VIII.F.41. Recess

A recess is a suspension of the Auction for a period of no less than twenty minutes, giving bidders more time to consider their bids. A bidder may request a recess in the calculating phase or the reporting phase of a round. Each bidder is allowed one recess during the Auction. A bidder cannot request a recess in the first ten rounds of the Auction. Starting in the eleventh round, a bidder may request a recess in a round if the total excess supply in the Auction reported in the previous round was 20 or fewer tranches.

VIII.F.42. Reductions

Reductions in the number of tranches bid on an EDC for which the price has ticked down can be in the form of a withdrawal (i.e., reducing the number of tranches bid on an EDC without increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total) or a switch (i.e., re-assigning the total number of tranches bid so that the number of tranches bid on one or several EDCs is reduced, but the number of tranches bid on other EDCs is increased by the same amount).

VIII.F.43. Registered Bidder

A qualified bidder that submits a Part 2 Application and that satisfies all conditions of the Part 2 Application becomes a registered bidder. Conditions of the Part 2 Application include the submission of an indicative offer and financial guarantees. These conditions also include making certifications relating to the Association and Confidential Information Rules or, if not all these certifications can be made, providing additional information to the Auction Manager and abiding by the course of action decided by the Auction Manager.

VIII.F.44. Reporting Phase

The reporting phase is the third and final phase of a round, during which the Auction Manager informs the bidders of the results of that round's bidding phase. All bidders are informed of the going prices for the next round's bidding phase and are provided with a range of total excess supply. Each bidder privately receives the results of the bidder's own bid from that round, indicating to each bidder its obligation at this point in the Auction.

VIII.F.45. Round

The Auction runs in discrete time periods called rounds. Each round has a bidding phase, a calculating phase, and a reporting phase.

VIII.F.46. Rounding

Prices and exit prices are in dollars per MW-day and are rounded off to the nearest cent.

VIII.F.47. RSCP Load

RSCP Load is the retail load excluding the CIEP Load.

VIII.F.48. Session

Each day in the Auction will consist of two bidding sessions of roughly equal length. Each bidding session will consist of a number of rounds.

VIII.F.49. Starting Prices

The starting prices are announced three days before the Auction starts. The starting prices are the going prices in round 1.

VIII.F.50. Statewide Load Cap

The statewide load cap is a maximum on the number of tranches that a bidder can bid and win statewide.

VIII.F.51. Switches and Switching

Switching involves an increase in the number of tranches bid on some EDCs while at the same time a reduction in the number of tranches bid on other EDCs. Switching (without a simultaneous reduction of tranches through a withdrawal) occurs when a bidder reallocates its tranches bid without changing the total number of tranches bid on all EDCs. Switching alone has no effect on eligibility.

VIII.F.52. Switching Priority

A switching priority is a preference assigned to one of the EDCs for which a bidder is increasing its number of tranches bid. The bidder is required to assign a switching priority when the bidder is increasing the number of tranches bid for more than one EDC.

VIII.F.53. Target Eligibility Ratio

The target eligibility ratio is a desired ratio of eligibility to the Auction volume.

VIII.F.54. Time-Out

A time-out is a pause in the Auction. A time-out suspends activity in the Auction for a period of up to four hours. The Auction Manager can call a time-out at any time during a round.

VIII.F.55. Total Excess Supply

The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus tranches of free eligibility.

VIII.F.56. Tranche

A tranche of one EDC is a full-requirements tranche. A tranche is a fixed percentage share of the BGS-CIEP Load of an EDC for the period June 1, 2019 to May 31, 2020.

VIII.F.57. Tranche Size

The tranche size of an EDC in this Auction is the percentage share of the BGS-CIEP Load of the EDC represented by one tranche.

VIII.F.58. Tranche Target

The tranche target is the number of tranches available at the Auction for an EDC.

VIII.F.59. Withdrawal

A bidder reduces the number of tranches bid on an EDC through a withdrawal when the bidder is reducing the number of tranches bid on an EDC, while not increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total. A bidder who does not bid a tranche of free eligibility in the round when it becomes available is making a withdrawal.

VIII.G. Appendix

VIII.G.1. Tranches

Data for sample calculations are provided in the table below. The second column of the table provides, for each EDC, the 2018 CIEP Peak Load Share on a proxy day. The third column provides the number of tranches for each EDC, which is the tranche target. The number of tranches is rounded to the nearest non-zero integer. The last column provides the MW-measure, which is the size of the tranche multiplied by the CIEP Peak Load Share.

Table VIII-4. Sample Data

EDC	CIEP Peak Load Share (MW)	Number of tranches	Size of tranche (%)	MW-Measure
PSE&G	1,797.77	24	4.17%	74.91
JCP&L	805.89	11	9.09%	73.26
ACE	318.40	4	25.00%	79.60
RECO	56.00	1	100.00%	56.00
Total	2,978.06	40		

For ACE, the CIEP Peak Load Share is 318.40 MW. The size of a tranche is 25%. The MW-measure is then 79.60 MW, or 25% of 318.40 MW.

VIII.G.2. Decrements

The calculation of the size of the decrement, Δ , is based on the oversupply ratio, γ , which is the ratio of the excess supply on an EDC to an estimate of the maximum excess supply:

$$\gamma = \frac{B - TT}{\min(\overline{RES}, n \cdot \min\{SWLC, TT\} - TT)}$$

The numerator is the excess supply on an EDC, which is the number of tranches bid at the going price (B) minus the tranche target (TT). The denominator is a measure of maximum possible excess supply. The excess supply on an EDC must be less than or equal to the total excess supply in the Auction. \overline{RES} is the upper bound of the range of total excess supply

reported to bidders and serves as a measure of total excess supply in the Auction. The excess supply on an EDC must also be less than or equal to the excess supply that would result from all bidders bidding the maximum possible number of tranches on the EDC. The maximum possible number of tranches that can be bid on an EDC is either the statewide load cap (SWLC) or the tranche target (TT), whichever is lower. Thus, the excess supply that would result from all bidders bidding the maximum possible number of tranches on the EDC would be $n \cdot \min\{SWLC, TT\} - TT$ tranches, namely the number of registered bidders (n) times the statewide load cap (SWLC) or the tranche target (TT), minus the tranche target (to get a measure of excess supply). The estimate of maximum possible excess supply for the EDC used for the decrement rule is \overline{RES} , or the measure based on the number of registered bidders and the load cap ($n \cdot \min\{SWLC, TT\} - TT$), whichever is smaller.

Regime 1

The Auction starts in Regime 1. In Regime 1, the following decrement formulas will be used.

If an EDC's tranche target is 20 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0050 & if & \gamma \le 0.11 \\ 0.0175 & if & 0.11 < \gamma \le 0.28 \\ 0.0300 & if & 0.28 < \gamma \le 0.44 \\ 0.0400 & if & 0.44 < \gamma \le 0.60 \\ 0.0500 & if & \gamma > 0.60 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1100 (but above 0), the decrement is set at 0.50%. When the oversupply ratio is at or below a value of 0.2800 (but above 0.1100), the decrement is set at 1.75%. When the oversupply ratio is at or below a value of 0.4400 (but

above 0.2800), the decrement is set at 3%. When the oversupply ratio is at or below a value of 0.6000 (but above 0.4400), the decrement is set at 4%. When the oversupply ratio is above 0.6000, which means that the excess supply on the EDC exceeds 60.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is between 10 and 19 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0050 & if & \gamma \le 0.11 \\ 0.0175 & if & 0.11 < \gamma \le 0.28 \\ 0.0300 & if & 0.28 < \gamma \le 0.44 \\ 0.0400 & if & 0.44 < \gamma \le 0.60 \\ 0.0500 & if & \gamma > 0.60 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1100 (but above 0), the decrement is set at 0.50%. When the oversupply ratio is at or below a value of 0.2800 (but above 0.1100), the decrement is set at 1.75%. When the oversupply ratio is at or below a value of 0.4400 (but above 0.2800), the decrement is set at 3%. When the oversupply ratio is at or below a value of 0.6000 (but above 0.4400), the decrement is set at 4%. When the oversupply ratio is above 0.6000, which means that the excess supply on the EDC exceeds 60.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is between 3 tranches and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.75% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a predetermined value (but above 0). The decrement is never more than 5% (subject to rounding

off). The largest decrement would be in effect when the oversupply ratio is above that predetermined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0175 & if & \gamma \le 0.22 \\ 0.03 & if & 0.22 < \gamma \le 0.44 \\ 0.05 & if & \gamma > 0.44 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2200 (but above 0), the decrement is set at 1.75%. When the oversupply ratio is at or below a value of 0.4400 (but above 0.2200), the decrement is set at 3%. When the oversupply ratio is above 0.4400, which means that the excess supply on the EDC exceeds 44.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is 2 tranches or fewer, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 3% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.03 & if & \gamma \le 0.20 \\ 0.05 & if & \gamma > 0.20 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2000 (but above 0), the decrement is set at 3%. When the oversupply ratio is above 0.2000, which means that the excess supply on the EDC exceeds 20.00% of its maximum, the decrement is set at 5%.

Change from Regime 1 to 2

Decrements continue to be calculated by the decrement formulas of Regime 1 until the going prices for round 4 are calculated. After that time, in the first round in which the upper bound of the total excess supply range (\overline{RES}) reported to bidders is

Greater than 20 tranches; and

• At least 10 tranches fewer than the upper bound of the total excess supply range reported to bidders in round 1;

then the decrement formulas of Regime 2 will be used for the going prices in the next round. If, in the first round in which \overline{RES} is at least 10 tranches fewer than \overline{RES} in round 1, it is also the case that the upper bound of the total excess supply range reported to bidders is 20 tranches, the decrement formulas of Regime 3 will be used for the going prices in the next round and for the remainder of the Auction.

Regime 2

In Regime 2, the following decrement formulas will be used.

If an EDC's tranche target is 20 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.375% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.00375 & if & \gamma \le 0.11 \\ 0.01250 & if & 0.11 < \gamma \le 0.28 \\ 0.02250 & if & 0.28 < \gamma \le 0.44 \\ 0.03000 & if & 0.44 < \gamma \le 0.60 \\ 0.03750 & if & \gamma > 0.60 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1100 (but above 0), the decrement is set at 0.375%. When the oversupply ratio is at or below a value of 0.2800 (but above 0.1100), the decrement is set at 1.25%. When the oversupply ratio is at or below a value of 0.4400 (but above 0.2800), the decrement is set at 2.25%. When the oversupply ratio is at or below a value of 0.6000 (but above 0.4400), the decrement is set at 3%. When the oversupply ratio is above 0.6000, which means that the excess supply on the EDC exceeds 60.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is between 10 and 19 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.375% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a predetermined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that predetermined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.00375 & if & \gamma \le 0.11 \\ 0.01250 & if & 0.11 < \gamma \le 0.28 \\ 0.02250 & if & 0.28 < \gamma \le 0.44 \\ 0.03000 & if & 0.44 < \gamma \le 0.60 \\ 0.03750 & if & \gamma > 0.60 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1100 (but above 0), the decrement is set at 0.375%. When the oversupply ratio is at or below a value of 0.2800 (but above 0.1100), the decrement is set at 1.25%. When the oversupply ratio is at or below a value of 0.4400 (but above 0.2800), the decrement is set at 2.25%. When the oversupply ratio is at or below a value of 0.6000 (but above 0.4400), the decrement is set at 3%. When the oversupply ratio is above 0.6000, which means that the excess supply on the EDC exceeds 60.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is between 3 tranches and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a predetermined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that predetermined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.01250 & if & \gamma \le 0.22 \\ 0.02250 & if & 0.22 < \gamma \le 0.44 \\ 0.03750 & if & \gamma > 0.44 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2200 (but above 0), the decrement is set at 1.25%. When the oversupply ratio is at or below a value of 0.4400 (but above 0.2200), the decrement is set at 2.25%. When the oversupply ratio is above 0.4400, which means that the excess supply on the EDC exceeds 44.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is 2 tranches or fewer, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 2.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0225 & if & \gamma \le 0.20 \\ 0.0375 & if & \gamma > 0.20 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2000 (but above 0), the decrement is set at 2.25%. When the oversupply ratio is above 0.2000, which means that the excess supply on the EDC exceeds 20.00% of its maximum, the decrement is set at 3.75%.

Change from Regime 2 to 3

In the first round:

- After the going prices for round 4 are calculated; and
- In which the upper bound of the total excess supply range reported to bidders is 20 tranches;

then the decrement formulas of Regime 3 will be used for the going prices in the next round and for the remainder of the Auction.

Regime 3

In Regime 3, the following decrement formulas will be used.

If an EDC's tranche target is 20 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0025 & \text{if} & \gamma \le 0.16 \\ 0.0100 & \text{if} & 0.16 < \gamma \le 0.36 \\ \\ 0.0150 & \text{if} & 0.36 < \gamma \le 0.56 \\ \\ 0.0250 & \text{if} & \gamma > 0.56 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1600 (but above 0), the decrement is set at 0.25%. When the oversupply ratio is at or below a value of 0.3600 (but above 0.1600), the decrement is set at 1%. When the oversupply ratio is at or below a value of 0.5600 (but above 0.3600), the decrement is set at 1.5%. When the oversupply ratio is above 0.5600, which means that the excess supply on the EDC exceeds 56.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is between 10 and 19 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a predetermined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that predetermined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0025 & \text{if} & \gamma \le 0.16 \\ 0.0100 & \text{if} & 0.16 < \gamma \le 0.36 \\ \\ 0.0150 & \text{if} & 0.36 < \gamma \le 0.56 \\ \\ 0.0250 & \text{if} & \gamma > 0.56 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1600 (but above 0), the decrement is set at 0.25%. When the oversupply ratio is at or below a value of 0.3600 (but above 0.1600), the decrement is set at 1%. When the oversupply ratio is at or below a value of 0.5600 (but above 0.3600), the decrement is set at 1.5%. When the oversupply ratio is above 0.5600, which means that the excess supply on the EDC exceeds 56.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is between 3 tranches and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.00% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a predetermined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that predetermined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.010 & \text{if} & \gamma \le 0.27 \\ \\ 0.025 & \text{if} & \gamma > 0.27 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2700 (but above 0), the decrement is set at 1%. When the oversupply ratio is above 0.2700, which means that the excess supply on the EDC exceeds 27.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is 2 tranches or fewer, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be

in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.015 & if & \gamma \le 0.20 \\ 0.025 & if & \gamma > 0.20 \end{cases}$$

When the oversupply ratio is at or below a value of 0.2000 (but above 0), the decrement is set at 1.5%. When the oversupply ratio is above a maximum value of 0.2000, which means that the excess supply on the EDC exceeds 20.00% of its maximum, the decrement is set at 2.5%.

IX. APPENDIX B

Provisional BGS-RSCP Auction Rules

IX. APPENDIX B: PROVISIONAL BGS-RSCP AUCTION RULES

Table of Con		
IX.	Appendix B: Provisional BGS-RSCP Auction Rules	1
IX.A.	Introduction	3
IX.A.1.	Overview	3
IX.A.2.	Basic Generation Service and Load to Be Procured	5
IX.A.3.	Payment to BGS-RSCP Suppliers	8
IX.A.4.	Reliance on Product Definition, Payment Information, Rates, and Customer S	witching
	in These Rules	9
IX.B.	Before the Auction	10
IX.B.1.	Information Provided to Bidders	10
IX.B.2.	Qualification Process	11
IX.B.3.	Starting Prices	16
IX.B.4.	Extraordinary Events	
IX.C.	Bidding Rules	
IX.C.1.	Overview of Auction Format	18
IX.C.2.	Round Phases and Bidding Day	
IX.C.3.	Round 1 of the Auction	
IX.C.4.	Bidding in Round 2	
IX.C.5.	Calculating and Reporting Phases in Round 2	
IX.C.6.	Round 3 and All Subsequent Rounds	
IX.C.7.	Reporting of Total Excess Supply	
IX.C.8.	Calculation of Next Round Prices	
IX.C.9.	Pauses in the Auction	
IX.C.10.	Failure to Submit a Bid	43
IX.C.11.	End of Auction	48
IX.D.	Post Auction	49
IX.E.	Association and Confidential Information Rules	
IX.E.1.	Process for Reporting Associations, Identifying Concerns and Remedies	
IX.E.2.	Association Criteria	
IX.E.3.	Definitions of Confidential Information.	
IX.E.4.	Certifications and Disclosures to Be Made	
IX.E.5.	Actions to Be Taken If Certifications Cannot Be Made	
IX.F.	Glossary of Auction Terms	
IX.F.1.	Associated With	
IX.F.2.	Auction Volume	
IX.F.3.	BGS	
IX.F.4.	BGS Load	
IX.F.5.	BGS-CIEP Load	
IX.F.6.	BGS-RSCP Load	
IX.F.7.	BGS-RSCP Peak Load Share	
IX.F.8.	Bid	
IX.F.9.	Bidding Phase	
IX.F.10.	Calculating Phase	
IX.F.11.	CIEP Load	
IX.F.12.	Contingency Plan	
IX.F.13.	Decrement	
IX.F.14.	Denied Switches	
	EDC Load Can	62

IX.F.16.	EDC-Specific Summer Factor	62
IX.F.17.	EDC-Specific Winter Factor	62
IX.F.18.	Eligibility and Initial Eligibility	
IX.F.19.	Eligibility Ratio	
IX.F.20.	End of Auction	
IX.F.21.	Excess Supply	63
IX.F.22.	Exit Price	63
IX.F.23.	Extension	
IX.F.24.	Financial Guarantee	64
IX.F.25.	Free Eligibility	
IX.F.26.	Full-Requirements Service	
IX.F.27.	Going Price	
IX.F.28.	Indicative Offer	
IX.F.29.	Initial Eligibility	
IX.F.30.	Minimum and Maximum Starting Prices	
IX.F.31.	MW-Measure	
IX.F.32.	Outbid (Denied Switches)	65
IX.F.33.	Oversupply Ratio	
IX.F.34.	Part 1 Application	65
IX.F.35.	Part 1 Application Date	66
IX.F.36.	Part 2 Application	
IX.F.37.	Part 2 Application Date	66
IX.F.38.	Product in the Auction	66
IX.F.39.	Qualified Bidder	66
IX.F.40.	Recess	66
IX.F.41.	Reductions	
IX.F.42.	Registered Bidder	
IX.F.43.	Reporting Phase	
IX.F.44.	Round	67
IX.F.45.	Rounding	
IX.F.46.	RSCP Load	
IX.F.47.	RSCP Peak Load Share	
IX.F.48.	Session	68
IX.F.49.	Starting Prices	
IX.F.50.	Statewide Load Cap	
IX.F.51.	Switches and Switching	
IX.F.52.	Switching Priority	68
IX.F.53.	Target Eligibility Ratio	
IX.F.54.	Time-Out	69
IX.F.55.	Total Excess Supply	
IX.F.56.	Tranche	
IX.F.57.	Tranche Size	
IX.F.58.	Tranche Target	
IX.F.59.	Withdrawal	
IX.G.	Appendix	
IX.G.1.	Tranches	70
TXZ CC 2	Decrease	7.1

IX.A. Introduction

IX.A.1. Overview

The four New Jersey Electric Distribution Companies ("EDCs") are Atlantic City Electric Company ("ACE"), Jersey Central Power & Light Company ("JCP&L"), Public Service Electric and Gas Company ("PSE&G"), and Rockland Electric Company ("RECO"). Approximately one-third of each EDC's Basic Generation Service Residential Small Commercial Pricing ("BGS-RSCP") Load was procured in an Auction held in February 2018 and is under contract until May 31, 2021. Approximately one-third of each EDC's BGS-RSCP Load was procured through an Auction held a year earlier in February 2017 and is under contract until May 31, 2020. The remainder of each EDC's BGS-RSCP Load was procured through an Auction held in February 2016 and these contracts will expire on May 31, 2019. The EDCs are proposing to procure full-requirements service for the period June 1, 2019 to May 31, 2022, through a single, statewide Auction to be held in February 2019. Each EDC's BGS-RSCP Load includes the load of all BGS customers except large customers in rate classes excluded from BGS-RSCP Load as described in Table IX-1 below.

An EDC's BGS-RSCP Load will be divided into a number of tranches, each representing the same percentage of that EDC's BGS-RSCP Load. The MW-measure of each tranche is announced prior to the Auction, and is expected to be close to 100 MW of RSCP Peak Load Share. Table IX-2 provides, for each EDC, the number of tranches that were procured in the 2017 BGS-RSCP Auction, the number of tranches that were procured in the 2018 BGS-RSCP Auction, and the number of tranches to be procured in the 2019 BGS-RSCP Auction.

Suppliers will be bidding for the right to serve a portion of BGS-RSCP Load for one or more EDCs over the three-year period from June 1, 2019 to May 31, 2022. The EDCs will procure their BGS-RSCP Loads through a simultaneous descending clock auction ("clock auction"). The clock auction proceeds in a series of rounds. During the bidding phase of each round, each bidder must indicate how many tranches of each EDC's BGS-RSCP Load the bidder wishes to serve given the prices announced by the Auction Manager. The prices will be in cents per kilowatt-hour. A bid is the number of tranches of each EDC's BGS-RSCP Load that a bidder wants to serve. After the bidding phase of a round, the Auction Manager reduces the price for the tranches of an EDC by a decrement if the number of tranches bid by all bidders is greater

than the number of tranches that are needed for that EDC. The Auction Manager then announces new prices for each EDC before bidding in the next round opens. The Auction continues and the prices tick down until, for each EDC, the total number of tranches subscribed falls to the point where it equals the number of tranches needed. When the Auction ends, the bidders holding tranches in the final round are the winners. All winners for an EDC's tranches receive the same price, as explained in section IX.C.11.

The payments to suppliers for an EDC are shaped using seasonal factors. Payments from June 1 through September 30 are shaped by the use of a multiplicative summer factor on the Auction price, and payments for the remaining months are shaped by the use of a winter multiplicative factor on the Auction price. The factors by which the Auction prices will be multiplied to obtain the summer and winter payments are specific to each EDC. Provisional seasonal factors are provided in section IX.A.3.

To the extent that the transmission rates change and that the Board of Public Utilities ("Board") approves a corresponding change to the BGS-RSCP customers' rates: (1) if such a change results in an increase in BGS-RSCP customers' rates, then any additional revenues collected as a result of such an increase, once approved by the FERC in a final order and no longer subject to refund, would be to the benefit of BGS-RSCP suppliers; or (2) if such a change results in a decrease in BGS-RSCP customers' rates, then any decrease in revenues would result in a reduction of payments to BGS-RSCP suppliers. The parameters of what constitutes a change in transmission rates are specified in Section 15.9 of the BGS-RSCP Supplier Master Agreement.

At a given point in time, BGS-RSCP Load is served by suppliers that have won the right to supply BGS-RSCP for rolling three-year terms. On June 1, 2019, one-third of the BGS-RSCP Load will be served by suppliers with a three-year contract ending on May 31, 2020, one third of the BGS-RSCP Load will be served by suppliers with a three-year contract ending May 31, 2021, and the remaining third will be served by suppliers with a three-year contract ending May 31, 2022. The rates paid by BGS-RSCP customers are a function of the tranche-weighted average of the Auction prices obtained for the various contract terms adjusted for the seasonal payment factors. BGS-RSCP is not an hourly service. BGS-RSCP rates will be established on the basis of the Auction prices, filed with the Board and will be communicated to customers well in

advance of the supply period. Each EDC describes, in its Company Specific Addendum, the rate design methodology that it applies to its particular rate classes and rate structure. The EDCs will provide to potential bidders all necessary information concerning how Auction prices are translated into customer rates, including a calculating tool that displays the rate structure for each EDC that would result from given prices in the 2019 BGS-RSCP Auction. The rate design will apply from June 1, 2019 to May 31, 2020. The rate design may be revised for the supply period starting June 1, 2020.

Following the end of the Auction, the Board will decide whether to approve the Auction results and if it does, winners will become authorized BGS-RSCP suppliers. All winners for an EDC's tranches receive the same price, as explained in section IX.C.11.

IX.A.2. Basic Generation Service and Load to Be Procured

Basic Generation Service, or BGS, is the electric supply for those retail customers who are not served by a third party supplier. BGS Load is the load associated with these customers, and it is obtained by subtracting the load of third party suppliers from the retail load in an EDC zone. For each EDC, BGS Load is divided into two portions, BGS Commercial and Industrial Energy Pricing Load ("BGS-CIEP Load") and BGS Residential Small Commercial Pricing Load ("BGS-RSCP Load"). An EDC's BGS-CIEP Load is defined to include the sum of the hourly load of all BGS-CIEP customers, adjusted for losses. An EDC's BGS-RSCP Load is obtained by subtracting BGS-CIEP Load from BGS Load. Table IX-1 provides the list of rate classes for which either all customers, or only customers with a peak load contribution of 500 kW or greater, must currently take BGS on a CIEP tariff or rate and that are thus excluded from each EDC's BGS-RSCP Load. All remaining customers are RSCP customers and such customers may, if they choose BGS, take this service on an RSCP (rather than a CIEP) tariff or rate. On any meter reading date and with requisite prior notice, a customer taking BGS-RSCP may switch to a third party supplier, and an RSCP customer taking service from a third party supplier may switch to BGS-RSCP.

Table IX-1. Customers Who Must Take BGS on a CIEP Tariff (Excluded from BGS-RSCP)

EDC	Rate Class	Customers Included	Description	Voltage Level
	High Tension Service ("HTS-HV")	All	General purposes at high voltages	138,000V to 230,000V
PSE&G	High Tension Service ("HTS-Sub")	All	General purposes at subtransmission voltages	26,400V to 69,000V
	Large Power and Lighting, Primary Service ("LPL-P")	All	General purposes at primary distribution voltages.	2,400V to 13,200V
	Large Power and Lighting, Secondary Service ("LPL-S")	500 kW or greater	General purposes at secondary distribution voltages	208V to 480V
	General Service Primary ("GP")	All	General purposes at primary distribution voltages	2,400V to 34,500 YV
	General Service Transmission ("GT")	All	General purposes at transmission voltages	34,500∆V
JCP&L		All	High Tension Service	230,000V
	General Service Secondary ("GS")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	General Service Secondary Time-of-Day ("GST")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	Transmission General Service ("TGS")	All	General purposes at high voltages	23,000V or higher
	Annual General Service – Primary ("AGS-Primary")	500 kW or greater	General purposes at primary distribution voltages	4,000 & 12,000V
ACE	Annual General Service – Secondary ("AGS- Secondary")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
	Monthly General Service Primary ("MGS – Primary")	500 kW or greater	General purposes at primary distribution voltages	4,000 & 12,000V
	Monthly General Service Secondary ("MGS – Secondary")	500 kW or greater	General purposes at secondary distribution voltages	120 – 480V
RECO	Service Classification No. 7 – Primary TOU Service and Separately Metered Space Heating Service	All	Primary service customers with demands exceeding 500 kW	2,400V or higher
	Service Classification No. 2 – General Service	500 kW or greater	General primary and secondary service	All primary and secondary voltages

For purposes of the Auction, the BGS-RSCP Load for an EDC is called a *product in the Auction*: the BGS-RSCP Load for PSE&G for a three-year supply period is a product in the Auction, and the BGS-RSCP Load for RECO for a three-year supply period is another product in the Auction, etc. Each product in the Auction is divided into units called *tranches*, each representing the same percentage of that EDC's BGS-RSCP load in this Auction. The number of tranches that each EDC will procure in this Auction and the size of each tranche (i.e., the percentage of the BGS-RSCP Load that each tranche represents for an EDC) are provided below in Table IX-2. Table IX-2 also provides a MW-measure for each of an EDC's tranches. The MW-measure of each tranche is determined on the basis of the obligation attributable to all RSCP customers (those who receive BGS and those who do not) during 2018 (called the *RSCP Peak Load Share*). The data in Table IX-2 are provisional. Following the October 2018 update of this obligation by PJM, this table will be updated. The Auction Manager will provide the update no later than 10 days before the suppliers must first apply to participate in the Auction.

Table IX-2. Number of Tranches and Provisional MW-Measures of Tranches per EDC

		Tranches already procured		Tranches to be procured in 2019 for a three-year term			
EDC	RSCP Peak Load Share (MW)	Total number of tranches	2017 (one year remaining)	2018 (two years remaining)	Number of tranches	Size of tranche (%)	MW- Measure
PSE&G	7,641.16	85	28	29	28	1.18%	89.90
JCP&L ¹	4,508.97	53	15	20	18	1.89%	85.40
ACE	2,027.00	22	8	7	7	4.55%	92.14
RECO	400.00	4	2	1	1	25.00%	100.00
Total	14,577.13	164	53	57	54		

The EDCs do not represent that each tranche will have the loads shown above or any particular value. The actual BGS-RSCP Load will depend upon many factors including customer migration to third party suppliers and weather conditions. Bidders are responsible for evaluating the uncertainties associated with BGS-RSCP Load.

The Public Power Association of New Jersey ("PPANJ") has petitioned the Board requesting authorization to receive the entire St. Lawrence allocation on behalf of PPANJ's members. JCP&L will continue to serve the 10 MW of load associated with the St. Lawrence allocation pending a final determination by the Board.

IX.A.3. Payment to BGS-RSCP Suppliers

Winners at the Auction are authorized to become BGS-RSCP suppliers by the Board. A BGS-RSCP supplier provides *full-requirements service* for the percentage of the EDC's BGS-RSCP Load for a given supply period corresponding to the number of tranches won by the BGS-RSCP supplier for that supply period. Full-requirements service means that the BGS-RSCP supplier is responsible for fulfilling all the requirements of a PJM Load Serving Entity ("LSE") including capacity, energy, ancillary services and transmission, and any other service as PJM may require. A winning supplier may win one or more tranches for one or more EDCs.

To the extent that the rates change and that the Board approves a corresponding change to the BGS-RSCP customers' rates: (1) if such a change results in an increase in BGS-RSCP customers' rates, then any additional revenues collected as a result of such an increase, once approved by the FERC in a final order and no longer subject to refund, would be to the benefit of BGS-RSCP suppliers; or (2) if such a change results in a decrease in BGS-RSCP customers' rates, then any decrease in revenues would result in a reduction of payments to BGS-RSCP suppliers. The parameters of what constitutes a change in transmission rates are specified in Section 15.9 of the BGS-RSCP Supplier Master Agreement.

In the summer, each BGS-RSCP supplier for an EDC receives the Auction price for that EDC times an *EDC-specific summer factor* for every kWh of load served. In the winter, each BGS-RSCP supplier for an EDC receives the Auction price for that EDC times an *EDC-specific winter factor* for every kWh of load served. The load served by a BGS-RSCP supplier is equal to the BGS-RSCP supplier's share (the number of tranches won times the size of the tranche) times the EDC's BGS-RSCP Load.

Table IX-3 provides provisional EDC-specific seasonal factors. The final EDC specific seasonal factors will be determined no later than 6 days² before the Auction and, once determined, will apply for the entire three years of the supply period. The Auction Manager will announce any revisions to the EDC-specific seasonal factors.

² Unless otherwise specified, "days" refers to business days.

Table IX-3. Provisional EDC-Specific Summer Factors and Winter Factors

EDC	PSE&G	JCP&L	ACE	RECO
Summer Factor	1.0000	1.0000	1.0000	1.0000
Winter Factor	1.0000	1.0000	1.0000	1.0000

If, for an EDC, the rate design methodology results in an EDC-specific summer factor less than 1 and an EDC-specific winter factor greater than 1, then both EDC-specific factors will be set to 1. When both the EDC-specific summer factor and the EDC-specific winter factor are set to 1 for an EDC, BGS-RSCP suppliers for that EDC are paid the Auction price for that EDC for every kWh of load served in all months of the three-year supply period.

Example 1.3

Suppose that an EDC's summer factor is 1.0230 and its winter factor is 0.9857.

A bidder at the Auction wins 2 of the EDC's tranches at a price of 10.000¢/kWh. The bidder receives 10.230¢ (10.000¢ x 1.0230) for each kWh of load served in the summer and 9.857¢ (10.000¢ x 0.9857) for each kWh of load served in the winter.

IX.A.4. Reliance on Product Definition, Payment Information, Rates, and Customer Switching in These Rules

Information regarding the definition of a product in the Auction and information regarding the payment bases given in these Auction Rules are solely for the convenience of bidders and are not to be relied upon by bidders. The BGS-RSCP Supplier Master Agreement posted on the BGS Auction website is the document that provides the official definitions of products in the Auction and payment terms. Information regarding rates paid by customers and information on rules regarding the ability of customers to switch from or back to BGS-RSCP given in these Auction Rules are solely for the convenience of bidders and are not to be relied upon by bidders. Board Orders as well as the tariffs of each EDC are the documents that provide official information in this regard.

³ All examples are for illustrative purposes. Prices, EDC load caps, the statewide load cap, and bids are illustrative only.

IX.B. Before the Auction

IX.B.1. Information Provided to Bidders

The EDCs make data available to potential bidders in advance of qualification. The data series are posted to the BGS Auction website, http://www.bgs-auction.com/bgs.dataroom.asp.

The EDCs provide historical data, consisting of hourly load and daily capacity and transmission peak load allocations, for the following load categories: total retail, BGS, CIEP, BGS-CIEP, RSCP, and BGS-RSCP. The data include associated zonal losses.

The EDCs provide generally three years of historical data. All data series at a given point in time use the contemporaneous definition of CIEP customers. The BGS-CIEP data for a given month includes customers who take BGS on a CIEP tariff in force in that month. The CIEP data includes customers who take BGS on a CIEP tariff in force in that month as well as CIEP customers (as defined for that month) who are served by a third party supplier. BGS-RSCP and RSCP data are derived as residuals; for example, RSCP Load is equal to total retail load less CIEP Load, and BGS-RSCP Load is equal to the BGS Load less BGS-CIEP Load. The EDCs provide monthly customer switching data (number of customers and estimated load) as currently provided to the Board. Hourly load data series, peak load allocations, and switching statistics are updated monthly. Some historical data series older than three years remain posted to the BGS Auction website strictly for the convenience of bidders.

To the extent practicable, the EDCs will also provide additional data to assist bidders. The EDCs will provide one-time customer counts and historical aggregate energy usage for several groupings of customers who in the past were eligible to take BGS on an RSCP tariff but who are or will be required to take BGS on a CIEP tariff. These groupings may include: customers 500 to 749 kW, 750 to 999 kW, 1,000 to 1,249 kW, 1,250 to 1,499 kW, and 1,500 kW or greater. The EDCs may provide limited series of historical hourly load for specific customer classes or customer groupings for which customers have interval meters. The EDCs may provide load profiles for specific customer classes or customer groupings as well as additional historical customer switching data.

No later than 10 days before interested parties first apply to participate in the Auction, the Auction Manager will announce the *EDC load caps*, a *statewide load cap*, a statewide

maximum starting price, and a statewide minimum starting price. At the same time, the Auction Manager will provide the MW-measure of each tranche for each EDC, based on the percentage of the RSCP Peak Load Share that a tranche represents. An EDC load cap is a maximum number of tranches of BGS-RSCP Load that any one bidder can bid and serve for that EDC. The statewide load cap is a maximum number of tranches of BGS-RSCP Load that any one bidder can bid in the Auction and serve statewide. The statewide load cap cannot exceed the sum of the EDC load caps. An EDC load cap cannot exceed the statewide load cap. The statewide load cap limits the impact that any one bidder may have on the Auction; an EDC load cap limits an EDC's exposure to default by any single supplier in a given supply period. The minimum and maximum starting prices establish the range of possible starting prices for the Auction: each EDC will choose a starting level for its price for round 1 of the Auction that is between the minimum and the maximum starting prices. The EDCs will agree on the statewide load cap, and on the statewide minimum and maximum starting prices. Each EDC will set its EDC load cap. Board Staff and the Board Consultant will review these decisions.

The EDCs will update the provisional EDC-specific seasonal factors provided in these Auction Rules at the time of the compliance filing. This update will be included in the final Auction rules; however, the EDC-specific seasonal factors will still be provisional at that time. No later than 6 days before the Auction, the Auction Manager will announce the final EDC-specific seasonal factors that will apply for the entire three years of the supply period.

IX.B.2. Qualification Process

The application process is in two parts. All interested parties that have no impediments to meeting the PJM LSE requirements can submit a *Part 1 Application*. There is no state licensing requirement. Interested parties will be asked to submit financial information so that the EDCs can assess their creditworthiness. In addition, each interested party will be asked to comply with other qualification criteria that will have been agreed upon by all EDCs, including agreeing to comply with the BGS-RSCP Auction Rules and agreeing to the terms of the BGS-RSCP Supplier Master Agreement. Each interested party will also be asked to agree that if the interested party is successful in its Part 1 Application it will keep confidential the list of other successful applicants and it will not assign its rights or substitute another entity in its place. This is to ensure that the entity that agrees to the BGS-RSCP Auction Rules in the Part 1 Application

is also the entity submitting bids in the BGS-RSCP Auction, and to ensure that the entity that agrees to the terms of the BGS-RSCP Supplier Master Agreement is the entity that will execute the BGS-RSCP Supplier Master Agreement should the interested party become an Auction winner. In accordance with these Auction Rules, execution of the BGS-RSCP Supplier Master Agreement must occur within three days of Board certification of the Auction results and within that period the Auction winner will demonstrate compliance with the creditworthiness requirements set forth in the BGS-RSCP Supplier Master Agreement. Such creditworthiness requirements will take into consideration all BGS obligations held by the Auction winner, including those from past BGS Auctions.

Applications must be submitted no later than noon⁴ on the *Part 1 Application Date*, which will be no earlier than ten (10) days after the maximum and minimum starting prices have been announced. All interested parties will have at least ten (10) business days to complete the Part 1 Application. Interested parties will be notified as to whether they succeeded in qualifying to participate in the Auction no later than three days after the Part 1 Application Date. An interested party that has qualified becomes a *qualified bidder*. The Auction Manager will send simultaneously to each qualified bidder a list of all qualified bidders, but the list of qualified bidders will not be publicly disclosed. Interested parties, in their Part 1 Applications, will have undertaken to maintain the confidentiality of the list of qualified bidders, and to destroy documents with this information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results, as explained further in this document in section IX.E.4 in the "Association and Confidential Information Rules".

Qualified bidders that wish to participate in the Auction must submit a *Part 2 Application* to the Auction Manager. Only qualified bidders may submit Part 2 Applications. Part 2 Applications must be submitted no later than noon on the *Part 2 Application Date*, which will be no later than 10 days before the start of the Auction. In the Part 2 Application, qualified bidders will make a number of certifications to ensure compliance with the association and confidential information portion of these rules. These certifications, provided in section IX.E.4 below, ensure that each qualified bidder is bidding independently of other qualified bidders and ensure the confidentiality of information regarding the Auction. Each qualified bidder is also

⁴ Unless otherwise specified, all times are Eastern Time Zone times.

asked to agree to keep confidential the list of other successful applicants; to agree that the submission of any bid creates a binding and irrevocable offer to provide service under the terms of the BGS-RSCP Supplier Master Agreement; and not to assign its rights or substitute another entity in its place. On an exceptional basis, the Auction Manager may consider requests for substitution of another entity for a qualified bidder when the qualified bidder consolidates into, amalgamates into, or merges into another corporate entity. Such a request may only be considered if it is received prior to the Part 2 Application Date. This is to ensure that the entity that agrees to comply with the Association and Confidential Information Rules is also the entity submitting bids in the BGS-RSCP Auction, and to ensure that the entity that agrees that its bids create a binding and irrevocable offer to provide service under the terms of the BGS-RSCP Supplier Master Agreement is also the entity that will execute the BGS-RSCP Supplier Master Agreement should the qualified bidder become an Auction winner. With their Part 2 Application, qualified bidders will also be required to submit an *indicative offer* and to submit a *financial guarantee* in proportion to their indicative offer.

A qualified bidder is associated with another qualified bidder if the two bidders have ties that could allow them to act in concert or that could prevent them from competing actively against each other in the Auction. The competitiveness of the Auction and the ability of the Auction Process to deliver competitive prices may be harmed by the coordinated or collusive behavior that associations facilitate. The Auction Manager, who may rely, among other factors, on the number of independent competitors to set the Auction volume, would be using inaccurate information unless associations are duly disclosed in the Part 2 Application. The Auction volume is the number of tranches that the EDCs plan to purchase through the Auction. Associations may be considered in setting the Auction volume and may be used in the application of load caps. See section IX.E "Association and Confidential Information Rules" later in this document for precise criteria.

Sanctions can be imposed on a bidder for failing to disclose information relevant to determining associations, for coordinating with another bidder, or for failing to abide by any of the certifications that the bidder will have made in its Part 1 and Part 2 Applications. Such sanctions can include, but are not limited to, loss of all rights to serve any BGS-RSCP Load won in the Auction by such bidder, forfeiture of financial guarantees and other fees posted or paid,

prosecution under applicable state and federal laws, debarment from participation in future BGS Auctions, and other sanctions that the Board may consider appropriate. For any failure to disclose information or any violation of the certifications, the Auction Manager will make a recommendation to the Board on a possible sanction and the Board will make the final determination.

An indicative offer specifies two numbers of tranches. The first number represents the amount that the qualified bidder is willing to serve at the maximum starting price on a statewide basis (i.e., for all EDCs combined). The second number represents the amount that the qualified bidder is willing to supply at the minimum starting price on a statewide basis. At each of the maximum and the minimum starting prices, the number of tranches indicated by the qualified bidder cannot exceed the statewide load cap.

Indicative offers are important in two respects. First, the EDCs may use the indicative offers to inform their decision in setting the round 1 prices. Second, the number of tranches indicated by the qualified bidder at the maximum starting price determines the qualified bidder's *initial eligibility*. As explained in the bidding rules in section IX.C.3, a bidder will never be able to bid in the Auction on a number of tranches greater than the bidder's initial eligibility. Thus, the qualified bidder is encouraged to state the maximum possible number of tranches that it would be willing to serve.

All qualified bidders will also be required to provide their preliminary maximum interest in each EDC. The number of tranches for an EDC cannot exceed the EDC load cap. However, the sum of the qualified bidder's preliminary maximum interest in each EDC at a given price *can* exceed the statewide number of tranches that the qualified bidder is willing to serve at that price, and it can also exceed the statewide load cap. Information that a qualified bidder provides regarding its interest in any particular EDC has no effect on initial eligibility or subsequent bidding in the Auction.

Example 2.

Suppose that the EDC load caps are: 13 tranches for PSE&G; 8 tranches for JCP&L; 3 tranches for ACE; and 1 tranche for RECO (i.e., RECO has no EDC load cap). Suppose that the statewide load cap is 20 tranches.

A qualified bidder submits an indicative offer of 9 tranches at the maximum starting price. The qualified bidder's initial eligibility is 9 tranches, and the qualified bidder will never be able to bid on more than a total of 9 tranches during the Auction.

At the maximum starting price, the qualified bidder may indicate a preliminary maximum interest in as many as 9 tranches for PSE&G, 8 tranches for JCP&L, 3 tranches for ACE, and 1 tranche for RECO.

At the maximum starting price, the qualified bidder actually submits, as its preliminary maximum interest in each EDC:

- 9 tranches for PSE&G.
- 4 tranches for JCP&L, and
- 2 tranches for ACE.

The bidder does not have a preliminary maximum interest for RECO.

This means that of the 9 tranches that the qualified bidder is willing to serve statewide, the qualified bidder is willing to serve at most 9 of them for PSE&G, 4 for JCP&L, and 2 for ACE. The sum of the bidder's preliminary maximum interest in each EDC at the maximum starting price, 15 (9 + 4 + 2), exceeds the amount of the indicative offer at the maximum starting price (9).

Each qualified bidder must post a financial guarantee, in the form of a letter of credit (or bid bond), proportional to its initial eligibility. A financial guarantee of \$500,000 per tranche is required. Letters of credit (or bid bonds) must be in a form acceptable to the EDCs. Sample letters of credit and sample bid bonds that are acceptable to the EDCs will be posted to the Auction website. Depending upon the creditworthiness assessment made at the time of the Part 1 Application, additional security may be required. Any such additional security must be submitted in a form acceptable to the EDCs. Sample credit instruments for this additional security will be posted to the Auction website.

Example 3.

Suppose that the maximum starting price is 17.000¢/kWh and suppose that the statewide load cap is 20 tranches. A qualified bidder submits an indicative offer of 3 tranches at the maximum starting price. The qualified bidder is advised at the time of qualification that it is required to submit a financial guarantee of \$500,000 per tranche but that no additional security is required.

The qualified bidder posts a financial guarantee of \$1,500,000, in the form of a letter of credit (or bid bond). The letter of credit is in the standard form provided on the website and is thus acceptable to all EDCs. The number of tranches of the indicative offer does not exceed the maximum of the statewide load cap of 20 tranches.

For a Part 2 Application to be accepted, it must be complete, including its indicative offer, preliminary maximum interest in each EDC, financial guarantees and additional security (if necessary). The financial guarantees and additional security must be provided in a form acceptable to all EDCs and must be sufficient to cover the indicative offer at the maximum starting price. If its Part 2 Application is accepted, a qualified bidder becomes a *registered bidder*. The Auction Manager will send simultaneously to each registered bidder a list of all registered bidders and the total initial eligibility in the Auction. Neither the list of registered bidders nor the total initial eligibility in the Auction will be released publicly. Qualified bidders, in their Part 2 Applications, will have undertaken to maintain the confidentiality of the list of registered bidders and the total initial eligibility in the Auction, and to destroy documents with this information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results, as explained further in this document in section IX.E.4.

Financial guarantees and additional security (if required) will remain in full force until the Board renders a decision on the Auction results and the bidder has won no tranches, or until the Board renders a decision on the Auction results and the bidder has won tranches, has signed the BGS-RSCP Supplier Master Agreement, and has complied with all creditworthiness requirements of that agreement. The EDCs can collect the financial guarantees if bidders fail to comply with their obligations.

BGS-RSCP suppliers must meet PJM LSE requirements by the start of the supply period.

IX.B.3. Starting Prices

Three days before the Auction starts, the Auction Manager informs all registered bidders of each EDC's starting price, which will be the price in round 1 of the Auction. Each EDC's starting price will be no higher than the statewide maximum starting price and no lower than the statewide minimum starting price. Each EDC will set its own starting price in consultation with the Auction Manager, Board Staff and the Board Consultant.

IX.B.4. Extraordinary Events

The EDCs, in consultation with the Auction Manager, Board Staff and the Board Consultant, may determine that, due to extraordinary events, the statewide maximum starting price and the statewide minimum starting price require revision. In this event, the schedule may

also be revised. If the indicative offers have already been received, the Auction Manager would request that the registered bidders (or the qualified bidders if registration had not been completed) revise their indicative offers on the basis of the revised statewide maximum starting price and the revised statewide minimum starting price.

For such a revision to be necessary, an extraordinary event must occur between the time at which the statewide maximum starting price and the statewide minimum starting price are announced (no later than 10 days before the Part 1 Application is due) and the day on which the Auction starts. The EDCs, in consultation with the Auction Manager, Board Staff and the Board Consultant, will agree that an event constitutes an extraordinary event. Such events could include, for instance, the advent of war.

If an extraordinary event occurs during that time, the EDCs will determine a revised statewide maximum starting price and a revised statewide minimum starting price, and may also determine a revised schedule. Board Staff and the Board Consultant will review these decisions. New indicative offers will be required from bidders. The determination of new maximum and minimum starting prices, the submission of new indicative offers, and if necessary the announcement of new starting prices, will be carried out so as to afford bidders sufficient time.

The EDCs, in consultation with the Auction Manager, Board Staff, and the Board Consultant, may determine that, due to extraordinary events, one or more of the EDC-specific winter and summer factors require revision. In this event, the schedule may also be revised.

For a revision of one or more of the EDC-specific summer and winter factors to be necessary, an extraordinary event must occur after the final EDC-specific summer and winter factors have been announced but before the Auction starts. The Auction Manager would provide the opportunity for new indicative offers to be submitted on the basis of this revision. The Auction Manager will provide bidders sufficient time to revise their indicative offers.

IX.C. Bidding Rules

We first present an overview of the Auction format. We then proceed to explain the bidding and other rules in detail.

IX.C.1. Overview of Auction Format

The Auction is a simultaneous, multiple round, descending clock auction. We can explain this format's features by simply "unpacking" this terminology.

The Auction is called simultaneous because tranches for all the EDCs are put on offer through the same auction. The Auction proceeds in rounds. In a round, the Auction Manager announces a price for each EDC. Bidders bid by providing the number of tranches that they are willing to serve for each EDC at the prices announced by the Auction Manager. If the number of tranches bid is greater than number of tranches needed for an EDC, the price for that EDC is reduced for the next round. In the next round, bidders are given an opportunity to bid again.

The Auction is called a descending clock auction because prices "tick down" throughout the Auction, starting high and being reduced gradually until the supply bid is just sufficient to meet the load to be procured. Prices that tick down in a round decrease by a *decrement*; a decrement is a given percentage of the previous price. Bidders holding the final bids when the Auction closes are the winners.

Example 4.

There are 21 bidders in the Auction. Suppose that the EDC load caps are: 13 tranches for PSE&G, 8 tranches for JCP&L, 3 tranches for ACE, and 1 tranche for RECO (i.e., RECO has no EDC load cap). Consider the following sample round.

ROUND 1	5.000.57				
EDC	Price (¢/kWh)	# Tranches bid	# Tranches desired	Excess supply	Oversupply ratio
PSE&G	15.000	78	28	50	0.714
JCP&L	15.000	35	18	17	0.243
ACE	15.000	9	7	2	0.036
RECO	15.000	1	1	0	0.000

The Auction Manager reduces the price for an EDC if the number of tranches bid is greater than the number of tranches desired. The amount of the price reduction depends on the oversupply ratio, which is the ratio of the excess supply on that EDC to an estimate of the maximum possible excess supply on that EDC, taking into account the total excess supply in the Auction. Roughly speaking, the larger the oversupply ratio for an EDC, the larger is the portion of maximum excess supply that is actually on that EDC, and the larger is the price decrease. A formula for the oversupply ratio is provided in section IX.G.2.

In round 1, all bidders combined stand ready to supply 78 tranches of PSE&G at a price of 15.000¢/kWh. The number of tranches bid (78) exceeds the number of tranches desired (28) by 50 tranches. The price for PSE&G will tick down.

The actual excess supply on PSE&G is 50 tranches. The maximum possible excess supply is the total excess supply for all the EDCs in the Auction, or 69 (50 + 17 + 2 + 0). If all the excess bids in the Auction had been bid on PSE&G, PSE&G would have 69 tranches of excess supply. The estimate of the maximum excess supply used to calculate the oversupply ratio is the upper bound of the range of total excess supply at the Auction reported to bidders, which is 70. Thus, the oversupply ratio for PSE&G is 0.714 (50 / 70). Roughly speaking, 71% of the excess supply in the Auction is bid on PSE&G.

The calculation for ACE is similar. In round 1, all bidders combined stand ready to supply 9 tranches of ACE at a price of 15.000¢/kWh. The number of tranches bid (9) exceeds the number of tranches desired (7) by 2 tranches. The price for ACE will tick down.

The actual excess supply on ACE is 2 tranches. For ACE, the maximum possible excess supply is not the total excess supply for all the EDCs in the Auction (69 calculated above). Indeed, with only twenty-one bidders in the Auction and a load cap of 3 tranches on ACE, the maximum possible number of tranches bid on ACE is 63 (3 x 21). The maximum possible excess supply on ACE is 56 (63 - 7 = 56). For ACE, the oversupply ratio is 0.036 (2 / 56).

The oversupply ratio is calculated in the same way for all other EDCs.

The Auction Manager will lower the price in round 2 for every EDC except RECO, since for every EDC except RECO the number of tranches bid exceeds the number of tranches needed.

The largest decrement will be for PSE&G, which has the largest oversupply ratio, and the smallest decrement will be for ACE, which has the smallest oversupply ratio.

In round 2 below, prices have fallen from round 1 for all but RECO. The price for PSE&G, which had the largest decrement from round 1, fell the most; the price for ACE, which had the smallest decrement from round 1, fell the least. Bidders submit new bids at these prices. The total excess supply range reported to bidders is 56-60 (so that 60 is used as the measure of total excess supply in calculating the oversupply ratio).

ROUND 2	•	*			
EDC	Price (¢/kWh)	# Tranches bid	# Tranches desired	Excess supply	Oversupply ratio
PSE&G	14.250	60	28	32	0.533
JCP&L	14.637	38	18	20	0.333
ACE	14.813	9	7	2	0.036
RECO	15.000	5	1	4	0.200

Each EDC has more tranches bid than tranches desired. The Auction Manager will calculate the decrement for each EDC from that EDC's oversupply ratio and lower the price of each EDC for the next round accordingly.

Although we describe the main points of the Auction Rules in more detail below, four aspects should be briefly highlighted at the outset. These are as follows:

1. Winners for each EDC are not determined until the bidding has closed for all EDCs. When the number of tranches bid in a round for an EDC does not exceed the number of

tranches desired, the price for that EDC will not tick down for the next round. However, as the Auction progresses and the prices for the other EDCs tick down, some bidders may re-assign tranches and increase the number of tranches bid on that EDC, which may cause the price for the EDC to tick down again. Hence, the winners cannot be determined for any one EDC until bidding stops for all EDCs. There is no matching of suppliers to customers during the course of the Auction or selection of winners by the Auction Manager. It is only at the end of the Auction that suppliers reveal themselves to be winners by not withdrawing from the Auction.

- 2. If a bidder bid on an EDC in the preceding round and the price for the EDC's tranches did not tick down for the current round, the bidder cannot reduce the number of tranches bid for that EDC in the current round, either through a withdrawal or a switch (these terms are defined in sections IX.C.4.b and IX.C.4.c). Any bid is a binding obligation to accept the BGS-RSCP supply responsibility at the price at which the bid was made for the three-year supply period. The bidder may be able to reduce the number of tranches bid on that EDC later in the Auction. If other bidders increase their number of tranches bid on the EDC (because its price has remained high relative to the prices for other EDCs) and cause the price for that EDC to tick down, the bidder will once again be able to reduce the number of tranches bid on that EDC.
- 3. Bidders can never increase the total number of tranches they bid during the Auction. If a bidder does not bid a tranche in the first round, that tranche cannot be bid later on. Once a tranche is withdrawn, it can never be bid again.
- 4. All bidders that win tranches for a particular EDC and are authorized as suppliers by the Board receive the same price per kWh of load served for that EDC during the supply period.

IX.C.2. Round Phases and Bidding Day

Each round of the Auction is divided into three phases: a bidding phase, a calculating phase, and a reporting phase.

In the bidding phase of the round, bidders place bids. To be valid, a bid must be submitted and verified in the bidding phase and processed by the Auction software. Bidders should allow

time within the bidding phase of the round for submission, verification, and confirmation of the bid by the Auction software. The time-stamp of a bid is the time at which the bid is confirmed. A bidder that submits a bid in a round may change this bid as long as the bidding phase of the round is still open. The last valid bid by the bidder in the round becomes a firm offer to supply that cannot be rescinded.

In the calculating phase of the round, the Auction Manager tabulates the results of that round's bidding phase and calculates the prices for the next round. During this phase, bidders cannot submit bids and bidders do not yet have access to the results from that round's bidding phase.

In the reporting phase of the round, the Auction Manager informs bidders that are still in a position to win at the Auction of the results of that round's bidding phase. All Auction results are confidential as explained below in section IX.E.3. Bidders are informed of the *going prices* for the next round's bidding phase and are provided with a range of total excess supply left in the Auction. The going prices in a round are the prices at which the Auction Manager solicits bids in the bidding phase of that round. (Total excess supply is defined more precisely in section IX.C.5 and information provided concerning total excess supply is specified in more detail in section IX.C.7 below.) A bidder receives no information regarding another bidder's bid. Each bidder privately receives the results of its own bid from that round, indicating to each bidder its obligation at this point in the Auction.

A typical schedule for a bidding day will have a number of rounds in a morning session, a lunch break, and then a similar number of rounds scheduled in an afternoon session. The round times will speed up over the course of the Auction as bidders become more familiar with the process and bidding becomes more routine.

IX.C.3. Round 1 of the Auction

IX.C.3.a. Definition of a Bid

A bidder selects how many tranches it wants to serve for each EDC at the round 1 prices. In round 1, a *bid* then consists of four numbers:

a number of tranches that the bidder wants to supply for PSE&G at the PSE&G round 1 price;

- a number of tranches that the bidder wants to supply for JCP&L at the JCP&L round 1 price;
- a number of tranches that the bidder wants to supply for ACE at the ACE round 1 price;
 and
- a number of tranches that the bidder wants to supply for RECO at the RECO round 1 price.

The EDCs are always ranked in decreasing order of the *tranche targets*; a tranche target is the number of tranches needed for a given EDC.

The number of tranches that a bidder chooses for one EDC may or may not be the same as the number of tranches that the bidder chooses for another EDC. A number of tranches is an integer (0,1,2,...). A number of 0 (zero) for one EDC means that at the round 1 price for the EDC the bidder does not want to supply any of the BGS-RSCP Load for that EDC.

Example 5.

zampie si								
Suppose that the round 1 prices are:								
EDC	PSE&G	JCP&L	ACE	RECO				
Price (¢/kWh)	14.750	15.000	15.000	15.000				
Then (10, 0, 3, 1) is a round 1 bid and it indicates that the bidder stands ready to supply 10 PSE&G tranches, no JCP&L tranches, 3 ACE tranches, and 1 RECO tranche at the round 1 prices.								
EDC	PSE&G	JCP&L	ACE	RECO				
Price (¢/kWh)	14.750	15.000	15.000	15.000				
Bid (tranches)	10	0	3	1				

IX.C.3.b. Bidding Phase

The Auction Manager informs bidders of the starting prices for each EDC three days prior to the Auction; these starting prices are the prices in force, or the going prices, for round 1 of the Auction. The going prices in a round are the prices at which the Auction Manager solicits bids in the bidding phase of that round.

Any bid submitted in round 1 must satisfy two conditions; the first condition relates to the total number of tranches bid while the second relates to the number of tranches bid for any one EDC. First, the total number of tranches bid cannot exceed the bidder's initial eligibility. As explained in section IX.B.2, the bidder's initial eligibility is equal to the number of tranches in the bidder's indicative offer at the maximum starting price. As the bidder's indicative offer

cannot exceed the statewide load cap, the bidder's bid automatically satisfies the statewide load cap. Second, the number of tranches bid for any one EDC cannot exceed the EDC load cap.

Example 6.

Suppose that the EDC load caps are: 13 tranches for PSE&G, 8 tranches for JCP&L, 3 tranches for ACE, and 1 tranche for RECO (i.e., RECO has no EDC load cap). Suppose that the statewide load cap is 20 tranches.

Bidder A submitted an indicative offer of 20 tranches at the maximum starting price. Bidder A's initial eligibility is 20 tranches.

Bidder A's bid in round 1 is (10, 3, 3, 1). This bid satisfies all requirements:

- In total, Bidder A is bidding on 17 tranches, which does not exceed Bidder A's initial eligibility of 20 tranches.
- For each EDC, Bidder A is bidding on a number of tranches that does not exceed that EDC load cap. For example, Bidder A is bidding 3 tranches for ACE, which does not exceed the EDC load cap of 3.

As explained further in section IX.C.9, the bidding phase in round 1 is automatically extended for the convenience of bidders. If a bidder requests an extension in round 1, it will run concurrently with the automatic extension.

IX.C.3.c. Calculating Phase, Reporting Phase and Potential Volume Cutback

The calculating phase of round 1 immediately follows the extended bidding phase. In the ordinary course of events, the Auction Manager reviews the results and sets the prices that will be in force in round 2 of the Auction. Round 1 moves to the reporting phase and the Auction Manager reports to bidders the results of bidding in round 1 as well as the round 2 prices. The price for an EDC is the same as the round 1 price when the number of tranches bid in round 1 is insufficient to fill the tranche target for the EDC, or when the number of tranches bid for the EDC is just equal to the tranche target for the EDC. The price for an EDC ticks down when the number of tranches bid on that EDC exceeds its tranche target. Details on the amount by which the price in each round is reduced are given in section IX.G.2. The Auction Manager also provides to bidders an indication of the total excess supply in the Auction in round 1, as explained in section IX.C.7. A bidder is not provided any information regarding any other individual bidder's bids.

The Auction Manager may call a pause in the Auction during the calculating phase of round 1. This pause is called a *time-out* (see section IX.C.9 for additional details on time-outs).

The Auction Manager will call this time-out if the Auction Manager needs to consider whether to cut back the Auction volume to ensure the competitiveness of the Auction. It is not expected that the Auction Manager would revise EDC load caps at that time except if the new tranche target would fall below the EDC load cap, but the Auction Manager retains the discretion to make revisions to the EDC load caps based on the revised Auction volume. It is not expected that the Auction Manager would revise the statewide load cap, but the Auction Manager retains the discretion to make such a revision based on the revised Auction volume, and will do so when the reduced Auction volume falls below the statewide load cap.

As soon as practicable during the time-out, the Auction Manager will either announce that the Auction volume will not change, or will announce the revised Auction volume, the revised tranche target for each EDC, the revised statewide load cap (if applicable), and the EDC load caps (if applicable).

As soon as practicable during the time-out, the Auction Manager will start the reporting phase of round 1. The Auction Manager will report to the bidders the prices in force for round 2 as well as an indication of the total excess supply in the Auction in round 1. (Total excess supply is defined in section IX.C.5.) The manner in which the tranche targets for the EDCs will be changed on the basis of the revised Auction volume will be announced no later than 10 days before the Part 2 Application is due.

If an EDC load cap is reduced, and if a bidder's number of tranches bid on that EDC is greater than the reduced EDC load cap, then the Auction Manager reports that the bidder has a number of tranches bid on the EDC just equal to the reduced EDC load cap. The Auction Manager also reports that the tranches that the bidder had bid on that EDC in excess of the reduced EDC load cap can be freely bid on any EDC in round 2. From round 1 to round 2, the bidder does not lose the eligibility associated with the tranches that were bid in excess of the reduced EDC load cap. If the statewide load cap is reduced because it exceeds the reduced Auction volume, the Auction Manager reports to a bidder whose eligibility is greater than the reduced statewide load cap that the bidder's bid has been adjusted to conform to the reduced statewide load cap. The bidder withdraws any excess eligibility in round 2.

If the Auction volume is cut back, each EDC will implement a *Contingency Plan* for its tranches that have been removed from the Auction. Under their Contingency Plans, the EDCs

will purchase necessary services to serve BGS-RSCP Load for one year, including installed capacity, energy, transmission, and ancillary services through PJM-administered markets. Pursuant to these Contingency Plans, which are described in more detail in the Company Specific Addenda, the EDCs will not negotiate contracts with suppliers for BGS-RSCP Load after the Auction.

The Auction Manager will use a confidential set of guidelines approved by the Board to decide whether to cut back the Auction volume and to determine the magnitude of any necessary cutback. If the Auction volume is cut back, it will be cut back to the number of tranches bid in round 1 divided by a parameter called the *target eligibility ratio* (a desired ratio of tranches bid to the Auction volume). The precise value of this parameter depends on various factors, such as the number of bidders and characteristics of individual bids.

Example 7.

In the bidding phase of round 1, the Auction volume is 54 tranches (this is the sum of the tranche targets for all EDCs in the Auction: 28 + 18 + 7 + 1 = 54). Suppose that a total of 72 tranches are bid. Further, suppose that given the number of bidders and the characteristics of the bids, the Auction Manager sets the target eligibility ratio at 1.5.

The actual eligibility ratio is approximately 1.33 (72 / 54). The Auction Manager reduces the Auction volume to attain the target eligibility ratio of 1.5 by cutting back the volume by 6 tranches to 48 tranches (72 / 1.5 = 48).

After the volume cutback, there will be 1.5 tranches bid for every tranche to be purchased through the Auction (72 / 48 = 1.5).

The Auction Manager may further cut back the Auction volume on the basis of the bids as the Auction progresses, in accordance with the confidential guidelines approved by the Board. If such a revision is necessary to ensure a competitive bidding environment, the Auction Manager will call a time-out during the calculating phase of a round. As soon as practicable during the time-out, the Auction Manager will advise the bidders of the revised Auction volume, the revised tranche target for each EDC, the revised statewide load cap (if applicable), and the revised EDC load caps (if applicable).

No later than three days before the start of the Auction, the Auction Manager may release further information regarding the possible values of the target eligibility ratio and the circumstances under which a second volume cutback may be undertaken.

IX.C.4. Bidding in Round 2

IX.C.4.a. Bid and Eligibility

A bidder selects how many tranches to serve from each EDC at the round 2 prices.

As stated in the introduction to the Auction format in section IX.C.1, a bidder cannot increase its total number of tranches bid at the round 2 prices from its total number of tranches bid in round 1. This is more generally expressed by saying that the bidder cannot exceed its *eligibility*. A bidder's eligibility in round 2 is the bidder's total number of tranches bid⁵ in round 1. As in round 1, the number of tranches bid on any one EDC cannot exceed the EDC load cap for the EDC.

A bidder can always select the same number of tranches for each EDC in round 2 as in round 1. A bidder that wishes to change the number of tranches bid on a given EDC can reduce the number of tranches bid as long as the price for that EDC has ticked down. Such *reductions* in the number of tranches bid on an EDC for which the price has ticked down can be in the form of a *withdrawal* or a *switch*. A bidder reduces the number of tranches bid on an EDC through a withdrawal when the bidder is reducing the number of tranches bid on an EDC, not increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total. A bidder reduces the number of tranches bid on an EDC through a switch when the bidder is reassigning the total number of tranches bid so that the number of tranches bid on one or several EDCs is reduced, but the number of tranches bid on other EDCs is increased by the same amount. If a bidder re-assigns tranches, the total number of tranches that the bidder bids on at least one EDC will increase, but the total number of tranches bid on all EDCs cannot increase. This is explained in more detail below in sections IX.C.4.b and IX.C.4.c.

In addition to specifying the number of tranches that a bidder is prepared to serve for each EDC, fully specifying a bid in round 2 may require the bidder to provide *switching priorities* or *exit prices* (defined below). A bidder is required to provide switching priorities when increasing the total number of tranches bid on two or more EDCs, as explained in section IX.C.4.c. A bidder

⁵ Given that the bidder's round 1 bid satisfied the statewide load cap, and given that the bidder cannot increase its total number of tranches bid, the bidder's bid will satisfy the statewide load cap in all subsequent rounds.

is required to provide exit prices if a bidder is reducing the number of tranches bid on an EDC through a withdrawal, as explained in section IX.C.4.b.

In reporting to the bidder the result of its bid in the reporting phase, the Auction Manager will always report that it accepts a bid that is unchanged. The Auction Manager may disallow reductions that a bidder wants to make from an EDC. The Auction Manager retains withdrawn tranches if, by accepting all reductions from withdrawals and switches, the tranche target for that EDC would no longer be filled. The Auction Manager denies reductions from switches if, after retaining all withdrawn tranches from that EDC, accepting all reductions from switches would prevent the tranche target for that EDC from being filled. To fill the tranche target of an EDC, the Auction Manager first takes tranches that are bid at the round 2 price, then retains tranches that bidders want to withdraw, and finally denies reductions from switches that bidders have requested, as necessary. As explained in more detail below, the Auction Manager retains tranches that a bidder has withdrawn at the exit price named by the bidder and keeps switches that must be denied at the last price at which the bidder freely bid these tranches.

IX.C.4.b. Reductions through Withdrawals

A bidder can withdraw tranches from an EDC only if the price for that EDC has decreased from round 1 to round 2. If, instead, the price for an EDC has not changed from round 1 to round 2, then the bidder's offer in round 1 at that price is still standing, and the bidder cannot bid on fewer tranches for that EDC at the (unchanged) going price. As explained below, a bidder can always bid more tranches for an EDC whose price has not changed from round 1 by reducing the number of tranches from other EDCs whose prices have ticked down and switching them to the EDC whose price has not ticked down.

A bidder that withdraws one or more tranches from an EDC must name an exit price for those tranches. A bidder names an exit price when it bids one or more tranches at the previous going price but is unwilling to bid these tranches at the current going price. A bidder that withdraws several tranches previously bid at the round 1 price for a given EDC must specify the same exit price for all tranches from that EDC. An exit price must be less than or equal to the last price at which the tranches were freely bid (in round 2, this is the price in round 1) and must be higher than the EDC's going price (in round 2, this is the round 2 price, a price at which the

bidder is no longer willing to bid the tranches being withdrawn). A bidder that withdraws tranches from more than one EDC can specify a different exit price for each EDC.

An exit price enables the Auction Manager to determine which bidder would have remained ready to serve an EDC had the price ticked down continuously rather than in lumpy, discrete decrements. The Auction Manager relies on exit prices when tranches are withdrawn from an EDC and when the number of tranches bid on that EDC at the round 2 price falls short of that EDC's tranche target due to reductions from withdrawals and possibly also due to reductions from switches. The tranches with lower exit prices are retained first, and they are retained at the exit price that the bidder has named.

If, to fill the last tranches of the tranche target of an EDC, the Auction Manager must retain some but not all the tranches from two or more bidders that named the same exit price, then the Auction Manager, for each tranche to be retained, will choose at random the bidder whose tranche is retained. Thus, for the first tranche needed at the tied exit price, the probability that a particular bidder is chosen is the number of tranches that the bidder has bid at the exit price divided by the total number of tranches bid at the exit price. If a second tranche is needed at the exit price, the Auction Manager again will choose at random a bidder whose tranche will be retained. The probability that any one bidder is chosen is the number of tranches that the bidder has bid at the exit price and that have not yet been retained divided by the total number of tranches bid at the exit price and that have not yet been retained. The Auction Manager repeats this procedure until the tranche target for the EDC is filled.

A bidder that withdraws tranches from an EDC loses the eligibility associated with these tranches, and forfeits the right to bid these tranches for the remainder of the Auction on any EDC. Eligibility is lost even if the withdrawn tranches are retained. Eligibility is always lost as a result of withdrawing tranches.

Example 8.

The round 1 prices and the round 1 bid for Bidder A are:						
EDC	PSE&G	JCP&L	ACE	RECO		
Round 1 Price (¢/kWh)	15.000	15.000	14.750	15.000		
Bid (tranches) 10 1 3 1						

The round 2 prices are:				
EDC	PSE&G	JCP&L	ACE	RECO
Round 2 Price (¢/kWh)	15.000	14.680	14.650	15.000

Bidder A cannot bid fewer tranches for PSE&G or RECO in round 2 because the prices for these two EDCs have not fallen from round 1.

Bidder A reduces the number of tranches bid for ACE from 3 to 2. The bidder enters an exit price of 14.700¢/kWh, which is between 14.750¢/kWh and 14.650¢/kWh. (The exit price could have been equal to 14.750¢/kWh but not equal to 14.650¢/kWh.)

IX.C.4.c. Reductions through Switches

When a bidder is switching and not withdrawing, a bidder is simultaneously decreasing the number of tranches bid for one or more EDCs and increasing the number of tranches for one or more EDCs while leaving the total number of tranches bid the same. As in the case when the bidder is reducing the number of tranches bid on an EDC because the bidder is withdrawing tranches, the bidder can reduce the number of tranches bid on an EDC by switching only if the price for that EDC has decreased in round 2. Again, if instead the price for an EDC has not ticked down, then the bidder cannot bid fewer tranches on that EDC (i.e., the bidder cannot withdraw tranches or switch tranches from that EDC).

Example 9.

Suppose that the EDC load caps are: 13 tranches for PSE&G, 8 tranches for JCP&L, 3 tranches for ACE, and 1 tranche for RECO (i.e., RECO has no EDC load cap). The following are the round 1 and round 2 prices and the bids for Bidder A:

EDC	PSE&G	JCP&L	ACE	RECO
Round 1 Price (¢/kWh)	15.000	15.000	14.750	15.000
Bid Round 1 (tranches)	7	4	3	1
Round 2 Price (¢/kWh)	15.000	14.680	14.650	15.000
Bid Round 2 (tranches)	9	4	1	1

In round 2, Bidder A is increasing by 2 the number of tranches bid on PSE&G. Also, Bidder A is reducing by 2 the number of tranches bid on ACE. Since the total number of tranches bid is the same (15), the bidder is switching. Bidder A can reduce the number of tranches bid on ACE since its price has decreased in round 2. Bidder A can bid for more tranches of PSE&G (the number of tranches bid for PSE&G in round 1 is less than the EDC load cap). If the price for an EDC does not change, a bidder cannot bid *fewer* tranches but can bid *more* tranches for that EDC.

When (and only when) a bidder is increasing the number of tranches bid at the going price on more than one EDC, the bidder must specify a *switching priority*. A switching priority is a rank assigned to each of the EDCs on which the bidder is increasing the number of tranches bid. The Auction Manager will use this switching priority only when accepting some, but not all, reductions involved in a switch that the bidder has requested. The Auction Manager will use this switching priority only when, to keep an EDC's tranche target filled, the Auction Manager must retain all tranches that were withdrawn out of that EDC and must deny some, but not all, reductions from that EDC that come from a bidder's switch. The Auction Manager first increases the number of tranches bid of the EDC to which the bidder has assigned the highest priority (1 is the highest priority). The switches that the Auction Manager denies to fill the tranche target of an EDC are retained at the price at which they were last freely bid (in round 2, this is the round 1 price).

If a bidder intended to reduce the number of tranches bid on an EDC through a switch, some or all of these reductions may be denied. The Auction Manager will deny reductions from switches if the number of tranches bid at the going price, and the number of tranches that can be retained from withdrawals, are not together sufficient to fill the tranche target of an EDC. If the Auction Manager must deny one or several reductions from switches from two or more bidders, the Auction Manager chooses at random the bidder whose switch will be denied.

For each tranche of the target that must be filled by denying a reduction from a switch, the Auction Manager chooses at random the bidder whose switch is denied. Thus, for the first switch that must be denied, the probability that the Auction Manager chooses a tranche bid by a particular bidder requesting a switch is the number of tranches by which the bidder's bid on the EDC is reduced by the switch and that could be denied, divided by the total number of tranches by which the number of tranches bid on the EDC is reduced by switches from all bidders and that could be denied. If a second switch must be denied, the Auction Manager again chooses at random the bidder whose switch will be denied. The probability that the Auction Manager chooses a tranche bid by a bidder requesting a switch is the number of tranches by which the bidder's bid on the EDC is reduced by the switch and that could have, but have not yet been denied, divided by the total number of tranches by which the number of tranches bid on the EDC is reduced by all switches from bidders and that could have, but have not yet been denied. The Auction Manager repeats this procedure until the tranche target for the EDC is filled. The

Auction Manager continues to report that some or all of these switches are being denied in subsequent rounds as long as they are still needed to fill the EDC's tranche target.

Example 10.

Bidder B submits the following bid in round 2 after this history in round 1:				
EDC	PSE&G	JCP&L	ACE	RECO
Round 1 Price (¢/kWh)	15.000	15.000	14.750	15.000
Round 1 Bid (tranches)	7	4	2	1
Round 2 Price (¢/kWh)	15.000	14.680	14.650	15.000
Round 2 Bid (tranches)	8	2	3	1

In round 2, Bidder B is increasing the number of tranches bid on ACE and PSE&G while reducing the number of tranches bid on JCP&L. Since the total number of tranches bid is the same (14) in both rounds, the bidder is switching. Bidder B gives the switching priority to the increase on ACE.

If the Auction Manager will allow one of Bidder B's reductions but needs to deny the other, then the increase for ACE will be allowed and the increase for PSE&G will not be allowed. The denied switch will remain a JCP&L tranche with a price of 15.000¢/kWh, which is the last price at which the tranche was freely bid.

EDC	PSE&G	JCP&L	ACE	RECO
Round 2 Price (¢/kWh)	15.000	14.680	14.650	15.000
Round 2 Accepted Bid	7	2 @ 14.680	3	1
(tranches)		1 @ 15.000		

IX.C.4.d. Withdrawing and Switching Tranches Simultaneously

If a bidder is both switching and withdrawing, a bidder can reduce tranches from a particular EDC only if the price for that EDC has decreased in round 2. Additionally, if the bidder is switching and withdrawing, the bidder may be asked to specify which tranches are being withdrawn and which tranches are being switched. The bidder would be asked for this additional information, for instance, when the bidder is switching and withdrawing, and the bidder is reducing the number of tranches bid for more than one EDC. The tranche or tranches that the bidder specifies to be withdrawn are the tranche(s) for which the bidder will name an exit price.

Example 11.

Bidder C submits the following bid in round 2 after this history in round 1:				
EDC	PSE&G	JCP&L	ACE	RECO
Round 1 Price (¢/kWh)	15.000	15.000	14.750	15.000
Round 1 Bid (tranches)	10	3	3	0
Round 2 Price (¢/kWh)	14.760	15.000	14.650	15.000
Round 2 Bid (tranches)	9	4	1	1

Bidder C bids a total of 16 tranches in round 1 and 15 tranches in round 2. Bidder C is withdrawing 1 tranche and will name an exit price for that tranche.

Since Bidder C reduces the number of tranches bid for both PSE&G and ACE, Bidder C's intentions are not clear unless the Auction Manager asks Bidder C for additional information. Indeed, it could be that Bidder C is withdrawing 1 tranche from PSE&G and switching the remaining tranches. Or it could be that Bidder C is withdrawing 1 tranche from ACE and switching the remaining tranches.

Bidder C is asked to select whether it is withdrawing a tranche from PSE&G or ACE. Bidder C selects that it is withdrawing a tranche from PSE&G and is asked to name an exit price for this tranche. Since Bidder C is increasing the number of tranches bid for more than one EDC (JCP&L and RECO), Bidder C is also asked for switching priorities. Bidder C assigns the first priority to RECO.

IX.C.5. Calculating and Reporting Phases in Round 2

The calculating phase starts immediately after the bidding phase. Once the Auction Manager has tabulated and reviewed the results, the reporting phase begins. The Auction Manager informs all bidders of the round 3 price for each EDC. The Auction Manager provides to all bidders a range for the *total excess supply* in the Auction for round 2. The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus free eligibility, which is defined below in section IX.C.6. The range of total excess supply reported to bidders will change as the Auction progresses. When total excess supply is high, this range will count as few as 5 integers; as total excess supply decreases, the range will become wider and count 10 or more integers. The exact ranges of total excess supply provided as the Auction progresses are specified in detail in section IX.C.7.

In addition to what the Auction Manager tells all bidders about the general progress of the Auction, the Auction Manager reports privately to each bidder the outcome of its own bid. (The Auction Manager never reports the bid of one bidder to another bidder.) If the round 2 bid is unchanged from round 1, or if requested reductions (from switches or withdrawals) all have been granted, the Auction Manager reports the bid made in round 2.

If the bidder intended to reduce the number of tranches bid on an EDC by withdrawing tranches and some or all of these tranches are retained, the Auction Manager informs the bidder of the number of withdrawn tranches that are being retained and the price at which these tranches are retained. This can occur only if the EDC's number of tranches bid in round 1 exceeded its tranche target while the number of tranches bid at the round 2 price is not sufficient to fill its tranche target. The price at which the withdrawn tranches are retained is the exit price. The Auction Manager will continue to report that some or all of these tranches are being retained in subsequent rounds as long as they are needed to fill the EDC's tranche target.

If a bidder intended to reduce the number of tranches bid on an EDC through a switch, but if some or all of these reductions are denied, the Auction Manager informs the bidder of the number of tranches for which the switch is denied. The price at which a switch is denied is the last price at which the tranche was freely bid. In round 2, this price is the round 1 price. A switch can be denied only if, for the EDC from which the number of tranches bid is reduced, the number of tranches bid in round 1 exceeded its tranche target, and the number of tranches bid at the round 2 price plus the retention of all the withdrawals from the EDC are not sufficient to fill the tranche target.

Example 12.

Bidders A, B and C are the only bidders bidding for tranches of PSE&G, for which the tranche target is 28. None of these bidders are bidding on RECO. Their bids in rounds 1 and 2 are:

EDC	PSE&G	JCP&L	ACE
Tranche Target	28	18	7
Round 1 Price (¢/kWh)	15.000	15.000	14.750
Bidder A Bid (tranches)	10	0	0
Bidder B Bid (tranches)	10	0	0
Bidder C Bid (tranches)	9	0	0
Round 2 Price (¢/kWh)	14.920	14.680	14.750
Bidder A Bid (tranches)	9	1	0
Bidder B Bid (tranches)	8	1	1
Bidder C Bid (tranches)	9	0	0

Bidder A switches, increasing the number of tranches bid on JCP&L to 1 and reducing the number of tranches bid on PSE&G. Bidder B is also switching. Bidder C's bid remains the

same. In total, 3 fewer tranches are bid on PSE&G. The number of tranches bid on JCP&L and ACE are both increased. Bidder B gives ACE the switching priority.

No other bidders bid on tranches for PSE&G in round 1 or in round 2. In round 1, PSE&G's tranche target is more than filled with 29 tranches bid at the round 1 price. In round 2, the 26 tranches bid at the round 2 price are insufficient to fill the tranche target. The Auction Manager must deny 2 of the intended switches.

Two of the three reductions from PSE&G must be denied. For the first tranche that must be filled by denying a switch, the probability that each bidder is chosen is 1/3 for Bidder A and 2/3 for Bidder B. Bidder B is chosen at random. The Auction Manager repeats the procedure for the second tranche that must be filled by denying a switch. The probability that each bidder is chosen is 1/2 for Bidder A and 1/2 for Bidder B. Bidder A is chosen at random. This means that the Auction Manager denies Bidder A its intended switch and that Bidder B is allowed to switch 1 of the 2 tranches requested. Bidder B gave ACE the switching priority. Adding the 2 denied switches to the 26 tranches bid at the round 2 price fills PSE&G's tranche target.

EDC	PSE&G	JCP&L	ACE
Report to Bidder A	9 @ 14.920¢ 1 denied switch @ 15.000¢	0	0
Report to Bidder B	8 @ 14.920¢ 1 denied switch @ 15.000¢	0	1@14.750¢
Report to Bidder C	9 @ 14.920¢	0	0

The Auction Manager reports to Bidder A that its switch was denied. The Auction Manager also reports to Bidder A that it has 9 PSE&G tranches bid at the round 2 price of 14.920¢/kWh and 1 denied switch, a PSE&G tranche at the round 1 price of 15.000¢/kWh. The Auction Manager reports to Bidder B that Bidder B has 8 tranches of PSE&G bid at the round 2 price of 14.920¢/kWh and 1 denied switch, a PSE&G tranche at the round 1 price of 15.000¢/kWh. The Auction Manager reports to Bidder B that Bidder B has 1 ACE tranche bid at 14.750¢/kWh (because of the switching priority). The Auction Manager also reports to Bidder C that Bidder C has 9 tranches of PSE&G bid at the round 2 price.

IX.C.6. Round 3 and All Subsequent Rounds

Rounds start with a bidding phase. When an EDC's price in a round has ticked down from the previous round, a bidder is free to maintain its number of tranches bid on the EDC or reduce its number of tranches bid on the EDC. Also, the bidder can increase its number of tranches bid on an EDC if the bidder has tranches bid on another EDC for which the price has ticked down. When an EDC's price has not ticked down from the previous round, a bidder must maintain its number of tranches bid on the EDC, or the bidder can increase its number of tranches bid on the EDC (subject to the condition just noted). To fully specify a bid, in certain

circumstances the bidder will be required to provide exit prices and switching priorities as explained in sections IX.C.4.b and IX.C.4.c.

At all times, the total number of tranches bid by a bidder cannot exceed the bidder's eligibility. The bidder's eligibility for a round is its eligibility at the start of the previous round minus the number of tranches that the bidder withdrew from the Auction in the bidding phase of the previous round. The total number of tranches bid by a bidder includes tranches bid at the going prices and denied switches that are kept at the last price at which they were freely bid.

At all times, the number of tranches that a bidder could win for an EDC cannot exceed the EDC load cap. In any given round, the number of tranches bid by a bidder at the going price for an EDC, the number of reductions from switches denied from that EDC, and the number of withdrawals retained for that EDC cannot together exceed the EDC load cap. The number of withdrawals retained for an EDC counts toward determining whether a bidder is within the EDC load cap, but a bidder can always choose to replace retained withdrawals with tranches bid at the going price (i.e., the current round price at which the Auction Manager is soliciting bids). If a bidder has retained withdrawals for the EDC and the bidder wishes to switch into the EDC in a manner that would exceed the EDC load cap, then the bidder's tranches bid at the going price will replace just enough of the bidder's own retained withdrawals to ensure that the bidder does not exceed the EDC load cap.

In the reporting phase that follows the bidding phase and the calculating phase, the Auction Manager reports to all bidders a range for the total excess supply left in the Auction, as described in section IX.C.7. The Auction Manager reports privately to a bidder on its bid. The Auction Manager reports any withdrawn tranches that are retained and any switches that are denied along with their associated prices, as described above in sections IX.C.4.b and IX.C.4.c. If a switch that had been denied in a previous round is freed to be bid once again, then the Auction Manager will report the number of tranches that are being freed.

To fill the tranche target for an EDC, the Auction Manager takes bids in increasing order of price. The Auction Manager first takes tranches bid at the current round price, then the Auction Manager retains withdrawn tranches, and finally the Auction Manager denies switches. Thus, when new tranches are bid at the current round price for an EDC from which switches had been denied, the new tranches replace the denied switches (partially or completely) in filling the

EDC's tranche target, and therefore free up some or all of these denied switches. A denied switch that is being replaced by a new tranche at the going price is then said to be *outbid*. If switches from more than one bidder are retained, and if not all denied switches are outbid, the Auction Manager chooses at random, for each denied switch that will be outbid, the bidder whose switch will be outbid. Thus, for the first denied switch that is outbid, the probability that the Auction Manager chooses a particular bidder's denied switch is the bidder's number of denied switches divided by the total number of denied switches for that EDC. If a second denied switch must be outbid, the Auction Manager again will choose at random the bidder whose denied switch will be outbid. The probability that the Auction Manager chooses a bidder's denied switch is the bidder's number of denied switches that have not yet been outbid divided by the total number of denied switches that have not yet been outbid. The Auction Manager repeats this procedure until the required number of denied switches has been outbid.

Once some denied switches are outbid for a bidder, the Auction Manager reports this fact to the bidder, along with the number of tranches that have been outbid. A denied switch that is outbid becomes *free eligibility* for the bidder in the next round. A tranche of free eligibility can be bid on any one of the EDCs in the next round, or it can be withdrawn in the next round, as explained further below.

The Auction Manager reports privately to a bidder if a withdrawn tranche that had been retained is now being released and thereby irrevocably removed from the Auction. Withdrawn tranches that had been retained at the highest exit prices are the first to be released and removed from the Auction. If withdrawn tranches from more than one bidder had been retained at the same exit price, and if not all retained tranches at that exit price are being released, the Auction Manager chooses at random the bidder or bidders whose tranches are released and thereby irrevocably removed from the Auction. Thus, for the first retained tranche that should be released, the probability that a particular bidder is chosen is the bidder's number of retained tranches at that exit price. If a second retained tranche needs to be released, the Auction Manager again will choose at random the bidder whose retained tranche will be released, and the probability that a particular bidder is chosen is the bidder's number of retained tranches at the tied exit price that have not yet been released divided by the total number of retained tranches at the tied exit price that have

not yet been released. The Auction Manager repeats this procedure until the required number of tranches has been released.

Two additional bidding rules come into play when a bidder has been denied a switch.

First, if a bidder holds tranches for an EDC from a switch that had been denied in a previous round, and if this bidder bids new tranches for this same EDC at the price for the current round, the bidder will be deemed to have bid any tranches from a previously denied switch at the current round price for that EDC. That is, tranches from the denied switch become tranches that are bid at the price for the current round. The Auction Manager, in filling the tranche target for the EDC, will take first tranches bid at the going price; in these tranches at the going price, the Auction Manager will include any denied switches that have become tranches bid at the current round price (because the bidder has bid new tranches for this same EDC at the current round price).

Second, if a bidder's denied switches are partially or totally outbid in a round and become free eligibility, and if the bidder withdraws one or more of these tranches of free eligibility in the very next round, then the bidder does not name an exit price for these tranches. No exit price is needed to withdraw a tranche of free eligibility because such tranches are not associated with any one EDC. The Auction Manager always accepts a withdrawal of tranches of free eligibility, and these tranches are always irrevocably removed from the Auction.

Example 13.

Bidder A's bids in rounds 6 and EDCs):	7 are given below (Bidder A do	oes not bid for any other
EDC	PSE&G	JCP&L
Price Round 6 (¢/kWh)	13.312	13.659
Bidder A Bid (tranches)	4	0
Price Round 7 (¢/kWh)	13.031	13.345
Bidder A Bid (tranches)	0	4

In round 7, Bidder A requests to switch 4 tranches. The Auction Manager denies part of the switch. Bidder A, in the reporting phase of round 7, is informed that its bid consists of 2 tranches of JCP&L at 13.345¢/kWh and 2 denied switches of PSE&G at 13.312¢/kWh (the price at which the tranches of PSE&G were last freely bid).

EDC	PSE&G	JCP&L
Report to Bidder A	2 denied switches @ 13.312¢	2 @ 13.345¢

In round 8, Bidder A reduces its number of tranches bid on JCP&L by 1 and increases its number of tranches bid on PSE&G. At the round 8 prices, Bidder A bids 1 tranche of JCP&L and 1 tranche of PSE&G at the going price. The denied switches are kept on PSE&G and cannot be freely bid.

EDC	PSE&G	JCP&L
Price Round 8 (¢/kWh)	13.031	13.058
Bidder A Bid (tranches)	1	1
Denied Switches	2 denied switches @	
Defined Switches	13.312¢	

Bidder A has bid a new tranche of PSE&G at the round 8 price while having switches denied on the same EDC at a higher price. Bidder A is then deemed to have bid all 3 PSE&G tranches at the round 8 price of 13.031¢/kWh, as shown below. All 3 tranches of PSE&G become tranches bid at the round 8 price.

EDC	PSE&G	JCP&L
Price Round 8 (¢/kWh)	13.031	13.058
Report to Bidder A	3 @ 13.031¢	1

IX.C.7. Reporting of Total Excess Supply

In addition to the next round prices, the Auction Manager provides all bidders that could still be winners with a range for total excess supply in the Auction. The actual total excess supply in the Auction lies within the reported range. The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus all tranches of free eligibility.

The manner in which the total excess supply in the Auction is reported changes as the Auction progresses. In general, when total excess supply is high, the Auction Manager announces a range for the total excess supply that includes exactly 5 integers. The largest integer of each such range will be divisible by 5. When the total excess supply falls to 40 or fewer tranches, the Auction Manager announces a range for the total excess supply that includes at least 10 integers. The largest integer of each such range will be divisible by 5. When the total excess supply falls to 20 or fewer tranches, the Auction Manager will report to all bidders only that the total excess supply is between 0 and 20 tranches.

In summary, the ranges are expected to be the following:

Table IX-4. Possible Ranges of Total Excess Supply

Total excess supply falls to 20 tranches or fewer: a single range remains	0 20
Total excess supply is between 21 and 40 tranches:	21 – 30
ranges count 10 integers	31 – 40
	41 – 45
Total excess supply is 41 tranches or more: ranges count 5 integers	46 – 50
	51 – 55 (etc.)

The Auction Manager may revise these ranges before the start of the Auction. The Auction Manager will inform the registered bidders no later than 3 days after registration of any changes in the possible ranges of total excess supply.

When the actual total excess supply is zero, the Auction ends, as explained in section IX.C.11.

A bidder that has no remaining obligation loses its access to the Auction software and is no longer provided with the range of total excess supply and the next round prices. A bidder has no remaining obligation when the bidder has zero eligibility and has no retained withdrawals. A bidder that has no remaining obligation can no longer win at the Auction. Such a bidder loses its access to the Auction software and is no longer provided with the range of total excess supply and the next round prices no earlier than the round after the bidder has been first informed that it has no remaining obligation. The bidder will lose its access as soon as practicable, but in no event later than eight rounds after it has no remaining obligation.

Round results that are reported to all bidders will also be reported to a list of representatives from the EDCs, the Board and the Board Consultant. The bidders, the EDCs, the Auction Manager, the Board and the Board Consultant hold any Auction results to which they have access to be confidential. Before being registered to participate in the Auction, the bidders will agree to keep all Auction results, as well as the list of qualified bidders, the list of registered bidders, and the total initial eligibility in the Auction confidential. The bidders will agree not to

disclose any such confidential information about the Auction Process, except for any aspects of the Auction results that the Board releases as part of rendering its decision on the Auction results, or that the Board explicitly authorizes can be released (see section IX.E.3). Bidders will also agree to destroy documents with Auction information provided by the Auction Manager within five days of the Board rendering a decision on the Auction results.

IX.C.8. Calculation of Next Round Prices

An EDC's price in a round is calculated so that it ticks down more quickly the larger is the excess supply on the EDC. The excess supply on an EDC is measured against an estimate of the maximum possible excess supply for the EDC. This estimate takes into account the total excess supply in the Auction, the EDC load cap, the tranche target of each EDC, and the number of registered bidders. A decrease in price for an EDC that has a number of tranches bid greater than the tranche target is called the decrement; the decrement is a percentage of the previous going price. The decrements are calculated using formulas. The Auction Manager has the discretion to override the calculated decrement for any EDC in any round. The Auction Manager does not expect to use such discretion, but may do so. When the Auction Manager, in the reporting phase of a round, uses discretion to override the decrement formulas and determine the going prices for the next round, the Auction Manager notifies bidders of that fact during the reporting phase of that round.

The decrement formulas allow the decrements to be larger at the start of the Auction than in later rounds. At the start of the Auction, when the number of tranches bid on an EDC exceeds the tranche target, the decrement is between 0.5% and 5% of the previous round price. Decrements for EDCs for which the number of tranches bid is greater than the tranche target continue to be between 0.5% and 5% of the previous round price until the going prices for round 4 are calculated. After that time, in the first round in which the upper bound of the total excess supply range reported to bidders is at least 20 tranches fewer than the upper bound of the total excess supply range reported to bidders in round 1 and the total excess supply is greater than 30 tranches, decrements (for EDCs for which the number of tranches bid is greater than the tranche target) will be between 0.375% and 3.75% of the previous round price. Decrements for EDCs for which the number of tranches bid is greater than the tranche target continue to be between 0.375% and 3.75% of the previous round price until total excess supply in the Auction is 30 or

fewer tranches. At that point, decrements for EDCs for which the number of tranches bid is greater than the tranche target will be between 0.25% and 2.5% of the previous round price and will remain so until the end of the Auction.

The excess supply of an EDC is the number of tranches bid at the going price minus the EDC's tranche target, or it is zero, whichever is greater. Prices are rounded off to the nearest thousandth of a cent per kWh.

Provisional decrement formulas are provided in section IX.G.2. The formulas that will be used in the Auction, which depend on the EDC load caps and the number of registered bidders, will be released to all bidders no later than three days after bidder registration.

IX.C.9. Pauses in the Auction

Any one bidder can pause the Auction by requesting an *extension* or a *recess*. A bidder requests an extension during the bidding phase of a round; such a request extends the bidding phase of the round by 15 minutes for all bidders. A bidder requests a recess during the calculating or reporting phase of the round; such a request suspends the Auction for a period of no less than 20 minutes. A recess allows a bidder additional time to consider its bid.

The bidding phase of round 1 is automatically extended. The bidding phase of a round can be extended only once by 15 minutes. Each bidder is allowed two extensions during the Auction. A bidder with positive eligibility is automatically deemed to have requested an extension when — by the scheduled ending time of the bidding phase — the bidder has not submitted a bid and when the bidder has not already used the two allowable extensions. A bidder with positive eligibility that has already used the two allowable extensions and that does not submit a bid during a round will be assigned a default bid as explained in section IX.C.10. Extension requests from all bidders are granted but all extensions run concurrently. All bidders that have requested an extension during the bidding phase of a round will see their available number of extensions reduced. The bidding phase in a round will be extended only once by 15 minutes. The Auction Manager reports to all bidders at the end of the planned bidding phase that the bidding phase has been extended.

A recess must be requested during the calculating phase or during the reporting phase. If the reporting phase of a round is scheduled to be shorter than 5 minutes, a recess must be requested during the scheduled calculating phase of the round. If the reporting phase is scheduled to be 5 minutes or longer, a recess must be requested before the scheduled last half of the reporting phase of a round starts, or before the last five minutes of the reporting phase of a round, whichever is earlier. (That is, if the reporting phase of a round is scheduled to be 8 minutes, then the recess must be requested before the last 5 minutes of the reporting phase; if the reporting phase of a round is scheduled to be 14 minutes, then the recess must be requested before the last 7 minutes of the reporting phase.) As soon as is feasible after the time at which a recess can be requested has passed, the Auction Manager reports to all bidders that a recess has been called. The Auction Manager retains the discretion to set the length of a recess but the Auction Manager will not set the recess time to be less than 20 minutes.

Each bidder is allowed to request at most one recess during the Auction. A bidder cannot request a recess in the first ten rounds of the Auction. Starting in the eleventh round, a bidder may request a recess in a round only if the total excess supply in the Auction reported in the previous round was 40 or fewer tranches. All recess requests are granted, but all requested recesses run concurrently. All bidders making a request in a given calculating or reporting phase will be deemed to have used a recess request. All bidders that have requested a recess will see their available number of recesses reduced to zero and the recess will last a length of time determined by the Auction Manager for one recess and will be no less than 20 minutes.

Example 14.

The total number of tranches in the Auction is 54.

In round 5, total excess supply is reported to be in the 96-100 range. Bidders are not able to request a recess in round 5 since round 11 has not yet been reached.

In round 12, the total excess supply is reported to be in the 31-40 range for the first time. Bidders can request a recess in round 13, since the total excess supply reported in the previous round (round 12) was at 40 tranches or below, and round 11 had been reached.

The Auction Manager can call a time-out to the Auction at any time during a round. It is intended that a time-out suspend activity in the Auction for a period of no more than four hours; however, the Auction Manager retains the discretion to suspend activity for a longer period, if necessary. Whenever a time-out is called, the Auction Manager reports to all bidders how long the time-out is expected to last.

During any bidding phase, the Auction Manager may call a time-out and extend bidding in case of technical difficulties. Such a time-out would typically last fifteen minutes. The Auction Manager reports to all bidders as soon as practicable that the Auction Manager has extended the bidding phase.

During the calculating phase of round 1, the Auction Manager may call a time-out to the Auction to evaluate whether the Auction volume should be adjusted. The Auction Manager has the discretion to call additional time-outs during the Auction. Such discretion could be used, for example, in case of an extraordinary event occurring during the Auction. The Auction Manager expects to exercise this discretion rarely.

IX.C.10. Failure to Submit a Bid

A bidder with positive eligibility must submit a bid in every round (even when the bidder's bid does not change or the bidder is bidding only on EDCs whose prices have not ticked down). As explained in section IX.C.9, if a bidder with positive eligibility does not submit a bid during the bidding phase of a round, the bidder is granted an extension whenever possible. If the bidder has previously used both extensions, or if the bidder does not submit a bid during the extension to the bidding phase, then the bidder has failed to submit a bid.

When a bidder with positive eligibility has failed to submit a bid in a round, the bidder is assigned a *default bid*. A default bid is the number of tranches on each EDC equal to the minimum number of tranches that the bidder could have bid on each EDC, as explained below. Any tranches of free eligibility that could have been bid in the round are deemed to be withdrawn and are irrevocably removed from the Auction. Any tranche that was bid on an EDC and that could be withdrawn is deemed to be withdrawn at the highest possible exit price for the round, which is equal to the going price in the previous round. A tranche that is withdrawn as a result of a default bid can be retained only if the number of tranches bid by other bidders at the going price, and the number of withdrawn tranches from bidders that have submitted a bid, together are not sufficient to fill the tranche target. In breaking any ties between tranches withdrawn at the same exit price, default bids will be disadvantaged compared to bids actually submitted by bidders. In general, in any circumstance where a tie must be broken, default bids will be disadvantaged compared to bids actually submitted by bidders, also as explained below.

As a consequence of a bidder being assigned a default bid, a bidder may lose the ability to submit bids for the remainder of the Auction. For example, if the prices of all the EDCs for which a bidder bid tranches tick down, then the default bid is zero on all EDCs. Indeed, the minimum number of tranches that can be bid on an EDC whose price has ticked down is zero. Any tranche that had previously been bid is withdrawn. After such a default bid, the bidder will be assigned zero eligibility in the next round, and the bidder will lose the ability to bid in the next and in all future rounds.

The default bid assigned to a bidder that has not submitted a bid during a round or during its extension, and the treatment of this default bid in any required tie-breaking procedure, are determined during the calculating phase of a round. The default bid for a bidder in round 1 is 0 tranches on each and every EDC. The default bid for a bidder in round 2 and all subsequent rounds is described in detail as follows.

If the bidder has some tranches of free eligibility in the current round, these tranches are deemed to be withdrawn and are irrevocably removed from the Auction.

If, in the previous round, a bidder did not bid any tranches on an EDC at the going price and in the reporting phase of that round the Auction Manager reported that the bidder did not have any retained withdrawals or denied switches for that EDC, then the bidder is assigned zero tranches for that EDC.

If, as of the reporting phase in the previous round, a bidder had tranches on a particular EDC at the going price for the previous round and if the EDC's price ticked down from the previous round to the current round, then the bidder is deemed to have withdrawn all tranches at the highest exit price, namely the price from the previous round. The bidder loses the eligibility associated with these tranches. Tranches with a lower exit price named by bidders that have submitted a bid in the current round are retained first. If needed, tranches with the same exit price named by bidders that have submitted a bid in the current round are retained next. If all the withdrawn tranches by the bidder and by other bidders that were assigned a default bid are needed to fill the tranche target, these tranches are retained. If some but not all of the tranches submitted by the bidder and other bidders that were assigned a default bid are needed, tranches are chosen at random to fill the tranche target, in a procedure analogous to that used for bidders that submitted a bid, as described in section IX.C.4.

If, as of the reporting phase in the previous round, a bidder had some tranches on a particular EDC at the going price for the previous round, and/or retained withdrawals, and/or denied switches; if the EDC's price did not tick down from the previous round to the current round; and if there is excess supply for the EDC in the current round, so that the price will tick down from the current round to the next round, then:

- all withdrawals that were previously retained are released and the bidder has no remaining obligation from those tranches;
- all switches that had previously been denied are outbid and the bidder is assigned free
 eligibility for those tranches in the next round; and
- all tranches bid in the previous round at the going price are bid in the current round on the EDC at the going price. The bidder has eligibility associated with these tranches in the next round. If the bidder is assigned a default bid in the next round these tranches will be withdrawn and assigned the highest possible exit price.

If, as of the reporting phase in the previous round, a bidder had some tranches on a particular EDC bid at the going price, and/or retained withdrawals, and/or denied switches; if the EDC's price did not tick down from the previous round to the current round; and if there is no excess supply on the EDC in the current round so that the price will not tick down in the next round; then:

- any tranches bid at the going price in the previous round are deemed to be bid in the current round at the current going price;
- if any new tranches were bid by other bidders on the EDC at the going price in the current round, the denied switches (if any) of bidders that have been assigned default bids are outbid first, before the denied switches of bidders that have submitted a bid in the current round. If more than one bidder has been assigned a default bid, and if some but not all denied switches from such bidders are outbid, then for each denied switch that must be outbid, the Auction Manager chooses at random among the default bidders the bidder whose switch is outbid, in a procedure analogous to that used for bidders that submitted a bid, as described in section IX.C.6; and

if any new tranches were bid by other bidders on the EDC at the going price in the current round, and if all denied switches from default bidders and from bidders that submitted a bid are outbid, retained withdrawals are released, starting with the highest named exit price. For a given exit price, tranches from bidders that have been assigned default bids (if any) are released first, before the retained withdrawals of bidders that have submitted a bid in the current round. If more than one bidder has been assigned a default bid, and if some but not all of the retained withdrawals from such bidders must be released at a given exit price, then for each retained withdrawal that must be released, the Auction Manager chooses at random among the default bidders the bidder whose withdrawn tranche is released, in a procedure analogous to that used for bidders that submitted a bid, as described in section IX.C.6.

The bidder can lose its ability to bid in all future rounds by failing to bid during the bidding phase of a round or during its extension. It is the responsibility of the bidder to ensure that bids are submitted on time.

Example 15.

Bidder A's bid in round 5 is given below:					
EDC	PSE&G	JCP&L	ACE	RECO	
Price Round 5 (¢/kWh)	13.890	14.445	13.950	14.610	
Bidder A Bid (tranches)	0	4	2	0	
Bidder A's bid in round 6 is given below:					
EDC	PSE&G	JCP&L	ACE	RECO	
Price Round 6 (¢/kWh)	13.816	14.252	13.688	14.246	
Bidder A Bid (tranches)	2	4	0	0	

This bid represents a switch, whereby Bidder A has increased the number of tranches on PSE&G while decreasing the number of tranches on ACE.

The Auction Manager reports the following:

EDC	PSE&G	JCP&L	ACE	RECO
Price Round 7 (¢/kWh)	13.816	14.075	13.688	13.895
Report to Bidder A (tranches)	0	4	2 denied switches @ 13.950¢	0

In round 7, the minimum number of tranches that Bidder A can bid on PSE&G and RECO is zero, because Bidder A did not have any tranches accepted for these EDCs in round 6. The minimum number of tranches that Bidder A can bid on JCP&L is zero, since the price for JCP&L has ticked down from 14.252¢/kWh to 14.075¢/kWh. The minimum number of tranches that Bidder A can bid on ACE is 2, since the price for ACE has not ticked down and Bidder A has two denied switches on ACE.

In round 7, Bidder A fails to submit a bid in the bidding phase of the round. Bidder A is granted an extension but does not submit a bid during the extension. Bidder A will be assigned the following bid, which is its default bid. Bidder A is the only bidder that is assigned a default bid.

EDC	PSE&G	JCP&L	ACE	RECO
Price Round 7 (¢/kWh)	13.816	14.075	13.688	13.895
Default Bid for Bidder A (tranches)	0	0	2 denied switches @ 13.950¢	0

None of Bidder A's withdrawn tranches are retained; tranches from other bidders that actually submitted bids are sufficient to fill the tranche target for JCP&L. Three new tranches are bid on ACE by other bidders. Both of Bidder A's tranches become outbid denied switches. Bidder A's eligibility in round 8 is 2. Should Bidder A again fail to bid in round 8, this free eligibility will be lost. Bidder A would be assigned eligibility of zero in round 9 and would lose the ability to bid for the remainder of the Auction.

IX.C.11. End of Auction

The Auction ends for all EDCs at the same time. The Auction ends in the reporting phase before the first round in which no bidder could change its bid from the previous round. For the Auction to end, the total excess supply in the Auction must be equal to zero and it must be that no price can tick down. At the end of the Auction, tranches are allocated to the winners and all winners for an EDC's tranches, should the Board authorize them as suppliers, will receive the same price for that EDC. The price given to the winners is determined as follows.

If, to fill the tranche target for an EDC in the final round, only tranches bid at the price from the final round are used, the winners are those that submitted bids at the price from the final round. The final price given to all winners is the price from the final round.

If, to fill the tranche target for an EDC in the last round, withdrawn tranches must be retained, then the winners are the bidders that submitted bids at the price from the last round and the bidders that submitted the lowest of the exit prices. If, to fill the last tranches of the tranche target of an EDC in the last round, the Auction Manager must use some but not all the tranches from two or more bidders tied at the same exit price, then the Auction Manager, for each tranche, will choose at random the bidder whose tranche is retained. For the first tranche needed at the tied exit price, the probability that a bidder is chosen is the number of tranches that the bidder has bid at the exit price divided by the total number of tranches bid at the tied exit price. If a second tranche is needed at the exit price, the Auction Manager again will choose at random the bidder whose tranche will be retained. The probability that any one bidder is chosen is the number of tranches that the bidder has bid at the exit price and that have not yet been retained divided by the total number of tranches bid at the exit price and that have not yet been retained. The Auction Manager repeats this procedure until the tranche target for the EDC is filled. The final price given to all winners is the last accepted price.

If, to fill the tranche target for an EDC in the last round the Auction Manager must disallow reductions from both withdrawals and switches, then the winners are the bidders that submitted bids at the last round price, the bidders that withdrew their tranches, and the bidders with reductions for that EDC from switches that were denied. The final price received by all winners is the price at which the denied switches were last freely bid.

Example 16.

The tranche target for PSE&G is 28.

In round 24, 29 tranches for PSE&G are bid at a price of 9.600¢/kWh. In round 25, 24 tranches for PSE&G are bid at a price of 9.471¢/kWh.

Bidder A bids 8 tranches for PSE&G in round 24 and 5 tranches in round 25. Bidder A enters an exit price equal to 9.595¢/kWh for the 3 tranches it is withdrawing.

Bidder B bids 5 tranches for PSE&G in round 24 and 3 tranches in round 25. Bidder B enters an exit price of 9.593¢/kWh for the 2 tranches it is withdrawing.

No other bidder changes its number of tranches bid on PSE&G. All other bidders collectively bid 16 tranches for PSE&G in both rounds 24 and 25. The total excess supply in round 25 is zero and the Auction ends in round 25. Twenty-four tranches for PSE&G are allocated to the bidders that bid at the going price of 9.471¢/kWh. Two additional tranches are allocated to Bidder B since it submitted a lower exit price. Finally, Bidder A wins 2 additional tranches so that the tranche target is filled. All winning bidders will receive a price of 9.595¢/kWh, which is the lowest price at which the tranche target is filled.

IX.D. Post Auction

The Board will decide whether or not to approve the results of this Auction (the BGS-RSCP Auction) by the end of the second business day following the calendar day on which the BGS-RSCP or BGS-CIEP Auction closes, whichever comes later.

If the Board approves the Auction results and authorizes the winners to become BGS-RSCP suppliers, the Auction Manager will notify each winner of the tranches it has won and the final prices. The Auction Manager will notify each EDC of the following for its territory: the identity of the winners, the number of tranches won, and the final price.

Each winner and each EDC will have three days from the time at which the Board approves the Auction results to execute the BGS-RSCP Supplier Master Agreement. Each winner will have these three days to demonstrate compliance with the creditworthiness requirements set forth in the BGS-RSCP Supplier Master Agreement and to submit the executed BGS-RSCP Supplier Master Agreement to the Board for information. A winner's financial guarantee posted before the Auction may be forfeited if the winner does not execute the Agreement within three days, if the winner fails to demonstrate compliance with the creditworthiness requirements set forth in the BGS-RSCP Supplier Master Agreement, or if the winner fails to agree to any of the terms of the Agreement. Effective with the exercise by an EDC of its right to collect on the financial guarantees, any contractual rights or other entitlements

of the winners shall immediately terminate without further notice by the EDC. In addition, winners shall be liable for damages incurred by the EDCs, which damages shall be determined in accordance with the terms of the BGS-RSCP Supplier Master Agreement as if the winner were a defaulting party to that Agreement.

Each winner will have to pay a fee per tranche won. This fee will be set to recover all the costs associated with the Auction Process. The Auction Manager will announce the fee per winning tranche no later than 6 days before the Auction. The fee will be netted against the first payment made to the winner during the supply period.

IX.E. Association and Confidential Information Rules

IX.E.1. Process for Reporting Associations, Identifying Concerns and Remedies

Interested parties applying to qualify to bid in the Auction will be required to indicate in their Part 1 Applications whether they are part of a bidding agreement, a joint venture for purposes of participating in the Auction, a bidding consortium, or other arrangements pertaining to bidding in the Auction. Interested parties will also be required to certify in their Part 1 Application that, should they qualify to participate in the Auction, they will not disclose information regarding the list of qualified bidders. In addition to certifications regarding bidding agreements and confidential information, each interested party makes a number of other certifications signifying its agreement with the terms of the BGS-RSCP Auction Rules, the terms of the BGS-RSCP Supplier Master Agreement, as well as signifying its agreement that it will not assign its rights or substitute another entity in its place. These additional certifications are discussed in section IX.B.2.

Once parties are qualified to bid in the Auction, each qualified bidder will be asked in its Part 2 Application to make a number of certifications, each detailed below, and each qualified bidder may be asked to provide additional information to the Auction Manager if these certifications cannot be made. In particular, each qualified bidder will be informed of the list of qualified bidders and will be asked to certify that it is not associated with any other qualified bidder. If a qualified bidder cannot make such a certification, it will be asked to identify associations it may have with other qualified bidders. The criteria that determine whether two bidders are associated with one another are given below. If two qualified bidders are associated

with one another, the Auction Manager will determine whether the two qualified bidders can both participate in the Auction, as well as the terms and conditions of such participation. The Auction Manager may require qualified bidders that are associated with one another to bid as one entity or to reorganize so as to no longer be associated with one another.

Qualified bidders will be asked to certify that they will undertake to appropriately restrict their disclosure of confidential information relative to their bidding strategy and confidential information regarding the Auction Process (both of which are defined in section IX.E.3). Qualified bidders will also be asked to certify that they have not come and will not come to any agreement with another qualified bidder with respect to bidding at the Auction, except as disclosed and approved by the Auction Manager in their Part 1 Application. In addition to certifications regarding associations and confidential information, each qualified bidder makes a number of other certifications signifying its agreement that a bid is a binding offer to provide service under the terms of the BGS-RSCP Supplier Master Agreement, as well as signifying its agreement that it will not assign its rights or substitute another entity in its place. These additional certifications are discussed in section IX.B.2.

Before obtaining final documentation necessary to participate in the Auction, registered bidders will be required to certify that they will continue to maintain the confidentiality of any information that they will have acquired through their participation in the Auction Process.

IX.E.2. Association Criteria

1. Preliminary Definitions

- a. A party *controls* an entity directly if the party holds a majority of shares, majority voting power, a majority of common directors, can appoint a majority of directors, or if the party in fact controls the entity's affairs through some other means. A party controls an entity indirectly if the party controls another entity that controls the entity in question (or through a longer line of control; e.g., if the party controls another entity that controls an entity that controls the entity in question, etc.).
- b. A party *participates* directly in another entity Z if the party holds any class of listed shares, if it holds the right to acquire such shares, if it holds any option to purchase shares or if it has voting power. The participation is indirect if the party participates

in another entity that participates in Z (with potentially a longer line of "indirect participation"). When the participation is indirect, the percentage of participation of the party in the entity is obtained by multiplying the percentages of participation at each level.

c. A party is concerned with the bid of a bidder if the party has confidential information relative to the bidders' bidding strategy (see definition in the next section), has agreed to provide assistance with financing or has agreed to provide assistance in another way.

2. Bidder A and Bidder B are associated with each other if Bidder A

- a. Controls bidder B, directly or indirectly; or
- b. Has at least a 10% participation in Bidder B and is concerned with Bidder B's bid; or
- c. Controls an entity that has at least a 10% participation, direct or indirect, in Bidder B and that is concerned with Bidder B's bid; or
- d. Is controlled by an entity that controls Bidder B directly or indirectly; or
- e. Is controlled by an entity that has at least a 10% participation, direct or indirect, in Bidder B and that is concerned with Bidder B's bid.

3. Bidder A and Bidder B are associated if there is a party which

- a. Controls Bidder A, directly or indirectly; or
- b. Has at least a 10% participation in Bidder A, directly or indirectly, and is concerned with Bidder A's bid; or
- c. Controls an entity that has at least a 10% participation in Bidder A, direct or indirect, and is concerned with Bidder A's bid; or
- d. Has confidential information about Bidder A's bid and is controlled by Bidder A; or
- e. Has confidential information about Bidder A's bid and is controlled by an entity or person that controls Bidder A directly or indirectly; or

f. Has confidential information about Bidder A's bid and is controlled, directly or indirectly, by an entity that has at least a 10% participation in Bidder A and is concerned with Bidder A's bid;

and if this same party has one of relationships a. to f. with Bidder B.

- 4. Bidder A and Bidder B are associated if there is a party that has at least a 20% participation, directly or indirectly, in both bidders.
- 5. Bidder A and Bidder B are associated if there is a party that has at least a 20% participation, directly or indirectly, in Bidder A and that:
 - a. Has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - b. Is controlled by Bidder B; or
 - c. Controls a person or entity that controls Bidder B; or
 - d. Controls a person or entity that: has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - e. Is controlled by a person or entity that controls Bidder B directly or indirectly; or
 - f. Is controlled by a person or entity that has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid; or
 - g. Is controlled by a person or entity who controls a person has at least 10% participation in Bidder B, directly or indirectly, and is concerned with Bidder B's bid.

IX.E.3. Definitions of Confidential Information

Confidential information relative to the bidding strategy means information relating to a bidder's bid, whether in writing or verbally, which if it were to be made public would be likely to have an effect on any of the bids that another bidder would be willing to submit at the Auction. Confidential information relative to the bidding strategy includes (but is not limited to) a bidder's Auction strategy; a bidder's indicative offer; the bidder's preference to bid for one EDC's system rather than another; the quantities that a bidder wishes to serve of one or more EDCs' systems; the bidder's estimation of the value of a tranche of the various EDCs' systems; the bidder's

estimation of the risks associated with serving BGS-RSCP Load or of serving a particular EDC; and a bidder's contractual arrangements for power with a party to serve the BGS-RSCP Load were the bidder to be a winner at the Auction.

Confidential information regarding the Auction Process means information that is not released publicly by the Board or the Auction Manager and that a bidder acquires as a result of participating in the Auction Process, whether in writing or verbally, which if it were to be made public could impair the integrity of current or future Auctions, impair the ability of the EDCs to hold future Auctions, harm consumers, or injure bidders or applicants. Confidential information regarding the Auction Process includes (but is not limited to) the list of qualified bidders, the list of registered bidders, the initial eligibility in the Auction, the status of a bidder's participation in the Auction, and all reports of results and announcements made by the Auction Manager to all or any one bidder during the Auction.

IX.E.4. Certifications and Disclosures to Be Made

An interested party will be required in its Part 1 Application to disclose any bidding agreement or arrangement in which the interested party may have entered. In addition, the following certification will be required and will apply from the time that the Part 1 Application has been submitted.

1. An applicant must certify that if it qualifies to participate in the Auction, the applicant will not disclose at any time information regarding the list of qualified bidders, including the number of qualified bidders, the identity of any one or all entities that have been qualified, or the fact that an entity has not been qualified for participation in the Auction. The applicant must certify that it will destroy any document distributed by the Auction Manager that lists the qualified bidders within five days of the Board rendering a decision on the Auction results.

The following certifications will be required of each qualified bidder in its Part 2 Application and will apply from the time of qualification until the Board certification of the Auction results. Each qualified bidder must consult the list of all qualified bidders and attest to the following:

2. A qualified bidder must certify that it is not associated with another qualified bidder according to the criteria given above.

A qualified bidder unable to make certification 2 must identify the qualified bidders with which it is associated and the nature of the association.

3. A qualified bidder must certify that, other than qualified bidders that were explicitly named in its Part 1 Application as parties with whom the bidder has entered into a bidding agreement, joint venture for the purpose of bidding in the Auction, bidding consortium, or other arrangement pertaining to bidding in the Auction, the bidder has not entered into any agreement with another qualified bidder, directly or indirectly, regarding bids at the Auction, including, but not limited to, the amount to bid at certain prices, the system on which bids are placed, when or at what prices bids are withdrawn or switched, or the amount of exit prices.

An *Advisor* is an entity or person(s) that will be advising or assisting the qualified bidder with bidding strategy in the BGS-RSCP Auction, with estimation of the value of a system's tranches, or with the estimation of the risks associated with serving BGS-RSCP load.

4. A qualified bidder must certify to one of the following: (i) the qualified bidder has not retained an Advisor; or (ii) the qualified bidder has retained an Advisor, the Advisor will not provide any similar advice or assistance to any other qualified bidder, and the Advisor will not be privy to confidential information relative to another qualified bidder's bidding strategy; or (iii) the qualified bidder has retained an Advisor who will provide similar advice or assistance to another qualified bidder, or who will be privy to confidential information relative to any other qualified bidder's bidding strategy, but appropriate protections have been put into place to ensure that the Advisor does not serve as a conduit of information between, or as a coordinator of the bidding strategies of, multiple bidders.

A qualified bidder unable to make certification 4 must name the Advisor and the other bidder(s) concerned.

5. A qualified bidder must certify that the qualified bidder is not a party to any contract for the purchase of power that might be used as source of supply for BGS-RSCP, and that (i) would require the disclosure of any confidential information (confidential information

relative to the bidding strategy or confidential information regarding the Auction Process) to the counterparty under such a contract; or (ii) that would require the disclosure of any confidential information (confidential information relative to the bidding strategy or confidential information regarding the Auction Process) to any other party; or (iii) that would provide instructions, direct financial incentives, or other inducements for the bidder to act in a way determined by the counterparty in the agreement and/or in concert with any other bidder in the Auction. Notwithstanding the above, a qualified bidder may, during negotiations prior to the Auction for contractual arrangements for power to serve BGS-RSCP Load were the bidder to be a winner at the Auction, discuss with the counterparty to such arrangements the nature of the products to be purchased, the volume, and the price at which it is willing to buy these products.

A qualified bidder unable to make certification 5 must disclose the contractual terms that prevent the qualified bidder from making the certification.

6. A qualified bidder must certify that it does not have any knowledge of confidential information relative to the bidding strategy of any other qualified bidder.

A qualified bidder unable to make certification 6 will be asked to name the other qualified bidder and the nature of the confidential information.

7. A qualified bidder must certify that it will not disclose confidential information relative to its own bidding strategy except to bidders that were explicitly named in its Part 1 Application as parties with whom the bidder has entered into a bidding agreement, joint venture for the purpose of bidding in the Auction, bidding consortium, or other arrangement pertaining to bidding in the Auction, bidders with which it is associated as disclosed through certification 2, to its Advisor, and to its financial institution.

A qualified bidder unable to make certification 7 will be asked to state all reasons.

8. A qualified bidder must certify that, other than entities affiliated with the bidder, and other than bidders with which the bidder has entered a bidding agreement, joint venture for purposes of the Auction, bidding consortium, or other arrangement pertaining to the Auction, no party has agreed to defray any of its costs of participating in the Auction,

including the cost of preparing the bid, the cost of any financial guarantees, the cost to be paid upon winning a tranche, or any other participation cost.

A qualified bidder unable to make certification 8 must identify the party that has agreed to defray some or all of the qualified bidder's cost of participating in the Auction, and the nature of the participation costs that the party has agreed to defray.

Furthermore, in its Part 2 Application a qualified bidder will have to certify that it will not disclose any confidential information regarding the Auction Process that it has acquired or will acquire through its participation. These certifications will apply from the time of submission of the Part 2 Application.

- 9. A qualified bidder must certify that if it is registered to participate in the Auction, the qualified bidder will not disclose at any time information regarding the initial eligibility in the Auction or the list of registered bidders, including the number of registered bidders, the identity of any one or all entities that have been registered, or the fact that an entity has not been registered for participation in the Auction.
- 10. A qualified bidder must certify that it will not disclose any confidential information regarding the Auction Process to any party except to its Advisor and bidders with which it is associated.

A qualified bidder unable to make certification 9 or certification 10 will be asked to state all reasons.

Following a successful Part 2 Application, the registered bidder will be required to certify that it will continue to abide by its prior commitment to maintain the confidentiality of information regarding the Auction Process. The registered bidder will be required to do so before obtaining manuals and procedures essential to submit bids in the Auction. The registered bidder also certifies that it will destroy all documents provided by the Auction Manager that contain confidential information within five days of the Board rendering a decision on the Auction results. Such certification will apply from the time at which it is made, no later than five days before the start of the Auction.

11. A registered bidder certifies that it continues to abide by its prior confidentiality certifications. The registered bidder will not disclose any confidential information

regarding the Auction Process to any party except to its Advisor and bidders with which it is associated. Further, the registered bidder certifies that it will destroy all documents written or electronic provided by the Auction Manager that contain confidential information regarding the Auction Process within five days of the Board rendering a decision on the Auction results.

Once the Auction has been concluded, if the Board approves the Auction results, the Board may choose to release information regarding final BGS-RSCP prices and the names of the winners. At that point, a winner may itself release information only regarding the number of tranches it has won and the territories the winner will be serving, and a losing bidder may itself release information only regarding the fact that it participated in the Auction. The winners and losing bidders otherwise continue to be bound by their certifications as described previously.

IX.E.5. Actions to Be Taken If Certifications Cannot Be Made

If a party cannot make all above certifications, the Auction Manager will decide within five days on a course of action on a case-by-case basis. To decide on this course of action, the Auction Manager may make additional inquiries to understand the reason for the inability of the bidder to make all the certifications.

In general, qualified bidders that are associated with one another, or that have entered into agreements regarding bidding at the Auction, are considered as one bidder for the purposes of application of the load caps and for the administration of the Auction. Bidders can be allowed to bid independently or can be asked to end their association or agreement as a condition of participation, as circumstances warrant. If qualified bidders are asked to end their associations they will be given five days to do so.

If qualified bidders do not comply with additional information requests by the Auction Manager regarding certifications required in the Part 2 Application, or do not comply with a request from the Auction Manager to end their associations, this may be sufficient grounds for the Auction Manager to reject the application.

Sanctions can be imposed on a qualified bidder for failing to disclose information relevant to determining associations, for coordinating with another bidder without disclosing this fact, for releasing confidential information except as provided in 3, 5, and 7 of section IX.E.4, or for

disclosing information during the Auction to a person other than those specified in 10 of section IX.E.4. Such sanctions can include, but are not limited to, loss of all rights to serve any BGS-RSCP Load won in the Auction by such bidder, forfeiture of bid bonds and other fees posted or paid, liquidated damages of \$100,000, action under state or federal laws, attorneys' fees and court costs incurred in any litigation that arises out of the bidder's improper disclosure, debarment from participation in future BGS Auctions, prosecution under applicable state and federal laws, or other sanctions that the Board may consider appropriate. The Auction Manager will make a recommendation to the Board on a possible sanction and the Board will make the final determination.

IX.F. Glossary of Auction Terms

IX.F.1. Associated With

A bidder is associated with another if the two bidders have ties that could allow them to act in concert or that prevent them from competing actively against each other in the Auction. Specific criteria for associations are provided in section IX.E.2.

IX.F.2. Auction Volume

The Auction volume is a number of tranches that the EDCs taken together will purchase through the Auction (representing one-third of BGS-RSCP Load). The Auction Manager, in accordance with confidential guidelines approved by the Board, may cut back the Auction volume.

IX.F.3. BGS

Basic Generation Service is the electric supply for those retail customers who are not served by a third party supplier.

IX.F.4. BGS Load

The BGS Load is the full electricity requirement (i.e., including energy, capacity, ancillary services, transmission, etc.) of retail customers who have not chosen a third party supplier, as measured and reported to PJM. It is the EDC aggregate zonal requirements less the wholesale requirements less the third party supplier requirements.

IX.F.5. BGS-CIEP Load

An EDC's BGS-CIEP Load includes the sum of the hourly load of all BGS-CIEP customers, adjusted for losses.

IX.F.6. BGS-RSCP Load

The BGS-RSCP Load is the BGS Load excluding the BGS-CIEP Load.

IX.F.7. BGS-RSCP Peak Load Share

BGS-RSCP Peak Load Share is the portion of PJM-determined EDC zonal peak load on one proxy day attributable to customers in RSCP rate classes that are taking BGS on that proxy day. The BGS-RSCP Peak Load Share is a single value measured in MW.

IX.F.8. Bid

A bid consists of four numbers: a number of PSE&G tranches, a number of JCP&L tranches, a number of ACE tranches, and a number of RECO tranches. A bid represents the number of tranches that a bidder wishes to serve for each EDC at the going prices in force at that point in the Auction. In all rounds except the first, to fully specify a bid, a bidder may be asked for information in addition to the number of tranches that it wishes to bid for each EDC, such as switching priorities and exit prices. A bid is valid when it is submitted and verified in the bidding phase, and processed by the Auction software.

IX.F.9. Bidding Phase

The bidding phase is the first phase of a round, during which bidders place bids. A bidder that submits a bid in a round may revise or change this bid as long as the bidding phase of the round is still open.

IX.F.10. Calculating Phase

The calculating phase is the second phase of the round, during which the Auction Manager tabulates the results of that round's bidding phase and calculates the prices for the next round.

IX.F.11. CIEP Load

CIEP Load for an EDC is defined as the sum of the hourly load of all BGS-CIEP customers, as well as the hourly load of retail customers served by a third party supplier who, were they to take BGS, would do so under a CIEP tariff, adjusted for losses.

IX.F.12. Contingency Plan

If the number of tranches procured through the Auction is less than the Auction volume set at the start of the Auction, each EDC whose tranche target is not filled at the end of the Auction will implement a Contingency Plan for the remaining tranches. Under their Contingency Plans, the EDCs will purchase necessary services including installed capacity, energy, and ancillary services, through PJM-administered markets.

IX.F.13. Decrement

If the number of tranches bid exceeds the number of tranches needed for an EDC, the price for the EDC falls by a decrement in the next round. The decrement varies in each round based on the excess supply on an EDC measured against a measure of maximum possible excess supply.

IX.F.14. Denied Switches

The Auction Manager denies reductions in the number of tranches bid for an EDC from switches when the tranches bid at the going price and the withdrawn tranches that can be retained are not sufficient to fill that EDC's tranche target. Denied switches are retained at the last price at which they were freely bid.

IX.F.15. EDC Load Cap

An EDC load cap is a maximum number of tranches that a bidder can bid and win for that EDC. For each EDC, the bidder's combined number of tranches bid at the going price, retained withdrawals, and denied switches cannot exceed the EDC load cap in any given round.

IX.F.16. EDC-Specific Summer Factor

The factor by which an EDC's Auction price is multiplied to determine the price that each BGS-RSCP supplier for that EDC receives for every kWh of load it serves in each summer month of the supply period. The summer months are from June 1 through September 30.

IX.F.17. EDC-Specific Winter Factor

The factor by which an EDC's Auction price is multiplied to determine the price that each BGS-RSCP supplier for an EDC receives for every kWh of load it serves in each winter month of the supply period. The winter months are from October 1 through May 31.

IX.F.18. Eligibility and Initial Eligibility

Initial eligibility is equal to the number of tranches that a bidder has financially guaranteed with a letter of credit (or bid bond) with its Part 2 Application. Initial eligibility can never exceed the statewide load cap.

A bidder's initial eligibility is the bidder's eligibility in round 1. A bidder's eligibility in round 2 is the total number of tranches bid for all EDCs at the going prices in round 1. For any subsequent round, a bidder's eligibility in a round is the bidder's eligibility at the start of the previous round minus the number of tranches that the bidder withdrew in the bidding phase of the previous round (whether or not the Auction Manager retains these withdrawn tranches).

IX.F.19. Eligibility Ratio

An eligibility ratio is obtained by dividing the total eligibility in the Auction by the number of tranches available in the Auction.

IX.F.20. End of Auction

The Auction ends in the reporting phase of the first round in which the total excess supply is zero.

IX.F.21. Excess Supply

The excess supply on an EDC is the number of tranches bid at the going price for the EDC minus the EDC's tranche target, or it is zero, whichever is greater.

IX.F.22. Exit Price

If a bidder is reducing eligibility in a round, in general the bidder names an exit price. The bidder names an exit price for the tranches of an EDC that the bidder is no longer willing to serve at the price for the current round. An exit price must be a price higher than the going price in the current round and less than or equal to the previous going price for the EDC. No exit price is named when withdrawing tranches of free eligibility.

IX.F.23. Extension

An extension extends the bidding phase of a round by 15 minutes. Each bidder is allowed two extensions during the Auction. A bidder is automatically deemed to have requested an

extension whenever no bid has been received from the bidder by the end of the scheduled bidding phase of a round and the bidder has not used the two allowable extensions. The bidding phase in round 1 is automatically extended for the convenience of bidders.

IX.F.24. Financial Guarantee

Each qualified bidder must post a financial guarantee, in the form of a letter of credit (or bid bond), proportional to its indicative offer at the maximum starting price. Some bidders may be required to post additional pre-Auction security depending on a creditworthiness assessment.

IX.F.25. Free Eligibility

A denied switch that is replaced in filling the tranche target by a new tranche at the going price is said to be outbid. An outbid denied switch becomes a tranche of free eligibility. The bidder can assign a tranche of free eligibility to any EDC(s) and, if the bidder wishes to withdraw such a tranche, no exit price will be named and this tranche will not be retained.

IX.F.26. Full-Requirements Service

Full-requirements service means that the supplier is responsible for fulfilling all the requirements of a PJM LSE including capacity, energy, ancillary services, transmission, and any other service as may be required by PJM.

IX.F.27. Going Price

The going price for an EDC in a round is the price at which the Auction Manager is soliciting bids in that round. A bidder that submits a bid in a round declares itself ready to supply the number of tranches bid for each EDC at the going price for the EDC.

IX.F.28. Indicative Offer

An indicative offer specifies two numbers of tranches. The first number represents the amount that the qualified bidder is willing to serve at the maximum starting price on a statewide basis (i.e., for all EDCs combined). The second number represents the amount that the qualified bidder is willing to supply at the minimum starting price on a statewide basis. At each of the maximum and the minimum starting prices, the number of tranches indicated by the qualified bidder cannot exceed the statewide load cap. Indicative offers are provided with the Part 2 Application.

IX.F.29. Initial Eligibility

See Eligibility and Initial Eligibility.

IX.F.30. Minimum and Maximum Starting Prices

The minimum and maximum starting prices establish the range of possible starting prices for the Auction: each EDC will choose a starting price for round 1 of the Auction that is between the minimum and the maximum starting prices. The EDCs will agree on the statewide minimum and maximum starting prices.

IX.F.31. MW-Measure

The approximate measure in MW of a single tranche for an EDC, given the RSCP Peak Load Share for the EDC and the percentage of load represented by the tranche.

IX.F.32. Outbid (Denied Switches)

A denied switch that is replaced in filling the tranche target by a new tranche at the going price is said to be outbid.

IX.F.33. Oversupply Ratio

The oversupply ratio is the ratio of the excess supply on an EDC to a measure of the maximum possible excess supply on that EDC. The measure of the maximum possible excess supply on an EDC takes into account the total excess supply in the Auction, the EDC load cap, and the number of registered bidders.

IX.F.34. Part 1 Application

In their Part 1 Applications, interested parties will be asked to submit financial information so that the EDCs can assess their creditworthiness. In addition, interested parties will be asked to comply with other qualification criteria, including agreeing to comply with the BGS-RSCP Auction Rules and agreeing to the terms of the BGS-RSCP Supplier Master Agreement. Each interested party will also be asked to agree that if the interested party is successful in its Part 1 Application it will keep confidential the list of other successful applicants and it will not assign its rights or substitute another entity in its place.

IX.F.35. Part 1 Application Date

Date at which Part 1 Applications are due. This date will be set no earlier than 10 days after the maximum and minimum starting prices have been announced.

IX.F.36. Part 2 Application

In the Part 2 Application, qualified bidders will make a number of certifications to ensure compliance with the association and confidential information portion of these rules. Each qualified bidder is also asked to agree to keep confidential the list of other successful applicants; to agree that the submission of any bid creates a binding and irrevocable offer to provide service under the terms of the BGS-RSCP Supplier Master Agreement; and not to assign its rights or substitute another entity in its place.

IX.F.37. Part 2 Application Date

Date at which Part 2 Applications are due. This date will be set no later than 10 days before the start of the Auction.

IX.F.38. Product in the Auction

A product in the (BGS-RSCP) Auction is the BGS-RSCP Load for an EDC for a three-year supply period.

IX.F.39. Qualified Bidder

An interested party that has submitted a Part 1 Application and that has satisfied all conditions of the Part 1 Application becomes a qualified bidder. Interested parties will be notified that they have qualified no later than three days after the Part 1 Application Date.

IX.F.40. Recess

A recess is a suspension of the Auction for a period of no less than twenty minutes, giving bidders more time to consider their bids. A bidder may request a recess in the calculating phase or the reporting phase of a round. Each bidder is allowed one recess during the Auction. A bidder cannot request a recess in the first ten rounds of the Auction. Starting in the eleventh round, a bidder may request a recess in a round if the total excess supply in the Auction reported in the previous round was 40 or fewer tranches.

IX.F.41. Reductions

Reductions in the number of tranches bid on an EDC for which the price has ticked down can be in the form of a withdrawal (i.e., reducing the number of tranches bid on an EDC without increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total) or a switch (i.e., re-assigning the total number of tranches bid so that the number of tranches bid on one or several EDCs is reduced, but the number of tranches bid on other EDCs is increased by the same amount).

IX.F.42. Registered Bidder

A qualified bidder that submits a Part 2 Application and that satisfies all conditions of the Part 2 Application becomes a registered bidder. Conditions of the Part 2 Application include the submission of an indicative offer and financial guarantees. These conditions also include making certifications relating to the Association and Confidential Information Rules or, if not all these certifications can be made, providing additional information to the Auction Manager and abiding by the course of action decided by the Auction Manager.

IX.F.43. Reporting Phase

The reporting phase is the third and final phase of a round, during which the Auction Manager informs the bidders of the results of that round's bidding phase. All bidders are informed of the going prices for the next round's bidding phase and are provided with a range of total excess supply. Each bidder privately receives the results of the bidder's own bid from that round, indicating to each bidder its obligation at this point in the Auction.

IX.F.44. Round

The Auction runs in discrete time periods called rounds. Each round has a bidding phase, a calculating phase, and a reporting phase.

IX.F.45. Rounding

Prices and exit prices are in cents per kWh and are rounded off to the nearest thousandth of a cent.

IX.F.46. RSCP Load

RSCP Load is the retail load excluding the CIEP Load.

IX.F.47. RSCP Peak Load Share

RSCP Peak Load Share is the portion attributable to those customers on RSCP service of PJM-determined EDC zonal peak load on one proxy day. The RSCP Peak Load Share is a single value measured in MW that consists of the BGS-RSCP Peak Load Share and the peak load attributable to customers who are served by third party suppliers on the proxy day and who are in RSCP rate classes.

IX.F.48. Session

Each day in the Auction will consist of two bidding sessions of roughly equal length. Each bidding session will consist of a number of rounds.

IX.F.49. Starting Prices

The starting prices are announced three days before the Auction starts. The starting prices are the going prices in round 1.

IX.F.50. Statewide Load Cap

The statewide load cap is a maximum on the number of tranches that a bidder can bid and win statewide.

IX.F.51. Switches and Switching

Switching involves an increase in the number of tranches bid on some EDCs while at the same time a reduction in the number of tranches bid on other EDCs. Switching (without a simultaneous reduction of tranches through a withdrawal) occurs when a bidder reallocates its tranches bid without changing the total number of tranches bid on all EDCs. Switching alone has no effect on eligibility.

IX.F.52. Switching Priority

A switching priority is a preference assigned to one of the EDCs for which a bidder is increasing its number of tranches bid. The bidder is required to assign a switching priority when the bidder is increasing the number of tranches bid for more than one EDC.

IX.F.53. Target Eligibility Ratio

The target eligibility ratio is a desired ratio of eligibility to the Auction volume.

IX.F.54. Time-Out

A time-out is a pause in the Auction. A time-out suspends activity in the Auction for a period of up to four hours. The Auction Manager can call a time-out at any time during a round.

IX.F.55. Total Excess Supply

The total excess supply in the Auction is the sum, over all EDCs for which the number of tranches bid exceeds the tranche target, of the excess supplies for the individual EDCs, plus tranches of free eligibility.

IX.F.56. Tranche

A tranche of one EDC is a full-requirements tranche. A tranche is a fixed percentage share of the BGS-RSCP Load of an EDC for the period June 1, 2019 to May 31, 2022.

IX.F.57. Tranche Size

The tranche size of an EDC in this Auction is the percentage share of the BGS-RSCP Load of the EDC represented by one tranche.

IX.F.58. Tranche Target

The tranche target is the number of tranches available at the Auction for an EDC.

IX.F.59. Withdrawal

A bidder reduces the number of tranches bid on an EDC through a withdrawal when the bidder is reducing the number of tranches bid on an EDC, while not increasing the number of tranches bid on other EDCs, and thus reducing the number of tranches bid in total. A bidder who does not bid a tranche of free eligibility in the round when it becomes available is making a withdrawal.

IX.G. Appendix

IX.G.1. Tranches

Data for sample calculations are provided in the table below. The second column of the table provides, for each EDC, the 2018 RSCP Peak Load Share on a proxy day. The fourth column provides the number of tranches for each EDC, which is the tranche target. The last column provides the MW-measure, which is generally the size of the tranche times the RSCP Peak Load Share (please see the note about JCP&L below).

Table IX-5. Sample Data

EDC	RSCP Peak Load Share (MW)	RSCP Peak Load Share in 2019 Auction (MW)	Tranche Target	Size of tranche (%)	MW- Measure
PSE&G	7,641.16	2,517.09	28	1.18%	89.90
JCP&L ⁶	4,508.97	1,537.15	18	1.89%	85.40
ACE	2,027.00	644.95	7	4.55%	92.14
RECO	400.00	100.00	1	25.00%	100.00
Total	14,577.13	4,799.19	54		

For RECO, the RSCP Peak Load Share is 400.00 MW. The size of a tranche is 25%. The MW-measure is then 100.00 MW, or 25% of 400.00 MW.

With the 2010 Auction, JCP&L began transitioning to procuring more tranches so that the MW-measure of a tranche is closer to approximately 100 MW. Three tranches were added to each of the 2010, 2011, and 2012 Auctions. JCP&L now procures 53 tranches. As a consequence, the size of a tranche for JCP&L varies across the three Auctions used to supply JCP&L's BGS-RSCP customers at any point in time, as shown in the table below. For the other three EDCs, the size of a tranche is the same across all Auctions.

Table IX-6. JCP&L Tranche Numbers and Sizes

Auction	Number of Tranches	Size of Tranche
2017	15	(12/44) x (1/15) or approximately 1.82%
2018	20	(17/44) x (1/20) or approximately 1.93%
2019	18	(15/44) x (1/18) or approximately 1.89%

The Public Power Association of New Jersey ("PPANJ") has petitioned the Board requesting authorization to receive the entire St. Lawrence allocation on behalf of PPANJ's members. JCP&L will continue to serve the 10 MW of load associated with the St. Lawrence allocation pending a final determination by the Board.

IX.G.2. Decrements

The calculation of the size of the decrement, Δ , is based on the oversupply ratio, γ , which is the ratio of the excess supply on an EDC to an estimate of the maximum excess supply:

$$\gamma = \frac{B - TT}{\min\left(\overline{RES}, n \cdot LC - TT\right)}$$

The numerator is the excess supply on an EDC, which is the number of tranches bid at the going price (B) minus the tranche target (TT). The denominator is a measure of maximum possible excess supply for that EDC. The excess supply on an EDC must be less than or equal to the total excess supply in the Auction. \overline{RES} is the upper bound of the range of total excess supply reported to bidders or 30 tranches, whichever is larger, and it serves as a measure of total excess supply in the Auction. The excess supply on an EDC must also be less than or equal to the excess supply that would result from all bidders bidding the load cap on the EDC. This is represented by $n \cdot LC - TT$, namely the number of registered bidders (n) times the load cap (LC) minus the tranche target (TT). The estimate of maximum possible excess supply for the EDC used for the decrement rules is \overline{RES} , or the measure based on the number of registered bidders and the load cap $n \cdot LC - TT$, whichever is smaller.

Regime 1

The Auction starts in Regime 1. In Regime 1, the following decrement formulas will be used.

If an EDC's tranche target is 25 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0050 & if & \gamma \le 0.15 \\ 0.0150 & if & 0.15 < \gamma \le 0.29 \\ 0.0300 & if & 0.29 < \gamma \le 0.41 \\ 0.0425 & if & 0.41 < \gamma \le 0.53 \\ 0.0500 & if & \gamma > 0.53 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1500 (but above 0), which means that the excess supply on the EDC is at or below 15.00% of its maximum, the decrement is set at 0.50%. When the oversupply ratio is at or below a value of 0.2900 (but above 0.1500), the decrement is set at 1.5%. When the oversupply ratio is at or below a value of 0.4100 (but above 0.2900), the decrement is set at 3%. When the oversupply ratio is at or below a value of 0.5300 (but above 0.4100), the decrement is set at 4.25%. When the oversupply ratio is above 0.5300, which means that the excess supply on the EDC exceeds 53.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is between 10 and 24 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0050 & \text{if} & \gamma \leq 0.12 \\ 0.0150 & \text{if} & 0.12 < \gamma \leq 0.24 \\ 0.0300 & \text{if} & 0.24 < \gamma \leq 0.36 \\ 0.0425 & \text{if} & 0.36 < \gamma \leq 0.47 \\ 0.0500 & \text{if} & \gamma > 0.47 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1200 (but above 0), which means that the excess supply on the EDC is at or below 12.00% of its maximum, the decrement is set at 0.50%. When the oversupply ratio is at or below a value of 0.2400 (but above 0.1200), the decrement is

set at 1.5%. When the oversupply ratio is at or below a value of 0.3600 (but above 0.2400), the decrement is set at 3%. When the oversupply ratio is at or below a value of 0.4700 (but above 0.3600), the decrement is set at 4.25%. When the oversupply ratio is above 0.4700, which means that the excess supply on the EDC exceeds 47.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is between 5 and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.5% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0150 & if & \gamma \le 0.15 \\ 0.0300 & if & 0.15 < \gamma \le 0.27 \\ 0.0425 & if & 0.27 < \gamma \le 0.40 \\ 0.0500 & if & \gamma > 0.40 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1500 (but above 0), the decrement is set at 1.50%. When the oversupply ratio is at or below a value of 0.2700 (but above 0.1500), the decrement is set at 3%. When the oversupply ratio is at or below a value of 0.4000 (but above 0.2700), the decrement is set at 4.25%. When the oversupply ratio is above 0.4000, which means that the excess supply on the EDC exceeds 40.00% of its maximum, the decrement is set at 5%.

If an EDC's tranche target is 4 tranches or fewer, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 3.00% (and the amount of the price decrease would be rounded off to the nearest thousandth of a cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.03 & if & \gamma \leq 0.10 \\ 0.05 & if & \gamma > 0.10 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1000 (but above 0), the decrement is set at 3.00%. When the oversupply ratio is above 0.1000, which means that the excess supply on the EDC exceeds 10.00% of its maximum, the decrement is set at 5%.

Change from Regime 1 to 2

Decrements continue to be calculated by the decrement formulas of Regime 1 until the going prices for round 4 are calculated. After that time, in the first round in which the upper bound of the total excess supply range (\overline{RES}) reported to bidders is

- Greater than 30 tranches; and
- At least 15 tranches fewer than the upper bound of the total excess supply range reported to bidders in round 1;

then the decrement formulas of Regime 2 will be used for the going prices in the next round. If, in the first round in which \overline{RES} is at least 15 tranches fewer than \overline{RES} in round 1, it is also the case that the upper bound of the total excess supply range reported to bidders is 30 or fewer tranches, the decrement formulas of Regime 3 will be used for the going prices in the next round and for the remainder of the Auction.

Regime 2

In Regime 2, the following decrement formulas will be used.

If an EDC's tranche target is 25 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.375% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The

decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.003750 & if & \gamma \le 0.15 \\ 0.011250 & if & 0.15 < \gamma \le 0.29 \\ 0.022500 & if & 0.29 < \gamma \le 0.41 \\ 0.031875 & if & 0.41 < \gamma \le 0.53 \\ 0.037500 & if & \gamma > 0.53 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1500 (but above 0), which means that the excess supply on the EDC is at or below 15.00% of its maximum, the decrement is set at 0.375%. When the oversupply ratio is at or below a value of 0.2900 (but above 0.1500), the decrement is set at 1.125%. When the oversupply ratio is at or below a value of 0.4100 (but above 0.2900), the decrement is set at 2.25%. When the oversupply ratio is at or below a value of 0.5300 (but above 0.4100), the decrement is set at 3.1875%. When the oversupply ratio is above 0.5300, which means that the excess supply on the EDC exceeds 53.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is between 10 and 24 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.375% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.003750 & if & \gamma \le 0.12 \\ 0.011250 & if & 0.12 < \gamma \le 0.24 \\ 0.022500 & if & 0.24 < \gamma \le 0.36 \\ 0.031875 & if & 0.36 < \gamma \le 0.47 \\ 0.037500 & if & \gamma > 0.47 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1200 (but above 0), which means that the excess supply on the EDC is at or below 12.00% of its maximum, the decrement is set at 0.375%. When the oversupply ratio is at or below a value of 0.2400 (but above 0.1200), the decrement is set at 1.125%. When the oversupply ratio is at or below a value of 0.3600 (but above 0.2400), the decrement is set at 2.25%. When the oversupply ratio is at or below a value of 0.4700 (but above 0.3600), the decrement is set at 3.1875%. When the oversupply ratio is above 0.4700, which means that the excess supply on the EDC exceeds 47.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is between 5 and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 1.125% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.011250 & if & \gamma \le 0.15 \\ 0.022500 & if & 0.15 < \gamma \le 0.27 \\ 0.031875 & if & 0.27 < \gamma \le 0.41 \\ 0.037500 & if & \gamma > 0.41 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1500 (but above 0), the decrement is set at 1.125%. When the oversupply ratio is at or below a value of 0.2700 (but above 0.1500), the decrement is set at 2.25%. When the oversupply ratio is at or below a value of 0.4100 (but above 0.2700), the decrement is set at 3.1875%. When the oversupply ratio is above 0.4100, which means that the excess supply on the EDC exceeds 41.00% of its maximum, the decrement is set at 3.75%.

If an EDC's tranche target is 4 tranches or fewer, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 2.25% (and the amount of the

price decrease would be rounded off to the nearest thousandth of a cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 3.75% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.0225 & \text{if } \gamma \leq 0.10 \\ 0.0375 & \text{if } \gamma > 0.10 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1000 (but above 0), the decrement is set at 2.25%. When the oversupply ratio is above 0.1000, which means that the excess supply on the EDC exceeds 10.00% of its maximum, the decrement is set at 3.75%.

Change from Regime 2 to 3

In the first round:

- After the going prices for round 4 are calculated; and
- In which the upper bound of the total excess supply range reported to bidders is 30 or fewer tranches;

then the decrement formulas of Regime 3 will be used for the going prices in the next round and for the remainder of the Auction.

Regime 3

In Regime 3, the following decrement formulas will be used.

If an EDC's tranche target is 25 tranches or more, the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The

decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.002500 & if & \gamma \le 0.15 \\ 0.007500 & if & 0.15 < \gamma \le 0.31 \\ 0.015000 & if & 0.31 < \gamma \le 0.47 \\ 0.021250 & if & 0.47 < \gamma \le 0.62 \\ 0.025000 & if & \gamma > 0.62 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1500 (but above 0), which means that the excess supply on the EDC is at or below 15.00% of its maximum, the decrement is set at 0.25%. When the oversupply ratio is at or below a value of 0.3100 (but above 0.1500), the decrement is set at 0.75%. When the oversupply ratio is at or below a value of 0.4700 (but above 0.3100), the decrement is set at 1.5%. When the oversupply ratio is at or below a value of 0.6200 (but above 0.4700), the decrement is set at 2.125%. When the oversupply ratio is above 0.6200, which means that the excess supply on the EDC exceeds 62.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is between 10 and 24 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.25% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.002500 & \text{if} & \gamma \leq 0.12 \\ 0.007500 & \text{if} & 0.12 < \gamma \leq 0.22 \\ 0.015000 & \text{if} & 0.22 < \gamma \leq 0.36 \\ 0.021250 & \text{if} & 0.36 < \gamma \leq 0.48 \\ 0.025000 & \text{if} & \gamma > 0.48 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1200 (but above 0), which means that the excess supply on the EDC is at or below 12.00% of its maximum, the decrement is set at 0.25%. When the oversupply ratio is at or below a value of 0.2200 (but above 0.1200), the decrement is set at 0.75%. When the oversupply ratio is at or below a value of 0.3600 (but above 0.2200), the decrement is set at 1.5%. When the oversupply ratio is at or below a value of 0.4800 (but above 0.3600), the decrement is set at 2.125%. When the oversupply ratio is above 0.4800, which means that the excess supply on the EDC exceeds 48.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is between 5 and 9 tranches (inclusive), the decrement for that EDC is set as a series of steps. Using this rule, the smallest decrement would be 0.75% (and the amount of the decrease in price would be rounded off to the nearest cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined maximum value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.00750 & if & \gamma \le 0.11 \\ 0.01500 & if & 0.11 < \gamma \le 0.21 \\ 0.02125 & if & 0.21 < \gamma \le 0.31 \\ 0.02500 & if & \gamma > 0.31 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1100 (but above 0), which means that the excess supply on the EDC is at or below 11.00% of its maximum, the decrement is set at 0.75%. When the oversupply ratio is at or below a value of 0.2100 (but above 0.1100), the decrement is set at 1.5%. When the oversupply ratio is at or below a value of 0.3100 (but above 0.2100), the decrement is set at 2.125%. When the oversupply ratio is above 0.3100, which means that the excess supply on the EDC exceeds 31.00% of its maximum, the decrement is set at 2.5%.

If an EDC's tranche target is 4 tranches or fewer, the decrement for that EDC is set as a series of steps.

Using this rule, the smallest decrement would be 1.50% (and the amount of the decrease in price would be rounded off to the nearest thousandth of a cent). The smallest decrement would be in effect when the oversupply ratio is at or below a pre-determined value (but above 0). The decrement is never more than 2.5% (subject to rounding off). The largest decrement would be in effect when the oversupply ratio is above that pre-determined value.

The following series of steps will be used:

$$\Delta = \begin{cases} 0.015 & \text{if} \quad \gamma \leq 0.10 \\ \\ 0.025 & \text{if} \quad \gamma > 0.10 \end{cases}$$

When the oversupply ratio is at or below a value of 0.1000 (but above 0), the decrement is set at 1.50%. When the oversupply ratio is above 0.1000, which means that the excess supply on the EDC exceeds 10.00% of its maximum, the decrement is set at 2.5%.

X. APPENDIX C

BGS-CIEP Supplier Master Agreement

BGS-CIEP SUPPLIER MASTER AGREEMENT

TABLE OF CONTENTS

ARTIC	LE 1:	DEFINITIONS	
ARTIC	LE 2:	GENERAL TERMS AND CONDITIONS	11
2.1	Capacit	y In Which Company Is Entering Into This Agreement	11
2.2	Parties'	Obligations	12
2.3		rvices	
2.4	Commi	unications and Data Exchange	16
2.5	Record	Retention	17
2.6	Verifica	ation	
ARTIC	LE 3:	REPRESENTATIONS AND WARRANTIES	18
3.1		r's Representations and Warranties	
3.2		ny's Representations and Warranties	
3.3	Surviva	l of Obligations	22
ARTIC		COMMENCEMENT AND TERMINATION OF AGREEMENT	
4.1	Comme	encement and Termination	22
4.2		ation of Right to Supply BGS-CIEP	
4.3		d of Obligations	
4.4		Termination	
ARTIC		BREACH AND DEFAULT	
5.1		of Default	
5.2		Upon Default	
5.3		es Resulting From an Event Of Default	
5.4		ation of an Early Termination Date and Calculation of Settlement Amount a	
		ation Payment	
5.5		Provision	
5.6		of Payment Obligations of the Non-Defaulting Party	
5.7		ation of Rights of Non-Defaulting Party	35
ARTIC		CREDITWORTHINESS	
6.1	Applica	bilityorthiness Determination	35
6.2			
6.3		Exposure/ Security Calculation	
6.4		y Instruments	
6.5		nance of Creditworthiness	
6.6		On Security	
6.7		on Cash Held by Company	
6.8 6.9	No End	orsement of BGS-CIEP Supplier	.4J
	No End	a DCS Sympley A agreements	43 45
6.10		e BGS Supply AgreementsROCEDURES FOR ENERGY SCHEDULING, CAPACITY RESOUR	
		AND TRANSMISSION PROCUREMENT	
7.1		bligationsbligations	
7.1	Doto To	ansmission	4 0
ADTTC	ne o an	Scheduling	۱۳۰۰ 10

8.1	Energy Settlement By PJM	48
8.2	Energy Settlement by the Company	48
ARTIC	LE 9: BILLING AND PAYMENT	49
9.1	The Company Payment of Obligations to the BGS-CIEP Supplier	49
9.2	Billing for BGS-CIEP Supplier's Obligations to Other Parties	53
9.3	The BGS-CIEP Supplier Payment of Obligations to the Company	53
ARTIC	LE 10: SYSTEM OPERATION	55
10.1	Disconnection and Curtailment By the Company	55
10.2	Inadvertent Loss of Service To BGS-CIEP Customers	55
10.3	Good Faith Efforts	56
10.4	PJM Requirements	
10.5	Compliance With Governmental Directives	
	LE 11: DISPUTE RESOLUTION	
11.1	Informal Resolution of Disputes	
11.2	Recourse to Agencies or Courts of Competent Jurisdiction	57
	LE 12: REGULATORY AUTHORIZATIONS AND JURISDICTION	
12.1	Compliance With Applicable Legal Authorities	
12.2	FERC Jurisdictional Matters	
	LE 13: LIMITATION OF LIABILITY	
	Limitations on Liability	
	Risk of Loss	
	LE 14: INDEMNIFICATION	
14.1	Indemnification	
14.2	Survives Agreement	
	LE 15: MISCELLANEOUS PROVISIONS	
15.1	Notices	
15.2	No Prejudice of Rights	
15.3	Assignment	
15.4	Governing Law and Venue	
15.5	Headings	
15.6	Third Party Beneficiaries	
15.7	General Miscellaneous Provisions	
15.8	Taxes	
	Changes in Transmission Charges for Firm Transmission Service	
	Use of the Word "Including"	
	Federal Acquisition Regulation	
	Binding Terms	
	Amendment	
	Counterparts	
	BGS-CIEP Letter of Credit	
Guaran	ty	88

BGS-CIEP SUPPLIER MASTER AGREEMENT

THIS BGS-CIEP SUPPLIER MASTER AGREEMENT, made and entered
into this day of, 2019, by and between (the "Company"), a
corporation and a public utility organized and existing under the laws of the State of New
Jersey, on its own behalf and as agent as more fully set forth below, and each of the
suppliers listed on Appendix A hereto, severally and not jointly (each a "BGS-CIEP
Supplier" and, collectively, the "BGS-CIEP Suppliers"), the Company and each BGS-
CIEP Supplier hereinafter sometimes referred to collectively as the "Parties", or
individually as a "Party",

WITNESSETH:

WHEREAS, the Company is a public utility engaged, <u>inter alia</u>, in the transmission and distribution of electric Energy within its Service Territory located in the State of New Jersey; and

WHEREAS, pursuant to Section 9(c) of the Electric Discount and Energy Competition Act, N.J.S.A. 48:3-49 et. seq. ("EDECA"), the New Jersey Board of Public Utilities ("BPU") has been authorized to make available to any power supplier on a competitive basis the opportunity to provide Basic Generation Service ("BGS"); and

WHEREAS, in its Decision and Order dated_____, in Docket No. ER18040356 (the "_____ Order"), the BPU found that for periods after May 31, 2019, it would serve the public interest for the Company to continue to secure BGS Supply through a bid process; and

WHEREAS, in the _____ Order, the BPU approved an auction design for bidding out two Basic Generation Service products, one a product termed Basic Generation Service – Residential Small Commercial Pricing ("BGS-RSCP"), and one a product termed Basic Generation Service – Commercial and Industrial Energy Pricing ("BGS-CIEP"); and

WHEREAS, on ______, 2019 through ______, 2019, a successful auction for bidding out BGS-CIEP was held; and,

WHEREAS, each BGS-CIEP Supplier was one of the winning bidders in the auction for the provision of BGS-CIEP; and

WHEREAS, the _____ Order authorized the Company to contract with winning bidders, on behalf of the consumers of electricity located on the Company's distribution system, for the provision of the share of BGS-CIEP Load covered by the winning bid; and

WHEREAS, pursuant to the auction procedures approved in the _____ Order, the Company and each of the BGS-CIEP Suppliers desire to enter into this BPU-approved BGS-CIEP Supplier Master Agreement ("Agreement") setting forth their respective obligations concerning the provision of BGS-CIEP;

NOW, THEREFORE, in consideration of the mutual covenants and promises set forth below, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound, hereby covenant, promise and agree as follows:

ARTICLE 1: DEFINITIONS

Any capitalized or abbreviated term not elsewhere defined in this Agreement shall have the definition set forth in this Article 1.

Ancillary Services – shall have the meaning ascribed thereto in the PJM Tariff.

Ancillary Service Charge – the price for Ancillary Services in dollars per MWh equal to the amount shown in the Company Specific Addendum approved by the BPU on

Applicable Legal Authorities – generally, those federal and New Jersey statutes and administrative rules and regulations that govern the electric utility industry in New Jersey.

Auction Website - www.bgs-auction.com

Bankruptcy Code – those laws of the United States of America related to bankruptcy, codified and enacted as Title 11 of the United States Code, entitled "Bankruptcy" and found at 11 U.S.C. § 101 et seq., as such laws may be amended, modified, replaced or superseded from time to time.

Basic Generation Service or "BGS" – electric generation service that is provided at retail pursuant to the Applicable Legal Authorities under the Company's retail electric tariffs and under any other agreements or arrangements between the Company and Customers, to any Customer that is not being served by a Third Party Supplier.

Basic Generation Service – Commercial and Industrial Energy Pricing or "BGS-CIEP" – electric generation service that is provided pursuant to the Applicable Legal Authorities at prices that include an energy charge component that varies on an hourly basis in accordance with changes in the load-weighted average Residual Metered Load aggregate Real-Time Locational Marginal Price (including energy, congestion and marginal losses) for the Company's Transmission Zone under the Company's BGS-CIEP Tariffs.

Basic Generation Service – Residential Small Commercial Pricing or "BGS-RSCP" – electric generation service that is provided at retail pursuant to the Applicable Legal Authorities under the Company's BGS-RSCP Tariffs.

Basic Generation Service Supply or "BGS Supply" – either BGS-RSCP Supply or BGS-CIEP Supply.

BGS-CIEP Price – the price resulting from the Company's ____ 2019 BGS-CIEP auction, equal to \$____ per MW-day, and paid to BGS-CIEP Suppliers based on the daily BGS-CIEP Capacity obligation, as specified in Article 9 of this Agreement.

BGS-CIEP Customer – a Customer who is being served under any one of the Company's BGS-CIEP Tariffs and is not taking electric generation service from a Third Party Supplier.

BGS-CIEP-Eligible Customer – any Customer who, if that Customer was taking BGS, would be served under the Company's BGS-CIEP Tariffs, whether or not such Customer receives BGS-CIEP or receives electric generation service from a Third Party Supplier.

BGS-CIEP Load – the full electricity requirement (including without limitation, Energy, Capacity, Ancillary Services and Firm Transmission Service) of BGS-CIEP Customers.

BGS-CIEP Peak Load Share – is the portion of load attributable to those Customers on BGS-CIEP of the PJM-determined EDC zonal peak load.

BGS-CIEP Supplier – an entity that has been selected through the BGS-CIEP solicitation and has accepted the obligations and associated rights to provide BGS-CIEP Supply to retail customers within the State of New Jersey in accordance with the Applicable Legal Authorities and has entered into this Agreement with the Company as a Party. Each BGS-CIEP Supplier under this Agreement must be an LSE and shall have the obligations of an LSE under the PJM Agreements. The term "supplier" also refers generically to any entity authorized by the BPU to provide BGS Supply, as opposed to a specific signatory to this Agreement, where the context makes it appropriate to do so. The distinction can be derived from the context, but is also generally reflected in the use of lower case type ("supplier") to reflect the generic usage, and an initial capital ("Supplier") to reflect a Party to this Agreement.

BGS-CIEP Supplier Responsibility Share — for each BGS-CIEP Supplier, the fixed percentage share of the Company's BGS-CIEP Load for which the BGS-CIEP Supplier is responsible as set forth in Appendix A. The stated percentage share was determined by multiplying the number of Tranches won by the BGS-CIEP Supplier in the auction times the BGS-CIEP Tranche size percentage share.

BGS-CIEP Supply – unbundled Energy, Capacity, Ancillary Services and Firm Transmission Service, including all losses and/or congestion costs associated with the provision of such services, and such other services or products that a BGS-CIEP Supplier may be required, by PJM or other governmental body having jurisdiction, to provide in order to meet the BGS-CIEP Supplier Responsibility Share under this Agreement.

BGS-CIEP Tariffs – Rate Schedules ______ of the Company.

BGS-RSCP Customer – a Customer who is being served under any one of the Company's BGS-RSCP Tariffs.

BGS-RSCP Load – the full electricity requirement (including, without limitation, Energy, Capacity, Ancillary Services and Firm Transmission Service) of BGS-RSCP Customers.

BGS-RSCP Supplier – an entity that has been selected through the BGS-RSCP solicitation and has accepted the obligations and associated rights to provide BGS-RSCP Supply to retail customers within the State of New Jersey in accordance with the Applicable Legal Authorities and has entered into a BGS-RSCP Supplier Master Agreement with the Company as a Party. Each BGS-RSCP Supplier must be an LSE and shall have the obligations of an LSE under the PJM Agreements.

BGS-RSCP Supply – unbundled Energy, Capacity, Ancillary Services and Firm Transmission Service, including all losses and/or congestion costs associated with the provision of such services, and such other services or products that a BGS-RSCP Supplier may be required, by PJM or other governmental body having jurisdiction, to provide in order to meet the BGS-RSCP Supplier Responsibility Share of BGS-RSCP Load.

BGS-RSCP Tariffs – Rate Schedules ______ of the Company.

Billing Month – each calendar month during the term of this Agreement.

BPU or "Board" - the New Jersey Board of Public Utilities or its successor.

Business Day – any day on which the Company's and PJM's corporate offices are open for business and commercial banks are not authorized or required to close in New York, New York.

Capacity – shall mean "Unforced Capacity" as set forth in the PJM RAA or in successor, superseding or amended versions of the PJM RAA that may take effect from time to time over the term of this Agreement, and any successor measurement of generating capacity as may be employed in PJM (whether set forth in the PJM RAA or elsewhere) for the purpose of stating the capacity obligation of an LSE.

Charge – any fee, charge or other amount that is billable by the Company to the BGS-CIEP Supplier under this Agreement.

CIEP Standby Fee – the price, equal to \$0.00015 per kWh, and paid to compensate the BGS-CIEP Supplier for being available to provide BGS-CIEP Supply to the BGS-CIEP Supplier Responsibility Share under this Agreement.

Competitive Electricity Supply – unbundled Energy, Capacity and Firm Transmission Service, including all losses and/or congestion associated with the provision of the foregoing services, and such other services or products that are provided by a Third Party Supplier to fulfill its obligations to serve customer load. The provision of Competitive Electricity Supply by Third Party Suppliers entails fulfillment of all obligations associated with service to Customers, including the obligations of a Load Serving Entity under the PJM Tariff, procedures, agreements and manuals.

Costs – means, with respect to the Non-Defaulting Party, brokerage fees, commissions and other similar transaction costs and expenses reasonably incurred by such Party either

in terminating any arrangement pursuant to which it has hedged its obligations or entering into new arrangements which replace this Agreement; and all reasonable attorneys' fees and expenses incurred by the Non-Defaulting Party in connection with the termination of this Agreement.

Customer – a Company customer as defined in the Company's retail tariffs, eligible to receive Competitive Electricity Supply from a Third Party Supplier or BGS-RSCP or BGS-CIEP from a BGS-RSCP Supplier or a BGS-CIEP Supplier, respectively, in accordance with the Applicable Legal Authorities.

Damages – financial compensation from the Defaulting Party to the Non-Defaulting Party associated with the occurrence of an Event of Default or an Early Termination of this Agreement. This compensation shall be assessed pursuant to Article 5 of this Agreement.

Early Termination – termination of this Agreement prior to the end of the term due to the occurrence of an Event of Default as specified in Section 5.1 of this Agreement and the declaration of Early Termination.

Early Termination Date – the date upon which an Early Termination becomes effective as specified in Section 5.2 of this Agreement.

Effective Date - the date first written above.

Electric Distribution Company or "EDC" – the applicable of the four New Jersey Electric Distribution Companies.

Emergency – (i) an abnormal system condition requiring manual or automatic action to maintain system frequency, or to prevent loss of firm load, equipment damage, or tripping of system elements that could adversely affect the reliability of an electric system or the safety of persons or property; or (ii) a fuel shortage requiring departure from normal operating procedures; or (iii) a condition that requires implementation of Emergency Operations Procedures as defined in the PJM OATT or PJM manuals; or (iv) any other condition or situation that the Company or PJM deems imminently likely to endanger life or property or to affect or impair the Company's electrical system or the electrical system(s) of other(s) to which the Company's electrical system is directly or indirectly connected (a "Connected Entity"). Such a condition or situation may include, but shall not be limited to, potential overloading of the Company's transmission and/or distribution circuits, PJM minimum generation ("light load") conditions, or unusual operating conditions on either the Company's or a Connected Entity's electrical system, or conditions such that the Company is unable to accept Energy from the BGS-CIEP Supplier without jeopardizing the Company's electrical system or a Connected Entity's electrical system.

Energy – three-phase, 60-cycle alternating current electric energy, expressed in units of kilowatt-hours or megawatt-hours.

Energy Portfolio Standards – (i) those standards imposed by the BPU pursuant to N.J.S.A. 48:3-87(d) requiring that specified levels of electric power sold for the provision of Basic Generation Service shall be produced from renewable energy resources as set forth in the Renewable Energy Portfolio Standards, N.J.A.C. 14:8-2 et seq., as modified or superseded, or in such other regulations of the BPU as may be adopted or amended from time to time, and (ii) any standards imposed under any federal, state or local legislation or by any federal, interstate, state or local court, tribunal or governmental agency or authority or regulatory body that have the effect of requiring that specified levels of electric power sold for the provision of Basic Generation Service shall be produced from, or be otherwise attributable to, renewable energy resources or shall be produced from, or be otherwise attributable to, resources emitting only specified amounts of carbon or other substances, as the same may be adopted or amended from time to time.

Event of Default - a breach of obligations under this Agreement as set forth in Section 5.1 of this Agreement.

FERC - the Federal Energy Regulatory Commission.

Final FERC Order – a final order issued by FERC in connection with a request for an increase or decrease in the rates for Firm Transmission Service, or, if FERC has issued an order authorizing Company's recovery of transmission costs through formula rates, the acceptance by operation of law or otherwise of the Company's rate tariff filing seeking an increase or decrease for Firm Transmission Service, which is no longer subject (either actually or potentially) to rehearing or judicial review in which the amount of the increase or decrease is in dispute or in question and is not the subject of proceedings at FERC on remand from any court in which the amount of the specific increase or decrease in the Firm Transmission Service rate is in dispute or in question.

Final Hourly Energy Allocation or "FHEA" – is a quantity in kWh which, for any hour, is the PHEA, adjusted for any billing or metering errors found subsequent to the calculation of PHEA, of which PJM is notified within 60 days.

Final Monthly Energy Allocation or "FMEA" – is a quantity in kWh which, for any Billing Month, is the PMEA adjusted for any billing or metering data received subsequent to the calculation of PMEA of which PJM is notified within 60 days.

Firm Transmission Service – shall mean "Network Integration Transmission Service" under the PJM OATT in effect as of the Effective Date of this Agreement, or in successor, superseding or amended versions of the PJM OATT that may take effect from time to time over the term of this Agreement. In the event the PJM OATT is modified such that "Network Integration Transmission Service" is no longer offered, Firm Transmission Service shall mean the type of transmission service offered under the PJM OATT that is accorded the highest level of priority for scheduling and curtailment purposes.

Gains – means, with respect to any Party, an amount equal to the present value of the economic benefit to it, if any (exclusive of Costs), resulting from an Early Termination of this Agreement, determined in a commercially reasonable manner.

Guaranty – means a guaranty, hypothecation agreement, margins or security agreement or any other document (whether in the form attached to this BGS-CIEP Supplier Master Agreement or other form approved by the Company).

Guarantor – any party who has the authority and may agree to guarantee a BGS-CIEP Supplier's financial obligations under this Agreement, recognizing that such a party will be obligated to meet the Company's creditworthiness requirements for BGS-CIEP Suppliers.

Interest Index – the average Federal Funds Effective Rate for the period of time the funds are on deposit. The Federal Funds Effective Rate is published daily on the Federal Reserve website (http://www.federalreserve.gov/releases/h15/update/).

Kilowatt or "kW" – unit of measurement of useful power equivalent to 1000 watts.

Kilowatt-hour or "kWh" - one kilowatt of electric power used over a period of one hour.

Load Serving Entity or "LSE" – an entity that has been granted the authority or has an obligation pursuant to state or local law, regulation or franchise to sell electricity to retail customers located within the PJM Control Area as that term is defined in the PJM RAA or in successor, superseding or amended versions of the PJM RAA that may take effect from time to time over the term of this Agreement.

Losses – means, with respect to any Party, an amount equal to the present value of the economic loss to it, if any (exclusive of Costs), resulting from an Early Termination of this Agreement, determined in a commercially reasonable manner.

Margin – the amount by which the Total Exposure Amount exceeds a BGS-CIEP Supplier's or Guarantor's credit limit as defined in Section 6.2 of this Agreement.

Maximum Credit Limit - the lesser of the applicable % of TNW and the applicable credit limit cap as specified in Section 6.2 of this Agreement.

Megawatt or MW – one thousand kilowatts.

Megawatt-hour or MWh – one megawatt of electric power used over a period of one hour.

Merger Event – when a Party consolidates or amalgamates with, or merges into or with, or transfers all or substantially all of its assets to another entity and either (i) the resulting entity fails to assume all of the obligations of such Party hereunder or (ii) the benefits of any credit support provided pursuant to Article 6 of this Agreement fail to extend to the performance by such resulting, surviving or transferee entity of the Party's obligations

hereunder, and the resulting entity or its guarantor fails to meet the creditworthiness requirements of this Agreement. Transfer of all or substantially all of the Company's generation assets does not qualify as a Merger Event.

Meter Reading – the process whereby the Company takes notice of the information presented on a Customer's meter. A Meter Reading may be obtained manually, through telemetry, or by estimation, in accordance with the Company's established procedures and practices.

Minimum Rating – a minimum senior unsecured debt rating as defined in Section 6.2(a)(i) of this Agreement.

NERC – the North American Electric Reliability Corporation or its successor.

PJM – the Pennsylvania-New Jersey-Maryland Interconnection L.L.C. or its successor.

PJM Agreements – shall have the meaning ascribed in Section 2.3 of this Agreement.

PJM Control Area – that certain Control Area encompassing systems in Pennsylvania, New Jersey, Maryland, Delaware, Virginia and the District of Columbia, as may be modified from time to time, and which is recognized by the North American Electric Reliability Council as the "PJM Control Area".

PJM OA – the PJM Operating Agreement or its successor.

PJM OATT – the prevailing PJM Open Access Transmission Tariff on file with the FERC, which sets forth the rates, terms and conditions of transmission service over transmission facilities located in the PJM Control Area, as is in effect on the Effective Date and as modified from time to time.

PJM OI – the PJM Office of Interconnection, the system operator for the PJM Control Area.

PJM RAA – the PJM Reliability Assurance Agreement or its successor.

Preliminary Hourly Energy Allocation or "PHEA" – is a quantity in kWh which, for any hour, is the preliminary calculation of the product of the BGS-CIEP Supplier's BGS-CIEP Supplier Responsibility Share and the Energy requirements portion of the BGS-CIEP Load for that hour.

Preliminary Monthly Energy Allocation or "PMEA" – is a quantity in kWh which, for any Billing Month, is the preliminary calculation of the product of the BGS-CIEP Supplier's BGS-CIEP Supplier Responsibility Share and the Energy requirements portion of the BGS-CIEP Load.

PHEA/FHEA Adjustment Amount – for any Billing Month, the monetary amount due to the BGS-CIEP Supplier or the Company, as the case may be, in order to reconcile any

difference between the PHEA used for the purpose of calculating estimated payments made to the BGS-CIEP Supplier for a given hour and the FHEA used for calculating the final payments due to the BGS-CIEP Supplier for such hour as more fully described in Article 9 hereof.

PMEA/FMEA Adjustment Amount – for any Billing Month, the monetary amount due to the BGS-CIEP Supplier or the Company, as the case may be, in order to reconcile any difference between the PMEA used for the purpose of calculating estimated payments made to the BGS-CIEP Supplier for a given month and the FMEA used for calculating the final payments due to the BGS-CIEP Supplier for such month as more fully described in Article 9 hereof.

RFC - the Reliability First Corporation of NERC or its successor.

Service Territory - the geographic areas of the State of New Jersey in which the Company serves electric Customers.

Settlement Amount — with respect to a Non-Defaulting Party, the net amount of the Losses or Gains, and Costs, expressed in U.S. Dollars, which such party incurs as a result of Early Termination, as set forth in Section 5.4(a) of this Agreement. For the purposes of calculating the Termination Payment, the Settlement Amount shall be considered an amount due to the Non-Defaulting Party under this Agreement if the total of the Losses and Costs exceeds the Gains and shall be considered an amount due to the Defaulting Party under this Agreement if the Gains exceed the total of the Losses and Costs.

Statement – a monthly report prepared by the Company for the BGS-CIEP Supplier indicating the amount due to the BGS-CIEP Supplier by the Company in compensation for kWhs supplied to BGS-CIEP Customers by the BGS-CIEP Supplier during the current Billing Month, in accordance with the BGS-CIEP Supplier's obligations under this Agreement.

Supply Day – any calendar day during the term of this Agreement on which the BGS-CIEP Supplier is providing, or is obligated by this Agreement to provide, BGS-CIEP Supply to the Company's BGS-CIEP Customers.

Tangible Net Worth or "TNW" – the total assets less intangible assets and total liabilities. Intangible assets include benefits such as goodwill, patents, copyrights and trademarks.

Termination Payment – shall have the meaning ascribed in Section 5.4 of this Agreement.

Total Exposure Amount – an amount calculated daily for each BGS-CIEP Supplier reflecting the total credit exposure to the Company and consisting of the sum of (i) amount designated as the credit exposure under this Agreement, (ii) any amount(s) designated as the "Mark-to-Market Exposure Amount" arising under any other BGS Supply agreements providing for "BGS-RSCP Supply" or similar BGS service; and (iii) the amount designated as the "credit exposure" under any other BGS Supply agreements providing for "BGS-CIEP Supply" or similar BGS service; provided that in the event the

amount calculated for any day is a negative number, it shall be deemed to be zero for such day.

Third Party Supplier or "TPS" – a person or entity that is duly licensed by the Board to offer and to assume the contractual and legal responsibility to provide electric generation service to retail customers located in the state of New Jersey pursuant to retail open access programs approved by the Board.

Tranche – a fixed percentage share of the BGS-CIEP Load of the Company as determined for the purposes of the auction of the Company's BGS-CIEP Load. The fixed percentage is the Tranche size for the Company.

Transmission Charge – A rate of \$_____ per MW-day based on the applicable OATT for Firm Transmission Service, and in effect on _____, 2019 subject to adjustment under Section 15.9 of this Agreement.

Wholesale Customer – an entity (e.g., a municipality or borough) authorized to take electric service for resale to retail customers under a wholesale contract filed with the FERC.

ARTICLE 2: GENERAL TERMS AND CONDITIONS

2.1 Capacity In Which Company Is Entering Into This Agreement

Each BGS-CIEP Supplier agrees and acknowledges that the Company is contracting for the provision of BGS-CIEP Supply from such Supplier as the agent for Customers receiving Basic Generation Service — Commercial and Industrial Energy Pricing on the Company's distribution system pursuant to the authorizations provided to the Company under the _____ Order. The BGS-CIEP Supplier further agrees and acknowledges that the Company will administer and monitor the BGS-CIEP Supplier's performance in providing BGS-CIEP Supply under this Agreement and that the Company shall be entitled, on behalf of Customers, to enforce BGS-CIEP Suppliers' obligations related to the provision of BGS-CIEP Supply. The BGS-CIEP Supplier hereby permanently and irrevocably waives any claim that Company is not entitled to seek enforcement of this Agreement on behalf of Customers. However, no Customer or group

of Customers may seek enforcement of this Agreement directly against the BGS-CIEP Supplier on their own behalf, including independently or by joining in any legal action by the Company. This Agreement does not create, assign or grant to any Customer or group of Customers any rights in or claims to damages or remedies against the BGS-CIEP Supplier independent of or different from the rights expressly granted to the Company hereunder as agent for Customers.

The Company is also contracting on its own behalf and not as agent for Customers insofar as this Agreement requires the Company: (i) to pay the BGS-CIEP Suppliers as required under this Agreement; and (ii) to provide data or to otherwise cooperate with a BGS-CIEP Supplier in connection with such BGS-CIEP Supplier's provision of BGS-CIEP Supply. The Parties acknowledge that the Agreement is a forward contract and, accordingly, the Parties hereto are entitled to the protections of section 556 of the Bankruptcy Code. The Parties therefore agree that the Agreement may be terminated by either Party upon the commencement of a proceeding by the other Party under any chapter of the Bankruptcy Code in accordance with Section 5.2 of this Agreement.

2.2 Parties' Obligations

(a) Obligations of BGS-CIEP Supplier

Each BGS-CIEP Supplier hereby agrees severally, but not jointly, as follows:

(i) to provide sufficient quantities of BGS-CIEP Supply on an instantaneous basis at all times to meet the BGS-CIEP Supplier Responsibility Share; without limitation, the BGS-CIEP Supplier shall be obligated to procure those services provided by the PJM OI and to perform such functions as may be required by the PJM OI

that are necessary for the delivery of BGS-CIEP Supply required hereunder;

- (ii) to cooperate with the Company in any regulatory compliance efforts that may be required to maintain the ongoing legitimacy and enforceability of the terms of this Agreement and to fulfill any regulatory reporting requirement associated with the provision of BGS-CIEP Supply, before the BPU, FERC or any other regulatory body asserting jurisdiction, including meeting the reporting requirements of any Energy Portfolio Standards and the BPU's Environmental Information Disclosure Standards, N.J.A.C. 14:8-3 et seq.;
- (iii) to become the Load Serving Entity with respect to the provision of BGS-CIEP Supply for the BGS-CIEP Supplier Responsibility Share and to comply with all requirements of a Load Serving Entity with respect to such BGS-CIEP Supplier Responsibility Share;
- (iv) to pay to the Company the PMEA/FMEA Adjustment Amount for any Billing Month in which the PMEA exceeds the FMEA, and to pay to the Company the PHEA/FHEA Adjustment Amount for any Billing Month in which a PHEA/FHEA Adjustment Amount is due to the Company as more fully described in Article 9 of this Agreement;
- (v) to pay to the Company every Billing Month an amount equal to the CIEP Standby Fee times the difference between all BGS-CIEP-Eligible Customers' preliminary monthly Energy usage and all BGS-CIEP-Eligible Customers' final Energy usage as measured at the BGS-CIEP-Eligible Customers' meters, multiplied by the BGS-CIEP Supplier Responsibility Share, if the preliminary monthly Energy usage exceeds final Energy usage for that Billing Month, as more fully described in Article 9 of this

Agreement;

- (vi) to pay to the Company a Charge of \$_____ per Tranche, which amount was announced prior to the auction for each Tranche comprising the BGS-CIEP Supplier's BGS-CIEP Supplier Responsibility Share, in order to reimburse the Company for the total costs of the BGS-CIEP auction and related costs associated with providing BGS-CIEP administration;
- (vii) to satisfy the Energy Portfolio Standards with respect to its BGS-CIEP Supplier Responsibility Share; and
- (viii) to comply in a timely manner with all obligations under this Agreement imposed upon a BGS-CIEP Supplier.

(b) Obligations of the Company

The Company hereby agrees as follows:

- (i) to pay to each BGS-CIEP Supplier every Billing Month for the charges as more fully described in Article 9 of this Agreement and subject to the exceptions set forth therein;
- (ii) to pay to each BGS-CIEP Supplier the PMEA/FMEA Adjustment Amount for any Billing Month in which the FMEA exceeds the PMEA, and the PHEA/FHEA Adjustment Amount for any Billing Month in which a PHEA/FHEA Adjustment Amount is due to the BGS-CIEP Supplier as more fully described in Article 9 of this Agreement and subject to the exceptions set forth therein;
- (iii) to pay to each of the BGS-CIEP Suppliers every Billing Month an amount equal to the CIEP Standby Fee times the difference between all BGS-CIEP-Eligible Customers' preliminary monthly Energy usage and all BGS-CIEP-Eligible

Customers' final Energy usage as measured at the BGS-CIEP-Eligible Customers' meters, multiplied by the BGS-CIEP Supplier Responsibility Share, if the final estimated monthly Energy usage exceeds preliminary Energy usage for that Billing Month, as more fully described in Article 9 of this Agreement and subject to the exceptions set forth therein;

- (iv) to provide to each BGS-CIEP Supplier its estimated aggregate load obligation for each Supply Day twenty (20) days prior to the Supply Day, and its final estimated load obligation for each Supply Day five (5) days prior to the Supply Day;
- (v) to pay each BGS-CIEP Supplier, as approved by the BPU, a percentage share equal to the BGS-CIEP Supplier Responsibility Share of such BGS-CIEP Supplier, of all amounts received from a TPS as damages, penalties or forfeited security due to the failure of such TPS to provide adequate notice in conformance with applicable BPU requirements that a Customer previously served by the TPS is switching to BGS-CIEP service or forfeited as a result of an event of default by a TPS under the Company's Third Party Supplier Agreement; provided that the amounts paid to BGS-CIEP Suppliers shall be net of any amounts retained by the Company to offset costs or losses of the Company associated with the failure of the TPS to provide adequate notice or the occurrence of an event of default under the Company's Third Party Supplier Agreement; and further provided, that the Company shall have no obligation to seek the recovery of any damages, penalties or forfeited security due from a TPS through collection efforts, judicial procedures or otherwise; and
- (vi) to comply in a timely manner with all obligations under this Agreement imposed upon the Company.

(c) Obligation of the BGS-CIEP Customers

The Company hereby agrees on behalf of the BGS-CIEP Customers to accept the delivery of BGS-CIEP Supply necessary to meet the BGS-CIEP Load.

2.3 PJM Services

Each BGS-CIEP Supplier shall make all necessary arrangements for the delivery of BGS-CIEP Supply through the PJM OI. The Company will advise the PJM OI of the magnitude and location of each BGS-CIEP Supplier's actual BGS-CIEP Supplier Responsibility Share, as required by the PJM OI, for the purpose of calculating such BGS-CIEP Supplier's appropriate Firm Transmission Service obligation, Capacity obligation, Energy obligation, or other requirements related to the provision of service under this Agreement by BGS-CIEP Suppliers arising under the PJM OATT, PJM RAA, PJM OA and any other applicable PJM agreement (collectively, the "PJM Agreements"). Each BGS-CIEP Supplier shall remain responsible to PJM for the performance of its LSE obligations associated with the provision of BGS-CIEP Supply under this Agreement until the effective date of the transfer of such LSE obligations.

2.4 Communications and Data Exchange

Each BGS-CIEP Supplier and the Company shall supply to each other all data, materials or other information that is specified in this Agreement, or that may otherwise reasonably be required by BGS-CIEP Suppliers or by the Company in connection with the provision of BGS-CIEP Supply by the BGS-CIEP Supplier to BGS-CIEP Customers, if required, in a thorough and timely manner.

Electronic information exchange between each BGS-CIEP Supplier and the Company under this Agreement shall employ a BGS-CIEP Supplier identification

number, assigned by the Company, which shall be consistent with the BGS-CIEP Supplier's Dunn & Bradstreet Business number. Each BGS-CIEP Supplier must be equipped with the communications capabilities necessary to comply with the communications and data exchange standards that are set by and as may, from time to time, be modified by PJM, and must bear the costs of putting in place and successfully testing all required information technology systems that will enable it to send to and receive data from the Company and PJM and to satisfy its obligations under this Agreement, the PJM Agreements and all other relevant agreements.

2.5 Record Retention

The Company shall retain, for a period of two (2) years following the expiration of the term of this Agreement, necessary records so as to permit BGS-CIEP Suppliers to confirm the validity of payments due to BGS-CIEP Suppliers hereunder; provided that if a BGS-CIEP Supplier has provided notice within two (2) years of the expiration of the term of this Agreement that it disputes the validity of any payments, the Company agrees that it shall retain all records related to such dispute until the dispute is finally resolved.

2.6 Verification

In the event of a good faith dispute regarding any invoice issued or payment due under this Agreement, and provided that a mutually acceptable confidentiality agreement is executed by the Parties, each Party will have the right to verify, at its sole expense, the accuracy of the invoice or the calculation of the payment due by obtaining copies of relevant portions of the books and records of the other Party. The right of verification will survive the termination of this Agreement for a period of two (2) years after termination.

ARTICLE 3: REPRESENTATIONS AND WARRANTIES

3.1 Supplier's Representations and Warranties

Each BGS-CIEP Supplier hereby represents, warrants and covenants to the Company as follows:

- a) such BGS-CIEP Supplier is a corporation, partnership, limited liability company or other legal entity, as set forth in Appendix A hereto, duly organized, validly existing and in good standing under the laws of the State of New Jersey or, if another jurisdiction, under the laws of such jurisdiction and, in such case, is duly registered and authorized to do business and is in good standing in the State of New Jersey;
- b) such BGS-CIEP Supplier has all requisite power and authority to execute and deliver this Agreement and to carry on the business to be conducted by it under this Agreement and to enter into and perform its obligations hereunder, including satisfaction of all applicable FERC requirements;
- c) the execution and delivery of this Agreement and the performance of such BGS-CIEP Supplier's obligations hereunder have been duly authorized by all necessary action on the part of the BGS-CIEP Supplier and do not and will not conflict with, or constitute a breach of or default under, any of the terms, conditions, or provisions of the BGS-CIEP Supplier's certificate of incorporation or bylaws or other constituent instruments or any indenture, mortgage, other evidence of indebtedness, or other agreement or instrument or any statute or rule, regulation, order, judgment, or decree of any judicial or administrative body to which the BGS-CIEP Supplier is a party or by which the BGS-CIEP Supplier or any of its properties is bound or subject;

- d) all necessary and appropriate action that is required on the BGS-CIEP Supplier's part to execute this Agreement has been completed;
- e) this Agreement is the legal, valid and binding obligation of such BGS-CIEP Supplier, enforceable in accordance with its terms;
- f) there are no actions at law, suits in equity, proceedings or claims pending or, to such BGS-CIEP Supplier's knowledge, threatened against the BGS-CIEP Supplier before any federal, state, foreign or local court, tribunal or governmental agency or authority that might materially delay, prevent or hinder the BGS-CIEP Supplier's performance of its obligations hereunder;
- g) it has entered into this Agreement with a full understanding of the material terms and risks of the same, and it is capable of assuming those risks;
- h) the BGS-CIEP Supplier is in good standing as an LSE in PJM, is a signatory to all applicable PJM Agreements, and is in compliance with, and will continue to comply with, all obligations, rules and regulations, as established and interpreted by the PJM OI, that are applicable to LSEs as defined by the PJM Agreements; provided that the BGS-CIEP Supplier shall not be obligated to become an LSE in PJM until the date it begins providing BGS-CIEP Supply to Customers;
- i) it has made its trading and investment decisions (including regarding the suitability thereof) based upon its own judgment and any advice from such advisors as it has deemed necessary and not in reliance upon any view expressed by the Company; and
- j) the BGS-CIEP Supplier will comply with any and all information and data transfer protocols that may be adopted by the Company or that are set by, and from time

to time modified by, the Board; provided that each BGS-CIEP Supplier shall be entitled to exercise its reserved right to challenge any such protocols in the appropriate forum.

3.2 Company's Representations and Warranties

The Company hereby represents, warrants and covenants to the BGS-CIEP Suppliers as follows:

- a) the Company is an electric utility corporation duly organized, validly existing and in good standing under the laws of the State of New Jersey;
- b) the Company has all requisite power and authority to carry on the business to be conducted by it under this Agreement and to enter into and perform its obligations hereunder;
- c) the execution and delivery of this Agreement and the performance of the Company's obligations hereunder have been duly authorized by all necessary action on the part of the Company and do not and will not conflict with, constitute a breach of or default under, any of the terms, conditions, or provisions of the Company's certificate of incorporation or bylaws or any indenture, mortgage, other evidence of indebtedness, or other agreement or instrument or any statute or rule, regulation, order, judgment, or decree of any judicial or administrative body to which the Company is a party or by which the Company or any of its properties is bound or subject;
- d) all necessary and appropriate action that is required on the Company's part to execute this Agreement has been completed;
- e) this Agreement is the legal, valid and binding obligation of the Company, enforceable in accordance with its terms;

- f) the ability of the Company to pay any and all amounts due and payable under this Agreement, or upon any potential breach thereof, is not conditioned upon any governmental or administrative appropriation by the Board, the State of New Jersey or any other governmental authority;
- g) it has entered into this Agreement with a full understanding of the material terms and risks of the same, and it is capable of assuming those risks;
- h) there are no actions at law, suits in equity, proceedings or claims pending or, to the Company's knowledge, threatened against the Company before any federal, state, foreign or local court, tribunal or governmental agency or authority that might materially delay, prevent or hinder the Company's performance of its obligations under this Agreement;
- i) with respect to those rights and entitlements conferred on Customers under this Agreement as set forth in Section 2.1 of this Agreement, the Board has conferred on the Company all requisite power and authority to execute this Agreement on behalf of such Customers;
- j) that the Company's performance under this Agreement is not contingent upon the performance of Customers or the ability of Customers to pay rates;
- k) that the Company shall have full responsibility for metering, billing and delivery with respect to Customers and BGS-CIEP Suppliers shall have no responsibility with respect thereto; and
- that the Company shall be responsible for distribution services and that the
 BGS-CIEP Supplier shall not be responsible for distribution charges.

3.3 Survival of Obligations

All representations and warranties contained in this Article are of a continuing nature and shall be maintained during the term of this Agreement. If a Party learns that any of the representations, warranties or covenants in this Agreement are no longer true during the term of this Agreement, the Party shall immediately notify the other Party via facsimile, with a hard copy of the notice delivered by overnight mail.

ARTICLE 4: COMMENCEMENT AND TERMINATION OF AGREEMENT

4.1 Commencement and Termination

The term of this Agreement shall commence upon the date first written above (the "Effective Date"); provided that the provision of BGS-CIEP Supply by BGS-CIEP Suppliers to BGS-CIEP Customers on the Company's system shall commence at 00:01 a.m. on June 1, 2019 and shall end at midnight on May 31, 2020, unless this Agreement is terminated earlier in accordance with the provisions hereof.

4.2 Termination of Right to Supply BGS-CIEP

Each BGS-CIEP Supplier agrees that termination of this Agreement for reason of an Event of Default shall terminate any right of such BGS-CIEP Supplier to provide BGS-CIEP Supply to the BGS-CIEP Customers and nullify any of the entitlements to which such BGS-CIEP Supplier became entitled as a result of being selected as a winning bidder in the competitive solicitation for BGS-CIEP Supply.

4.3 Survival of Obligations

Termination of this Agreement for any reason shall not relieve the Company or any BGS-CIEP Supplier of any obligation accrued or accruing prior to such termination. Applicable provisions of this Agreement shall continue in effect after termination to the extent necessary to provide for final billings and adjustments including, without limitation, any obligation to pay amounts tracked and retained by the Company during the term of this Agreement for the benefit of the BGS-CIEP Suppliers under Section 15.9 of this Agreement.

4.4 Mutual Termination

The Company and the BGS-CIEP Supplier may agree at any time during the term of this Agreement to terminate their respective rights and obligations hereunder on such terms and under such conditions that they mutually deem to be appropriate as set forth in a mutual termination agreement acceptable in form and substance to the Company and the BGS-CIEP Supplier ("Mutual Termination Agreement"); provided that Company agrees that it shall enter into such a Mutual Termination Agreement, which will discharge the terminating BGS-CIEP Supplier (the "Terminating BGS-CIEP Supplier") with respect to liabilities arising after the effective date of the Mutual Termination Agreement if the following conditions precedent are met: (i) the Terminating BGS-CIEP Supplier identifies a replacement supplier willing to assume all obligations of the Terminating BGS-CIEP Supplier hereunder for the remaining term of this Agreement (the "Replacement BGS Supplier"); (ii) the Replacement BGS Supplier demonstrates its compliance with Article 6 of this Agreement, "Creditworthiness", as of the effective date of the Mutual Termination Agreement; (iii) the Replacement BGS Supplier executes a counterpart signature page to this Agreement and thereby becomes a Party under this Agreement, effective immediately following the effective date of the Mutual Termination Agreement; and (iv) the Terminating BGS Supplier is not, to the belief or knowledge of the Company, subject to an Event of Default as of the effective date of the Mutual

Termination Agreement or, if the Company believes that the Terminating BGS Supplier may be subject to an Event of Default, either (a) the Company has determined that, as of the effective date of the Mutual Termination Agreement, it has not incurred any Damages as a result of the Event of Default or (b) if the Company has determined, as of the effective date of the Mutual Termination Agreement, that it may have incurred Damages as a result of the Event of Default, that the Replacement BGS Supplier has agreed in writing to be responsible for the payment of such Damages or to otherwise cure the Event of Default, in either case to the satisfaction of the Company.

ARTICLE 5: BREACH AND DEFAULT

5.1 Events of Default

An Event of Default under this Agreement shall occur if a Party (the "Defaulting Party"):

- (i) is the subject of a voluntary bankruptcy, insolvency or similar proceeding;
- (ii) makes an assignment for the benefit of its creditors;
- (iii) applies for, seeks consent to, or acquiesces in the appointment of a receiver, custodian, trustee, liquidator or similar official to manage all or a substantial portion of its assets;
- (iv) is dissolved (other than pursuant to a consolidation, amalgamation or merger) or is the subject of a Merger Event;
- (v) has a secured party take possession of all or substantially all of its assets or has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all of its assets;

- (vi) has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger);
- (vii) in the case of a BGS-CIEP Supplier, PJM terminates the BGS-CIEP Supplier's ability to make purchases from PJM markets or PJM holds the Company responsible for the provision of Energy, Capacity, Firm Transmission Service or Ancillary Services to meet the BGS-CIEP Supplier's BGS-CIEP Supplier Responsibility Share under this Agreement;
- (viii) fails to comply with the creditworthiness requirements as set forth in Article 6 of this Agreement, within the time frames set forth in this Agreement;
 - (ix) fails to pay the other Party when payment is due;
- (x) in the case of the Company, fails to implement any measures that it is directed to implement by the Board pursuant to the creditworthiness review procedure adopted by the Board in its Order dated December 4, 2002 (Docket No. EX01110754);
- (xi) violates any federal, state or local code, regulation or statute applicable to the supply of Energy in a manner that materially, and adversely, affects the Party's performance under this Agreement, including by way of failure to continually satisfy all applicable FERC requirements, or, in the case of a BGS-CIEP Supplier, by way of failure to maintain any other governmental approvals required for participation in the New Jersey retail Energy market as a BGS-CIEP Supplier, default on any obligation or other failure to comply with PJM requirements under the PJM Agreements or fails to comply with the Energy Portfolio Standards with respect to the BGS Supplier's BGS-CIEP Supplier Responsibility Share;
 - (xii) is the subject of an involuntary bankruptcy or similar proceeding;

- (xiii) subject to Section 5.3 (b) of this Agreement, in the case of the Company acting on behalf of the BGS-CIEP Customers, fails to accept BGS-CIEP Supply properly tendered by the BGS-CIEP Supplier under this Agreement;
- (xiv) fails to satisfy any other material obligation under this Agreement not listed above;
- (xv) makes a materially incorrect or misleading representation or warranty under this Agreement; or
- (xvi) commits an act or makes an omission that constitutes an "Event of Default" under any other agreement(s) for the provision of BGS Supply between the Company and the BGS-CIEP Supplier;

and fails to remedy such condition, event or delinquency herein above described such that the other Party (the "Non-Defaulting Party") is completely made whole with respect to such condition, event or delinquency, within three (3) Business Days of receipt of written notice thereof from such Non-Defaulting Party; provided, however, that an Event of Default shall be deemed to have occurred immediately, without any need for the provision of notice thereof by the Non-Defaulting Party and without any right of cure on the part of the Defaulting Party, in the event of the occurrence of a condition, event or delinquency described in subsections "i", "ii", "ii", "iv", "v", "vi", "vii" or "viii" above.

5.2 Rights Upon Default

Upon and during the continuation of an Event of Default, the Non-Defaulting Party shall be entitled to:

(i) pursue any and all available legal and equitable remedies;

- (ii) declare an Early Termination Date of this Agreement with respect to the obligations of the Defaulting Party without any liability or responsibility whatsoever except for obligations arising prior to the date of termination, by providing written notice to the Defaulting Party; provided, however, that this Agreement shall immediately terminate automatically and without notice in the case of any Event of Default in which a Supplier is the Defaulting Party occurring under subsections (i), (ii), (iii), (iv), (v), (vii) or (viii) of Section 5.1 of this Agreement and such date of automatic termination shall be deemed the Early Termination Date of this Agreement with respect to such Supplier; and
- (iii) receive Damages in accordance with Section 5.3 of this Agreement.
 The Non-Defaulting Party shall be entitled to elect or pursue one or more of the above remedies.

5.3 Damages Resulting From an Event Of Default

Declaration of Early Termination By Company: Damages resulting from (i) a BGS-CIEP Supplier's failure to (A) provide BGS-CIEP Supply in conformance with Section 2.2 hereof or (B) pay PJM for purchases of any products or services from PJM, or other failure to comply with PJM requirements, such that PJM holds the Company responsible for the provision of Energy, Capacity, Firm Transmission Service or Ancillary Services to meet such BGS-CIEP Supplier's BGS-CIEP Supplier Responsibility Share under this Agreement or (ii) the occurrence of any Event of Default attributable to a BGS-CIEP Supplier resulting in Early Termination, shall include all Costs incurred by the Company, acting in a commercially reasonable manner consistent with any statutory or regulatory

requirement imposed by the Applicable Legal Authorities, in obtaining replacement services or in obtaining a replacement supplier, which Costs exceed the amounts that would have been payable to the defaulting BGS-CIEP Supplier under this Agreement. Costs incurred by the Company for the purpose of calculating Damages hereunder will consist of:

- (i) the cost of Energy (including all charges for losses and congestion), Firm Transmission Service, Capacity, Ancillary Services or other elements of BGS-CIEP Supply allocated to the Company by the PJM OI due to the failure of a BGS-CIEP Supplier to meet obligations owing to the PJM OI in connection with its obligations under this Agreement;
- (ii) the cost of Energy (including all charges for losses and congestion), Firm Transmission Service, Capacity, Ancillary Services or other elements of BGS-CIEP Supply purchased by the Company to replace BGS-CIEP Supply that a BGS-CIEP Supplier was obligated to supply under this Agreement during the term hereof;
- (iii) administrative and legal costs associated with procuring replacement BGS-CIEP Supply; and
- (iv) financial hedging costs incurred by the Company on behalf of BGS-CIEP Customers as a result of having to procure BGS-CIEP Supply not provided by a BGS-CIEP Supplier.

Without limitation of the foregoing, Damages calculated hereunder shall constitute the ultimate liability of a BGS-CIEP Supplier in the event of an Early Termination caused by an Event of Default attributable to such BGS-CIEP Supplier

regardless of the reason or basis for such Early Termination. The Parties recognize, however, that the final calculation of Damages hereunder may not be known for some time since the level of such Damages may be dependent upon the arrangements made by the Company to obtain replacement services or a replacement supplier. The Company and each BGS-CIEP Supplier agree that, until the calculation of Damages under this provision is completed, the amount and payment to the Company of the Settlement Amount on behalf of BGS-CIEP Customers in the event of an Early Termination as set forth in Section 5.4 of this Agreement shall be immediately due and owing as an estimate of all Damages ultimately determined to be due and owing. After Damages have been finally determined under this Section 5.3, the amounts of Damages due and owing will be reconciled with payments already made by the BGS-CIEP Supplier under Section 5.4 of this Agreement.

Supply Tendered By BGS-CIEP Supplier: Damages resulting from the failure of the Company on behalf of Customers to accept BGS-CIEP Supply tendered by the BGS-CIEP Supplier necessary to meet the BGS-CIEP Supplier Responsibility Share of BGS-CIEP Load under this Agreement shall consist of the positive difference (if any) between the amounts that would have been payable to the BGS-CIEP Supplier hereunder had the Company accepted the BGS-CIEP Supplier Responsibility Share of BGS-CIEP Supplier necessary to meet the BGS-CIEP Supplier Responsibility Share of BGS-CIEP Load under this Agreement minus the amount realized by the BGS-CIEP Supplier in disposing, in a commercially reasonable manner, of the BGS-CIEP Supply not accepted by the Company; provided, however, that the Company shall not be required to accept on behalf

of any Customer, quantities of unbundled Energy, Ancillary Services or other component of BGS-CIEP Supply utilized by Customers on an instantaneous basis as a function of electrical load, in excess of such Customer's instantaneous consumption of such component of BGS-CIEP Supply.

- Default Attributable To the Company: Damages resulting from Early Termination due to an Event of Default attributable to the Company shall be as set forth in Section 5.4 of this Agreement. Damages calculated in accordance with said Section 5.4 shall be the exclusive remedy available to the BGS-CIEP Supplier in the event of Early Termination resulting from an Event of Default attributable to the Company.
- (d) Other Damages: Damages for Events of Default not specified above shall consist of the direct Damages incurred by the Non-Defaulting Party.
 - 5.4 Declaration of an Early Termination Date and Calculation of Settlement Amount and Termination Payment
- (a) <u>Settlement Amount</u>: If an Event of Default with respect to a Defaulting Party shall have occurred and be continuing, the Non-Defaulting Party (in the case of an Event of Default by the Company, each BGS-CIEP Supplier shall be considered a "Non-Defaulting Party") shall have the right (i) to designate a day, no earlier than the day such notice is effective and no later than twenty (20) days after such notice is effective, as a date for Early Termination ("Early Termination Date") to accelerate all amounts owing between the Parties and to liquidate and terminate the undertakings set forth in this Agreement, (ii) to withhold any payments due to the Defaulting Party under this Agreement, and (iii) to suspend performance; provided however, that an Early Termination Date shall be deemed to occur automatically and concurrently with the

Event of Default, without any requirement for the provision of notice by the Non-Defaulting Party, with respect to an Event of Default under subsections "i", "ii", "iii", "iv", "v", "vi", "vii" and "viii" of Section 5.1 of this Agreement. The Non-Defaulting Party shall calculate, in a commercially reasonable manner, a Settlement Amount with respect to the obligations under this Agreement. For the purposes of such determination, the quantity amounts of Energy (including all charges for losses and congestion), Capacity and other services provided for under this Agreement for the period following the Early Termination Date through the remainder of the term of this Agreement shall be deemed to be those quantity amounts that would have been delivered on an hourly basis, had this Agreement been in effect during the previous calendar year adjusted for such BGS-CIEP Load changes as may have occurred since the previous calendar year.

(b) Net Out of Settlement Amounts: The Non-Defaulting Party shall calculate a Termination Payment by aggregating all Settlement Amounts due under this Agreement or any other agreements between the Company and the BGS-CIEP Supplier for the provision of BGS Supply into a single amount: by netting out (a) all Settlement Amounts that are due or will become due to the Defaulting Party, plus at the option of the Non-Defaulting Party, any cash or other form of security then available to the Non-Defaulting Party and actually received, liquidated and retained by the Non-Defaulting Party, plus any or all other amounts due to the Defaulting Party under this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply against (b) all Settlement Amounts that are due or will become due to the Non-Defaulting Party, plus any or all other amounts due to the Non-Defaulting Party under this Agreement or any other agreement(s) between the Company and the

BGS-CIEP Supplier for the provision of BGS Supply so that all such amounts shall be netted out to a single liquidated amount; provided however, that if the BGS-CIEP Supplier is the Defaulting Party and the Termination Payment is due to the BGS-CIEP Supplier, the Company shall be entitled to retain a commercially reasonable portion of the Termination Payment, which may be equal to the entire amount of the Termination Payment, as security for additional amounts that may be determined to be due and owing by the BGS-CIEP Supplier as Damages and further provided that any previously attached security interest of the Company in such retained amounts shall continue. The Termination Payment shall be due to or due from the Non-Defaulting Party as appropriate. If the Termination Payment has been retained by the Company as security for additional amounts that may be determined to be due and owing by the BGS-CIEP Supplier, and if, upon making a final determination of Damages, the Termination Payment, or any portion thereof, is to be made to the BGS-CIEP Supplier, the Company will pay simple interest on the Termination Payment amount being made to the BGS-CIEP Supplier. Simple interest will be calculated at the lower of the Interest Index or six (6) percent per annum.

(c) Notice of Termination Payment: As soon as practicable after calculation of a Termination Payment, notice shall be given by the Non-Defaulting Party to the Defaulting Party of the amount of the Termination Payment and whether the Termination Payment is due to or due from the Non-Defaulting Party. The notice shall include a written statement explaining in reasonable detail the calculation of such amount. Subject to Section 5.4(b) above, the Termination Payment shall be made by the Party that owes it within three (3) Business Days after such notice is effective.

- Party disputes the Non-Defaulting Party's calculation of the Termination Payment, in whole or in part, the Defaulting Party shall, within three (3) Business Days of receipt of Non-Defaulting Party's calculation of the Termination Payment, provide to the Non-Defaulting Party a detailed written explanation of the basis for such dispute; provided, however, that if the Termination Payment is due from the Defaulting Party, the Defaulting Party shall first provide commercially reasonable financial assurances to the Non-Defaulting Party in an amount equal to the Termination Payment.
- (e) Multiple BGS Supply Agreements: It is the intention of the Company and the BGS-CIEP Supplier that, in the event the BGS-CIEP Supplier is a party to other agreements with the Company for the provision of BGS Supply that existed prior to the Effective Date of this Agreement or are entered into after the Effective Date of this Agreement, the Company will calculate a single Termination Payment applicable to all such agreements as set forth herein. Each BGS-CIEP Supplier that is a party to such other agreements with the Company for the provision of BGS Supply hereby agrees that such other agreements are deemed amended by this Agreement for the purpose of calculating a single Termination Payment as described herein.

5.5 Step-up Provision

If any one or more BGS-CIEP Suppliers defaults in its obligations hereunder resulting in the exercise of the right of Early Termination by the Company with respect to such BGS-CIEP Supplier(s), then the Company, consistent with its Contingency Plan approved by the Board in its Order dated_______, may, at the Company's option, offer some or all Non-Defaulting Supplier(s) the optional right to assume under

this Agreement additional Tranches of BGS-CIEP Load, subject to compliance with the creditworthiness provisions of Article 6 of this Agreement. The provision of any such offer by the Company to Non-Defaulting Suppliers shall indicate the duration of the offer and the manner of acceptance thereof. Following the assumption by BGS-CIEP Supplier(s) of additional Tranches hereunder, the Company shall prepare a modified Appendix A which shall set forth the revised BGS-CIEP Supplier Responsibility Shares of the BGS-CIEP Load of the participating Non-Defaulting BGS-CIEP Supplier(s) following such assumption. This modified Appendix A shall be initialed (as a single document or in counterparts) by the Company and any affected BGS-CIEP Supplier(s) and shall thereafter be deemed a part of this Agreement, as to such affected BGS-CIEP Supplier(s), from the effective date of such modified Appendix A. A BGS-CIEP Supplier will not suffer any prejudice if it declines an offer to assume additional Tranches upon the default by another BGS-CIEP Supplier.

5.6 Setoff of Payment Obligations of the Non-Defaulting Party

Any payment obligations of the Non-Defaulting Party to the Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply shall be set off: (i) first, to satisfy any payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply that are unsecured and not subject to any Guaranty; (ii) second, to satisfy any payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply that are

unsecured, but which are subject to a Guaranty; and (iii) third, to satisfy any remaining payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply.

5.7 Preservation of Rights of Non-Defaulting Party

The rights of the Non-Defaulting Party under this Agreement, including without limitation Sections 5.4 and 5.6 of this Agreement, shall be supplemental to, and not in lieu of, any right of recoupment, lien, or set-off afforded by applicable law, and all such rights are expressly preserved for the benefit of the Non-Defaulting Party.

ARTICLE 6: CREDITWORTHINESS

6.1 Applicability

Each BGS-CIEP Supplier agrees that it shall meet the creditworthiness requirements of this Article 6 at all times during the term of this Agreements and shall inform the Company immediately of any changes in its credit rating or financial condition. Without limitation of the foregoing, each BGS-CIEP Supplier, upon written request, shall affirmatively demonstrate to the Company, its compliance with the creditworthiness requirements set forth hereunder. The Company may establish less restrictive creditworthiness requirements under this Article 6 in a non-discriminatory manner.

6.2 Creditworthiness Determination

The BGS-CIEP Supplier may submit and maintain a security deposit, in accordance with Section 6.3 below, in lieu of submitting to or being qualified under a

creditworthiness evaluation. The BGS-CIEP Supplier shall have the opportunity to petition the Company to re-evaluate its creditworthiness whenever an event occurs that the BGS-CIEP Supplier believes would improve the determination made by the Company of its creditworthiness. The Company's credit re-evaluation must be completed as soon as possible but no longer than thirty (30) days after receiving a fully documented request. The Company must provide the rationale for its determination of the credit limit and any resulting security requirement. The Company must perform its credit re-evaluation and associated security calculation in a non-discriminatory manner. BGS-CIEP Suppliers shall provide unrestricted access to audited financial statements; provided that if audited financial statements are not available, the Company may specify other types of financial statements that will be accepted.

- (a) The following criteria constitute the Company's creditworthiness requirements for BGS-CIEP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or the District of Columbia and may be utilized by BGS-CIEP Suppliers or Guarantors of BGS-CIEP Suppliers that have not been incorporated or otherwise formed under the laws of the United States. In all instances, the most current senior unsecured debt rating (or, if unavailable, the most current corporate issuer rating discounted by one notch) will be used.
- (i) For BGS-CIEP Suppliers to be granted an unsecured line of credit, to cover the Total Exposure Amount the Supplier shall meet the following requirements: (1) must be rated by at least two of the following rating agencies: S&P Global Ratings ("S&P"), Moody's Investors Service, Inc. ("Moody's"), Fitch, Inc. ("Fitch") or A.M. Best Company ("A.M. Best") and (2) must have a minimum senior unsecured debt rating

(or, if unavailable, corporate issuer rating discounted one notch) of at least "BBB-" from S&P, "Baa3" from Moody's, "BBB-" from Fitch or "bbb" from A.M. Best (a "Minimum Rating"). If the BGS-CIEP Supplier is rated by only two rating agencies, and the ratings are split, the lower rating will be used. If the BGS-CIEP Supplier is rated by three or four rating agencies and the ratings are split, the lower of the two highest ratings will be used; however, in the event that the two highest ratings are common, then such common rating will be used. The Maximum Credit Limit to cover the Total Exposure Amount will be determined based on the following table:

Credit	Rating of the	Max. Credit Limit to be calculated as the lesser of the % of TNW and credit limit cap below			
S&P	Moody's	Fitch	A.M. Best	%	Credit Limit Cap
A- and above	A3 and above	A- and above	aaa	16% of TNW	\$60,000,000
BBB+	Baa1	BBB+	aa	10% of TNW	\$40,000,000
BBB	Baa2	BBB	a	8% of TNW	\$30,000,000
BBB-	Baa3	BBB-	bbb	6% of TNW	\$15,000,000
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW	0

where TNW is the Tangible Net Worth.

The BGS-CIEP Supplier will be required to post cash or a letter of credit at the time of, or prior to the execution of this Agreement in an acceptable form as defined in Section 6.4(b) of this Agreement (see standard format in Appendix B) for the Total Exposure Amount under this Agreement and any other BGS Supply agreements(s) between it and the Company, exceeding the credit limit; or

(ii) For BGS-CIEP Suppliers having a Guarantor, the Guarantor (1) must be rated by at least two of the following rating agencies: S&P, Moody's, Fitch or A.M. Best and (2) must have a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) equal to the Minimum Rating. If the Guarantor is rated by only two rating agencies, and the ratings are split the lower rating will be used. If the Guarantor is rated by three or four rating agencies and the ratings are split, the lower of the two highest ratings will be used; however, in the event that the two highest ratings are common, then such common rating will be used. The Maximum Credit Limit to cover the Total Exposure Amount that could be provided through the Guaranty (see standard format in Appendix C) will be determined based on the following table:

C	redit Rating o	Max. Credit Limit to be calculated as the lesser of the % of TNW and credit limit cap below			
S&P	Moody's	Fitch	A.M. Best	%	Credit Limit Cap
A- and above	A3 and above	A- and above	aaa	16% of TNW	\$60,000,000
BBB+	Baa1	BBB+	aa	10% of TNW	\$40,000,000
BBB	Baa2	BBB	a	8% of TNW	\$30,000,000
BBB-	Baa3	BBB-	bbb	6% of TNW	\$15,000,000
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW	0

The BGS-CIEP Supplier will be granted a credit limit equal to the lesser of (i) the amount of the Guaranty at the time this Agreement is executed as such amount may be modified in any amended or substitute Guaranty provided to the Company during the term of this Agreement or (ii) the Supplier's Maximum Credit Limit. The BGS-CIEP

Suppliers will be required to post cash or letter of credit in an acceptable form as defined in Section 6.4 (b) below (see standard format in Appendix B) for the Total Exposure Amount exceeding the amount of the credit limit granted to the BGS-CIEP Supplier; or

- (iii) The posting of cash or a letter of credit in an acceptable form as defined in Section 6.4 (b) of this Agreement (see standard format in Appendix B) for the Total Exposure Amount shall be required.
- (b) The following standards shall apply to BGS-CIEP Suppliers or Guarantors of BGS-CIEP Suppliers that have not been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia. For BGS-CIEP Suppliers who cannot meet the following requirements, the posting of cash or letter of credit in an acceptable form as defined in Section 6.4(b) below (see standard format in Appendix B) for the Total Exposure Amount shall be required.
- (i) The BGS-CIEP Supplier shall supply such evidence of creditworthiness so as to provide the Company with comparable assurances of creditworthiness as is applicable above for BGS-CIEP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia. The Company shall have full discretion, without liability or recourse to the BGS-CIEP Supplier, to evaluate the evidence of creditworthiness submitted by such BGS-CIEP Supplier; or
- (ii) The Guarantor of a BGS-CIEP Supplier shall supply such evidence of creditworthiness so as to provide the Company with comparable assurances of creditworthiness as is applicable above for Guarantors of BGS-CIEP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or of

the District of Columbia. The Company shall have full discretion, without liability or recourse to the Guarantor or the BGS-CIEP Supplier, to evaluate the evidence of creditworthiness submitted by such Guarantor.

- (c) All BGS-CIEP Suppliers or Guarantors of BGS-CIEP Suppliers that have not been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia shall, in addition to all documentation required elsewhere in this Section 6.2, supply the following as a condition of being granted a credit limit, up to a maximum level, for the purpose of covering the Total Exposure Amount:
- (i) For BGS-CIEP Suppliers: (i) a legal opinion of counsel qualified to practice in the foreign jurisdiction in which the BGS-CIEP Supplier is incorporated or otherwise formed that this Agreement has been duly authorized, executed and delivered and is the legal, valid and binding obligation of the BGS-CIEP Supplier in the jurisdiction in which it has been incorporated or otherwise formed; (ii) the sworn certificate of the corporate secretary (or similar officer) of such BGS-CIEP Supplier that the person executing this Agreement on behalf of the BGS-CIEP Supplier has the authority to execute the Agreement and that the governing board of such BGS-CIEP Supplier has approved the execution of this Agreement; and (iii) the sworn certificate of the corporate secretary (or similar officer) of such BGS-CIEP Supplier that the BGS-CIEP Supplier has been authorized by its governing board to enter into agreements of the same type as this Agreement. The Company shall have full discretion, without liability or obligation to the BGS-CIEP Supplier, to evaluate the sufficiency of the documents submitted by the BGS-CIEP Supplier.

- (ii) For the Guarantor of a BGS-CIEP Supplier: (i) a legal opinion of counsel qualified to practice in the foreign jurisdiction in which the Guarantor is incorporated or otherwise formed that the Guaranty has been duly authorized, executed and delivered and is the legal, valid and binding obligation of the Guarantor in the jurisdiction in which it has been incorporated or otherwise formed; (ii) the sworn certificate of the corporate secretary (or similar officer) of such Guarantor that the person executing the Guaranty on behalf of the Guarantor has the authority to execute the Guaranty and that the governing board of such Guarantor has approved the execution of the Guaranty; and (iii) the sworn certificate of the corporate secretary (or similar officer) of such Guarantor that the Guarantor has been authorized by its governing board to enter into agreements of the same type as this Guaranty. The Company shall have full discretion, without liability or obligation to the Guarantor or the BGS-CIEP Supplier, to evaluate the sufficiency of the documents submitted by such Guarantor.
- (d) Failure to provide the required security or deposit within three (3) Business Days of the Company's request shall constitute an Event of Default unless the Company agrees in writing to extend such period for providing security or deposit. In the event that a creditworthiness determination by the Company concludes that the security requirement may be reduced, the Company will notify the affected BGS-CIEP Supplier of this circumstance, and take whatever action the BGS-CIEP Supplier requests for return of the surplus.

6.3 Credit Exposure/ Security Calculation

The credit exposure per tranche that will be required of BGS-CIEP Supplier under this Agreement shall be \$70,000. The credit exposure under this Agreement shall be

\$70,000 times the number of Tranches shown in Appendix A hereto. If a BGS-CIEP Supplier meets the Minimum Rating listed above, no security will be required as long as the Total Exposure Amount does not exceed the maximum credit limit of the BGS-CIEP Supplier or its Guarantor.

If a BGS-CIEP Supplier meets the Minimum Rating listed above, and the Total Exposure Amount exceeds the maximum credit limit of the BGS-CIEP Supplier or its Guarantor, cash or letter of credit in an acceptable form as defined in Section 6.4(b) of this Agreement (see standard format in Appendix B), in an amount equal to the Total Exposure Amount above the maximum credit limit of the BGS-CIEP Supplier or Guarantor will be required.

6.4 Security Instruments

At each BGS-CIEP Supplier's choice, the following are deemed to be acceptable methods for posting security, if required:

- (a) Cash; or
- (b) A standby irrevocable letter of credit acceptable to the Company issued by a bank or other financial institution with a minimum "A" senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) from S&P or "A2" from Moody's (see standard format in Appendix B). The letter of credit shall state that it shall renew automatically for successive one-year or shorter periods, until terminated upon at least ninety (90) days prior written notice from the issuing financial institution. If the Company receives notice from the issuing financial institution that the letter of credit is being cancelled, the BGS-CIEP Supplier will be required to provide a substitute letter of credit from an alternative bank satisfying the minimum requirements. The receipt of the

substitute letter of credit must be effective as of the cancellation date and delivered to the Company thirty (30) days before the cancellation date of the original letter of credit. If the BGS-CIEP Supplier fails to supply a substitute letter of credit as required, then the Company will have the right to draw on the existing letter of credit and to hold the amount as security.

If the credit rating of a bank or other financial institution from which a BGS-CIEP Supplier has obtained a letter of credit falls below the levels specified in Article 6 of this Agreement, the BGS-CIEP Supplier shall have two (2) Business Days following written notice by the Company to obtain a suitable letter of credit from a bank or other financial institution that meets those standards.

6.5 Maintenance of Creditworthiness

(a) Reporting of Changes

Each BGS-CIEP Supplier shall promptly notify the Company of any change in its credit rating or financial condition or that of its Guarantor. The BGS-CIEP Supplier or Guarantor shall also furnish evidence of an acceptable credit rating or financial condition upon the request of the Company.

(b) Change in Credit Standing

The Company will re-evaluate the creditworthiness of a BGS-CIEP Supplier whenever it becomes aware of an adverse change, through the provision of notice by such Supplier or otherwise, in the BGS-CIEP Supplier's or Guarantor's credit standing. The Company shall not require any additional security from a BGS-CIEP Supplier due to an adverse change in such BGS-CIEP Supplier's credit standing provided that the BGS-CIEP Supplier continues to meet the Company's minimum standard for an unsecured line

of credit. When an adverse change in a BGS-CIEP Supplier's credit standing causes the BGS-CIEP Supplier to no longer qualify for an unsecured line of credit from the Company, the Company shall be entitled to require additional security, prepayment, or a deposit in accordance with Sections 6.2 and 6.3 of this Agreement. The additional security must be in a form acceptable to the Company as specified in Section 6.4 of this Agreement.

6.6 Calling On Security

The Company may call upon the security posted by the BGS-CIEP Supplier if the BGS-CIEP Supplier fails to pay amounts due to the Company pursuant to this Agreement or any other agreement(s) between the Company and the BGS-CIEP Supplier for the provision of BGS Supply after all of the following events occur:

- (a) Written Notice of Default is provided to the BGS-CIEP Supplier; and
- (b) Any applicable cure period ends.

The foregoing notwithstanding, the security posted by the BGS-CIEP Supplier shall become due automatically without prior notice or right of cure in the case of any Event of Default arising under subsections (i), (ii), (iii), (iv), (v), (vi), (vii) and (viii) of Section 5.1 of this Agreement.

6.7 Interest on Cash Held by Company

The Company will pay simple interest calculated at the lower of the Interest Index or six (6) percent per annum on all cash held by the Company pursuant to this Agreement. Each Billing Month, the Company will prepare a statement of interest amounts due to the BGS-CIEP Supplier. The statement will be sent to the BGS-CIEP Supplier within three (3) Business Days after the end of the Billing Month via overnight

mail or other expeditious means. The Company shall make interest payments on the first Business Day after the 5th day of each calendar month.

6.8 Confidentiality

Information supplied by a BGS-CIEP Supplier in connection with the creditworthiness process shall be deemed confidential and not subject to public disclosure, unless Applicable Legal Authorities require disclosure of the information. If information must be disclosed, then the confidentiality of the information shall be maintained consistent with the Applicable Legal Authority's rules and regulations pertaining to confidentiality. The BGS-CIEP Supplier will be given prompt notice of any request by a third party to obtain confidential information related to the BGS-CIEP Supplier's creditworthiness.

6.9 No Endorsement of BGS-CIEP Supplier

The Company's determination that a BGS-CIEP Supplier is creditworthy pursuant to the process set forth above, shall not be deemed to constitute an express or implied warranty or guarantee of any kind with respect to the financial or operational qualifications of the BGS-CIEP Supplier. The Company will treat all BGS-CIEP Suppliers in a non-discriminatory manner and shall provide no preference to any BGS-CIEP Supplier.

6.10 Multiple BGS Supply Agreements

It is the intention of the Company and the BGS Supplier that, in the event the BGS-CIEP Supplier is a party to BGS agreements with the Company for the provision of BGS Supply, the Company will calculate a single Margin applicable to all such agreements. Each BGS-CIEP Supplier that is a party to other agreements with the

Company for the provision of BGS Supply hereby agrees that such other agreements are deemed amended by this Agreement for the sole purpose of calculating the Margin.

ARTICLE 7: PROCEDURES FOR ENERGY SCHEDULING, CAPACITY RESOURCE SUBMISSION AND TRANSMISSION PROCUREMENT

The Parties must adhere to any applicable operational requirements of PJM necessary to protect the integrity of the transmission system within the PJM Control Area and the transmission systems of interconnected control areas, and must satisfy any and all PJM, RFC and NERC criteria, when applicable. The BGS-CIEP Supplier also must adhere to any applicable operational requirements of the Company necessary to protect the integrity of the Company's local distribution system.

The BGS-CIEP Peak Load Share will be determined by the Company based on the zonal peak load contributions utilized in the PJM determination of the zonal obligations for Capacity, adjusted for contributions associated with Customers served by Third Party Suppliers, with Customers served by BGS-RSCP Suppliers and with Wholesale Customers. Contributions for TPSs, shall be calculated as set forth in the Company's TPS operating procedures found on the Company's website at www._____.com. The BGS-CIEP Peak Load Share of a BGS-CIEP Supplier is based on the BGS-CIEP Supplier Responsibility Share.

7.1 Load Obligations

The Company and the BGS-CIEP Supplier acknowledge that the BGS-CIEP Customers are within the Company's metered boundaries and that the BGS-CIEP Load must be divided into BGS-CIEP Supplier obligations by applying the BGS-CIEP Supplier Responsibility Share for each BGS-CIEP Supplier as a Load Serving Entity that

must meet its PJM obligations. These load obligations include, but are not limited to, hourly Energy obligations, Capacity obligations, Ancillary Services obligations, and Firm Transmission Service obligations under the PJM Agreements.

7.2 Data Transmission

(a) Energy

The procedures for transmitting load obligation data for the BGS-CIEP Supplier's hourly Energy obligations shall be as set forth by PJM.

(b) Capacity

The procedures for transmitting the BGS-CIEP Supplier Responsibility Share data to be used by PJM to determine the BGS-CIEP Supplier's Capacity obligations shall be as set forth by PJM.

(c) Transmission

The procedures for transmitting the BGS-CIEP Supplier Responsibility Share data based upon which the BGS-CIEP Supplier will meet its obligations under the PJM OATT shall be as set forth by PJM.

(d) Ancillary Services

The procedures for transmitting data regarding the BGS-CIEP Supplier's Ancillary Services obligations shall be as set forth by PJM.

7.3 Energy Scheduling

The Company will not provide load-forecasting services. The BGS-CIEP Supplier shall schedule Energy resources to meet its obligations with PJM as provided for in the PJM Agreements, procedures, and manuals. The Company, through an e-schedule, shall provide PJM and the BGS-CIEP Supplier with the data regarding the BGS-CIEP

Supplier Responsibility Share of the Energy obligations, as set forth by PJM. The Energy obligations for each BGS-CIEP Supplier will be determined based on its BGS-CIEP Supplier Responsibility Share of the BGS-CIEP Load. The total preliminary BGS-CIEP Energy obligation will be equal to the total Energy loads for the BGS-CIEP Customers, as calculated by the Company, including losses and unaccounted for energy.

ARTICLE 8: THE ENERGY SETTLEMENT/RECONCILIATION PROCESS

8.1 Energy Settlement By PJM

The settlement process occurs at PJM to reflect the BGS-CIEP Supplier's actual Energy obligations in a supply/usage reconciliation process. The Energy obligations for each BGS-CIEP Supplier will be determined based on the BGS-CIEP Supplier Responsibility Share of the BGS-CIEP Load. The reconciled total BGS-CIEP Energy obligation will be equal to the final total Energy loads for the Customers receiving BGS-CIEP service, including losses.

Any adjustments for billing and metering errors reported subsequent to the calculation of FHEA and FMEA will be proportionally allocated by the Company to the BGS-CIEP Suppliers.

8.2 Energy Settlement by the Company

In the event that actual BGS-CIEP Customer consumption data is not available until after the PJM deadline for conducting the final settlement, the Company will conduct the settlement process with the BGS-CIEP Supplier. Should PJM impose penalties against the Company as a result of the BGS-CIEP Supplier's transactions or failure to meet PJM requirements, such penalties shall be passed through by the Company, to the BGS-CIEP Supplier as part of this settlement process. In addition, all other applicable charges from

PJM, including any billing adjustments, will be appropriately allocated to the BGS-CIEP Supplier.

ARTICLE 9: BILLING AND PAYMENT

9.1 The Company Payment of Obligations to the BGS-CIEP Supplier

The Company shall pay all amounts due to the BGS-CIEP Supplier hereunder in accordance with the following provisions:

- (a) Each Billing Month, the Company will prepare a Statement of amounts due to the BGS-CIEP Supplier. Line items on this Statement will show amounts due equal to the: (i) CIEP Standby Fee multiplied by the sum of the BGS-CIEP Supplier Responsibility Share of all BGS-CIEP-Eligible Customers' preliminary Energy usage as measured at the BGS-CIEP-Eligible Customers' meters; plus (ii) Energy Charges equal to the sum of the products of the load-weighted average Residual Metered Load aggregate Real-Time Locational Marginal Prices for the Company's Transmission Zone multiplied by the PHEA in each hour of the Billing Month; plus (iii) Ancillary Service Charge equal to the product of \$6.00 per MWh times the PMEA for the Billing Month; plus (iv)BGS-CIEP Price equal to the product of \$_____ per MW-day multiplied by the BGS-CIEP Supplier Responsibility Share of the BGS-CIEP Capacity obligation (expressed in MW) for each day of the Billing Month in question; plus (v) the BGS-CIEP Firm Transmission Service obligation for each day of the Billing Month.
- (b) The Statement will be sent to the BGS-CIEP Supplier within eight (8) Business Days after the end of the Billing Month via overnight mail or other expeditious means.

- (c) The Company shall make payment on the first Business Day after the 19th day of each calendar month provided that the Company maintains a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) of at least "BBB-" from S&P, "Baa3" from Moody's or "BBB-" from Fitch (the "Required Rating"). If the Company is rated by only two rating agencies, and the ratings are split, the lower rating will be used. If Company is rated by three rating agencies, and the ratings are split, the lower of the two highest ratings will be used, and, in the event that the two highest ratings are common, such common rating will be used.
- (d) In the event that the Company's minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) falls below the Required Rating, and until the Company's minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) becomes equal or higher than the Required Rating, (i) the Company shall make an initial payment on the first Business Day after the 5th day of the calendar month for approximately 50% of the amount due to the BGS-CIEP Supplier for the previous calendar month (the "Initial Payment"), and (ii) the Company shall make a second payment on the first Business Day after the 19th day of the calendar month for any remaining amounts associated with the previous calendar month, which will include the difference between the Initial Payment and any amounts due equal to the (A) CIEP Standby Fee multiplied by the sum of the BGS-CIEP Supplier Responsibility Share of all BGS-CIEP-Eligible Customers' preliminary Energy usage as measured at the BGS-CIEP-Eligible Customers' meters; plus (B) Energy Charges equal to the sum of the products of the load-weighted average Residual Metered Load aggregate Real-Time Locational Marginal Prices for the Company's Transmission Zone

multiplied by the PHEA in each hour of the Billing Month; plus (C) Ancillary Service Charges equal to the product of \$6.00 per MWh times the PMEA for the Billing Month; plus (D) the BGS-CIEP Price equal to the product of \$_____ per MW-day multiplied by the BGS-CIEP Supplier Responsibility Share of the BGS-CIEP Capacity obligation (expressed in MW) for each day of the Billing Month in question; plus (E) Transmission Charges equal to the PJM OATT daily rate for the Company Transmission Zone multiplied by the BGS-CIEP Supplier Responsibility Share of the BGS-CIEP Firm Transmission Service obligation for each day of the Billing Month.

- (e) To the extent that the PMEA differs from the FMEA, the Company will pay or charge the BGS-CIEP Supplier the PMEA/FMEA Adjustment Amount within the PJM deadline for conducting the final settlement. PMEA/FMEA Adjustment Amounts apply only to payments of the Ancillary Service Charge.
- (f) To the extent that the FHEA differs from the PHEA, the Company will calculate the PHEA/FHEA Adjustment Amount for each hour by multiplying the difference between the two amounts by the Residual Metered Load aggregate Real-Time Locational Marginal Price for the Company's zone, and will sum the negative and positive dollar values over all hours to arrive at a net PHEA/FHEA Adjustment Amount for the Billing Month. Based on the calculated net PHEA/FHEA Adjustment Amount, the Company will pay or charge the BGS-CIEP Supplier for billing adjustments (the "FHEA/PHEA Billing Adjustment") within the PJM deadline for conducting the final settlement.
- (g) To the extent that the preliminary monthly Energy usage of all BGS-CIEP-Eligible Customers differs from the final monthly Energy usage as measured at the

BGS-CIEP-Eligible Customers' meters, the Company will pay or charge each BGS-CIEP Supplier every Billing Month an amount equal to the CIEP Standby Fee times the difference between all BGS-CIEP-Eligible Customers' preliminary monthly Energy usage and all BGS-CIEP-Eligible Customers' final Energy usage, multiplied by the BGS-CIEP Supplier Responsibility Share.

- (h) To the extent that the daily Capacity and Firm Transmission Service obligations used in the calculation detailed in Section 9.1(a) are adjusted after the PJM deadline for conducting final settlement, the Company will pay or charge the BGS-CIEP Suppliers any net difference between the payment calculated and made within the PJM deadline for conducting final settlement, and the payment calculated using the adjusted values.
- (i) If each Party owes an amount to the other Party pursuant to this Agreement, including any related interest, and payments or credits, the Parties may satisfy their respective obligations to each other by netting the aggregate amounts due to one Party against the aggregate amounts due to the other Party, with the Party, if any, owing the greater aggregate amount paying the other Party the difference between the amounts.
- (j) Payments shall be subject to adjustment for any arithmetic errors, computation errors, Meter Reading errors, or other errors, provided that the errors become known within one (1) year of the termination of this Agreement.
- (k) The Company shall make payments of funds payable to the BGS-CIEP Supplier by electronic transfer to a bank designated by the BGS-CIEP Supplier.

- (I) If a good faith dispute arises between the Company and the BGS-CIEP Supplier regarding a Statement, the disputing Party shall be obligated to pay only the undisputed portion of the Statement, if any, and shall present the dispute in writing and submit supporting documentation to the non-disputing Party within one hundred and twenty (120) calendar days from the date of the Statement in dispute. Statement disputes shall be addressed promptly, and in accordance with the dispute resolution procedures set forth in Article 11 of this Agreement. Upon resolution of a Statement dispute, any payments made to either Party will include simple interest on the payment at the lower of the Interest Index or six (6) percent per annum payable from the date that notice of a Statement dispute was received by the non-disputing Party.
- (m) If payment is made to the BGS-CIEP Supplier after the due date shown on the Statement, a late fee will be added to the unpaid balance until the entire Statement is paid. This late fee will be calculated at the prime rate commercial borrowers are charged by J.P. Morgan Chase.

9.2 Billing for BGS-CIEP Supplier's Obligations to Other Parties

The Company shall have no responsibility for billing between: the BGS-CIEP Supplier and PJM; the BGS-CIEP Supplier and any Energy or Capacity source; or the BGS-CIEP Supplier and any other third party. The Company will be responsible for billing BGS-CIEP Customers for BGS-CIEP.

9.3 The BGS-CIEP Supplier Payment of Obligations to the Company

The BGS-CIEP Supplier shall pay all Charges it incurs hereunder in accordance with the following provisions:

- (a) Each Billing Month, the Company shall submit an invoice to the BGS-CIEP Supplier for all Charges owed by the BGS-CIEP Supplier under this Agreement. The BGS-CIEP Supplier shall make payment for Charges incurred on or before the due date shown on the invoice. The due date will be on the first Business Day after the 19th day of each calendar month. The invoice will be sent to the BGS-CIEP Supplier within eight (8) Business Days after the end of the Billing Month via overnight mail or other expeditious means.
- (b) Invoices shall be subject to adjustment for any arithmetic errors, computation errors, Meter Reading errors, or other errors, provided that the errors become known within one (1) year of the termination of this Agreement.
- (c) The BGS-CIEP Supplier shall make payments of funds payable to the Company by electronic transfer to a bank designated by the Company.
- (d) If a good faith dispute arises between the Company and the BGS-CIEP Supplier regarding an invoice, the disputing Party shall pay only the undisputed portion of the invoice, if any, and shall present the dispute in writing and submit supporting documentation to the non-disputing Party within one hundred twenty (120) calendar days from the due date of the invoice in dispute. Billing disputes shall be addressed promptly, and in accordance with the dispute resolution procedures set forth in Article 11 of this Agreement. Upon resolution of a billing dispute, any payments made to either Party will include simple interest on the payment at the lower of the Interest Index or six (6) percent per annum payable from the date that notice of a bill dispute was received by the non-disputing Party.

(e) If payment is made to the Company after the due date shown on the invoice, a late fee will be added to the unpaid balance until the entire invoice is paid. This late fee will be calculated at the prime rate commercial borrowers are charged by J.P. Morgan Chase.

ARTICLE 10: SYSTEM OPERATION

10.1 Disconnection and Curtailment By the Company

The Company shall have the right, without incurring any liability to BGS-CIEP Suppliers, to disconnect (or otherwise curtail, interrupt or reduce deliveries from) the BGS-CIEP Suppliers or to disconnect (or otherwise curtail, interrupt or reduce deliveries to) any Customer whenever the Company determines in the exercise of its good faith discretion, or when the Company is directed by PJM, that such a disconnection, curtailment, interruption or reduction is necessary to facilitate construction, installation, maintenance, repair, replacement or inspection of any of the Company's facilities; or due to any other reason affecting the safe and reliable operation of the Company's or a Customer's facilities, including Emergencies, forced outages or potential overloading of the Company's transmission and/or distribution circuits, potential damage to any Customer's facilities or any risk of injury to persons.

10.2 Inadvertent Loss of Service To BGS-CIEP Customers

The Parties agree and acknowledge that service to BGS-CIEP Customers may be inadvertently lost due to storms, weather, accidents, breakage of equipment or other events beyond the reasonable control of the Company affecting the transmission and distribution system of the Company. Neither Party will have any liability to the other Party for the occurrence of such events except for the Company's obligation to pursue

steps for the resumption of the disrupted service as set forth in Section 10.3 below. In no event will an inadvertent loss of service affect a Party's obligation to make any payments then due or becoming due with respect to performance rendered prior to such inadvertent loss of service.

10.3 Good Faith Efforts

The Company shall use good faith efforts to: (a) minimize any curtailment, interruption or reduction to the extent practicable under the circumstances; (b) provide the BGS-CIEP Supplier with prior notification of any curtailment, interruption or reduction, to the extent practicable; and (c) resume service as promptly as practicable.

10.4 PJM Requirements

The BGS-CIEP Supplier acknowledges and agrees that, as a member of PJM, the Company is bound by all PJM operating instructions, policies and procedures as are currently set forth in the PJM Operating Manual, which are available through the Internet on the PJM Home Page (http://www.pjm.com), as may be revised from time to time, which are needed to maintain the integrity of the PJM system. The BGS-CIEP Supplier acknowledges and agrees that it will cooperate with the Company so that the Company will be in compliance with all PJM Emergency Operations Procedures, which include, but are not limited to, those procedures pertaining to minimum and maximum generation Emergencies, and measures requiring involuntary Customer participation, such as supply voltage reduction or full interruption of Customer load by either manual or automatic means.

10.5 Compliance With Governmental Directives

The BGS-CIEP Supplier also acknowledges and agrees that the Company may need to act in response to governmental or civil authority directives that may affect BGS-CIEP Customer load. The BGS-CIEP Supplier agrees to cooperate with the Company in order to comply with said directives.

ARTICLE 11: DISPUTE RESOLUTION

11.1 Informal Resolution of Disputes

The Company and the BGS-CIEP Supplier shall use good faith and reasonable commercial efforts to informally resolve all disputes arising out of the implementation of this Agreement. The BGS-CIEP Supplier's point of contact for all information, operations, and questions shall be the Company's Basic Generation Service Unit and the Auction Website. Any dispute between the Company and the BGS-CIEP Supplier under this Agreement may be referred to a designated senior representative of each of the Parties for resolution on an informal basis as promptly as practicable.

11.2 Recourse to Agencies or Courts of Competent Jurisdiction

Nothing in this Agreement shall restrict the rights of either Party to file a complaint with the FERC under relevant provisions of the Federal Power Act ("FPA"), with the BPU under relevant provisions of the Applicable Legal Authorities, with a New Jersey State court of competent jurisdiction, or with a federal court of competent jurisdiction situated in the State of New Jersey. The Parties' agreement hereunder is without prejudice to any Parties' right to contest the jurisdiction of the agency or court to which a complaint is brought.

To the extent that this Agreement is deemed to be subject to FERC jurisdiction, absent the agreement of all parties to a proposed change, the standard of review for changes to any section of the Agreement specifying the rate(s) or other material economic terms and conditions agreed to by the Parties herein, whether proposed by a Party, a non-party, the BPU or FERC acting sua sponte, will be the "public interest" standard of review set forth in <u>United Gas Pipe Line Co. v. Mobile Gas Service Corp.</u>, 350 U.S. 332 (1956) and <u>Federal Power Commission v. Sierra Pacific Power Co.</u>, 350 U.S. 348 (1956).

ARTICLE 12: REGULATORY AUTHORIZATIONS AND JURISDICTION

12.1 Compliance With Applicable Legal Authorities

The Company and the BGS-CIEP Supplier are subject to, and shall comply with, all existing or future applicable federal, State and local laws, all existing or future duly-promulgated orders or other duly-authorized actions of PJM or of Applicable Legal Authorities.

12.2 FERC Jurisdictional Matters

The inclusion herein of descriptions of procedures or processes utilized by PJM or otherwise subject to the jurisdiction of FERC is intended solely for informational purposes. If anything stated herein is found by the FERC to conflict with or be inconsistent with any provision of the FPA, or any rule, regulation, order or determination of the FERC under the FPA or if any existing procedures or processes utilized by PJM are duly modified, the applicable FERC rule, regulation, order, determination or modification shall control. To the extent required under any provision of the FPA, or any rule, regulation, order or determination of the FERC under the FPA,

the Company and/or the BGS-CIEP Supplier, if applicable, shall use reasonable commercial efforts to secure, from time to time, all appropriate orders, approvals and determinations from the FERC necessary to support this Agreement.

ARTICLE 13: LIMITATION OF LIABILITY

13.1 Limitations on Liability

Except to the extent expressly set forth in this Agreement, each Party shall be liable to the other Party only for direct damages incurred as a result of such Party's failure to comply with this Agreement and no Party shall have any liability to the other Party for consequential, indirect, special or punitive damages, including lost profits or lost revenues, arising out of such Party's failure to comply with its obligations under this Agreement. Notwithstanding anything to the contrary in this Agreement, the BGS-CIEP Suppliers have no obligations or liability to other BGS-CIEP Suppliers.

This Agreement does not create any duty to any BGS-CIEP Supplier with respect to the administration of the Company's Third Party Supplier Agreements by the Company, including, without limitation, with respect to the credit and security provisions of the Company's Third Party Supplier Agreements and Article 11 thereof. Nor will the Company have any liability to any BGS-CIEP Supplier for any act or failure to act in connection with such administration. If any dispute arises among the BGS-CIEP Suppliers regarding the allocation of forfeited security (net of the Company's costs as described in Section 11.3 of the Company's Third Party Supplier Agreements) due to BGS-CIEP Suppliers pursuant to Article 11 of the Company's Third Party Supplier Agreements, then such forfeited security (net of the Company's costs as described in

Section 11.3 of the Company's Third Party Supplier Agreements) will be deposited into an interest-bearing account pending final disposition of such dispute.

13.2 Risk of Loss

Solely for purposes of determining risk of loss and for determining the indemnity obligations under Article 14 of this Agreement, the Company shall be deemed to have custody and control of the electric Energy delivered by a BGS-CIEP Supplier upon receipt thereof into the Company's distribution system and until delivery thereof at the retail electric meter of the Customer; and each BGS-CIEP Supplier shall be deemed to have custody and control of the electric Energy at all times prior to receipt thereof by the Company. Each BGS-CIEP Supplier shall at all times be deemed to hold title to electric Energy until delivery at the retail electric meter of the Customer at which time title shall be deemed to pass to such Customer. The Party deemed to have custody and control of electric Energy shall, among the Parties to this Agreement, be responsible for all loss or damage to property or injury or death to persons arising in connection with such electric Energy while in its custody and control and shall indemnify the other Parties with respect to same as set forth in Article 14 of this Agreement.

ARTICLE 14: INDEMNIFICATION

14.1 Indemnification

(a) Should the Company become the defendant in, or obligor for, any third party claims and/or liabilities for losses, penalties, expenses, damage to property, injury to or death of any person including a Party's employees or any third parties, that were caused by or occur in connection with an act or omission of a BGS-CIEP Supplier with respect to an obligation arising under or in connection with this Agreement, or for which

such BGS-CIEP Supplier has otherwise assumed liability under the terms of this Agreement, such BGS-CIEP Supplier shall defend (at the Company's option), indemnify and hold harmless the Company, its shareholders, board members, directors, officers and employees, from and against any and all such third party claims and/or liabilities, except to the extent that a court of competent jurisdiction determines that the losses, penalties, expenses or damages were caused wholly or in part by the gross negligence or willful misconduct of the Company. The Company may, at its own expense, retain counsel and participate in the defense of any such suit or action.

(b) Should a BGS-CIEP Supplier (the "Indemnified Supplier") become the defendant in, or obligor for, any third party claims and/or liabilities for losses, penalties, expenses, damage to property, injury to or death of any person including a Party's employees or any third parties, that were caused by or occur in connection with an act or omission of the Company or another BGS-CIEP Supplier with respect to an obligation arising under or in connection with this Agreement, or for which the Company or such other BGS-CIEP Supplier has otherwise assumed liability under the terms of this Agreement, the Company or such BGS-CIEP Supplier shall defend (at the option of the Indemnified Supplier), indemnify and hold harmless the Indemnified Supplier, its shareholders, board members, directors, officers and employees, from and against any and all such third party claims and/or liabilities, except to the extent that a court of competent jurisdiction determines that the losses, penalties, expenses or damages were caused wholly or in part by the gross negligence or willful misconduct of the Indemnified The Indemnified Supplier may, at its own expense, retain counsel and participate in the defense of any such suit or action.

14.2 Survives Agreement

The obligation of a Party to defend, indemnify, and hold harmless another Party under this Article shall survive termination of this Agreement, and shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for either Party under any statutory scheme, including any Worker's Compensation Acts, Disability Benefit Acts or other Employee Benefit Acts.

ARTICLE 15: MISCELLANEOUS PROVISIONS

15.1 Notices

Unless otherwise stated herein, all notices, demands or requests required or permitted under this Agreement shall be in writing and shall be personally delivered or sent by overnight express mail, courier service or facsimile transmission (with the original transmitted by any of the other aforementioned delivery methods) addressed as follows:

If to a BGS-CIEP Supplier:

Notification information for each BGS-CIEP Supplier is set forth on Appendix A hereto.

If to the Company to:

Copy to:

or to such other person at such other address as a Party shall designate by like notice to the other Party. Notice received after the close of the Business Day shall be deemed received on the next Business Day; provided that notice by facsimile transmission shall be deemed to have been received by the recipient if the recipient confirms receipt telephonically or in writing.

15.2 No Prejudice of Rights

The failure of a Party to insist on any one or more instances upon strict performance of any provisions of this Agreement, or to take advantage of any of its rights hereunder, shall not be construed as a waiver of any such provisions or the relinquishment of any such right or any other right hereunder, which shall remain in full force and effect. No term or condition of this Agreement shall be deemed to have been waived and no breach excused unless such waiver or consent to excuse is in writing and signed by the Party claimed to have waived or consented to excuse.

15.3 Assignment

Parties shall not assign any of their rights or obligations under this Agreement without obtaining (a) any necessary regulatory approval(s); and (b) the prior written consent of the non-assigning Party, which consent shall not be unreasonably withheld; provided that the Company agrees that it shall grant its consent to a proposed assignment by a BGS-CIEP Supplier if the proposed assignee meets all of the Company's creditworthiness requirements then in effect under Article 6 of this Agreement; and further provided that a BGS-CIEP Supplier wishing to assign its interests hereunder shall not be obligated to obtain the consent of any other BGS-CIEP Supplier. No assignment of this Agreement shall relieve the assigning Party of any of its obligations under this Agreement until such obligations have been assumed by the assignee and all necessary consents have been obtained. Any assignment in violation of this Section 15.3 shall be void; provided, however, the Company may assign any or all of its rights and obligations

under this Agreement, without the BGS-CIEP Supplier's consent, to any entity succeeding to all or substantially all of the assets of the Company, if such assignee agrees, in writing, to be bound by all of the terms and conditions hereof and all necessary regulatory approvals are obtained. The BGS-CIEP Supplier may, with prior written notice to the Company but without obtaining the approval of the Company, assign the accounts, revenues or proceeds under this Agreement to a third party. The Company agrees that, following receipt of such notice of the assignment of accounts, revenues or proceeds and such other documentation that the Company may reasonably request, the Company will pay amounts becoming due to the assigning BGS-CIEP Supplier under this Agreement directly to the designated assignee; provided, however, that nothing herein shall enlarge or expand the rights of such designated assignee beyond the rights granted to the BGS-CIEP Supplier and the right of such designated assignee to receive payments shall be subject to all defenses, offsets and claims of the Company arising under this Agreement. The Company further agrees that, in the event necessary regulatory approvals to effectuate an assignment have been sought in good faith but that action by the regulatory body is pending, the Company shall accept the performance of the proposed assignee as a Party to this Agreement, as co-obligor with the Party proposing to assign its interest, until such approvals are obtained; provided that in the event the regulatory body declines to grant its approval (or in the discretion of the Company, the application seeking approval is still pending without action by the regulatory body after ninety (90) days), the request for approval of the assignment shall be deemed to have been rejected.

15.4 Governing Law and Venue

To the extent not subject to the jurisdiction of the FERC, questions including those concerning the formation, validity, interpretation, execution, amendment, termination and construction of this Agreement shall be governed by the laws of the State of New Jersey, without regard to principles of conflicts of law. Any lawsuit arising in connection with this Agreement shall be brought only in the State or federal courts of New Jersey.

15.5 Headings

The headings and subheadings contained in this Agreement are used solely for convenience and do not constitute a part of the Agreement between the Parties hereto, nor should they be used to aid in any manner in the construction of this Agreement.

15.6 Third Party Beneficiaries

This Agreement is intended solely for the benefit of the Parties hereto including Customers for which the Company is executing this Agreement as agent. Nothing in this Agreement shall be construed to create any duty, or standard of care with reference to, or any liability to, any person not a Party to this Agreement.

15.7 General Miscellaneous Provisions

(a) This Agreement shall not be interpreted or construed to create an association, joint venture, or partnership between the Parties (or any of them), or to impose any partnership obligation or liability upon any Party. The obligations of the BGS-CIEP Suppliers are expressly agreed to be several and not joint. No Party shall have any right, power, or authority to enter into any agreement or undertaking for, or on

behalf of, or to act as or be an agent or representative of, or to otherwise bind, any other Party.

- (b) Cancellation, expiration or Early Termination of this Agreement shall not relieve the Parties of obligations that by their nature survive such cancellation, expiration or termination, including warranties, remedies, promises of indemnity and confidentiality.
- (c) Should any provision of this Agreement be held invalid or unenforceable, such provision shall be invalid or unenforceable only to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable any other provision hereof unless it materially changes the agreement of the Parties.
- (d) Each of the Parties acknowledges that it has read this Agreement, understands it, and agrees to be bound by its terms. This Agreement is intended by the Parties as a final expression of their agreement. The Parties further agree that this Agreement is the complete and exclusive statement of agreement and supersedes all proposals (oral or written), understandings, representations, conditions, warranties, covenants and all other communications between the Parties relating thereto.

15.8 Taxes

All present and future federal, state, municipal or other taxes imposed by any taxing authority by reason of the provision of BGS-CIEP Supply to BGS-CIEP Customers by a BGS-CIEP Supplier under this Agreement shall be the liability of the BGS-CIEP Supplier, except for New Jersey State Sales and Use Taxes, which will be the Company's responsibility to collect from BGS-CIEP Customers for remittance to the applicable taxing authority. Should a BGS-CIEP Supplier be required to remit any New

Jersey State Sales and Use Taxes directly to the applicable taxing authority, other than taxes previously collected by the BGS-CIEP Supplier on behalf of the Company, the Company will defend and indemnify the BGS-CIEP Supplier for such Sales and Use Taxes and will pay to the BGS-CIEP Supplier all such tax amounts upon demand. Each BGS-CIEP Supplier shall pay all such taxes to the applicable taxing authority to the extent required or permitted by law. If any transaction is exempt from the payment of any such taxes, the affected BGS-CIEP Supplier will, if requested, provide the Company with valid tax exemption certificates. Should the Company be required to remit any such taxes directly to any applicable taxing authority, other than taxes previously collected by the Company directly from a BGS-CIEP Supplier, the BGS-CIEP Supplier will defend and indemnify the Company and will pay to the Company all such tax amounts upon demand.

If new taxes are imposed on Energy, Capacity, Firm Transmission Service or Ancillary Services after the Effective Date of this Agreement, within forty-five (45) days of the final adoption of any such new taxes, the Company will notify the BGS-CIEP Suppliers that such new tax has been adopted, will seek approval from the Board to collect the new taxes from BGS-CIEP Customers, and will provide the BGS-CIEP Suppliers with a copy of the Company's petition seeking such approval from the Board. Upon receipt of Board approval of the collection of the new taxes from BGS-CIEP Customers, the BGS-CIEP Supplier will be excused from liability for payment of those new taxes.

15.9 Changes in Transmission Charges for Firm Transmission Service

- (a) If during the term of this Agreement, a filing is made with the FERC to increase or decrease the charges for Firm Transmission Service, including any charge or surcharge imposed on customers receiving Firm Transmission Service, or if the charges for Firm Transmission Service are adjusted pursuant to a FERC-authorized formula rate, then the following procedures shall apply:
- (i) within forty-five (45) days of the date upon which the rate change filing or formula rate informational filing is made with the FERC, the Company will notify the BGS-CIEP Suppliers that such rate filing has been made, and will seek approval from the Board to increase or decrease the charges to BGS-CIEP Customers by the amount of such rate adjustment for Firm Transmission Service; provided that, if the rate adjustment for the Firm Transmission Service is to the demand component of such rate, the rate adjustment sought by the Company shall be stated as a dollar per kilowatt-hour charge or reduction based on a load factor for transmission service equal to the load factor for transmission service over the previous twelve (12) months; and further provided that the Company will provide the BGS-CIEP Suppliers with a copy of the Company's petition seeking such approvals from the Board;
- (ii) upon receipt of Board approval for the adjustment in the charges to BGS-CIEP Customers, the Company shall begin collecting the new charges (calculated with respect to the increase or decrease) from the BGS-CIEP Customers;
- (iii) in the event of a decrease in Firm Transmission Service Charges, the Auction Price shall be deemed to be decreased by the decrease to BGS-CIEP Customers approved by the BPU;
 - (iv) in the event of an increase in Firm Transmission Service Charges

the following provisions shall apply: (a) commencing with the date that the Company begins collecting the rate increase approved by the BPU from the BGS-CIEP Customers, the Company shall track that portion of the charges to BGS-CIEP Customers comprised by the increase and shall retain such tracked amounts for the benefit of the BGS-CIEP Suppliers; (b) upon approval by the FERC, in a Final FERC Order and not subject to refund, of a proposed increase in charges for Firm Transmission Service, or at such time as an annual adjustment to the charges for Firm Transmission Service becomes effective pursuant to the applicable FERC-authorized formula rate, the following shall occur: (A) the Auction Price shall be deemed to be increased by the increase to BGS-CIEP Customers approved by the BPU; and (B) the Company shall promptly pay each BGS-CIEP Supplier, in proportion to its BGS-CIEP Supplier Responsibility Share, the amounts tracked and retained for the benefit of BGS-CIEP Suppliers during the term of this Agreement; provided, however, that in the event only a proportion of the increased charge for Firm Transmission Service is approved by FERC, in a Final FERC Order and not subject to refund, the Auction Price shall be increased only by that portion of the increase approved by the FERC and the BGS-CIEP Suppliers shall be paid only that portion of the tracked and retained amounts associated with the portion of the increase approved by the FERC; and further provided that in the event FERC approves less than the entirety of the increased charge or rejects the increase in its entirety, all amounts tracked and retained by the Company for the benefit of the BGS-CIEP Suppliers associated with the rejected increase for Firm Transmission Service (or the rejected portion of the increase for Firm Transmission Service), shall be credited by the Company against future rates paid by BGS-CIEP Customers and the BGS-CIEP Suppliers shall have no right in or to such amounts. The Company will pay simple interest on amounts tracked and retained hereunder, calculated at the lower of the Interest Index or six (6) percent per annum; and

- (v) Nothing herein shall prevent or restrict the participation of the Company or the BGS-CIEP Suppliers in any FERC or judicial proceedings associated with any proposed increase or decrease in charges for Firm Transmission Service or any FERC or judicial proceedings associated with any formula rate adjustment, in sponsoring, supporting, opposing, challenging or otherwise addressing such increases, decreases, or adjustments.
- (b) If, prior to the Effective Date of this Agreement, a filing is made with the FERC to increase or decrease the charges for Firm Transmission Services, including any charge or surcharge imposed on customers receiving Firm Transmission Services, and the Company seeks the approval of the BPU to increase or decrease the rates charged to BGS-CIEP Customers by the amount of such increase or decrease for Firm Transmission Services, then the following procedures shall apply:
- (i) upon receipt of Board approval for the increase or decrease in the rates charged to BGS-CIEP Customers, the Company shall begin collecting the new rate (calculated with respect to the increase or decrease) from the BGS-CIEP Customers;
- (ii) in the event of a rate decrease, the Auction Price shall be deemed to be decreased by the rate decrease to BGS-CIEP Customers approved by the BPU;
- (iii) in the event of a rate increase the following provisions shall apply:

 (a) commencing with the later of (x) the date that the Company begins collecting the rate increase approved by the BPU from the BGS-CIEP Customers or (y) June 1, 2019 if the

Company begins collecting the rate increase approved by the BPU from the BGS-CIEP Customers prior to June 1, 2019, the Company shall track that portion of the rates charged to BGS-CIEP Customers comprised by the rate increase and shall retain such tracked amounts for the benefit of the BGS-CIEP Suppliers; (b) upon approval by the FERC, in a Final FERC Order and not subject to refund, of the proposed increase in charges for Firm Transmission Service, the following shall occur: (A) the Auction Price shall be deemed to be increased by the increase in charges to BGS-CIEP Customers approved by the BPU; and (B) the Company shall promptly pay each BGS-CIEP Supplier, in proportion to its BGS-CIEP Supplier Responsibility Share, the amounts tracked and retained for the benefit of BGS-CIEP Suppliers during the term of this Agreement; provided, however, that in the event only a proportion of the proposed increase in charges for Firm Transmission Service is approved by FERC, in a Final FERC Order and not subject to refund, the Auction Price shall be increased only by that portion of the proposed increase approved by the FERC and the BGS-CIEP Suppliers shall be paid only that portion of the tracked and retained amounts associated with the increase approved by the FERC; and further provided that in the event FERC approves less than the entirety of the proposed increase or rejects the proposed increase in its entirety, all amounts tracked and retained by the Company for the benefit of the BGS-CIEP Suppliers associated with the rejected increase in charges for Firm Transmission Service (or the rejected portion of the proposed increase in charges for Firm Transmission Service), shall be credited by the Company against future rates paid by BGS-CIEP Customers and the BGS-CIEP Suppliers shall have no right in or to such amounts. The Company will pay simple interest on amounts tracked and retained hereunder, calculated at the lower of the Interest Index or six (6) percent per annum; and

- (iv) Nothing herein shall prevent or restrict the participation of the Company or the BGS-CIEP Suppliers in any FERC or judicial proceedings associated with any proposed increase or decrease in charges for Firm Transmission Service, in sponsoring, supporting, opposing, challenging or otherwise addressing such proposed rate increases or decreases.
- (c) In the event that, at the beginning of the term of this Agreement, PJM Transmission Charges include charges approved by FERC subject to refund without a Final FERC Order and such charges are being collected from BGS-CIEP Customers pursuant to BPU approval, these charges will be treated as rate increases pursuant to subsection 15.9(a)(iv) of this Agreement.
- (d) Nothing herein to the contrary withstanding, in the event that the Company, in the exercise of its good faith judgment, cannot determine whether, or the extent to which, a filing at the FERC seeking a change in charges (including a filing seeking a change in rate design) for Firm Transmission Service, should be considered an increase or decrease in rates for the purposes of the preceding paragraphs, the Company may seek a ruling requesting such a determination from the BPU within ten (10) Business Days of the filing at the FERC. The Company shall thereafter implement the rate increase or rate decrease for Firm Transmission Service, if any, through an appropriate adjustment in the Auction Price in accordance with the directives of the BPU in its order.

15.10 Use of the Word "Including"

The word "including", when following any general statement or term, is not to be construed as limiting the general statement or term to the specific items or matters set forth or to similar items or matters, but rather as permitting the general statement or term to refer to all other items or matters that could reasonably fall within its broadest possible scope.

15.11 Federal Acquisition Regulation

If any of the following clauses prescribed by the Federal Acquisition Regulation ("FAR"), 48 Code of Federal Regulations Chapter 1, should be deemed to apply to this Agreement, the BGS-CIEP Supplier shall comply with the requirements of such clause(s), and shall include the terms or substance of such clause(s) in its subcontracts, as and to the extent required by the FAR:

- 1) Clean Air and Water: §52.223-2;
- Contract Work Hours and Safety Standards Act-Overtime Compensation: §52.222-4;
- 3) Equal Opportunity: §52.222-26;
- 4) Affirmative Action for and Employment Reports on Special Disabled and Vietnam Era Veterans: §52.222-35 and §52.222-37;
- 5) Affirmative Action for Handicapped Workers: §52.222-36;
- 6) Utilization of Small Business Concerns and Small Disadvantaged
 Business Concerns and Small Business and Small Disadvantaged Business
 Subcontracting Plan: §52.219-8 and §52-219-9.

In case of a conflict between the provisions of the FAR and the balance of this Agreement, the requirements of the FAR shall prevail.

15.12 Binding Terms

This Agreement and the rates, terms and conditions herein shall remain in effect for the entire term hereof and each Party (including the Company acting on behalf of Customers) agrees not to seek any change to such rates, terms and conditions pursuant to the FPA, if the FPA is deemed to have jurisdiction over this Agreement, including on the grounds that they are not just and reasonable.

15.13 Amendment

This Agreement, including the appendices hereto, cannot be amended without the written agreement of all Parties and the approval of the Board prior to such amendment becoming effective.

15.14 Counterparts

This Agreement may be executed in counterparts, each of which will be considered an original, but all of which shall constitute one instrument.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be
executed by their duly authorized representatives as of the date first set forth above.
ATTEST:
By:
[SUPPLIER SIGNATURES APPEAR ON SUCCEEDING PAGES]

APPENDIX A TO BGS-CIEP SUPPLIER MASTER AGREEMENT, DATED _____, 2019, BY AND BETWEEN ___ AND THE BGS-CIEP SUPPLIERS

BGS Supplier	Auction Price (\$/MW-Day)	BGS Supplier Responsibility Share
Address for Notice		
The address for any notice toBGS-CIEP Supplier Master Agree	provided pu eement shall be the follow	rsuant to Section 15.1 of the ving:
For Credit Related Issues:		
Name Address Telephone Fax E-Mail		
For Notices, Section 15.1:		
Name Address Telephone Fax E-Mail		,

APPENDIX B TO BGS-CIEP SUPPLIER MASTER AGREEMENT,

DATED _____, 2019,
BY AND BETWEEN ___
AND THE BGS-CIEP SUPPLIERS

Sample BGS-CIEP Letter of Credit

(Date))	
Letter of Credit No.		
To: [One of the following: Atlantic City Electric Company or Jers Service Electric and Gas Company or Ro ("Beneficiary") Attention:	ockland Electric Company]	ny or Public
1. We hereby establish in your favor the	his irrevocable transferable Letter o	f Credit (this
"Letter of Credit") for the account o	f(the '	'Applicant"),
in the amount of \$, effective immediately and availa	ble to you at
sight upon demand at our counters a	t (Location)	. This Letter
of Credit shall expire 364 days from	m date of issuance, unless terminat	ed earlier or
otherwise extended in accordance w	ith the provisions hereof. It is a con	dition of this
Letter of Credit that it shall be deen	ned automatically extended, without	amendment,
for additional period(s) of 364 day	ys from the expiry date hereof, o	r any future
expiration date unless at least ninety	(90) days before its current expira	tion date, we
send notice to you in writing by reg	gistered mail or overnight courier a	t the address
above, ATTN .: [Title] and to the	Applicant, that we do not intend to	extend this
Letter of Credit.		

- 2. This Letter of Credit is issued at the request of the Applicant, and we hereby irrevocably authorize you to draw on us, in accordance with the terms and conditions hereof, up to the maximum amount of this Letter of Credit, subject to reduction as provided in Paragraph 11 hereof. This Letter of Credit may be drawn upon an Event of Default under the BGS-RSCP Supplier Master Agreement(s) between the Applicant and you, dated ______ and the BGS-CIEP Supplier Master Agreement(s) between the Applicant and you, dated ______.
- 3. A partial or full drawing hereunder may be made by you on any Business Day on or prior to the expiration of this Letter of Credit by delivering, by no later than 11:00 A.M. (New York, NY time¹) on such Business Day to _ (Bank), _____ (address), (i) a notice executed by you in the form of Annex 1 hereto, appropriately completed and duly signed by your Authorized Officer and (ii) your draft in the form of Annex 2 hereto, appropriately completed and duly signed by your Authorized Officer. Authorized Officer shall mean President, Treasurer, any Vice President or any Assistant Treasurer.
- 4. We hereby agree to honor a drawing hereunder made in compliance with the terms and provisions of this Letter of Credit by transferring in immediately available funds the amount specified in the draft delivered to us in connection with such drawing to such account at such bank in the United States as you may specify in your draft delivered to us pursuant to Paragraph 3 hereof, by 3:00 P.M. (New York, NY time) on the date of such drawing, if delivery of this requisite document is made prior to

¹ If the issuer of the Letter of Credit is located in an area that is not in the Eastern time zone, this time and all other times in this Letter of Credit, and the definition of a Business Day should be adjusted accordingly.

- 11:00 AM (New York, NY time) on a Business Day pursuant to Paragraph 3 herein above, but at the opening of business on the first Business Day next succeeding the date of such drawing if delivery of the requisite document is made on or after 11:00 AM (New York, NY time) on any Business Day pursuant to Paragraph 3 herein above.
- 5. If a demand for payment made by you hereunder does not, in any instance, conform to the terms and conditions of this Letter of Credit, we shall give you prompt notice (not exceeding three (3) Business Days following the date of receipt of the documents) that the demand for payment was not effected in accordance with the terms and conditions of this Letter of Credit, stating the reasons that the demand for payment was not effected in accordance with such terms and conditions, and that we will upon your instructions hold any documents at your disposal or return the same to you. Upon being notified that the demand for payment was not effected in conformity with this Letter of Credit, you may attempt to correct any such non-conforming demand for payment to the extent that you are entitled to do so, provided, however, in such event a conforming demand for payment must be timely made in accordance with the terms of this Letter of Credit.
- 6. This Letter of Credit shall automatically terminate and be delivered to us for cancellation on the earliest of (i) the making by you of the drawings in an amount equal to the maximum amount available to be made hereunder, (ii) the date we receive from you a Certificate of Expiration in the form of Annex 3 hereto, or (iii) the above-stated expiration date hereof.

7. As used herein:

- "Business Day" shall mean any day on which commercial banks are not authorized or required to close in New York, New York and any day on which payments can be effected on the Fedwire system.
- 8. This Letter of Credit is transferable one or more times, but in each instance only to a single transferee for the full amount available to be drawn under the Letter of Credit at the time of such transfer. Such transfer shall be effected by us upon your submission of this Letter of Credit including all amendments, if any, accompanied by the Letter of Full Transfer substantially in the form of Annex 4. This Letter of Credit may not be transferred to any person with which the United States are prohibited from doing business under the United States Foreign Assets Control Regulations or other applicable United States laws and regulations. Except for such transfer, this Letter of Credit may not otherwise be amended or modified without consent from us, you and the Applicant, and except as otherwise expressly stated herein, is subject to the Uniform Customs and Practice for Documentary Credits 2007 Revision, ICC Publication No. 600, or any successor publication thereto (the "UCP"). Any and all transfer fees, expenses and costs shall be borne by the Applicant. This Letter of Credit shall, as to matters not governed by the UCP, be governed and construed in accordance with New York law, without regard to principles of conflicts of law.
- 9. This Letter of Credit sets forth in full our undertaking, and such undertaking shall not in any way be modified, amended, changed, amplified or limited by reference to any document, instrument or agreement referred to herein, except for Annexes 1 through 4 hereto and the notices referred to herein; and any such reference shall not be

deemed to incorporate herein by reference any document, instrument or agreement
except as set forth above.
We ("Bank") certify that as of
(date) our senior unsecured debt is rated "A" or better by S&P Global Ratings
("S&P") if rated by S&P or "A2" or better by Moody's Investors Service Inc.
("Moody's") if rated by Moody's. We hereby certify that our senior unsecured debt
is rated by at least one of S&P and Moody's.
The amount which may be drawn by you under this Letter of Credit shall be
automatically reduced by the amount of any drawings paid through us referencing this
Letter of Credit No Partial drawings are permitted hereunder.
Faxed document(s) are acceptable. Presentation by fax must be made to fax number
confirmed by telephone to
In the event of act of God, riot, civil commotion, insurrection, war, terrorism or by
any strikes or lock outs, or any cause beyond our control, that interrupts our business,
and causes the place for presentation of this Letter of Credit to be closed for business
on the last day of presentation, the expiration date of this Letter of Credit shall be
automatically extended without amendment to a date thirty (30) calendar days after
the place for presentation reopens for business.
This original Letter of Credit has been sent to [name, title, address]
, the Beneficiary above (as per Applicant's instructions). The
aggregate amount paid to the Beneficiary during the validity of this Letter of Credit
will not exceed the amount of this Letter of Credit. Any demands or communications
in the form of the attached Annexes or other communications directed to us under this

Letter of Credit must be signed by an Authorized Officer of the Beneficiary.

Acceptance or rejection of any amendments to this Letter of Credit must be signed by an Authorized Officer of the Beneficiary.

Very truly yours, (Bank)	
By:	
Name:	
Title:	
By:	
Name:	
Title:	

DRAWING UNDER LETTER OF CREDIT NO. _____ , 20 (Bank) To: (Address) Attention: Standby Letter of Credit Unit Ladies and Gentlemen: The undersigned is making a drawing under the above-referenced Letter of Credit in the amount specified below and hereby certifies to you as follows: Capitalized terms used herein that are defined herein shall have the meanings ascribed thereto in the Letter of Credit. Pursuant to Paragraph 2 of the Letter of Credit No. 2. _____, 20___, the undersigned is entitled to make a drawing under the Letter dated of Credit in the amount of \$_____, inasmuch as there is an Event of Default under any BGS Supplier Master Agreement between the Applicant and us. 3. The undersigned acknowledges that, upon your honoring the drawing herein requested, the amount of the Letter of Credit available for drawing shall be automatically decreased by an amount equal to this drawing. Very truly yours, (Beneficiary) By Name: Title: Date:

(Applicant Name)

Annex 1 to Letter of Credit

Annex 2 to Letter of Credit DRAWING UNDER LETTER OF CREDIT NO. _____ , 20____ ON [Business Day immediately succeeding date of presentation] PAY TO: (Beneficiary) Attn: For credit to the account of ______. FOR VALUE RECEIVED AND CHARGE TO ACCOUNT OF LETTER OF CREDIT NO. _____OF (Bank) (Address) (Beneficiary) By _____ Name: Title:

Anne	x 3 to Letter of Credit
	TIFICATE OF EXPIRATION ETTER OF CREDIT NO

To:	(Bank) (Address)
	Attention: Standby Letter of Credit Unit
Ladie	s and Gentlemen:
	he undersigned hereby certifies to you that the above referenced Letter of Credit be cancelled without payment. Attached hereto is said Letter of Credit, marked lled.
	(Beneficiary)
	By:
	Name:
	Title:
	Date:
cc:	(Applicant Name)

Annex 4 to Letter of Credit LETTER OF FULL TRANSFER _____, 20 Bank Bank Address To Whom It May Concern: Re: Credit____ Issued by Advice No For value received, the undersigned Beneficiary hereby irrevocably transfers to: (Name of Transferee) (Address) all rights of the undersigned Beneficiary to draw under the above Letter of Credit in its entirety. By this transfer, all rights of the undersigned Beneficiary in such Letter of Credit are transferred to the transferee and the transferee shall have the sole rights as beneficiary thereof, including sole rights relating to any amendments whether increases or extensions or other amendments and whether now existing or hereafter made. All amendments are to be advised direct to the transferee without necessity of any consent of or notice to the undersigned Beneficiary. The original of such Letter of Credit is returned herewith, and we ask you to endorse the transfer on the reverse thereof, and forward it direct to the transferee with your customary notice of transfer. Enclosed is remittance in the amount of \$_____ in payment of your transfer commission and in addition we agree to pay to you on demand any expenses that may be incurred by you in connection with this transfer. Transfer Commission Charges

SIGNATURE AUTHENTICATED The signatory/ies of this concern is/are authorized to withdraw corporate funds.	Yours very truly,
(BANK)	Signature of Beneficiary
(Authorized Signature)	
SIGNATURE AUTHENTICATED The signatory/ies of this concern is/are authorized to withdraw corporate funds.	
(BANK)	Signature of Transferee
(Authorized Signature)	

APPENDIX C TO BGS-CIEP SUPPLIER MASTER AGREEMENT,
DATED _____, 2019,
BY AND BETWEEN ____ AND THE BGS-CIEP SUPPLIERS

Guaranty

GU	ARANTY (this "Guaranty"), dated as of, made by
	(the "Guarantor"), a corporation organized and existing under the
law	of in favor of (the "Guaranteed Party"), a corporation
org	nized and existing under the laws of the State of New Jersey.
	as not defined herein take on the meaning given to them in the BGS-RSCP Supplier
Ma	er Agreement(s) dated and/or the BGS-CIEP Supplier Master
Ag	ement(s) dated (the "Agreement(s)"). Guarantor enters into this
	anty in consideration of, and as an inducement for Guaranteed Party having entered
inte	or entering into the "Agreement(s)" with [Name],
	[State] corporation (the "BGS Supplier"), which may involve
	extension of credit by the Guaranteed Party. Guarantor, subject to the terms and
COI	itions hereof, hereby unconditionally and absolutely guarantees to the Guaranteed
	the full and prompt payment when due, subject to an applicable grace period and
	demand in writing from the Guaranteed Party to the Guarantor's attention at the
	ess for Guarantor set forth in Section 11 hereof of any and all amounts payable by
tne	GS Supplier to the Guaranteed Party arising out of the Agreement(s), and,
1.	The Guarantor, as primary obligor and not merely as surety, hereby irrevocably and inconditionally guarantees the full and prompt payment when due (whether by occeleration or otherwise) of the principal and interest on any sums due and payable by the BGS Supplier as a result of an Event of Default under the Agreement(s) including, without limitation, indemnities, damages, fees and interest thereon, ursuant to the terms of the Agreement(s)). Notwithstanding anything to the contrary erein, the maximum aggregate liability of the Guarantor under this Guaranty shall option 1 [in no event exceed] Option 2 [in no event exceed the esser of [the credit limit amount] or the sum of the Total Exposures Amounts under the Agreement(s).] All such principal, interest, obligations and liabilities, ollectively, are the "Guaranteed Obligations". This Guaranty is a guarantee of ayment and not of collection.
2.	The Guarantor hereby waives diligence, acceleration, notice of acceptance of this Guaranty and notice of any liability to which it may apply, and waives presentment

and all demands whatsoever except as noted herein, notice of protest, notice of

- dishonor or nonpayment of any such liability, suit or taking of other action by any Guaranteed Party against, and any other notice to, any party liable thereon (including the Guarantor or any other guarantor), filing of claims with a court in the event of the insolvency or bankruptcy of the BGS Supplier, and any right to require a proceeding first against the BGS Supplier.
- 3. The Guaranteed Party may, at any time and from time to time, without notice to or consent of the Guarantor, without incurring responsibility to the Guarantor and without impairing or releasing the obligations of the Guarantor hereunder, upon or without any terms or conditions: (i) take or refrain from taking any and all actions with respect to the Guaranteed Obligations, any Document or any person (including the BGS Supplier) that the Guaranteed Party determines in its sole discretion to be necessary or appropriate; (ii) take or refrain from taking any action of any kind in respect of any security for any Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party; or (iii) compromise or subordinate any Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party including any security for such Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party.
- 4. Subject to the terms and conditions hereof, the obligations of the Guarantor under this Guaranty are absolute and unconditional and, without limiting the generality of the foregoing, shall not be released, discharged or otherwise affected by: extension, renewal, settlement, compromise, waiver, consent, discharge or release by the BGS Supplier concerning any provision of the Agreement(s) in respect of any Guaranteed Obligations of the BGS Supplier; (ii) the rendering of any judgment against the BGS Supplier or any action to enforce the same; (iii) the existence, or extent of, any release, exchange, surrender, non-perfection or invalidity of any direct or indirect security for any of the Guaranteed Obligations; (iv) any modification, amendment, waiver, extension of or supplement to any of the Agreement(s) or the Guaranteed Obligations agreed to from time to time by the BGS Supplier and the Guaranteed Party; (v) any change in the corporate existence (including its constitution, laws, rules, regulations or powers), structure or ownership of the BGS Supplier or the Guarantor, or any insolvency, bankruptcy, reorganization or other similar proceedings affecting the BGS Supplier or its assets, the Guarantor or any other guarantor of any of the Guaranteed Obligations; (vi) the existence of any claim, set-off or other rights which the Guarantor may have at any time against the BGS Supplier, the Guaranteed Party or any other corporation or person, whether in connection herewith or in connection with any unrelated transaction; provided that nothing herein shall prevent the assertion of any such claim by separate suit or compulsory counterclaim; (vii) the invalidity, irregularity or unenforceability in whole or in part of the Agreement(s) or any Guaranteed Obligations or any instrument evidencing any Guaranteed Obligations or the absence of any action to enforce the same, or any provision of applicable law or regulation purporting to prohibit payment by the BGS Supplier of amounts to be paid by it under the Agreement(s) or any of the Guaranteed Obligations; and (viii) except for a failure to comply with any applicable statute of limitations, any other act or omission to act or

- delay of any kind of the BGS Supplier, any other guarantor, the Guaranteed Party or any other corporation or person or any other event, occurrence or circumstance whatsoever which might, but for the provisions of this paragraph, constitute a legal or equitable discharge of the Guarantor's obligations hereunder.
- 5. The Guarantor hereby irrevocably waives (a) any right of reimbursement or contribution, and (b) any right of salvage against the BGS Supplier of any collateral security or guaranty or right of offset held by the Guaranteed Party.
- 6. The Guarantor will not exercise any rights, which it may acquire by way of subrogation until all Guaranteed Obligations to the Guaranteed Party pursuant to the Agreement(s) have been paid in full.
- 7. Subject to the terms and conditions hereof, this Guaranty is a continuing one and all liabilities to which it applies or may apply under the terms hereof shall be conclusively presumed to have been created in reliance hereon. Except for a failure to comply with any applicable statute of limitations, no failure or delay on the part of the Guaranteed Party in exercising any right, power or privilege hereunder, and no course of dealing between the Guarantor and the Guaranteed Party, shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights, powers and remedies herein expressly provided are cumulative and not exclusive of any rights, powers or remedies, which the Guaranteed Party would otherwise have. No notice to or demand on the Guarantor in any case shall entitle the Guarantor to any other or further notice of demand in similar or other circumstances or constitute a waiver of the rights of the Guaranteed Party to any other or further action in any circumstances without notice or demand.
- 8. This Guaranty shall be binding upon the Guarantor and upon its successors and assigns and shall inure to the benefit of and be enforceable by the Guaranteed Party and its successors and assigns; provided, however, that the Guarantor may not assign or transfer any of its rights or obligations hereunder without the prior written consent of the Guaranteed Party. The assignment rights of the Guaranteed Party will be in accordance with the terms of the underlying Agreement(s).
- Neither this Guaranty nor any provision hereof may be changed, waived, discharged
 or terminated except upon written agreement of the Guaranteed Party and the
 Guarantor.
- 10. The Guarantor agrees that its liability as guarantor shall continue and remain in full force and effect in the event that all or any part of any payment made hereunder or any obligation or liability guaranteed hereunder is recovered (as a fraudulent conveyance, preference or otherwise) rescinded or must otherwise be reinstated or returned due to bankruptcy or insolvency laws or otherwise.

11. All notices and other communications hereunder shall be made at the addresses by hand delivery, by the next day delivery service effective upon receipt or by certified mail return receipt requested (effective upon scheduled weekday delivery day) or telefacsimile (effective upon receipt of evidence, including telefacsimile evidence, that telefacsimile was received)

If to the Guarantor: [To be completed]

If to the Guaranteed Party: [To be completed]

- 12. If claim is ever made upon the Guaranteed Party for repayment or recovery of any amount or amounts received in payment or on account of any of the Guaranteed Obligations and the Guaranteed Party repays all or part of such amount by reason of (a) any judgment, decree or order of any court or administrative body having jurisdiction over such payee or any of its property, or (b) any settlement or compromise of any such claim effected by such payee with any such claimant (including the Guarantor), then and in such event the Guarantor agrees that any such judgment, decree, order, settlement or compromise shall be binding upon it, notwithstanding any revocation hereof or the cancellation of the Agreement(s) or other instrument evidencing any liability of the Guarantor, and the Guarantor shall be and remain liable to the Guaranteed Party hereunder for the amount so repaid or recovered to the same extent as if such amount had never originally been received by any such payee.
- 13. The Guarantor hereby certifies that it satisfies the Minimum Rating as defined in the Agreement(s).
- 14. This Guaranty shall remain in full force and effect until all Guaranteed Obligations have been fully and finally performed, at which point it will expire. The Guarantor may terminate this Guaranty upon thirty (30) days prior written notice to the Guaranteed Party which termination shall be effective only upon receipt by the Guaranteed Party of alternative means of security or credit support, as specified in the Agreement(s) and in a form reasonably acceptable to the Guaranteed Party. Upon the effectiveness of any such expiration or termination, the Guarantor shall have no further liability under this Guaranty, except with respect to the Guaranteed Obligations entered into prior to the time the expiration or termination is effective, which Guaranteed Obligations shall remain guaranteed pursuant to the terms of this Guaranty until finally and fully performed.
- 15. The Guarantor represents and warrants that: (i) it is duly organized and validly existing under the laws of the jurisdiction in which it was organized and has the power and authority to execute, deliver, and perform this Guaranty; (ii) no

authorization, approval, consent or order of, or registration or filing with, any court or other governmental body having jurisdiction over the Guarantor is required on the part of the Guarantor for the execution, delivery and performance of this Guaranty except for those already made or obtained; (iii) this Guaranty constitutes a valid and legally binding agreement of the Guarantor, and is enforceable against the Guarantor, except as the enforceability of this Guaranty may be limited by the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditor's rights generally and by general principles of equity; and (iv) the execution, delivery and performance of this Guaranty by the Guarantor have been and remain duly authorized by all necessary corporate or comparable action and do not contravene any provision of its _______ [insert appropriate corporate organizational document, such as Declaration of Trust, Limited Liability Agreement, Articles of Incorporation or by-laws] or any law, regulation or contractual restriction binding on it or its assets.

- 16. This Guaranty and the rights and obligations of the BGS Supplier and the Guarantor hereunder shall be construed in accordance with and governed by the laws of the State of New Jersey. The Guarantor and Guaranteed Party jointly and severally agree to the exclusive jurisdiction of State and federal courts located in the State of New Jersey over any disputes arising or relating to this Guaranty and waive any objections to venue or inconvenient forum. The Guarantor and Guaranteed Party each hereby irrevocably waive any and all rights to trial by jury with respect to any legal proceeding arising out of or relating to this Guaranty.
- 17. This writing is the complete and exclusive statement of the terms of this Guaranty and supersedes all prior oral or written representations, understandings, and agreements between the Guaranteed Party and the Guarantor with respect to subject matter hereof. The Guaranteed Party and the Guarantor agree that there are no conditions to the full effectiveness of this Guaranty.
- 18. Every provision of this Guaranty is intended to be severable. If any term or provision hereof is declared to be illegal or invalid for any reason whatsoever by a court of competent jurisdiction, such illegality or invalidity shall not affect the balance of the terms and provisions hereof, which terms and provisions shall remain binding and enforceable. This Guaranty may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one instrument.
- 19. No Trustee or shareholder of Guarantor shall be held to any liability whatsoever for any obligation under this Guaranty, and such Guaranty shall not be enforceable against any such Trustee in their or his or her individual capacities or capacity. This Guaranty shall be enforceable against the Trustees of Guarantor only as such, and every person, firm, association, trust or corporation having any claim or demand arising under this Guaranty and relating to Guarantor, its shareholders or Trustee shall look solely to the trust estate of Guarantor for the payment or satisfaction thereof.

IN WITNESS WHEREOF, the Guarantor has caused this Guaranty to be executed and delivered as of the date first above written to be effective as of the earliest effective date of any of the Agreement(s).
[GUARANTOR]
By:
Title:
Accepted and Agreed to:
(EDC)
Ву:
Title:

XI. APPENDIX D

BGS-RSCP Supplier Master Agreement

BGS-RSCP SUPPLIER MASTER AGREEMENT

TABLE OF CONTENTS

ARTICLI	E 1: DEFINITIONS	. 3				
ARTICLI						
2.1	Capacity In Which Company Is Entering Into This Agreement					
2,2	Parties' Obligations					
2.3	PJM Services					
2.4	Communications and Data Exchange	16				
2.5	Record Retention					
2.6	Verification	17				
ARTICLI	E 3: REPRESENTATIONS AND WARRANTIES	17				
3.1	BGS-RSCP Supplier's Representations and Warranties					
3.2	Company's Representations and Warranties					
3.3	Survival of Obligations	21				
ARTICLI	E 4: COMMENCEMENT AND TERMINATION OF AGREEMENT	21				
4.1	Commencement and Termination.					
4.2	Termination of Right to Supply BGS-RSCP	21				
4.3	Survival of Obligations					
4.4	Mutual Termination	22				
ARTICLE	E 5: BREACH AND DEFAULT	23				
5.1	Events of Default	23				
5.2	Rights Upon Default	25				
5.3	Damages Resulting From an Event of Default	26				
5.4	Declaration of an Early Termination Date and Calculation of Settlement					
	Amount and Termination Payment	29				
5.5	Step-up Provision					
5.6	Setoff of Payment Obligations of The Non-Defaulting Party					
5.7	Preservation of Rights of Non-Defaulting Party					
ARTICLE						
6.1	Applicability					
6.2	Creditworthiness Determination					
6.3	Independent Credit Requirement	35				
6.4	Independent Credit Threshold					
6.5	Mark-to-Market Credit Exposure Methodology	40				
6.6	Credit Limit	41				
6.7	Posting Margin and Return of Surplus Margin	44				
6.8	Grant of Security Interest/Remedies					
6.9	Security Instruments					
6.10	Maintenance of Creditworthiness	48				
6.11	Calling on Security					
6.12	Interest on Cash Held by Company	50				
6.13	Confidentiality					
6.14	No Endorsement of BGS-RSCP Supplier					
6.15	Multiple BGS Supply Agreements					
ARTICLE		-				
RESOUR	CE SUBMISSION AND TRANSMISSION PROCUREMENT	51				
7.1	Load Obligations					

8.2	Energy Settlement by the Company	. 54					
ARTICL							
9.1	The Company Payment of Obligations to the BGS-RSCP Supplier						
9.2	Billing for BGS-RSCP Supplier's Obligations to Other Parties						
9.3	The BGS-RSCP Supplier Payment of Obligations to the Company						
ARTICL							
10.1	Disconnection and Curtailment By the Company						
10.2	Inadvertent Loss of Service to BGS-RSCP Customers						
10.3	Good Faith Efforts						
10.4	PJM Requirements						
10.5	Compliance With Governmental Directives						
ARTICL							
11.1	Informal Resolution of Disputes						
11.2 ARTICL	Recourse to Agencies or Courts of Competent Jurisdiction E 12: REGULATORY AUTHORIZATIONS AND JURISDICTION						
12.1	Compliance With Applicable Legal Authorities						
12.1	FERC Jurisdictional Matters						
	E 13: LIMITATION OF LIABILITY						
13.1	Limitations on Liability						
13.2	Risk of Loss						
ARTICLI							
14.1	Indemnification						
14.2	Survives Agreement	. 64					
ARTICLI	E 15: MISCELLANEOUS PROVISIONS	. 65					
15.1	Notices						
15.2	No Prejudice of Rights	. 65					
15.3	Assignment						
15.4	Governing Law and Venue						
15.5	Headings						
15.6	Third Party Beneficiaries						
15.7	General Miscellaneous Provisions						
15.8	Taxes						
15.9	Changes in Transmission Charges for Firm Transmission Service						
15.10	Use of the Word "Including"						
15.11	Federal Acquisition Regulation						
15.12 15.13	Binding Terms						
	•						
	15.14 Counterparts Schedule for ICRT						
	Sample BGS-RSCP Letter of Credit8						
Guaranty9							

BGS-RSCP SUPPLIER MASTER AGREEMENT

THIS BGS-RSCP SUPPLIER MASTER AGREEMENT, made and entered
into this, 2019, by and between
(the "Company"), a corporation and a public utility organized and existing under the laws
of the State of New Jersey, on its own behalf and as agent as more fully set forth below,
and each of the suppliers listed on Appendix A hereto, severally and not jointly (each a
"BGS-RSCP Supplier" and, collectively, the "BGS-RSCP Suppliers"), the Company and
each BGS-RSCP Supplier hereinafter sometimes referred to collectively as the "Parties",
or individually as a "Party",
WITNESSETH:
WHEREAS, the Company is a public utility engaged, inter alia, in the
transmission and distribution of electric Energy within its Service Territory located in the
State of New Jersey; and
WHEREAS, pursuant to Section 9(c) of the Electric Discount and Energy
Competition Act, N.J.S.A. 48:3-49 et. seq., the New Jersey Board of Public Utilities
("BPU") has been authorized to make available to any power supplier on a competitive
basis the opportunity to provide Basic Generation Service ("BGS"); and
WHEREAS, in its Decision and Order dated, in Docket No. ER18040356
(the " Order"), the BPU found that for periods after May 31, 2019, it would serve
the public interest for the Company to continue to secure BGS Supply through a bid
process; and
WHEREAS, in the Order, the BPU approved an auction design for
bidding out two Basic Generation Service products, one a product termed Basic
Generation Service - Residential Small Commercial Pricing ("BGS-RSCP"), and one a

product termed Basic Generation Service – Commercial and Industrial Energy Pricing ("BGS-CIEP"); and

WHEREAS, on ______, 2019 to ______, 2019, a successful auction for bidding out BGS-RSCP was held; and,

WHEREAS, each BGS-RSCP Supplier was one of the winning bidders in the auction for the provision of BGS-RSCP; and

WHEREAS, the _____ Order authorized the Company to contract with winning bidders, on behalf of the consumers of electricity located on the Company's distribution system, for the provision of the share of BGS-RSCP Load covered by the winning bid; and

WHEREAS, pursuant to the auction procedures approved in the _____ Order, the Company and each of the BGS-RSCP Suppliers desire to enter into this BPU-approved BGS-RSCP Supplier Master Agreement ("Agreement") setting forth their respective obligations concerning the provision of BGS-RSCP.

NOW, THEREFORE, in consideration of the mutual covenants and promises set forth below, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound, hereby covenant, promise and agree as follows:

ARTICLE 1: DEFINITIONS

Any capitalized or abbreviated term not elsewhere defined in this Agreement shall

have the definition set forth in this Article 1.

Ancillary Services – shall have the meaning ascribed thereto in the PJM Tariff.

Applicable Legal Authorities – generally, those federal and New Jersey statutes and administrative rules and regulations that govern the electric utility industry in New Jersey.

Auction Price – the price, set forth in Appendix A, resulting from the Company's competitive solicitation of the opportunity to supply BGS-RSCP subject to any adjustment made pursuant to Section 15.9 hereof. This price is the basis for financial settlement of BGS-RSCP Supply supplied by the BGS-RSCP Supplier to BGS-RSCP Customers under this Agreement.

Auction Website - www.bgs-auction.com

Bankruptcy Code – those laws of the United States of America related to bankruptcy, codified and enacted as Title 11 of the United States Code, entitled "Bankruptcy" and found at 11 U.S.C. § 101 et seq., as such laws may be amended, modified, replaced or superseded from time to time.

Basic Generation Service or "BGS" – electric generation service that is provided at retail pursuant to the Applicable Legal Authorities under the Company's retail electric tariffs and under any other agreements or arrangements between the Company and Customers, to any Customer that is not being served by a Third Party Supplier.

Basic Generation Service – Commercial and Industrial Energy Pricing or "BGS-CIEP" – electric generation service that is provided pursuant to the Applicable Legal Authorities at prices that include an energy charge component that varies on an hourly basis in accordance with changes in the load-weighted average Residual Metered Load aggregate Real-Time Locational Marginal Price (including energy, congestion and marginal losses) for the Company's Transmission Zone under the Company's BGS-CIEP Tariffs.

Basic Generation Service – Residential Small Commercial Pricing or "BGS-RSCP" – electric generation service that is provided at retail pursuant to the Applicable Legal Authorities under the Company's BGS-RSCP Tariffs.

Basic Generation Service Supply or "BGS Supply" – either BGS-RSCP Supply or BGS-CIEP Supply.

BGS-CIEP Customer – a Customer who is being served under any one of the Company's BGS-CIEP Tariffs and is not taking electric generation service from a Third Party Supplier.

BGS-CIEP Load – the full electricity requirement (including without limitation, Energy, Capacity, Ancillary Services and Firm Transmission Service) of BGS-CIEP Customers.

BGS-CIEP Supplier – an entity that has been selected through the BGS-CIEP solicitation and has accepted the obligations and associated rights to provide BGS-CIEP Supply to retail customers within the State of New Jersey in accordance with the Applicable Legal Authorities and has entered into a BGS-CIEP Supplier Master Agreement with the Company as a Party. Each BGS-CIEP Supplier must be an LSE and shall have the obligations of an LSE under the PJM Agreements.

BGS-CIEP Supplier Responsibility Share — for each BGS-CIEP Supplier, the fixed percentage share of the Company's BGS-CIEP Load for which the BGS-CIEP Supplier is responsible as set forth in Appendix A to the BGS-CIEP Supplier Master Agreement. The stated percentage share was determined by multiplying the number of tranches won by the BGS-CIEP Supplier in the auction times the BGS-CIEP Tranche size percentage share.

BGS-CIEP Supply – unbundled Energy, Capacity, Ancillary Services and Firm Transmission Service, including all losses and/or congestion costs associated with the provision of such services, and such other services or products that a BGS-CIEP Supplier may be required, by PJM or other governmental body having jurisdiction, to provide in order to meet the BGS-CIEP Supplier Responsibility Share under the BGS-CIEP Supplier Master Agreement.

BGS-CIEP	Tariffs -	Rate	Schedules	of	the
Company.					

BGS-RSCP Customer – a Customer who is being served under any one of the Company's BGS-RSCP Tariffs and is not taking electric generation service from a Third Party Supplier.

BGS-RSCP Load – the full electricity requirement (including, without limitation, Energy, Capacity, Ancillary Services and Firm Transmission Service) of BGS-RSCP Customers.

BGS-RSCP Peak Load Share – is the portion of load attributable to those Customers on BGS-RSCP of the PJM-determined EDC zonal peak load.

BGS-RSCP Supplier – an entity that has been selected through the BGS-RSCP solicitation and has accepted the obligations and associated rights to provide BGS-RSCP Supply to retail customers within the State of New Jersey in accordance with the Applicable Legal Authorities and has entered into this Agreement with the Company as a Party. Each BGS-RSCP Supplier under this Agreement must be an LSE and shall have the obligations of an LSE under the PJM Agreements. The term "supplier" also refers generically to any entity authorized by the BPU to provide BGS Supply, as opposed to a specific signatory to this Agreement, where the context makes it appropriate to do so. The distinction can be derived from the context, but is also generally reflected in the use of lower case type ("supplier") to reflect the generic usage, and an initial capital ("Supplier") to reflect a Party to this Agreement.

BGS-RSCP Supplier Responsibility Share – for each BGS-RSCP Supplier, the fixed percentage share of the Company's BGS-RSCP Load for which the BGS-RSCP Supplier is responsible as set forth in Appendix A. The stated percentage share was determined by multiplying the number of Tranches won by the BGS-RSCP Supplier in the auction times the Tranche size percentage share.

BGS-RSCP Supply – unbundled Energy, Capacity, Ancillary Services and Firm Transmission Service, including all losses and/or congestion costs associated with the provision of such services, and such other services or products that a BGS-RSCP Supplier may be required, by PJM or other governmental body having jurisdiction, to provide in order to meet the BGS-RSCP Supplier Responsibility Share under this Agreement.

BGS-RSCP Tariffs – Rate Schedules _____ of the Company.

Billing Month – each calendar month during the term of this Agreement.

BPU or "Board" - the New Jersey Board of Public Utilities or its successor.

Business Day – any day on which the Company's and PJM's corporate offices are open for business and commercial banks are not authorized or required to close in New York, New York.

Capacity – shall mean "Unforced Capacity" as set forth in the PJM RAA or in successor, superseding or amended versions of the PJM RAA that may take effect from time to time over the term of this Agreement, and any successor measurement of generating capacity as may be employed in PJM (whether set forth in the PJM RAA or elsewhere) for the purpose of stating the capacity obligation of an LSE.

Charge – any fee, charge or other amount that is billable by the Company to the BGS-RSCP Supplier under this Agreement.

Competitive Electricity Supply – unbundled Energy, Capacity, and Firm Transmission Service, including all losses and/or congestion associated with the provision of the foregoing services, and such other services or products that are provided by a Third Party Supplier to fulfill its obligations to serve customer load. The provision of Competitive Electricity Supply by Third Party Suppliers entails fulfillment of all obligations associated with service to Customers, including the obligations of a Load Serving Entity under the PJM Tariff, procedures, agreements and manuals.

Costs – means, with respect to the Non-Defaulting Party brokerage fees, commissions and other similar transaction costs and expenses reasonably incurred by such Party either in terminating any arrangement pursuant to which it has hedged its obligations or entering into new arrangements which replace this Agreement; and all reasonable attorneys' fees and expenses incurred by the Non-Defaulting Party in connection with the termination of this Agreement.

Customer – a Company customer as defined in the Company's retail tariffs, eligible to receive Competitive Electricity Supply from a Third Party Supplier or BGS-RSCP or BGS-CIEP from a BGS-RSCP Supplier or BGS-CIEP Supplier, respectively, in accordance with the Applicable Legal Authorities.

Damages – financial compensation from the Defaulting Party to the Non-Defaulting Party associated with the occurrence of an Event of Default or an Early Termination of this Agreement. This compensation shall be assessed pursuant to Article 5 of this Agreement.

Early Termination – termination of this Agreement prior to the end of the term due to the occurrence of an Event of Default as specified in Section 5.1 of this Agreement and the declaration of Early Termination.

Early Termination Date – the date upon which an Early Termination becomes effective as specified in Section 5.2 of this Agreement.

Effective Date – the date first written above.

Electric Distribution Company or "EDC" – the applicable of the four New Jersey Electric Distribution Companies.

Emergency – (i) an abnormal system condition requiring manual or automatic action to maintain system frequency, or to prevent loss of firm load, equipment damage, or tripping of system elements that could adversely affect the reliability of an electric system or the safety of persons or property; or (ii) a fuel shortage requiring departure from normal operating procedures; or (iii) a condition that requires implementation of Emergency Operations Procedures as defined in the PJM OATT or PJM manuals; or (iv) any other condition or situation that the Company or PJM deems imminently likely to endanger life or property or to affect or impair the Company's electrical system or the electrical system(s) of other(s) to which the Company's electrical system is directly or indirectly connected (a "Connected Entity"). Such a condition or situation may include, but shall not be limited to, potential overloading of the Company's transmission and/or distribution circuits, PJM minimum generation ("light load") conditions, or unusual operating conditions on either the Company's or a Connected Entity's electrical system, or conditions such that the Company is unable to accept Energy from the BGS-RSCP Supplier without jeopardizing the Company's electrical system or a Connected Entity's electrical system.

Energy – three-phase, 60-cycle alternating current electric energy, expressed in units of kilowatt-hours or megawatt-hours.

Energy Portfolio Standards – (i) those standards imposed by the BPU pursuant to N.J.S.A. 48:3-87(d) requiring that specified levels of electric power sold for the provision of Basic Generation Service shall be produced from renewable energy resources as set forth in the Renewable Energy Portfolio Standards, N.J.A.C. 14:8-2 et seq., as modified or superseded, or in such other regulations of the BPU as may be adopted or amended from time to time, and (ii) any standards imposed under any federal, state or local

legislation or by any federal, interstate, state or local court, tribunal or governmental agency or authority or regulatory body that have the effect of requiring that specified levels of electric power sold for the provision of Basic Generation Service shall be produced from, or be otherwise attributable to, renewable energy resources or shall be produced from, or be otherwise attributable to, resources emitting only specified amounts of carbon or other substances, as the same may be adopted or amended from time to time.

Event of Default – a breach of obligations under this Agreement as set forth in Section 5.1 of this Agreement.

FERC – the Federal Energy Regulatory Commission.

Final FERC Order – a final order issued by FERC in connection with a request for an increase or decrease in the rates for Firm Transmission Service, or, if FERC has issued an order authorizing Company's recovery of transmission costs through formula rates, the acceptance by operation of law or otherwise of the Company's rate tariff filing seeking an increase or decrease for Firm Transmission Service, which is no longer subject (either actually or potentially) to rehearing or judicial review in which the amount of the increase or decrease is in dispute or in question and is not the subject of proceedings at FERC on remand from any court in which the amount of the specific increase or decrease in the Firm Transmission Service rate is in dispute or in question.

Final Monthly Energy Allocation or "FMEA" – is a quantity in kWh which, for any Billing Month, is the PMEA adjusted for any billing or metering data received subsequent to the calculation of PMEA of which PJM is notified within 60 days.

Firm Transmission Service – shall mean "Network Integration Transmission Service" under the PJM OATT in effect as of the Effective Date of this Agreement, or in successor, superseding or amended versions of the PJM OATT that may take effect from time to time over the term of this Agreement. In the event the PJM OATT is modified such that "Network Integration Transmission Service" is no longer offered, Firm Transmission Service shall mean the type of transmission service offered under the PJM OATT that is accorded the highest level of priority for scheduling and curtailment purposes.

Forward Market Price – the price for On-peak Energy Forwards as determined by averaging concurrent quotations obtained by all of the EDCs from the same three independent brokers active in the electric markets as available.

Gains – means, with respect to any Party, an amount equal to the present value of the economic benefit to it, if any (exclusive of Costs), resulting from an Early Termination of this Agreement, determined in a commercially reasonable manner.

Guaranty – means a guaranty, hypothecation agreement, margins or security agreement or any other document (whether in the form attached to this BGS-RSCP Supplier Master Agreement or other form approved by the Company).

Guarantor – any party who has the authority and may agree to guarantee a BGS-RSCP Supplier's financial obligations under this Agreement, recognizing that such a party will be obligated to meet the Company's creditworthiness requirements for BGS-RSCP Suppliers.

Independent Credit Requirement or "ICR" – an amount required as security under Section 6.3 of this Agreement, to reflect the risk of Energy and Capacity price movements between the Early Termination Date caused by an Event of Default by a BGS-RSCP Supplier and the date the final calculation of Damages owing to the Company under Section 5.2 of this Agreement is made.

Interest Index – the average Federal Funds Effective Rate for the period of time the funds are on deposit. The Federal Funds Effective Rate is published daily on the Federal Reserve website (http://www.federalreserve.gov/releases/h15/update/).

Kilowatt or "kW" – unit of measurement of useful power equivalent to 1000 watts.

Kilowatt-hour or "kWh" – one kilowatt of electric power used over a period of one hour.

Load Serving Entity or "LSE" – an entity that has been granted the authority or has an obligation pursuant to state or local law, regulation or franchise to sell electricity to retail customers located within the PJM Control Area as that term is defined in the PJM RAA or in successor, superseding or amended versions of the PJM RAA that may take effect from time to time over the term of this Agreement.

Losses – means, with respect to any Party, an amount equal to the present value of the economic loss to it, if any (exclusive of Costs), resulting from an Early Termination of this Agreement, determined in a commercially reasonable manner.

Margin – the amount by which the Total Exposure Amount exceeds a BGS-RSCP Supplier's, or Guarantor's, credit limit as defined in Section 6.6.

Mark-to-Market Exposure Amount – an amount calculated daily for each BGS-RSCP Supplier reflecting the total MtM credit exposure to the Company due to fluctuations in market prices for Energy minus amounts due pursuant to this Agreement to such BGS-RSCP Supplier for the delivery of BGS-RSCP Supply. The total MtM credit exposure will be equal to 1.1 times the sum of the MtM credit exposures for each Billing Month as set forth in Section 6.5 of this Agreement. The methodology for calculation of the MtM credit exposure is illustrated in Appendix B.

Maximum Credit Limit – the lesser of the applicable % of TNW and the applicable credit limit cap as specified in Section 6.6 of this Agreement.

Megawatt or MW – one thousand kilowatts.

Megawatt-hour or MWh – one megawatt of electric power used over a period of one hour.

Merger Event – when a Party consolidates or amalgamates with, or merges into or with, or transfers all or substantially all of its assets to another entity and either (i) the resulting entity fails to assume all of the obligations of such Party hereunder or (ii) the benefits of any credit support provided pursuant to Article 6 of this Agreement fail to extend to the performance by such resulting, surviving or transferee entity of the Party's obligations hereunder, and the resulting entity or its guarantor fails to meet the creditworthiness requirements of this Agreement. Transfer of all or substantially all of the Company's generation assets does not qualify as a Merger Event.

Meter Reading – the process whereby the Company takes notice of the information presented on a Customer's meter. A Meter Reading may be obtained manually, through telemetry, or by estimation, in accordance with the Company's established procedures and practices.

Minimum Rating – a minimum senior unsecured debt rating as defined in Section 6.4(a)(i) of this Agreement.

NERC – the North American Electric Reliability Corporation or its successor.

On-peak Energy Forward – a standardized energy trading product representing the delivery of electric power in PJM, at the Western Hub, over a period from 7:00 a.m. up to the hour ending at 11:00 p.m. Monday through Friday, excluding NERC holidays. In the event that the PJM Western Hub price is no longer available or no longer representative of a transparent trading hub, the Parties will negotiate in good faith to agree upon an alternate liquid price.

PJM – the Pennsylvania-New Jersey-Maryland Interconnection L.L.C. or its successor.

PJM Agreements – shall have the meaning ascribed in Section 2.3 of this Agreement.

PJM Control Area – that certain Control Area encompassing systems in Pennsylvania, New Jersey, Maryland, Delaware, Virginia and the District of Columbia, as may be modified from time to time, and which is recognized by the North American Electric Reliability Council as the "PJM Control Area".

PJM OA – the PJM Operating Agreement or its successor.

PJM OATT – the prevailing PJM Open Access Transmission Tariff on file with the FERC, which sets forth the rates, terms and conditions of transmission service over transmission facilities located in the PJM Control Area, as is in effect on the Effective Date and as modified from time to time.

PJM OI – the PJM Office of Interconnection, the system operator for the PJM Control Area.

PJM RAA – the PJM Reliability Assurance Agreement or its successor.

Preliminary Monthly Energy Allocation or "PMEA" – is a quantity in kWh which, for any Billing Month, is the preliminary calculation of the BGS-RSCP Load associated with the BGS-RSCP Supplier's BGS-RSCP Supplier Responsibility Share.

PMEA/FMEA Adjustment Amount – for any Billing Month, the monetary amount due to the BGS-RSCP Supplier or the Company, as the case may be, in order to reconcile any difference between the PMEA used for the purpose of calculating estimated payments made to the BGS-RSCP Supplier for a given month and the FMEA used for calculating the final payments due to the BGS-RSCP Supplier for such month as more fully described in Article 9 hereof.

RFC - the Reliability First Corporation of NERC or its successor.

Seasonal Billing Factor – a numerical factor set forth in Appendix B hereto, one amount applicable during the summer months of June through September, and one amount applicable during the non-summer months of October through May, applied to the Auction Price in accordance with the provisions of Article 9 of this Agreement and thereby used to shape the Company's payments to BGS-RSCP Suppliers.

Service Territory – the geographic areas of the State of New Jersey in which the Company serves electric Customers.

Settlement Amount — with respect to a Non-Defaulting Party, the net amount of the Losses or Gains, and Costs, expressed in U.S. Dollars, which such party incurs as a result of Early Termination, as set forth in Section 5.4(a) of this Agreement. For the purposes of calculating the Termination Payment, the Settlement Amount shall be considered an amount due to the Non-Defaulting Party under this Agreement if the total of the Losses and Costs exceeds the Gains and shall be considered an amount due to the Defaulting Party under this Agreement if the Gains exceed the total of the Losses and Costs.

Statement – a monthly report prepared by the Company for the BGS-RSCP Supplier indicating the amount due to the BGS-RSCP Supplier by the Company in compensation for kWhs supplied to BGS-RSCP Customers by the BGS-RSCP Supplier during the current Billing Month, in accordance with the BGS-RSCP Supplier's obligations under this Agreement.

Supply Day – any calendar day during the term of this Agreement on which the BGS-RSCP Supplier is providing, or is obligated by this Agreement to provide, BGS-RSCP Supply to the Company's BGS-RSCP Customers.

Tangible Net Worth or "TNW" – total assets less intangible assets and total liabilities. Intangible assets include benefits such as goodwill, patents, copyrights and trademarks.

Termination Payment – shall have the meaning ascribed in Section 5.4 of this Agreement.

Third Party Supplier or "TPS" – a person or entity that is duly licensed by the Board to offer and to assume the contractual and legal responsibility to provide electric generation service to retail customers located in the state of New Jersey pursuant to retail open

access programs approved by the Board.

Total Exposure Amount – an amount calculated daily for each BGS-RSCP Supplier reflecting the total credit exposure to the Company and consisting of the sum of (i) the Mark-to-Market Exposure Amount arising under this Agreement; (ii) any amount(s) designated as the "Mark-to-Market Exposure Amount" arising under any other BGS Supply agreements providing for "BGS-RSCP Supply" or similar BGS service; and (iii) the amount designated as the "credit exposure" under any other BGS Supply agreements providing for "BGS-CIEP Supply" or similar BGS service; provided that in the event the amount calculated for any day is a negative number, it shall be deemed to be zero for such day.

Tranche – a fixed percentage share of the BGS-RSCP Load of the Company as determined for the purposes of the auction of the Company's BGS-RSCP Load. The fixed percentage is the Tranche size for the Company.

Wholesale Customer – an entity (e.g., a municipality or borough) authorized to take electric service for resale to retail customers under a wholesale contract filed with the FERC.

ARTICLE 2: GENERAL TERMS AND CONDITIONS

2.1 Capacity In Which Company Is Entering Into This Agreement

Each BGS-RSCP Supplier agrees and acknowledges that the Company is contracting for the provision of BGS-RSCP Supply from such BGS-RSCP Supplier as the agent for Customers receiving Basic Generation Service – Residential Small Commercial Pricing on the Company's distribution system pursuant to the authorizations provided to the Company under the _____ Order. The BGS-RSCP Supplier further agrees and acknowledges that the Company will administer and monitor the BGS-RSCP Supplier's performance in providing BGS-RSCP Supply under this Agreement and that the Company shall be entitled, on behalf of Customers, to enforce BGS-RSCP Suppliers' obligations related to the provision of BGS-RSCP Supply. The BGS-RSCP Supplier hereby permanently and irrevocably waives any claim that Company is not entitled to seek enforcement of this Agreement on behalf of Customers. However, no Customer or

group of Customers may seek enforcement of this Agreement directly against the BGS-RSCP Supplier on their own behalf, including independently or by joining in any legal action by the Company. This Agreement does not create, assign or grant to any Customer or group of Customers any rights in or claims to damages or remedies against the BGS-RSCP Supplier independent of or different from the rights expressly granted to the Company hereunder as agent for Customers.

The Company is also contracting on its own behalf and not as agent for Customers insofar as this Agreement requires the Company: (i) to pay the BGS-RSCP Suppliers as required under this Agreement; and (ii) to provide data or to otherwise cooperate with a BGS-RSCP Supplier in connection with such BGS-RSCP Supplier's provision of BGS-RSCP Supply. The Parties acknowledge that the Agreement is a forward contract and, accordingly, the Parties hereto are entitled to the protections of section 556 of the Bankruptcy Code. The Parties therefore agree that the Agreement may be terminated by either Party upon the commencement of a proceeding by the other Party under any chapter of the Bankruptcy Code in accordance with Section 5.2 of this Agreement.

2.2 Parties' Obligations

(a) Obligations of BGS-RSCP Supplier

Each BGS-RSCP Supplier hereby agrees severally, but not jointly, as follows:

(i) to provide sufficient quantities of BGS-RSCP Supply on an instantaneous basis at all times to meet the BGS-RSCP Supplier Responsibility Share; without limitation, the BGS-RSCP Supplier shall be obligated to procure those services provided by the PJM OI and to perform such functions as may be required by the PJM OI that are necessary for the delivery of BGS-RSCP Supply required hereunder;

- (ii) to cooperate with the Company in any regulatory compliance efforts that may be required to maintain the ongoing legitimacy and enforceability of the terms of this Agreement and to fulfill any regulatory reporting requirement associated with the provision of BGS-RSCP Supply, before the BPU, FERC or any other regulatory body asserting jurisdiction, including meeting the reporting requirements of any Energy Portfolio Standards and the BPU's Environmental Information Disclosure Standards, N.J.A.C. 14:8.3 et seq;
- (iii) to become the Load Serving Entity with respect to the provision of BGS-RSCP Supply for the BGS-RSCP Supplier Responsibility Share and to comply with all requirements of a Load Serving Entity with respect to such BGS-RSCP Supplier Responsibility Share;
- (iv) to pay to the Company the PMEA/FMEA Adjustment Amount for any Billing Month in which the PMEA exceeds the FMEA, as more fully described in Article 9 of this Agreement;
- (v) to pay to the Company a charge of \$_____ per Tranche, which amount was announced prior to the auction for each Tranche comprising the BGS-RSCP Supplier's BGS-RSCP Supplier Responsibility Share, in order to reimburse the Company for the total costs of the BGS-RSCP auction and related costs associated with providing BGS-RSCP administration;
- (vi) to satisfy the Energy Portfolio Standards with respect to its BGS-RSCP Supplier Responsibility Share; and
- (vii) to comply in a timely manner with all obligations under this Agreement imposed upon a BGS-RSCP Supplier.

(b) Obligations of the Company

The Company hereby agrees as follows:

- (i) to pay to each BGS-RSCP Supplier every month an amount equal to the Auction Price multiplied by the Seasonal Billing Factor multiplied by the PMEA, as detailed in Article 9 of this Agreement and subject to the exceptions set forth therein;
- (ii) to pay to each BGS-RSCP Supplier the PMEA/FMEA Adjustment Amount for any Billing Month in which the FMEA exceeds the PMEA, as more fully described in Article 9 of this Agreement;
- (iii) to provide to each BGS-RSCP Supplier its estimated aggregate load obligation for each Supply Day twenty (20) days prior to the Supply Day, and its final estimated load obligation for each Supply Day five (5) days prior to the Supply Day;
- (iv) to the extent that (a) the Company purchases Energy during the term of this Agreement from renewable energy resources that meet the eligibility requirements for satisfying the Energy Portfolio Standards; (b) the renewable energy attributes transfer to the Company under the terms of the Energy purchase arrangements; and (c) the Company is not utilizing the renewable energy attributes associated with such purchases to meet obligations as a Load Serving Entity within PJM, the Company shall apply such renewable energy attributes to help satisfy the BGS-RSCP Supplier's Energy Portfolio Standards obligations under Section 2.2(a)(vi) of this Agreement to the extent that the Applicable Legal Authorities expressly allow or authorize the Company to do so;
- (v) to pay to each BGS-RSCP Supplier, as approved by the BPU, a percentage share equal to the BGS-RSCP Supplier Responsibility Share of such BGS-RSCP Supplier of all amounts received from a TPS as damages, penalties or forfeited security due to the failure of such TPS to provide adequate notice in conformance with applicable BPU requirements that a Customer previously served by the TPS is switching

to BGS-RSCP or forfeited as a result of an event of default by a TPS under the Company's Third Party Supplier Agreement; provided that the amounts paid to BGS-RSCP Suppliers shall be net of any amounts retained by the Company to offset costs or losses of the Company associated with the failure of the TPS to provide adequate notice or the occurrence of an event of default under the Company's Third Party Supplier Agreement; and further provided, that the Company shall have no obligation to seek the recovery of any damages, penalties or forfeited security due from a TPS through collection efforts, judicial procedures or otherwise; and

(vi) to comply in a timely manner with all obligations under this Agreement imposed upon the Company.

(c) Obligation of the BGS-RSCP Customers

The Company hereby agrees on behalf of the BGS-RSCP Customers to accept the delivery of BGS-RSCP Supply necessary to meet the BGS-RSCP Load.

2.3 PJM Services

Each BGS-RSCP Supplier shall make all necessary arrangements for the delivery of BGS-RSCP Supply through the PJM OI. The Company will advise the PJM OI of the magnitude and location of each BGS-RSCP Supplier's actual BGS-RSCP Supplier Responsibility Share, as required by the PJM OI, for the purpose of calculating such BGS-RSCP Supplier's appropriate Firm Transmission Service obligation, Capacity obligation, Energy obligation, or other requirements related to the provision of service under this Agreement by BGS-RSCP Suppliers arising under the PJM OATT, PJM RAA, PJM OA and any other applicable PJM agreement (collectively, the "PJM Agreements"). Each BGS-RSCP Supplier shall remain responsible to PJM for the performance of its LSE obligations associated with the provision of BGS-RSCP Supply under this

Agreement until the effective date of the transfer of such LSE obligations.

2.4 Communications and Data Exchange

Each BGS-RSCP Supplier and the Company shall supply to each other all data, materials or other information that is specified in this Agreement, or that may otherwise reasonably be required by BGS-RSCP Suppliers or by the Company in connection with the provision of BGS-RSCP Supply by the BGS-RSCP Supplier to BGS-RSCP Customers, if required, in a thorough and timely manner.

Electronic information exchange between each BGS-RSCP Supplier and the Company under this Agreement shall employ a BGS-RSCP Supplier identification number, assigned by the Company, which shall be consistent with the BGS-RSCP Supplier's Dunn & Bradstreet Business number. Each BGS-RSCP Supplier must be equipped with the communications capabilities necessary to comply with the communications and data exchange standards that are set by and as may, from time to time, be modified by PJM, and must bear the costs of putting in place and successfully testing all required information technology systems that will enable it to send to and receive data from the Company and PJM and to satisfy its obligations under this Agreement, the PJM Agreements and all other relevant agreements.

2.5 Record Retention

The Company shall retain, for a period of two (2) years following the expiration of the term of this Agreement, necessary records so as to permit BGS-RSCP Suppliers to confirm the validity of payments due to BGS-RSCP Suppliers hereunder; provided that if a BGS-RSCP Supplier has provided notice within two (2) years of the expiration of the term of this Agreement that it disputes the validity of any payments, the Company agrees that it shall retain all records related to such dispute until the dispute is finally resolved.

2.6 Verification

In the event of a good faith dispute regarding any invoice issued or payment due under this Agreement, and provided that a mutually acceptable confidentiality agreement is executed by the Parties, each Party will have the right to verify, at its sole expense, the accuracy of the invoice or the calculation of the payment due by obtaining copies of relevant portions of the books and records of the other Party. The right of verification will survive the termination of this Agreement for a period of two (2) years after termination.

ARTICLE 3: REPRESENTATIONS AND WARRANTIES

3.1 BGS-RSCP Supplier's Representations and Warranties

Each BGS-RSCP Supplier hereby represents, warrants and covenants to the Company as follows:

- a) such BGS-RSCP Supplier is a corporation, partnership, limited liability company or other legal entity, as set forth in Appendix A hereto, duly organized, validly existing and in good standing under the laws of the State of New Jersey or, if another jurisdiction, under the laws of such jurisdiction and, in such case, is duly registered and authorized to do business and is in good standing in the State of New Jersey;
- b) such BGS-RSCP Supplier has all requisite power and authority to execute and deliver this Agreement and to carry on the business to be conducted by it under this Agreement and to enter into and perform its obligations hereunder, including satisfaction of all applicable FERC requirements;
- c) the execution and delivery of this Agreement and the performance of such BGS-RSCP Supplier's obligations hereunder have been duly authorized by all necessary

action on the part of the BGS-RSCP Supplier and do not and will not conflict with, or constitute a breach of or default under, any of the terms, conditions, or provisions of the BGS-RSCP Supplier's certificate of incorporation or bylaws or other constituent instruments or any indenture, mortgage, other evidence of indebtedness, or other agreement or instrument or any statute or rule, regulation, order, judgment, or decree of any judicial or administrative body to which the BGS-RSCP Supplier is a party or by which the BGS-RSCP Supplier or any of its properties is bound or subject;

- d) all necessary and appropriate action that is required on the BGS-RSCP Supplier's part to execute this Agreement has been completed;
- e) this Agreement is the legal, valid and binding obligation of such BGS-RSCP Supplier, enforceable in accordance with its terms;
- f) there are no actions at law, suits in equity, proceedings or claims pending or, to such BGS-RSCP Supplier's knowledge, threatened against the BGS-RSCP Supplier before any federal, state, foreign or local court, tribunal or governmental agency or authority that might materially delay, prevent or hinder the BGS-RSCP Supplier's performance of its obligations hereunder;
- g) it has entered into this Agreement with a full understanding of the material terms and risks of the same, and it is capable of assuming those risks;
- h) the BGS-RSCP Supplier is in good standing as an LSE in PJM, is a signatory to all applicable PJM Agreements, and is in compliance with, and will continue to comply with, all obligations, rules and regulations, as established and interpreted by the PJM OI, that are applicable to LSEs as defined by the PJM Agreements; provided that the BGS-RSCP Supplier shall not be obligated to become an LSE in PJM until the date it begins providing BGS-RSCP Supply to Customers;

- i) it has made its trading and investment decisions (including regarding the suitability thereof) based upon its own judgment and any advice from such advisors as it has deemed necessary and not in reliance upon any view expressed by the Company; and
- j) the BGS-RSCP Supplier will comply with any and all information and data transfer protocols that may be adopted by the Company or that are set by, and from time to time modified by, the Board; provided that each BGS-RSCP Supplier shall be entitled to exercise its reserved right to challenge any such protocols in the appropriate forum.

3.2 Company's Representations and Warranties

The Company hereby represents, warrants and covenants to the BGS-RSCP Suppliers as follows:

- a) the Company is an electric utility corporation duly organized, validly existing and in good standing under the laws of the State of New Jersey;
- b) the Company has all requisite power and authority to carry on the business to be conducted by it under this Agreement and to enter into and perform its obligations hereunder;
- c) the execution and delivery of this Agreement and the performance of the Company's obligations hereunder have been duly authorized by all necessary action on the part of the Company and do not and will not conflict with, constitute a breach of or default under, any of the terms, conditions, or provisions of the Company's certificate of incorporation or bylaws or any indenture, mortgage, other evidence of indebtedness, or other agreement or instrument or any statute or rule, regulation, order, judgment, or decree of any judicial or administrative body to which the Company is a party or by which the Company or any of its properties is bound or subject;

- d) all necessary and appropriate action that is required on the Company's part to execute this Agreement has been completed;
- e) this Agreement is the legal, valid and binding obligation of the Company, enforceable in accordance with its terms;
- f) the ability of the Company to pay any and all amounts due and payable under this Agreement, or upon any potential breach thereof, is not conditioned upon any governmental or administrative appropriation by the Board, the State of New Jersey or any other governmental authority;
- g) there are no actions at law, suits in equity, proceedings or claims pending or, to the Company's knowledge, threatened against the Company before any federal, state, foreign or local court, tribunal or governmental agency or authority that might materially delay, prevent or hinder the Company's performance of its obligations under this Agreement;
- h) it has entered into this Agreement with a full understanding of the material terms and risks of the same, and it is capable of assuming those risks;
- i) with respect to those rights and entitlements conferred on Customers under this Agreement as set forth in Section 2.1 of this Agreement, the Board has conferred on the Company all requisite power and authority to execute this Agreement on behalf of such Customers;
- j) the Company's performance under this Agreement is not contingent upon the performance of Customers or the ability of Customers to pay rates;
- k) the Company shall have full responsibility for metering, billing and delivery with respect to Customers and BGS-RSCP Suppliers shall have no responsibility with respect thereto; and

 the Company shall be responsible for distribution services and the BGS-RSCP Supplier shall not be responsible for distribution charges.

3.3 Survival of Obligations

All representations and warranties contained in this Article are of a continuing nature and shall be maintained during the term of this Agreement. If a Party learns that any of the representations, warranties or covenants in this Agreement are no longer true during the term of this Agreement, the Party shall immediately notify the other Party via facsimile, with a hard copy of the notice delivered by overnight mail.

ARTICLE 4: COMMENCEMENT AND TERMINATION OF AGREEMENT

4.1 Commencement and Termination

The term of this Agreement shall commence upon the Effective Date; provided that the provision of BGS-RSCP Supply by BGS-RSCP Suppliers to BGS-RSCP Customers on the Company's system shall commence at 00:01 a.m. on June 1, 2019 and shall end at midnight on May 31, 2022, unless this Agreement is terminated earlier in accordance with the provisions hereof.

4.2 Termination of Right to Supply BGS-RSCP

Each BGS-RSCP Supplier agrees that termination of this Agreement for reason of an Event of Default shall terminate any right of such BGS-RSCP Supplier to provide BGS-RSCP Supply to the BGS-RSCP Customers and nullify any of the entitlements to which such BGS-RSCP Supplier became entitled as a result of being selected as a winning bidder in the competitive solicitation for BGS-RSCP Supply.

4.3 Survival of Obligations

Termination of this Agreement for any reason shall not relieve the Company or any BGS-RSCP Supplier of any obligation accrued or accruing prior to such termination. Applicable provisions of this Agreement shall continue in effect after termination to the extent necessary to provide for final billings and adjustments including, without limitation, any obligation to pay amounts tracked and retained by the Company during the term of this Agreement for the benefit of the BGS-RSCP Suppliers under Section 15.9 of this Agreement.

4.4 Mutual Termination

The Company and the BGS-RSCP Supplier may agree at any time during the term of this Agreement to terminate their respective rights and obligations hereunder on such terms and under such conditions that they mutually deem to be appropriate as set forth in a mutual termination agreement acceptable in form and substance to the Company and the BGS-RSCP Supplier ("Mutual Termination Agreement"); provided that Company agrees that it shall enter into such a Mutual Termination Agreement, which will discharge the terminating BGS-RSCP Supplier (the "Terminating BGS-RSCP Supplier") with respect to liabilities arising after the effective date of the Mutual Termination Agreement if the following conditions precedent are met: (i) the Terminating BGS-RSCP Supplier identifies a replacement supplier willing to assume all obligations of the Terminating BGS-RSCP Supplier hereunder for the remaining term of this Agreement (the "Replacement BGS Supplier"); (ii) the Replacement BGS Supplier demonstrates its compliance with Article 6 of this Agreement, "Creditworthiness", as of the effective date of the Mutual Termination Agreement; (iii) the Replacement BGS Supplier executes a counterpart signature page to this Agreement and thereby becomes a Party under this

Agreement, effective immediately following the effective date of the Mutual Termination Agreement; and (iv) the Terminating BGS Supplier is not, to the belief or knowledge of the Company, subject to an Event of Default as of the effective date of the Mutual Termination Agreement or, if the Company believes that the Terminating BGS Supplier may be subject to an Event of Default, either (a) the Company has determined that, as of the effective date of the Mutual Termination Agreement, it has not incurred any Damages as a result of the Event of Default or (b) if the Company has determined, as of the effective date of the Mutual Termination Agreement, that it may have incurred Damages as a result of the Event of Default, that the Replacement BGS Supplier has agreed in writing to be responsible for the payment of such Damages or to otherwise cure the Event of Default, in either case to the satisfaction of the Company.

ARTICLE 5: BREACH AND DEFAULT

5.1 Events of Default

An Event of Default under this Agreement shall occur if a Party (the "Defaulting Party"):

- (i) is the subject of a voluntary bankruptcy, insolvency or similar proceeding;
- (ii) makes an assignment for the benefit of its creditors;
- (iii) applies for, seeks consent to, or acquiesces in the appointment of a receiver, custodian, trustee, liquidator or similar official to manage all or a substantial portion of its assets;
- (iv) is dissolved (other than pursuant to a consolidation, amalgamation or merger) or is the subject of a Merger Event;
 - (v) has a secured party take possession of all or substantially all of its assets or

has a distress, execution, attachment, sequestration or other legal process levied, enforced or sued on or against all or substantially all of its assets;

- (vi) has a resolution passed for its winding-up, official management or liquidation (other than pursuant to a consolidation, amalgamation or merger);
- (vii) in the case of a BGS-RSCP Supplier, PJM terminates the BGS-RSCP Supplier's ability to make purchases from PJM markets or PJM holds the Company responsible for the provision of Energy, Capacity, Firm Transmission Service or Ancillary Services to meet the BGS-RSCP Supplier's BGS-RSCP Supplier Responsibility Share under this Agreement;
- (viii) fails to comply with the creditworthiness requirements as set forth in Article 6 of this Agreement, including, without limitation, compliance with the creditworthiness requirements to cover the Margin calculated under Section 6.7 or post any Margin due under Section 6.7 of this Agreement, within the time frames set forth in this Agreement;
- (ix) in the case of the Company, fails to implement any measures that it is directed to implement by the Board pursuant to the creditworthiness review procedure adopted by the Board in its Order dated December 4, 2002 (Docket No. EX01110754);
 - (x) fails to pay the other Party when payment is due;
- (xi) violates any federal, state or local code, regulation or statute applicable to the supply of Energy in a manner that materially, and adversely, affects the Party's performance under this Agreement, including by way of failure to continually satisfy all applicable FERC requirements, or, in the case of a BGS-RSCP Supplier, by way of failure to maintain any other governmental approvals required for participation in the New Jersey retail Energy market as a BGS-RSCP Supplier, default on any obligation or

other failure to comply with PJM requirements under the PJM Agreements or fails to comply with the Energy Portfolio Standards with respect to the BGS Supplier's BGS-RSCP Supplier Responsibility Share;

- (xii) is the subject of an involuntary bankruptcy or similar proceeding;
- (xiii) subject to Section 5.3 (b) of this Agreement, in the case of the Company acting on behalf of the BGS-RSCP Customers, fails to accept BGS-RSCP Supply properly tendered by the BGS-RSCP Supplier under this Agreement;
- (xiv) fails to satisfy any other material obligation under this Agreement not listed above;
- (xv) makes a materially incorrect or misleading representation or warranty under this Agreement; or
- (xvi) commits an act or makes an omission that constitutes an "Event of Default" under any other agreement(s) for the provision of BGS Supply between the Company and the BGS-RSCP Supplier; and fails to remedy such condition, event or delinquency herein above described such that the other Party (the "Non-Defaulting Party") is completely made whole with respect to

the other Party (the "Non-Defaulting Party") is completely made whole with respect to such condition, event or delinquency, within three (3) Business Days of receipt of written notice thereof from such Non-Defaulting Party; provided, however, that an Event of Default shall be deemed to have occurred immediately, without any need for the provision of notice thereof by the Non-Defaulting Party and without any right of cure on the part of the Defaulting Party, in the event of the occurrence of a condition, event or delinquency described in subsections "i", "ii", "iii", "iv", "v", "vi", "vii" or "viii" above.

5.2 Rights Upon Default

Upon and during the continuation of an Event of Default, the Non-Defaulting

Party shall be entitled to:

- (i) pursue any and all available legal and equitable remedies;
- (ii) declare an Early Termination Date of this Agreement with respect to the obligations of the Defaulting Party without any liability or responsibility whatsoever except for obligations arising prior to the date of termination, by providing written notice to the Defaulting Party; provided, however, that this Agreement shall immediately terminate automatically and without notice in the case of any Event of Default in which a Supplier is the Defaulting Party occurring under subsections (i), (ii), (iii), (iv), (v), (vii) or (viii) of Section 5.1 of this Agreement and such date of automatic termination shall be deemed the Early Termination Date of this Agreement with respect to such Supplier; and
- (iii) receive Damages in accordance with Section 5.3 of this Agreement.

 The Non-Defaulting Party shall be entitled to elect or pursue one or more of the above remedies.
 - 5.3 Damages Resulting From an Event of Default
- Declaration of Early Termination By Company: Damages resulting from (i) a BGS-RSCP Supplier's failure to (A) provide BGS-RSCP Supply in conformance with Section 2.2 hereof or (B) pay PJM for purchases of any products or services from PJM, or other failure to comply with PJM requirements, such that PJM holds the Company responsible for the provision of Energy, Capacity, Firm Transmission Service or Ancillary Services to meet such BGS-RSCP Supplier's BGS-RSCP Supplier Responsibility Share under this Agreement or (ii) the occurrence of any Event of Default attributable to a BGS-RSCP Supplier resulting in Early Termination, shall include all Costs incurred by the Company,

acting in a commercially reasonable manner consistent with any statutory or regulatory requirements imposed by the Applicable Legal Authorities, in obtaining replacement services or in obtaining a replacement supplier, which Costs exceed the amounts that would have been payable to the defaulting BGS-RSCP Supplier under this Agreement. Costs incurred by the Company for the purpose of calculating Damages hereunder will consist of:

- (i) the cost of Energy (including all charges for losses and congestion), Firm Transmission Service, Capacity, Ancillary Services or other elements of BGS-RSCP Supply allocated to the Company by the PJM OI due to the failure of a BGS-RSCP Supplier to meet obligations owing to the PJM OI in connection with its obligations under this Agreement;
- (ii) the cost of Energy (including all charges for losses and congestion), Firm Transmission Service, Capacity, Ancillary Services or other elements of BGS-RSCP Supply purchased by the Company to replace BGS-RSCP Supply that a BGS-RSCP Supplier was obligated to supply under this Agreement during the term hereof;
- (iii) administrative and legal costs associated with procuring replacement BGS-RSCP Supply; and
- (iv) financial hedging costs incurred by the Company on behalf of BGS-RSCP Customers as a result of having to procure BGS-RSCP Supply not provided by a BGS-RSCP Supplier.

Without limitation of the foregoing, Damages calculated hereunder shall constitute the ultimate liability of a BGS-RSCP Supplier in the event of an Early Termination caused by an Event of Default attributable to such BGS-RSCP Supplier

regardless of the reason or basis for such Early Termination. The Parties recognize, however, that the final calculation of Damages hereunder may not be known for some time since the level of such Damages may be dependent upon the arrangements made by the Company to obtain replacement services or a replacement supplier. The Company and each BGS-RSCP Supplier agree that, until the calculation of Damages under this provision is completed, the amount and payment to the Company of the Settlement Amount on behalf of BGS-RSCP Customers in the event of an Early Termination as set forth in Section 5.4 of this Agreement shall be immediately due and owing as an estimate of all Damages ultimately determined to be due and owing. After Damages have been finally determined under this Section 5.3, the amounts of Damages due and owing will be reconciled with payments already made by the BGS-RSCP Supplier under Section 5.4 of this Agreement.

Supply Tendered By BGS-RSCP Supplier: Damages resulting from the failure of the Company on behalf of Customers to accept BGS-RSCP Supply tendered by the BGS-RSCP Supplier necessary to meet the BGS-RSCP Supplier Responsibility Share of BGS-RSCP Load under this Agreement shall consist of the positive difference (if any) between the amounts that would have been payable to the BGS-RSCP Supplier hereunder had the Company accepted the BGS-RSCP Supplier Responsibility Share of BGS-RSCP Supplier necessary to meet the BGS-RSCP Supplier Responsibility Share of BGS-RSCP Load under this Agreement minus the amount realized by the BGS-RSCP Supplier in disposing, in a commercially reasonable manner, of the BGS-RSCP Supply not accepted by the Company; provided, however, that the Company shall not be required to accept on behalf of any Customer, quantities of unbundled Energy, Ancillary Services or other

component of BGS-RSCP Supply utilized by Customers on an instantaneous basis as a function of electrical load, in excess of such Customer's instantaneous consumption of such component of BGS-RSCP Supply.

- Default Attributable To the Company: Damages resulting from Early Termination due to an Event of Default attributable to the Company shall be as set forth in Section 5.4 of this Agreement. Damages calculated in accordance with said Section 5.4 shall be the exclusive remedy available to the BGS-RSCP Supplier in the event of Early Termination resulting from an Event of Default attributable to the Company.
- (d) Other Damages: Damages for Events of Default not specified above shall consist of the direct Damages incurred by the Non-Defaulting Party.

5.4 Declaration of an Early Termination Date and Calculation of Settlement Amount and Termination Payment

(a) Settlement Amount. If an Event of Default with respect to a Defaulting Party shall have occurred and be continuing, the Non-Defaulting Party (in the case of an Event of Default by the Company, each BGS-RSCP Supplier shall be considered a "Non-Defaulting Party") shall have the right (i) to designate a day, no earlier than the day such notice is effective and no later than twenty (20) days after such notice is effective, as a date for Early Termination ("Early Termination Date") to accelerate all amounts owing between the Parties and to liquidate and terminate the undertakings set forth in this Agreement, (ii) to withhold any payments due to the Defaulting Party under this Agreement, and (iii) to suspend performance; provided however, that an Early Termination Date shall be deemed to occur automatically and concurrently with the Event of Default, without any requirement for the provision of notice by the Non-

Defaulting Party, with respect to an Event of Default under subsections "i", "ii", "iii", "iv", "v", "vi", "vii" and "viii" of Section 5.1 of this Agreement. The Non-Defaulting Party shall calculate, in a commercially reasonable manner, a Settlement Amount with respect to the obligations under this Agreement. For the purposes of such determination, the quantity amounts of Energy (including all charges for losses and congestion), Capacity and other services provided for under this Agreement for the period following the Early Termination Date through the remainder of the term of this Agreement shall be deemed to be those quantity amounts that would have been delivered on an hourly basis, had this Agreement been in effect during the previous calendar year adjusted for such BGS-RSCP Load changes as may have occurred since the previous calendar year.

(b) Net Out of Settlement Amounts. The Non-Defaulting Party shall calculate a Termination Payment by aggregating all Settlement Amounts due under this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply into a single amount: by netting out (a) all Settlement Amounts that are due or will become due to the Defaulting Party, plus at the option of the Non-Defaulting Party, any cash or other form of security then available to the Non-Defaulting Party and actually received, liquidated and retained by the Non-Defaulting Party, plus any or all other amounts due to the Defaulting Party under this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply against (b) all Settlement Amounts that are due or will become due to the Non-Defaulting Party, plus any or all other amounts due to the Non-Defaulting Party under this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply, so that all such amounts shall be netted out to a single liquidated amount; provided, however, that if the BGS-RSCP

Supplier is the Defaulting Party and the Termination Payment is due to the BGS-RSCP Supplier, the Company shall be entitled to retain a commercially reasonable portion of the Termination Payment, which may be equal to the entire amount of the Termination Payment, as security for additional amounts that may be determined to be due and owing by the BGS-RSCP Supplier as Damages and further provided that any previously attached security interest of the Company in such retained amounts shall continue. The Termination Payment shall be due to or due from the Non-Defaulting Party as appropriate. If the Termination Payment has been retained by the Company as security for additional amounts that may be determined to be due and owing by the BGS-RSCP Supplier, and if, upon making a final determination of Damages, the Termination Payment, or any portion thereof, is to be made to the BGS-RSCP Supplier, the Company will pay simple interest on the Termination Payment amount being made to the BGS-RSCP Supplier. Simple interest will be calculated at the lower of the Interest Index or six (6) percent per annum.

- (c) Notice of Termination Payment. As soon as practicable after calculation of a Termination Payment, notice shall be given by the Non-Defaulting Party to the Defaulting Party of the amount of the Termination Payment and whether the Termination Payment is due to or due from the Non-Defaulting Party. The notice shall include a written statement explaining in reasonable detail the calculation of such amount. Subject to Section 5.4(b) above, the Termination Payment shall be made by the Party that owes it within three (3) Business Days after such notice is effective.
- (d) <u>Disputes With Respect to Termination Payment</u>. If the Defaulting Party disputes the Non-Defaulting Party's calculation of the Termination Payment, in whole or in part, the Defaulting Party shall, within three (3) Business Days of receipt of

Non-Defaulting Party's calculation of the Termination Payment, provide to the Non-Defaulting Party a detailed written explanation of the basis for such dispute; provided, however, that if the Termination Payment is due from the Defaulting Party, the Defaulting Party shall first provide commercially reasonable financial assurances to the Non-Defaulting Party in an amount equal to the Termination Payment.

(e) Multiple BGS Supply Agreements. It is the intention of the Company and the BGS-RSCP Supplier that, in the event the BGS-RSCP Supplier is a party to other agreements with the Company for the provision of BGS Supply that existed prior to the Effective Date of this Agreement or are entered into after the Effective Date of this Agreement, the Company will calculate a single Termination Payment applicable to all such agreements as set forth herein. Each BGS-RSCP Supplier that is a party to such other agreements with the Company for the provision of BGS Supply hereby agrees that such other agreements are deemed amended by this Agreement for the purpose of calculating a single Termination Payment as described herein.

5.5 Step-up Provision

If any one or more BGS-RSCP Suppliers defaults in its obligations hereunder resulting in the exercise of the right of Early Termination by the Company with respect to such BGS-RSCP Supplier(s), then the Company, consistent with its Contingency Plan approved by the Board in its Order dated_______, may, at the Company's option, offer some or all Non-Defaulting Supplier(s) the optional right to assume under this Agreement additional Tranches of BGS-RSCP Load, subject to compliance with the creditworthiness provisions of Article 6 of this Agreement. The provision of any such offer by the Company to Non-Defaulting Suppliers shall indicate the duration of the offer and the manner of acceptance thereof. Following the assumption by BGS-RSCP

Supplier(s) of additional Tranches hereunder, the Company shall prepare a modified Appendix A which shall set forth the revised BGS-RSCP Supplier Responsibility Shares of the BGS-RSCP Load of the participating Non-Defaulting BGS-RSCP Supplier(s) following such assumption. This modified Appendix A shall be initialed (as a single document or in counterparts) by the Company and any affected BGS-RSCP Supplier(s) and shall thereafter be deemed a part of this Agreement, as to such affected BGS-RSCP Supplier(s), from the effective date of such modified Appendix A. A BGS-RSCP Supplier will not suffer any prejudice if it declines an offer to assume additional Tranches upon the default by another BGS-RSCP Supplier.

5.6 Setoff of Payment Obligations of The Non-Defaulting Party

Any payment obligations of the Non-Defaulting Party to the Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply shall be set off: (i) first, to satisfy any payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply that are unsecured and not subject to any Guaranty; (ii) second, to satisfy any payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply that are unsecured, but which are subject to a Guaranty; and (iii) third, to satisfy any remaining payment obligations of the Defaulting Party to the Non-Defaulting Party pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply.

5.7 Preservation of Rights of Non-Defaulting Party

The rights of the Non-Defaulting Party under this Agreement, including without limitation Sections 5.4 and 5.6 of this Agreement, shall be supplemental to, and not in lieu of, any right of recoupment, lien, or set-off afforded by applicable law, and all such rights are expressly preserved for the benefit of the Non-Defaulting Party.

ARTICLE 6: CREDITWORTHINESS

6.1 Applicability

Each BGS-RSCP Supplier agrees that it shall meet the creditworthiness requirements of this Article 6 at all times during the term of this Agreements and shall inform the Company immediately of any changes in its credit rating or financial condition. Without limitation of the foregoing, each BGS-RSCP Supplier shall, upon written request, affirmatively demonstrate to the Company, its compliance with the creditworthiness requirements set forth hereunder. The Company may establish less restrictive creditworthiness requirements under this Article 6 in a non-discriminatory manner.

6.2 Creditworthiness Determination

The BGS-RSCP Supplier may submit and maintain a security deposit in accordance with Section 6.3 and 6.6 of this Agreement in lieu of submitting to or being qualified under a creditworthiness evaluation. The BGS-RSCP Supplier shall have the opportunity to petition the Company to re-evaluate its creditworthiness whenever an event occurs that the BGS-RSCP Supplier believes would improve the determination made by the Company of its creditworthiness. The Company's credit re-evaluation must be completed as soon as possible but no longer than thirty (30) days after receiving a

fully documented request. The Company must provide the rationale for its determination of the credit limit and any resulting security requirement. The Company must perform its credit re-evaluation and associated security calculation in a non-discriminatory manner. BGS-RSCP Suppliers shall provide unrestricted access to audited financial statements; provided that if audited financial statements are not available, the Company may specify other types of financial statements that will be accepted.

6.3 Independent Credit Requirement

The Independent Credit Requirement per Tranche ("ICRT") that will be required of BGS-RSCP Suppliers under this Agreement shall initially be \$2.4 million per Tranche and shall decline in accordance with the schedule included as part of Appendix B throughout the term hereof. The ICR under this Agreement is the ICRT times the number of Tranches shown in Appendix A hereto.

6.4 Independent Credit Threshold

BGS-RSCP Suppliers that qualify under the following criteria will be granted an Independent Credit Threshold ("ICT"). The ICT will be used by the BGS-RSCP Supplier solely to partially or fully cover the aggregate ICR amounts under this Agreement and any other BGS Supply agreement(s) between it and the Company. In all instances, the most current senior unsecured debt rating (or, if unavailable, the most current corporate issuer rating discounted one notch) will be used.

(a) The following requirements shall apply to BGS-RSCP Suppliers or Guarantors of BGS-RSCP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia in order to be granted an ICT. For BGS-RSCP Suppliers who cannot meet the following requirements, the posting of cash or a letter of credit in an acceptable form as defined in Section 6.9(b)

below (see standard format in Appendix C) for the entire aggregate ICR amounts under this Agreement and any other BGS Supply agreement(s) between it and the Company will be required at the time of or prior to the execution of this Agreement.

(i) The BGS-RSCP Supplier shall meet the following requirements: (1) must be rated by at least two of the following rating agencies: S&P Global Ratings ("S&P"), Moody's Investors Service, Inc. ("Moody's"), Fitch, Inc. ("Fitch") or A.M. Best Company ("A.M. Best"), and (2) must have a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) of at least "BBB-" from S&P, "Baa3" from Moody's, "BBB-" from Fitch, or "bbb" from A.M. Best (a "Minimum Rating"). If the BGS-RSCP Supplier is rated by only two rating agencies, and the ratings are split, the lower rating will be used. If the BGS-RSCP Supplier is rated by three or four rating agencies, and the ratings are split, the lower of the two highest ratings will be used; however, in the event that the two highest ratings are common, such common rating will be used. The maximum level of the ICT will be determined based on the following table:

Credit Rating of the BGS-RSCP Supplier				Max. Independent Credit Threshold
S&P	Moody's	Fitch	A.M. Best	
A- and above	A3 and above	A- and above	aaa	16% of TNW ¹
BBB+	Baa1	BBB+	aa	10% of TNW
BBB	Baa2	BBB	a	8% of TNW
BBB-	Baa3	BBB-	bbb	6% of TNW
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW

The BGS-RSCP Supplier will be required to post cash or a letter of credit in an

^{1 &}quot;TNW" means Tangible Net Worth.

acceptable form as defined in Section 6.9(b) of this Agreement (see standard format in Appendix C) for the aggregate ICR amounts under this Agreement and any other BGS Supply agreement(s) between it and the Company, to the extent that the aggregate ICR exceeds the ICT at the time of or prior to the execution of this Agreement; or

(ii) For BGS-RSCP Suppliers having a Guarantor, the Guarantor (1) must be rated by at least two of the following rating agencies: S&P, Moody's, Fitch, or A.M. Best, and (2) must have a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) equal to the Minimum Rating. If the Guarantor is rated by only two rating agencies, and the ratings are split, the lowest rating will be used. If the Guarantor is rated by three or four rating agencies, and the ratings are split, the lower of the two highest ratings will be used; however, in the event that the two highest ratings are common, then such common rating will be used. The maximum level of the ICT that could be provided through the Guaranty (see standard format in Appendix D) will be determined based on the following table:

	Max. Independent Credit Threshold			
S&P	Moody's	Fitch	A.M. Best	
A- and above	A3 and above	A- and above	aaa	16% of TNW
BBB+	Baa1	BBB+	aa	10% of TNW
BBB	Baa2	BBB	a	8% of TNW
BBB-	Baa3	BBB-	bbb	6% of TNW
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW

The BGS-RSCP Supplier will be granted an ICT of up to the amount of the Guaranty, provided that the amount of the Guaranty is below the maximum ICT shown in the table above; provided that if a Guaranty is provided for an unlimited amount, the BGS-RSCP Supplier will be granted an ICT of up to the maximum ICT shown in the

table above. The Guaranty tendered by the BGS-RSCP Supplier to satisfy the ICT requirement arising under this Section 6.4 shall be a separate document from the Guaranty, if any, tendered by the BGS-RSCP Supplier to satisfy any requirement for a credit limit to cover the Total Exposure Amount arising under Section 6.6 of this Agreement; provided, however, that a single Guaranty may be provided if such Guaranty is for an unlimited amount. The BGS-RSCP Suppliers will be required to post cash or a letter of credit in an acceptable form as defined in Section 6.9 of this Agreement for the aggregate ICR amounts under this Agreement and any other BGS Supply agreement(s) between it and the Company, to the extent that the aggregate ICR exceeds the ICT at the time of or prior to the execution of this Agreement.

- (b) The following standards shall apply to BGS-RSCP Suppliers or Guarantors of BGS-RSCP Suppliers that have not been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia. For a BGS-RSCP Supplier who cannot meet the following requirements, the posting of cash or a letter of credit in an acceptable form as defined in Section 6.9(b) of this Agreement (see standard format in Appendix C) for the entire aggregate ICR amounts under this Agreement and any other BGS Supply agreement(s) between it and the Company will be required at the time of or prior to the execution of this Agreement.
- (i) The BGS-RSCP Supplier shall supply such evidence of creditworthiness so as to provide the Company with comparable assurances of creditworthiness as is applicable above for BGS-RSCP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia. The Company shall have full discretion, without liability or recourse to the BGS-RSCP Supplier, to evaluate the evidence of creditworthiness

submitted by such BGS-RSCP Supplier; or

- (ii) The Guarantor of a BGS-RSCP Supplier shall supply such evidence of creditworthiness so as to provide the Company with comparable assurances of creditworthiness as is applicable above for Guarantors of BGS-RSCP Suppliers that have been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia. The Company shall have full discretion, without liability or recourse to the Guarantor or the BGS-RSCP Supplier, to evaluate the evidence of creditworthiness submitted by such Guarantor.
- (c) All BGS-RSCP Suppliers or Guarantors of BGS-RSCP Suppliers that have not been incorporated or otherwise formed under the laws of a state of the United States or of the District of Columbia shall, in addition to all documentation required elsewhere in this Section 6.4, supply the following as a condition of being granted an ICT:
- (i) For BGS-RSCP Suppliers: (i) a legal opinion of counsel qualified to practice in the foreign jurisdiction in which the BGS-RSCP Supplier is incorporated or otherwise formed that this Agreement has been duly authorized, executed and delivered and is the legal, valid and binding obligation of the BGS-RSCP Supplier in the jurisdiction in which it has been incorporated or otherwise formed; (ii) the sworn certificate of the corporate secretary (or similar officer) of such BGS-RSCP Supplier that the person executing this Agreement on behalf of the BGS-RSCP Supplier has the authority to execute the Agreement and that the governing board of such BGS-RSCP Supplier has approved the execution of this Agreement; and (iii) the sworn certificate of the corporate secretary (or similar officer) of such BGS-RSCP Supplier that the BGS-RSCP Supplier has been authorized by its governing board to enter into agreements of the

same type as this Agreement. The Company shall have full discretion, without liability or obligation to the BGS-RSCP Supplier, to evaluate the sufficiency of the documents submitted by the BGS-RSCP Supplier.

(ii) For the Guarantor of a BGS-RSCP Supplier: (i) a legal opinion of counsel qualified to practice in the foreign jurisdiction in which the Guarantor is incorporated or otherwise formed that the Guaranty has been duly authorized, executed and delivered and is the legal, valid and binding obligation of the Guarantor in the jurisdiction in which it has been incorporated or otherwise formed; (ii) the sworn certificate of the corporate secretary (or similar officer) of such Guarantor that the person executing the Guaranty on behalf of the Guarantor has the authority to execute the Guaranty and that the governing board of such Guarantor has approved the execution of the Guaranty; and (iii) the sworn certificate of the corporate secretary (or similar officer) of such Guarantor that the Guarantor has been authorized by its governing board to enter into agreements of the same type as this Guaranty. The Company shall have full discretion, without liability or obligation to the Guarantor or the BGS-RSCP Supplier, to evaluate the sufficiency of the documents submitted by such Guarantor.

6.5 Mark-to-Market Credit Exposure Methodology

To calculate the daily exposure for each BGS-RSCP Supplier the Mark-to-Market ("MtM") credit exposure methodology will be used. The "mark" for each Billing Month will be determined at the time the auction is completed based on the available Forward Market Prices and for the remaining Billing Months will be derived based on historical data. At the time the auction is completed, the MtM credit exposure for each BGS-RSCP Supplier shall be equal to zero. Subsequently, the differences between the available Forward Market Prices on the valuation date and the "mark" prices for the corresponding

Billing Months will be used to calculate the daily credit exposures for each BGS-RSCP Supplier. The total MtM credit exposure will be equal to 1.1 times the sum of the MtM credit exposures for each Billing Month. The methodology for calculation of the MtM credit exposure is illustrated in the example (using hypothetical numbers) set forth in Appendix B hereto.

6.6 Credit Limit

The following criteria constitute the Company's creditworthiness requirements for the BGS-RSCP Suppliers to cover the Total Exposure Amount. In all instances, the most current senior unsecured debt rating (or, if unavailable, the most current corporate issuer rating discounted one notch) will be used.

(i) For a BGS-RSCP Supplier to be granted an unsecured line of credit, the BGS-RSCP Supplier shall meet the following requirements: (1) must be rated by at least two of the following rating agencies: S&P, Moody's, Fitch or A.M. Best, and (2) must have a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) equal to the Minimum Rating. If the BGS-RSCP Supplier is rated by only two rating agencies, and the ratings are split, the lowest rating will be used. If the BGS-RSCP Supplier is rated by three or four rating agencies, and the ratings are split, the lower of the two highest ratings will be used; provided that in the event that the two highest ratings are common, such common rating will be used. The Maximum Credit Limit to cover the Total Exposure Amount will be determined based on the following table:

Credit Rating of the BGS-RSCP Supplier				Max. Credit Limit to be calculated as the lesser of the % of TNW and credit limit cap below	
S&P	Moody's	Fitch	A.M. Best	%	Credit Limit Cap
A- and above	A3 and above	A- and above	aaa	16% of TNW	\$60,000,000
BBB+	Baal	BBB+	aa	10% of TNW	\$40,000,000
BBB	Baa2	BBB	a	8% of TNW	\$30,000,000
BBB-	Baa3	BBB-	bbb	6% of TNW	\$15,000,000
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW	0 .

The BGS-RSCP Supplier will be required to post cash or a letter of credit in an acceptable form as defined in Section 6.9 (b) of this Agreement (see standard format in Appendix C) for the Margin due the Company as set forth in Section 6.7 of this Agreement; or

(ii) For BGS-RSCP Suppliers having a Guarantor, the Guarantor (1) must be rated by at least two of the following rating agencies: S&P, Moody's, Fitch or A.M. Best, and (2) must have a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) equal to the Minimum Rating. If the Guarantor is rated by only two rating agencies, and the ratings are split, the lowest rating will be used. If the Guarantor is rated by three or four rating agencies, and the ratings are split, the lower of the two highest ratings will be used; provided that in the event that the two highest ratings are common, such common rating will be used. The Maximum Credit Limit to cover the Total Exposure Amount that could be provided through the Guaranty (see standard format in Appendix D) will be determined based on the following table:

Credit Rating of the Guarantor				Max. Credit Limit to be calculated as the lesser of the % of TNW and credit limit cap below	
S&P	Moody's	Fitch	A.M. Best	%	Credit Limit Cap
A- and above	A3 and above	A- and above	aaa	16% of TNW	\$60,000,000
BBB+	Baal	BBB+	aa	10% of TNW	\$40,000,000
BBB	Baa2	BBB	a	8% of TNW	\$30,000,000
BBB-	Baa3	BBB-	bbb	6% of TNW	\$15,000,000
Below BBB-	Below Baa3	Below BBB-	Below bbb	0% of TNW	0

The BGS-RSCP Supplier will be granted a credit limit equal to the lesser of (i) the amount of the Guaranty as provided to the Company at the time this Agreement is executed as such amount may be modified in any amended or substitute Guaranty provided to the Company during the term of this Agreement, or (ii) the Supplier's Maximum Credit Limit. The BGS-RSCP Supplier, however, may not increase or substitute its Guaranty for the purpose of increasing its applicable credit limit during the time period after the Company has made a Margin call but before the BGS-RSCP Supplier has posted the required Margin. Notwithstanding anything herein to the contrary, the BGS-RSCP Supplier may increase the limit of its Guaranty after satisfying a Margin call from the Company and upon the Company's receipt of an amended or substitute Guaranty increasing the limit of the Guaranty, the BGS-RSCP Supplier may request a return of Margin in accordance with Section 6.7 of this Agreement. The BGS-RSCP Supplier will be required to post cash or a letter of credit in an acceptable form as defined in Section 6.9(b) of this Agreement (see standard format in Appendix C) for the Margin due the Company as set forth in Section 6.7 of this Agreement; or

(iii) The posting of cash or a letter of credit as defined in Section 6.9 (b) below for the entire Total Exposure Amount as set forth in Section 6.7 of this Agreement.

6.7 Posting Margin and Return of Surplus Margin

(a) If at any time and from time to time during the term of this Agreement, the Total Exposure Amount exceeds the BGS-RSCP Supplier's or the Guarantor's credit limit, then the Company on any Business Day, may request that the BGS-RSCP Supplier provide cash or a letter of credit in an acceptable form as defined in Section 6.9(b) of this Agreement (see standard format in Appendix C), in an amount equal to the Margin (less any Margin posted by the BGS-RSCP Supplier and held by the Company pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply).

If the BGS-RSCP Supplier receives written notice for Margin from the Company by 1:00 p.m. New York time on a Business Day, then the BGS-RSCP Supplier shall post Margin the next following Business Day, if posting cash, and by the second Business Day following the date of notice, if posting a letter of credit, unless the Company agrees in writing to extend the period to provide Margin. If the BGS-RSCP Supplier receives notice for Margin from the Company after 1:00 p.m. New York time on a Business Day, whether posting cash or a letter of credit, then the BGS-RSCP Supplier must post Margin the second Business Day following the date of notice unless the Company agrees in writing to extend the period to provide Margin. The Company will not unreasonably deny a request for a one-day extension of such period. In the event that the BGS-RSCP Supplier fails to post Margin when due in accordance with this Section 6.7, then an Event of Default under Article 5 of this Agreement will be deemed to have occurred and the Company will be entitled to the remedies set forth in Article 5 of this Agreement.

(b) Surplus Margin being held by the Company that is not needed to satisfy the Total Exposure Amount, as determined above, will be returned to the BGS-RSCP Supplier upon receipt of a written request by the BGS-RSCP Supplier. Surplus Margin means cash or a letter of credit posted by the BGS-RSCP Supplier as a result of a request by the Company pursuant to Section 6.7(a) that exceeds the Total Exposure Amount less the BGS-RSCP Supplier's or the Guarantor's credit limit. The amount returned to the BGS-RSCP Supplier shall be the surplus Margin then held by the Company. If the BGS-RSCP Supplier posted cash and notice is received by 1:00 p.m. New York time on a Business Day, the surplus Margin will be returned by the next following Business Day and if the BGS-RSCP Supplier posted cash and notice is received by the Company after 1:00 p.m. New York time on a Business Day, the surplus Margin shall be returned by the second Business Day following the date of notice, unless the BGS-RSCP Supplier agrees in writing to extend the period to return the surplus Margin. If the BGS-RSCP Supplier posted a letter of credit, the surplus Margin shall be returned on the next Business Day following the Business Day on which the amendment to the letter of credit is received from the issuing bank, unless the BGS-RSCP Supplier agrees in writing to extend the period to return the surplus Margin. The BGS-RSCP Supplier will not unreasonably deny a request for a one-day extension of such period. In the event that the Company fails to return the surplus Margin when due in accordance with this Article, then an Event of Default under Article 5 of this Agreement will be deemed to have occurred and the BGS-RSCP Suppliers will be entitled to the remedies set forth in Article 5 of this Agreement.

6.8 Grant of Security Interest/Remedies

To secure its obligations under this Agreement and to the extent that the BGS-RSCP Supplier posted Margin/collateral hereunder, the BGS-RSCP Supplier hereby grants to the Company a present and continuing security interest in, and lien on (and right of setoff against), and assignment of, all cash collateral and cash equivalent collateral and any and all proceeds resulting therefrom or the liquidation thereof, whether now or hereafter held by, on behalf of, or for the benefit of, the Company, and the BGS-RSCP Supplier and the Company agree to take such action as is reasonably required to perfect the secured Party's first priority security interest in, and lien on (and right of setoff against), such collateral and any and all proceeds resulting therefrom or from the liquidation thereof. Upon or any time after the occurrence or deemed occurrence and during the continuation of an Event of Default or an Early Termination Date, the Company may do any one or more of the following: (i) exercise any of the rights and remedies of the Company with respect to all collateral, including any such rights and remedies under law then in effect; (ii) exercise its rights of setoff against any and all property of the BGS-RSCP Supplier in the possession of the Company whether held in connection with this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply; (iii) draw on any outstanding letter of credit issued for its benefit; and (iv) liquidate all security held by or for the benefit of the Company free from any claim or right of any nature whatsoever of the BGS-RSCP Supplier, including any equity or right of purchase or redemption by the BGS-RSCP Supplier. The Company shall apply the proceeds of the collateral realized upon the exercise of such rights or remedies to reduce the BGS-RSCP Supplier's obligation under this Agreement or any other agreement(s) between the Company and the

BGS-RSCP Supplier for the provision of BGS Supply (the BGS-RSCP Supplier remaining liable for any amounts owing to the Company after such application), subject to the Company's obligation to return any surplus proceeds remaining after such obligations are satisfied in full.

All notices, demands or requests regarding credit requirements and credit related security or deposit transfers shall be in writing and shall be personally delivered or sent by overnight express mail, courier service or facsimile transmission (with the original transmitted by any of the other aforementioned delivery methods) addressed as follows:

If to a BGS-RSCP Supplier:

Notification information for each BGS-RSCP Supplier is set forth on Appendix A hereto.

If to the Company to:

Copy to:

or to such other person at such other address as a Party shall designate by like notice to the other Party. Notice received after the close of the Business Day shall be deemed received on the next Business Day; provided that notice by facsimile transmission shall be deemed to have been received by the recipient if the recipient confirms receipt telephonically or in writing.

6.9 Security Instruments

At each BGS-RSCP Supplier's choice, the following are deemed to be acceptable methods for posting security, if required:

(a) Cash; or

(b) A standby irrevocable letter of credit acceptable to the Company issued by a bank or other financial institution with a minimum "A" senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) from S&P or "A2" from Moody's (see standard format in Appendix C). The letter of credit shall state that it shall renew automatically for successive one-year or shorter periods, until terminated upon at least ninety (90) days prior written notice from the issuing financial institution. If the Company receives notice from the issuing financial institution that the letter of credit is being cancelled, the BGS-RSCP Supplier will be required to provide a substitute letter of credit from an alternative bank satisfying the minimum requirements. The receipt of the substitute letter of credit must be effective as of the cancellation date and delivered to the Company thirty (30) days before the cancellation date of the original letter of credit. If the BGS-RSCP Supplier fails to supply a substitute letter of credit as required, then the Company will have the right to draw on the existing letter of credit and to hold the amount as Margin.

If the credit rating of a bank or other financial institution from which a BGS-RSCP Supplier has obtained a letter of credit falls below the levels specified in Article 6 of this Agreement, the BGS-RSCP Supplier shall have two (2) Business Days following written notice by the Company to obtain a suitable letter of credit from another bank or other financial institution that meets those standards.

6.10 Maintenance of Creditworthiness

(a) Reporting of Changes.

Each BGS-RSCP Supplier shall promptly notify the Company of any change in its credit rating or financial condition or that of its Guarantor. The BGS-RSCP Supplier or Guarantor shall also furnish evidence of an acceptable credit rating or financial condition

upon the request of the Company.

(b) Change in Credit Standing.

The Company will re-evaluate the creditworthiness of a BGS-RSCP Supplier whenever it becomes aware of an adverse change, through the provision of notice by such BGS-RSCP Supplier or otherwise, in the BGS-RSCP Supplier's or Guarantor's credit standing. If the lowest credit rating (whether senior unsecured debt rating or corporate issuer rating) used to determine the BGS-RSCP Supplier's ICT or its credit limit adversely changes, the Company will require additional security from the BGS-RSCP Supplier in accordance with Sections 6.4 and 6.6 of this Agreement. The additional security must be in a form acceptable to the Company, as specified in Section 6.9 of this Agreement and must be posted as set forth in Section 6.7 of this Agreement.

6.11 Calling on Security

The Company may call upon the security posted by the BGS-RSCP Supplier if the BGS-RSCP Supplier fails to pay amounts due to the Company pursuant to this Agreement or any other agreement(s) between the Company and the BGS-RSCP Supplier for the provision of BGS Supply after all of the following events occur:

- (a) Written Notice of Default is provided to the BGS-RSCP Supplier; and
- (b) Any applicable cure period ends.

The foregoing notwithstanding, the security posted by the BGS-RSCP Supplier shall become due automatically without prior notice or right of cure in the case of any Event of Default arising under subsections (i), (ii), (iii), (iv), (v), (vi), (vii) and (viii) of Section 5.1 of this Agreement.

6.12 Interest on Cash Held by Company

The Company will pay simple interest calculated at the lower of the Interest Index or six (6) percent per annum on all cash held by the Company pursuant to this Agreement. Each Billing Month, the Company will prepare a statement of interest amounts due to the BGS-RSCP Supplier. The statement will be sent to the BGS-RSCP Supplier within three (3) Business Days after the end of the Billing Month via overnight mail or other expeditious means. The Company shall make interest payments on the first Business Day after the 5th day of each calendar month.

6.13 Confidentiality

Information supplied by a BGS-RSCP Supplier in connection with the creditworthiness process shall be deemed confidential and not subject to public disclosure, unless Applicable Legal Authorities require disclosure of the information. If information must be disclosed, then the confidentiality of the information shall be maintained consistent with the Applicable Legal Authority's rules and regulations pertaining to confidentiality. The BGS-RSCP Supplier will be given prompt notice of any request by a third party to obtain confidential information related to the BGS-RSCP Supplier's creditworthiness.

6.14 No Endorsement of BGS-RSCP Supplier

The Company's determination that a BGS-RSCP Supplier is creditworthy pursuant to the process set forth above, shall not be deemed to constitute an express or implied warranty or guarantee of any kind with respect to the financial or operational qualifications of the BGS-RSCP Supplier. The Company will treat all BGS-RSCP Suppliers in a non-discriminatory manner and shall provide no preference to any BGS-RSCP Supplier.

6.15 Multiple BGS Supply Agreements

It is the intention of the Company and the BGS Supplier that, in the event the BGS-RSCP Supplier is a party to other agreements with the Company for the provision of BGS Supply that existed prior to the effective date of this Agreement or is entered into after the effective date of this Agreement, the Company will calculate the Margin applicable to all such agreements as set forth herein. Each BGS-RSCP Supplier that is a party to such other agreements with the Company for the provision of BGS Supply hereby agrees that such other agreements are deemed amended by this Agreement for the purpose of calculating the Margin as described herein.

ARTICLE 7: PROCEDURES FOR ENERGY SCHEDULING, CAPACITY RESOURCE SUBMISSION AND TRANSMISSION PROCUREMENT

The Parties must adhere to any applicable operational requirements of PJM necessary to protect the integrity of the transmission system within the PJM Control Area and the transmission systems of interconnected control areas, and must satisfy any and all PJM, RFC and NERC criteria, when applicable. The BGS-RSCP Supplier also must adhere to any applicable operational requirements of the Company necessary to protect the integrity of the Company's local distribution system.

The BGS-RSCP Peak Load Share will be determined by the Company based on the zonal peak load contributions utilized in the PJM determination of the zonal obligations for Capacity, adjusted for contributions associated with Customers served by Third Party Suppliers, BGS-CIEP Customers and Wholesale Customers. Contributions for TPSs shall be calculated as set forth in the Company's TPS operating procedures found on the Company's website at www.____.com. The BGS-RSCP Peak Load Share of a BGS-RSCP Supplier is based on the BGS-RSCP Supplier Responsibility Share.

7.1 Load Obligations

The Company and the BGS-RSCP Supplier acknowledge that the BGS-RSCP Customers are within the Company's metered boundaries and that the BGS-RSCP Load must be divided into BGS-RSCP Supplier obligations by applying the BGS-RSCP Supplier Responsibility Share for each BGS-RSCP Supplier as a Load Serving Entity that must meet its PJM obligations. These load obligations include, but are not limited to, hourly Energy obligations, Capacity obligations, Ancillary Services obligations, and Firm Transmission Service obligations under the PJM Agreements.

7.2 Data Transmission

(a) Energy

The procedures for transmitting load obligation data for the BGS-RSCP Supplier's hourly Energy obligations shall be as set forth by PJM.

(b) Capacity

The procedures for transmitting the BGS-RSCP Supplier Responsibility Share data to be used by PJM to determine the BGS-RSCP Supplier's Capacity obligations shall be as set forth by PJM.

(c) Transmission

The procedures for transmitting the BGS-RSCP Supplier Responsibility Share data based upon which the BGS-RSCP Supplier will meet its obligations under the PJM OATT shall be as set forth by PJM.

(d) Ancillary Services

The procedures for transmitting data regarding the BGS-RSCP Supplier's Ancillary Services obligations shall be as set forth by PJM.

7.3 Energy Scheduling

The Company will not provide load-forecasting services. The BGS-RSCP Supplier shall schedule Energy resources to meet its obligations with PJM as provided for in the PJM Agreements, procedures, and manuals. The Company, through an e-schedule, shall provide PJM and the BGS-RSCP Supplier with the data regarding the BGS-RSCP Supplier Responsibility Share of the Energy obligations, as set forth by PJM. The Energy obligations for each BGS-RSCP Supplier will be determined based on its BGS-RSCP Supplier Responsibility Share of the BGS-RSCP Load. The total preliminary BGS-RSCP Energy obligation will be based on the total Energy loads for all of the BGS-RSCP Customers as calculated by the Company, including all losses and unaccounted for energy.

ARTICLE 8: THE ENERGY SETTLEMENT/RECONCILIATION PROCESS

8.1 Energy Settlement By PJM

The settlement process occurs at PJM to reflect the BGS-RSCP Supplier's actual Energy obligations in a supply/usage reconciliation process. The Energy obligations for each BGS-RSCP Supplier will be determined based on the BGS-RSCP Supplier Responsibility Share of the BGS-RSCP Load. The reconciled total BGS-RSCP Energy obligation will be based on the final total Energy loads for the Customers receiving BGS-RSCP service, including losses.

Any adjustments for billing and metering errors reported subsequent to the calculation of FMEA will be proportionally allocated by the Company to the BGS-RSCP Suppliers.

8.2 Energy Settlement by the Company

In the event that actual BGS-RSCP Customer consumption data is not available until after the PJM deadline for conducting the final settlement, the Company will conduct the settlement process with the BGS-RSCP Supplier. Should PJM impose penalties against the Company as a result of the BGS-RSCP Supplier's transactions or failure to meet PJM requirements, such penalties shall be passed through by the Company, to the BGS-RSCP Supplier as part of this settlement process. In addition, all other applicable charges from PJM, including any billing adjustments, will be appropriately allocated to the BGS-RSCP Supplier.

ARTICLE 9: BILLING AND PAYMENT

9.1 The Company Payment of Obligations to the BGS-RSCP Supplier

The Company shall pay all amounts due to the BGS-RSCP Supplier hereunder in accordance with the following provisions:

- (a) Each Billing Month, the Company will prepare a Statement of amounts due to the BGS-RSCP Supplier. A line item on this Statement will show amounts due equal to the Auction Price multiplied by the applicable Seasonal Billing Factor multiplied by PMEA for the Billing Month in question.
- (b) The Statement will be sent to the BGS-RSCP Supplier within eight (8) Business Days after the end of the Billing Month via overnight mail or other expeditious means.
- (c) The Company shall make payment on the first Business Day after the 19th day of each calendar month provided that the Company maintains a minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) of

at least "BBB-" from S&P, "Baa3" from Moody's or "BBB-" from Fitch (the "Required Rating"). If the Company is rated by only two rating agencies, and the ratings are split, the lower rating will be used. If Company is rated by three rating agencies, and the ratings are split, the lower of the two highest ratings will be used, and, in the event that the two highest ratings are common, such common rating will be used.

- (d) In the event that the Company's minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) falls below the Required Rating, and until the Company's minimum senior unsecured debt rating (or, if unavailable, corporate issuer rating discounted one notch) becomes equal or higher than the Required Rating, (i) the Company shall make an initial payment on the first business day after the 5th day of the calendar month for approximately 50% of the amount due to the BGS-RSCP Supplier for the previous calendar month (the "Initial Payment"), and (ii) the Company shall make a second payment on the first business day after the 19th day of the calendar month for any remaining amounts associated with the previous calendar month, which will include the difference between the Initial Payment and any amounts due equal to the Auction Price multiplied by the applicable Seasonal Billing Factor multiplied by PMEA for the Billing Month in question.
- (e) To the extent that the FMEA differs from the PMEA the Company will pay or charge the BGS-RSCP Supplier for the PMEA/FMEA Adjustment Amount within the PJM deadline for conducting the final settlement.
- (f) If each Party owes an amount to the other Party pursuant to this Agreement, including any related interest, payments or credits, the Parties may satisfy their respective obligations to each other by netting the aggregate amounts due to one Party against the aggregate amounts due to the other Party, with the Party, if any, owing

the greater aggregate amount paying the other Party the difference between the amounts owed.

- (g) Payments shall be subject to adjustment for any arithmetic errors, computation errors, meter reading errors, or other errors, provided that the errors become known within one (1) year of the termination of this Agreement.
- (h) The Company shall make payments of funds payable to the BGS-RSCP Supplier by electronic transfer to a bank designated by the BGS-RSCP Supplier.
- (i) If a good faith dispute arises between the Company and the BGS-RSCP Supplier regarding a Statement, the disputing Party shall be obligated to pay only the undisputed portion of the Statement, if any, and shall present the dispute in writing and submit supporting documentation to the non-disputing Party within one hundred twenty (120) calendar days from the date of the Statement in dispute. Statement disputes shall be addressed promptly, and in accordance with the dispute resolution procedures set forth in Article 11 of this Agreement. Upon resolution of a Statement dispute, any payments made to either Party will include simple interest on the payment at the lower of the Interest Index or six (6) percent per annum payable from the date that notice of a Statement dispute was received by the non-disputing Party.
- (j) If payment is made to the BGS-RSCP Supplier after the due date shown on the Statement, a late fee will be added to the unpaid balance until the entire Statement is paid. This late fee will be calculated at the prime rate commercial borrowers are charged by J.P. Morgan Chase.

9.2 Billing for BGS-RSCP Supplier's Obligations to Other Parties

The Company shall have no responsibility for billing between: the BGS-RSCP Supplier and PJM; the BGS-RSCP Supplier and any Energy or Capacity source; or the

BGS-RSCP Supplier and any other third party. The Company will be solely responsible for billing BGS-RSCP Customers for BGS-RSCP.

9.3 The BGS-RSCP Supplier Payment of Obligations to the Company

The BGS-RSCP Supplier shall pay all Charges it incurs hereunder in accordance with the following provisions:

- (a) Each Billing Month, the Company shall submit an invoice to the BGS-RSCP Supplier for all Charges owed by the BGS-RSCP Supplier under this Agreement. The BGS-RSCP Supplier shall make payment for Charges incurred on or before the due date shown on the invoice. The due date will be on the first Business Day after the 19th day of each calendar month. The invoice will be sent to the BGS-RSCP Supplier within eight (8) Business Days after the end of the Billing Month via overnight mail or other expeditious means.
- (b) Invoices shall be subject to adjustment for any arithmetic errors, computation errors, Meter Reading errors, or other errors, provided that the errors become known within one (1) year of the termination of this Agreement.
- (c) The BGS-RSCP Supplier shall make payments of funds payable to the Company by electronic transfer to a bank designated by the Company.
- (d) If a good faith dispute arises between the Company and the BGS-RSCP Supplier regarding an invoice, the disputing Party shall pay only the undisputed portion of the invoice, if any, and shall present the dispute in writing and submit supporting documentation to the non-disputing Party within one hundred twenty (120) calendar days from the due date of the invoice in dispute. Billing disputes shall be addressed promptly, and in accordance with the dispute resolution procedures set forth in Article 11 of this Agreement. Upon resolution of a billing dispute, any payments made to either Party will

include simple interest on the payment at the lower of the Interest Index or six (6) percent per annum payable from the date that notice of a bill dispute was received by the non-disputing Party.

(e) If payment is made to the Company after the due date shown on the invoice, a late fee will be added to the unpaid balance until the entire invoice is paid. This late fee will be calculated at the prime rate commercial borrowers are charged by J.P. Morgan Chase.

ARTICLE 10: SYSTEM OPERATION

10.1 Disconnection and Curtailment By the Company

The Company shall have the right, without incurring any liability to BGS-RSCP Suppliers, to disconnect (or otherwise curtail, interrupt or reduce deliveries from) the BGS-RSCP Suppliers or to disconnect (or otherwise curtail, interrupt or reduce deliveries to) any Customer whenever the Company determines in the exercise of its good faith discretion, or when the Company is directed by PJM, that such a disconnection, curtailment, interruption or reduction is necessary to facilitate construction, installation, maintenance, repair, replacement or inspection of any of the Company's facilities; or due to any other reason affecting the safe and reliable operation of the Company's or a Customer's facilities, including Emergencies, forced outages or potential overloading of the Company's transmission and/or distribution circuits, potential damage to any Customer's facilities or any risk of injury to persons.

10.2 Inadvertent Loss of Service to BGS-RSCP Customers

The Parties agree and acknowledge that service to BGS-RSCP Customers may be inadvertently lost due to storms, weather, accidents, breakage of equipment or other

events beyond the reasonable control of the Company affecting the transmission and distribution system of the Company. Neither Party will have any liability to the other Party for the occurrence of such events except for the Company's obligation to pursue steps for the resumption of the disrupted service as set forth in Section 10.3 below. In no event will an inadvertent loss of service affect a Party's obligation to make any payments then due or becoming due with respect to performance rendered prior to such inadvertent loss of service.

10.3 Good Faith Efforts

The Company shall use good faith efforts to: (a) minimize any curtailment, interruption or reduction to the extent practicable under the circumstances; (b) provide the BGS-RSCP Supplier with prior notification of any curtailment, interruption or reduction, to the extent practicable; and (c) resume service as promptly as practicable.

10.4 PJM Requirements

The BGS-RSCP Supplier acknowledges and agrees that, as a member of PJM, the Company is bound by all PJM operating instructions, policies and procedures as are currently set forth in the PJM Operating Manual, which are available through the Internet on the PJM Home Page (http://www.pjm.com), as may be revised from time to time, which are needed to maintain the integrity of the PJM system. The BGS-RSCP Supplier acknowledges and agrees that it will cooperate with the Company so that the Company will be in compliance with all PJM Emergency Operations Procedures, which include, but are not limited to, those procedures pertaining to minimum and maximum generation Emergencies, and measures requiring involuntary Customer participation, such as supply voltage reduction or full interruption of Customer load by either manual or automatic means.

10.5 Compliance With Governmental Directives

The BGS-RSCP Supplier also acknowledges and agrees that the Company may need to act in response to governmental or civil authority directives which may affect BGS-RSCP Customer load. The BGS-RSCP Supplier agrees to cooperate with the Company in order to comply with said directives.

ARTICLE 11: DISPUTE RESOLUTION

11.1 Informal Resolution of Disputes

The Company and the BGS-RSCP Supplier shall use good faith and reasonable commercial efforts to informally resolve all disputes arising out of the implementation of this Agreement. The BGS-RSCP Supplier's point of contact for all information, operations, and questions shall be the Company's Basic Generation Service Unit and the Auction Website. Any dispute between the Company and the BGS-RSCP Supplier under this Agreement may be referred to a designated senior representative of each of the Parties for resolution on an informal basis as promptly as practicable.

11.2 Recourse to Agencies or Courts of Competent Jurisdiction

Nothing in this Agreement shall restrict the rights of either Party to file a complaint with the FERC under relevant provisions of the Federal Power Act ("FPA"), with the BPU under relevant provisions of the Applicable Legal Authorities, with a New Jersey State court of competent jurisdiction, or with a federal court of competent jurisdiction situated in the State of New Jersey. The Parties' agreement hereunder is without prejudice to any Parties' right to contest the jurisdiction of the agency or court to which a complaint is brought.

To the extent that this Agreement is deemed to be subject to FERC jurisdiction,

absent the agreement of all parties to a proposed change, the standard of review for changes to any section of the Agreement specifying the rate(s) or other material economic terms and conditions agreed to by the Parties herein, whether proposed by a Party, a non-party, the BPU or FERC acting sua sponte, will be the "public interest" standard of review set forth in <u>United Gas Pipe Line Co. v. Mobile Gas Service Corp.</u>, 350 U.S. 332 (1956) and <u>Federal Power Commission v. Sierra Pacific Power Co.</u>, 350 U.S. 348 (1956).

ARTICLE 12: REGULATORY AUTHORIZATIONS AND JURISDICTION

12.1 Compliance With Applicable Legal Authorities

The Company and the BGS-RSCP Supplier are subject to, and shall comply with, all existing or future applicable federal, State and local laws, all existing or future duly-promulgated orders or other duly-authorized actions of PJM or of Applicable Legal Authorities.

12.2 FERC Jurisdictional Matters

The inclusion herein of descriptions of procedures or processes utilized by PJM or otherwise subject to the jurisdiction of FERC is intended solely for informational purposes. If anything stated herein is found by the FERC to conflict with or be inconsistent with any provision of the FPA, or any rule, regulation, order or determination of the FERC under the FPA or if any existing procedures or processes utilized by PJM are duly modified, the applicable FERC rule, regulation, order, determination or modification shall control. To the extent required under any provision of the FPA, or any rule, regulation, order or determination of the FERC under the FPA, the Company and/or the BGS-RSCP Supplier, if applicable, shall use reasonable

commercial efforts to secure, from time to time, all appropriate orders, approvals and determinations from the FERC necessary to support this Agreement.

ARTICLE 13: LIMITATION OF LIABILITY

13.1 Limitations on Liability

Except to the extent expressly set forth in this Agreement, each Party shall be liable to the other Party only for direct damages incurred as a result of such Party's failure to comply with this Agreement and no Party shall have any liability to the other Party for consequential, indirect, special or punitive damages, including lost profits or lost revenues, arising out of such Party's failure to comply with its obligations under this Agreement. Notwithstanding anything to the contrary in this Agreement, the BGS-RSCP Suppliers have no obligations or liability to other BGS-RSCP Suppliers.

This Agreement does not create any duty to any BGS-RSCP Supplier with respect to the administration of the Company's Third Party Supplier Agreements by the Company, including, without limitation, with respect to the credit and security provisions of the Company's Third Party Supplier Agreements and Article 11 thereof. Nor will the Company have any liability to any BGS-RSCP Supplier for any act or failure to act in connection with such administration. If any dispute arises among the BGS-RSCP Suppliers regarding the allocation of forfeited security (net of the Company's costs as described in Section 11.3 of the Company's Third Party Supplier Agreements) due to BGS-RSCP Suppliers pursuant to Article 11 of the Company's Third Party Supplier Agreements, then such forfeited security (net of the Company's costs as described in Section 11.3 of the Company's Third Party Supplier Agreements) will be deposited into an interest-bearing account pending final disposition of such dispute.

13.2 Risk of Loss

Solely for purposes of determining risk of loss and for determining the indemnity obligations under Article 14 of this Agreement, the Company shall be deemed to have custody and control of the electric Energy delivered by a BGS-RSCP Supplier upon receipt thereof into the Company's distribution system and until delivery thereof at the retail electric meter of the Customer; and each BGS-RSCP Supplier shall be deemed to have custody and control of the electric Energy at all times prior to receipt thereof by the Company. Each BGS-RSCP Supplier shall at all times be deemed to hold title to electric Energy until delivery at the retail electric meter of the Customer at which time title shall be deemed to pass to such Customer. The Party deemed to have custody and control of electric Energy shall, among the Parties to this Agreement, be responsible for all loss or damage to property or injury or death to persons arising in connection with such electric Energy while in its custody and control and shall indemnify the other Parties with respect to same as set forth in Article 14 of this Agreement.

ARTICLE 14: INDEMNIFICATION

14.1 Indemnification

(a) Should the Company become the defendant in, or obligor for, any third party claims and/or liabilities for losses, penalties, expenses, damage to property, injury to or death of any person including a Party's employees or any third parties, that were caused by or occur in connection with an act or omission of a BGS-RSCP Supplier with respect to an obligation arising under or in connection with this Agreement, or for which such BGS-RSCP Supplier has otherwise assumed liability under the terms of this Agreement, such BGS-RSCP Supplier shall defend (at the Company's option), indemnify

and hold harmless the Company, its shareholders, board members, directors, officers and employees, from and against any and all such third party claims and/or liabilities, except to the extent that a court of competent jurisdiction determines that the losses, penalties, expenses or damages were caused wholly or in part by the gross negligence or willful misconduct of the Company. The Company may, at its own expense, retain counsel and participate in the defense of any such suit or action.

(b) Should a BGS-RSCP Supplier (the "Indemnified Supplier") become the defendant in, or obligor for, any third party claims and/or liabilities for losses, penalties, expenses, damage to property, injury to or death of any person including a Party's employees or any third parties, that were caused by or occur in connection with an act or omission of the Company or another BGS-RSCP Supplier with respect to an obligation arising under or in connection with this Agreement, or for which the Company or such other BGS-RSCP Supplier has otherwise assumed liability under the terms of this Agreement, the Company or such BGS-RSCP Supplier shall defend (at the option of the Indemnified Supplier), indemnify and hold harmless the Indemnified Supplier, its shareholders, board members, directors, officers and employees, from and against any and all such third party claims and/or liabilities, except to the extent that a court of competent jurisdiction determines that the losses, penalties, expenses or damages were caused wholly or in part by the gross negligence or willful misconduct of the Indemnified The Indemnified Supplier may, at its own expense, retain counsel and participate in the defense of any such suit or action.

14.2 Survives Agreement

The obligation of a Party to defend, indemnify, and hold harmless another Party under this Article shall survive termination of this Agreement, and shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for either Party under any statutory scheme, including any Worker's Compensation Acts, Disability Benefit Acts or other Employee Benefit Acts.

ARTICLE 15: MISCELLANEOUS PROVISIONS

15.1 Notices

Unless otherwise stated herein, all notices, demands or requests required or permitted under this Agreement shall be in writing and shall be personally delivered or sent by overnight express mail, courier service or facsimile transmission (with the original transmitted by any of the other aforementioned delivery methods) addressed as follows:

If to a BGS-RSCP Supplier:

Notification information for each BGS-RSCP Supplier is set forth on Appendix A hereto.

If to the Company to:

Copy to:

or to such other person at such other address as a Party shall designate by like notice to the other Party. Notice received after the close of the Business Day shall be deemed received on the next Business Day; provided that notice by facsimile transmission shall be deemed to have been received by the recipient if the recipient confirms receipt telephonically or in writing.

15.2 No Prejudice of Rights

The failure of a Party to insist on any one or more instances upon strict

performance of any provisions of this Agreement, or to take advantage of any of its rights hereunder, shall not be construed as a waiver of any such provisions or the relinquishment of any such right or any other right hereunder, which shall remain in full force and effect. No term or condition of this Agreement shall be deemed to have been waived and no breach excused unless such waiver or consent to excuse is in writing and signed by the Party claimed to have waived or consented to excuse.

15.3 Assignment

Parties shall not assign any of their rights or obligations under this Agreement without obtaining (a) any necessary regulatory approval(s) and (b) the prior written consent of the non-assigning Party, which consent shall not be unreasonably withheld; provided that the Company agrees that it shall grant its consent to a proposed assignment by a BGS-RSCP Supplier if the proposed assignee meets all of the Company's creditworthiness requirements then in effect under Article 6 of this Agreement; and further provided that a BGS-RSCP Supplier wishing to assign its interests hereunder shall not be obligated to obtain the consent of any other BGS-RSCP Supplier. No assignment of this Agreement shall relieve the assigning Party of any of its obligations under this Agreement until such obligations have been assumed by the assignee and all necessary consents have been obtained. Any assignment in violation of this Section 15.3 shall be void; provided, however, the Company may assign any or all of its rights and obligations under this Agreement, without the BGS-RSCP Supplier's consent, to any entity succeeding to all or substantially all of the assets of the Company, if such assignee agrees, in writing, to be bound by all of the terms and conditions hereof and all necessary regulatory approvals are obtained. The BGS-RSCP Supplier may, with prior written notice to the Company but without obtaining the approval of the Company, assign the

accounts, revenues or proceeds under this Agreement to a third party. The Company agrees that, following receipt of such notice of the assignment of accounts, revenues or proceeds and such other documentation that the Company may reasonably request, the Company will pay amounts becoming due to the assigning BGS-RSCP Supplier under this Agreement directly to the designated assignee; provided, however, that nothing herein shall enlarge or expand the rights of such designated assignee beyond the rights granted to the BGS-RSCP Supplier and the right of such designated assignee to receive payments shall be subject to all defenses, offsets and claims of the Company arising under this Agreement. The Company further agrees that, in the event necessary regulatory approvals to effectuate an assignment have been sought in good faith but that action by the regulatory body is pending, the Company shall accept the performance of the proposed assignee as a Party to this Agreement, as co-obligor with the Party proposing to assign its interest, until such approvals are obtained; provided that, in the event the regulatory body declines to grant its approval (or, in the discretion of the Company, in the event the application seeking approval is still pending without action by the regulatory body after ninety (90) days), the request for approval of the assignment shall be deemed to have been rejected.

15.4 Governing Law and Venue

To the extent not subject to the jurisdiction of the FERC, questions including those concerning the formation, validity, interpretation, execution, amendment, termination and construction of this Agreement shall be governed by the laws of the State of New Jersey, without regard to principles of conflicts of law. Any lawsuit arising in connection with this Agreement shall be brought only in the State or federal courts of New Jersey.

15.5 Headings

The headings and subheadings contained in this Agreement are used solely for convenience and do not constitute a part of the Agreement between the Parties hereto, nor should they be used to aid in any manner in the construction of this Agreement.

15.6 Third Party Beneficiaries

This Agreement is intended solely for the benefit of the Parties hereto including Customers for which the Company is executing this Agreement as agent. Nothing in this Agreement shall be construed to create any duty, or standard of care with reference to, or any liability to, any person not a Party to this Agreement.

15.7 General Miscellaneous Provisions

- (a) This Agreement shall not be interpreted or construed to create an association, joint venture, or partnership between the Parties (or any of them), or to impose any partnership obligation or liability upon any Party. The obligations of the BGS-RSCP Suppliers are expressly agreed to be several and not joint. No Party shall have any right, power, or authority to enter into any agreement or undertaking for, or on behalf of, or to act as or be an agent or representative of, or to otherwise bind, any other Party.
- (b) Cancellation, expiration or Early Termination of this Agreement shall not relieve the Parties of obligations that by their nature survive such cancellation, expiration or termination, including warranties, remedies, promises of indemnity and confidentiality.
- (c) Should any provision of this Agreement be held invalid or unenforceable, such provision shall be invalid or unenforceable only to the extent of such invalidity or unenforceability without invalidating or rendering unenforceable any other provision

hereof unless it materially changes the agreement of the Parties.

(d) Each of the Parties acknowledges that it has read this Agreement, understands it, and agrees to be bound by its terms. This Agreement is intended by the Parties as a final expression of their agreement. The Parties further agree that this Agreement is the complete and exclusive statement of agreement and supersedes all proposals (oral or written), understandings, representations, conditions, warranties, covenants and all other communications between the Parties relating thereto.

15.8 Taxes

All present and future federal, state, municipal or other taxes imposed by any taxing authority by reason of the provision of BGS-RSCP Supply to BGS-RSCP Customers by a BGS-RSCP Supplier under this Agreement shall be the liability of the BGS-RSCP Supplier, except for New Jersey State Sales and Use Taxes, which will be the Company's responsibility to collect from BGS-RSCP Customers for remittance to the applicable taxing authority. Should a BGS-RSCP Supplier be required to remit any New Jersey State Sales and Use Taxes directly to the applicable taxing authority, other than taxes previously collected by the BGS-RSCP Supplier on behalf of the Company, the Company will defend and indemnify the BGS-RSCP Supplier for such Sales and Use Taxes and will pay to the BGS-RSCP Supplier all such tax amounts upon demand. Each BGS-RSCP Supplier shall pay all such taxes to the applicable taxing authority to the extent required or permitted by law. If any transaction is exempt from the payment of any such taxes, the affected BGS-RSCP Supplier will, if requested, provide the Company with valid tax exemption certificates. Should the Company be required to remit any such taxes directly to any applicable taxing authority, other than taxes previously collected by the Company directly from a BGS-RSCP Supplier, the BGS-RSCP Supplier will defend

and indemnify the Company and will pay to the Company all such tax amounts upon demand.

If new taxes are imposed on Energy, Capacity, Firm Transmission Service or Ancillary Services after the Effective Date of this Agreement, within forty-five (45) days of the final adoption of any such new taxes, the Company will notify the BGS-RSCP Suppliers that such new tax has been adopted, will seek approval from the Board to collect the new taxes from BGS-RSCP Customers, and will provide the BGS-RSCP Suppliers with a copy of the Company's petition seeking such approval from the Board. Upon receipt of Board approval of the collection of the new taxes from BGS-RSCP Customers, the BGS-RSCP Supplier will be excused from liability for payment of those new taxes.

15.9 Changes in Transmission Charges for Firm Transmission Service

- (a) If during the term of this Agreement, a filing is made with the FERC to increase or decrease the charges for Firm Transmission Service, including any charge or surcharge imposed on customers receiving Firm Transmission Service, or if the charges for Firm Transmission Service are adjusted pursuant to a FERC-authorized formula rate, then the following procedures shall apply:
- (i) within forty-five (45) days of the date upon which the rate change filing or formula rate informational filing is made with the FERC, the Company will notify the BGS-RSCP Suppliers that such rate filing has been made, and will seek approval from the Board to increase or decrease the charges to BGS-RSCP Customers by the amount of such rate adjustment for Firm Transmission Service; provided that, if the rate adjustment for the Firm Transmission Service is to the demand component of such rate, the rate adjustment sought by the Company shall be stated as a dollar per kilowatt-

hour charge or reduction based on a load factor for transmission service equal to the load factor for transmission service over the previous twelve (12) months; and further provided that the Company will provide the BGS-RSCP Suppliers with a copy of the Company's petition seeking such approvals from the Board;

- (ii) upon receipt of Board approval for the adjustment in the charges to BGS-RSCP Customers, the Company shall begin collecting the new charges (calculated with respect to the increase or decrease) from the BGS-RSCP Customers;
- (iii) in the event of a decrease in Firm Transmission Service Charges, the Auction Price shall be deemed to be decreased by the decrease to BGS-RSCP Customers approved by the BPU;
- (iv) in the event of an increase in Firm Transmission Service Charges the following provisions shall apply: (a) commencing with the date that the Company begins collecting the rate increase approved by the BPU from the BGS-RSCP Customers, the Company shall track that portion of the charges to BGS-RSCP Customers comprised by the increase and shall retain such tracked amounts for the benefit of the BGS-RSCP Suppliers; (b) upon approval by the FERC, in a Final FERC Order and not subject to refund, of a proposed increase in charges for Firm Transmission Service, or at such time as an annual adjustment to the charges for Firm Transmission Service becomes effective pursuant to the applicable FERC-authorized formula rate, the following shall occur: (A) the Auction Price shall be deemed to be increased by the increase to BGS-RSCP Customers approved by the BPU; and (B) the Company shall promptly pay each BGS-RSCP Supplier, in proportion to its BGS-RSCP Supplier Responsibility Share, the amounts tracked and retained for the benefit of BGS-RSCP Suppliers during the term of this Agreement; provided, however, that in the event only a proportion of the increased

charge for Firm Transmission Service is approved by FERC, in a Final FERC Order and not subject to refund, the Auction Price shall be increased only by that portion of the increase approved by the FERC and the BGS-RSCP Suppliers shall be paid only that portion of the tracked and retained amounts associated with the portion of the increase approved by the FERC; and further provided that in the event FERC approves less than the entirety of the increased charge or rejects the increase in its entirety, all amounts tracked and retained by the Company for the benefit of the BGS-RSCP Suppliers associated with the rejected increase for Firm Transmission Service (or the rejected portion of the increase for Firm Transmission Service), shall be credited by the Company against future rates paid by BGS-RSCP Customers and the BGS-RSCP Suppliers shall have no right in or to such amounts. The Company will pay simple interest on amounts tracked and retained hereunder, calculated at the lower of the Interest Index or six (6) percent per annum; and

- (v) Nothing herein shall prevent or restrict the participation of the Company or the BGS-RSCP Suppliers in any FERC or judicial proceedings associated with any proposed increase or decrease in charges for Firm Transmission Service or any FERC or judicial proceedings associated with any formula rate adjustment, in sponsoring, supporting, opposing, challenging or otherwise addressing such increases, decreases, or adjustments.
- (b) If, prior to the Effective Date of this Agreement, a filing is made with the FERC to increase or decrease the charges for Firm Transmission Services, including any charge or surcharge imposed on customers receiving Firm Transmission Services, and the Company seeks the approval of the BPU to increase or decrease the rates charged to BGS-RSCP Customers by the amount of such increase or decrease for Firm Transmission

Services, then the following procedures shall apply:

- (i) upon receipt of Board approval for the increase or decrease in the rates charged to BGS-RSCP Customers, the Company shall begin collecting the new rate (calculated with respect to the increase or decrease) from the BGS-RSCP Customers;
- (ii) in the event of a rate decrease, the Auction Price shall be deemed to be decreased by the rate decrease to BGS-RSCP Customers approved by the BPU;
- (iii) in the event of a rate increase the following provisions shall apply: (a) commencing with the later of (x) the date that the Company begins collecting the rate increase approved by the BPU from the BGS-RSCP Customers or (y) June 1, 2019 if the Company begins collecting the rate increase approved by the BPU from the BGS-RSCP Customers prior to June 1, 2019, the Company shall track that portion of the rates charged to BGS-RSCP Customers comprised by the rate increase and shall retain such tracked amounts for the benefit of the BGS-RSCP Suppliers; (b) upon approval by the FERC, in a Final FERC Order and not subject to refund, of the proposed increase in charges for Firm Transmission Service, the following shall occur: (A) the Auction Price shall be deemed to be increased by the increase in charges to BGS-RSCP Customers approved by the BPU; and (B) the Company shall promptly pay each BGS-RSCP Supplier, in proportion to its BGS-RSCP Supplier Responsibility Share, the amounts tracked and retained for the benefit of BGS-RSCP Suppliers during the term of this Agreement; provided, however, that in the event only a proportion of the proposed increase in charges for Firm Transmission Service is approved by FERC, in a Final FERC Order and not subject to refund, the Auction Price shall be increased only by that portion of the proposed increase approved by the FERC and the BGS-RSCP Suppliers shall be paid only that portion of the tracked and retained amounts associated with the

increase approved by the FERC; and further provided that in the event FERC approves less than the entirety of the proposed increase or rejects the proposed increase in its entirety, all amounts tracked and retained by the Company for the benefit of the BGS-RSCP Suppliers associated with the rejected increase in charges for Firm Transmission Service (or the rejected portion of the proposed increase in charges for Firm Transmission Service), shall be credited by the Company against future rates paid by BGS-RSCP Customers and the BGS-RSCP Suppliers shall have no right in or to such amounts. The Company will pay simple interest on amounts tracked and retained hereunder, calculated at the lower of the Interest Index or six (6) percent per annum; and

- (iv) Nothing herein shall prevent or restrict the participation of the Company or the BGS-RSCP Suppliers in any FERC or judicial proceedings associated with any proposed increase or decrease in charges for Firm Transmission Service, in sponsoring, supporting, opposing, challenging or otherwise addressing such proposed rate increases or decreases.
- (c) In the event that, at the beginning of the term of this Agreement, PJM Transmission Charges include charges approved by FERC subject to refund without a Final FERC Order and such charges are being collected from BGS-RSCP Customers pursuant to BPU approval, these charges will be treated as rate increases pursuant to subsection 15.9(a)(iv) of this Agreement.
- (d) Nothing herein to the contrary withstanding, in the event that the Company, in the exercise of its good faith judgment, cannot determine whether, or the extent to which, a filing at the FERC seeking a change in charges (including a filing seeking a change in rate design) for Firm Transmission Service, should be considered an increase or decrease in rates for the purposes of the preceding paragraphs, the Company

may seek a ruling requesting such a determination from the BPU within ten (10) Business Days of the filing at the FERC. The Company shall thereafter implement the rate increase or rate decrease for Firm Transmission Service, if any, through an appropriate adjustment in the Auction Price in accordance with the directives of the BPU in its order.

15.10 Use of the Word "Including"

The word "including", when following any general statement or term, is not to be construed as limiting the general statement or term to the specific items or matters set forth or to similar items or matters, but rather as permitting the general statement or term to refer to all other items or matters that could reasonably fall within its broadest possible scope.

15.11 Federal Acquisition Regulation

If any of the following clauses prescribed by the Federal Acquisition Regulation ("FAR"), 48 Code of Federal Regulations Chapter 1, should be deemed to apply to this Agreement, the BGS-RSCP Supplier shall comply with the requirements of such clause(s), and shall include the terms or substance of such clause(s) in its subcontracts, as and to the extent required by the FAR:

- 1) Clean Air and Water: §52.223-2;
- Contract Work Hours and Safety Standards Act-Overtime Compensation: §52.222-4;
- 3) Equal Opportunity: §52.222-26;
- 4) Affirmative Action for and Employment Reports on Special Disabled and Vietnam Era Veterans: §52.222-35 and §52.222-37;
- 5) Affirmative Action for Handicapped Workers: §52.222-36;

6) Utilization of Small Business Concerns and Small Disadvantaged Business Concerns and Small Business and Small Disadvantaged Business Subcontracting Plan: §52.219-8 and §52-219-9.

In case of a conflict between the provisions of the FAR and the balance of this Agreement, the requirements of the FAR shall prevail.

15.12 Binding Terms

This Agreement and the rates, terms and conditions herein shall remain in effect for the entire term hereof and each Party (including the Company acting on behalf of Customers) agrees not to seek any change to such rates, terms and conditions pursuant to the FPA, if the FPA is deemed to have jurisdiction over this Agreement, including on the grounds that they are not just and reasonable.

15.13 Amendment

This Agreement, including the appendices hereto, cannot be amended without the written agreement of all Parties and the approval of the Board prior to such amendment becoming effective.

15.14 Counterparts

This Agreement may be executed in counterparts, each of which will be considered an original, but all of which shall constitute one instrument.

IN WITNESS WHEREOF, the Parties hereto have caused this Agreement to be
executed by their duly authorized representatives as of the date first set forth above.
ATTEST:
By:

[SUPPLIER SIGNATURES APPEAR ON SUCCEEDING PAGES]

APPENDIX A TO BGS-RSCP SUPPLIER MASTER AGREEMENT,

DATED ______, 2019, BY AND BETWEEN ___ AND THE BGS-RSCP SUPPLIERS

BGS Supplier	Auction Price (¢/kWh)	BGS Supplier Responsibility Share
Address for Notice		
The address for any notice to 15.1 of the BGS-RSCP Suppli	provided pro	oursuant to Section 6.8 and the following:
For Credit Related Issues, Sec	tion 6.8:	
Name Address Telephone Fax E-Mail		
For Notices, Section 15.1:		
Name Address Telephone Fax E-Mail		

APPENDIX B TO BGS-RSCP SUPPLIER MASTER AGREEMENT,

DATED _____, 2019, BY AND BETWEEN _ AND THE BGS-RSCP SUPPLIERS

Seasonal Billing Factor

The Seasonal Billing Factor is as follows:

June through September:

October through May:

Schedule for ICRT

MONTH	36-MONTH BGS-RSCP
MONTH	(\$/tranche)
FEB. 2019	2,400,000
MAR. 2019	2,400,000
APR. 2019	2,400,000
MAY 2019	2,400,000
JUNE 2019	2,400,000
JULY 2019	2,400,000
AUG. 2019	2,400,000
SEPT. 2019	2,000,000
OCT. 2019	2,000,000
NOV. 2019	2,000,000
DEC. 2019	1,900,000
JAN. 2020	1,900,000
FEB. 2020	1,900,000
MAR. 2020	1,800,000
APR. 2020	1,800,000
MAY 2020	1,800,000
JUNE 2020	1,650,000
JULY 2020	1,650,000
AUG. 2020	1,650,000
SEPT. 2020	1,400,000
OCT. 2020	1,400,000
NOV. 2020	1,400,000
DEC. 2020	1,300,000
JAN. 2021	1,300,000
FEB. 2021	1,300,000
MAR. 2021	1,200,000
APR. 2021	1,200,000
MAY 2021	1,200,000
JUNE 2021	1,000,000
JULY 2021	1,000,000
AUG. 2021	1,000,000
SEPT. 2021	850,000
OCT. 2021	850,000
NOV. 2021	850,000
DEC. 2021	650,000
JAN. 2022	650,000
FEB. 2022	650,000
MAR, 2022	400,000
APR. 2022	400,000
MAY 2022	400,000

MtM Exposure Amount Calculation Information:

Table 1 contains the illustrative marks¹ for each month of the BGS-RSCP auction period. Monthly marks (example only, to be updated in January 2019) are provided for June 2019 through May 2022. For the months, two-month blocks,² or quarterly blocks² where broker³ quotes are available, broker quotes will be used for those months. For all the remaining months the EDCs will use a proprietary method that reflects forward market conditions.

The method described above will be employed from ______, 2019 through _______, 2019. From ______, 2019 through the day of the close of the BGS-RSCP Auction, only the months for which a monthly, a two-month block or a quarterly block quote is available will be updated. The mark for each Billing Month is the mark that was calculated on the date that the BGS-RSCP Auction closes and will not change over the life of the contract. After the close of the BGS-RSCP Auction Forward Market Prices will change. In addition, the on-peak and off-peak loads used to calculate the MtM Exposure Amount will be adjusted monthly to reflect the most current information.

Forward Market Prices for the months, two-month blocks or quarterly blocks where at least two broker quotes are available will be equal to the broker quotes. In case quotes for a component of a block and for the block are both available, the EDCs reserve the right not to use both the component of a block and the block if they are inconsistent with each other. However, when this inconsistency occurs the EDC must use either the component or the block. Forward Market Prices for the months, two-month blocks or quarterly blocks where broker quotes are unavailable will be equal to the last available broker quotes or in case they have not been quoted on the broker sheets since the BGS-RSCP Auction closed, they will be equal to the marks set at the close of the BGS-RSCP

¹ Illustrative marks represent PJM Western Hub on-peak prices.

For two-month and quarterly blocks for which the average for the block and a component of the block are both quoted, the component will be equal to its quoted price and the other months in the block will be constructed so that the weighted average (weighted by on-peak hours in each month) of the block equals the quote for the block; e.g., Q4 2019 = \$50 and Oct 2019 = \$40; therefore, Oct 2019 = \$40 and Nov-Dec 2019 = \$55 (\$50*(336+336+368)-\$40*336)/(336+368)=\$54.77. If only the block is quoted, that price will be used for all relevant months; e.g., Jan/Feb 2020 = \$35, then Jan 2020 = \$35 and Feb 2020 = \$35.

The term broker as used in this Agreement encompasses both independent brokers or exchanges active in the energy market.

Auction.

MtM Calculation Example

Parameters

On the closing day of the auction, the following parameters are set

- 1. The expected On-Peak Load per tranche for each EDC.
- 2. The expected Off-Peak Load per tranche for each EDC.
- 3. A table of monthly on-peak forward prices (to be used as the inception price "mark" for each month of the supply period).
- 4. A table of monthly historical on-and off-peak energy prices to determine the ratio of off-peak price to on-peak prices.

Indicative on-peak and off-peak loads per tranche for each EDC will be made available 14 days prior to the auction.

ALL ENERGY PRICES ARE BASED ON THE PJM WESTERN HUB.

EXAMPLE Table 1 - Initial Data **Broker Sheets (Quotes from Auction Closing Day)**

Broker	Bid/Offer	Jun-19	Jul-19	Aug-19	Sep-19	Oct-19	Nov-19
Broker 1	Bid						***************************************
Broker 1	Offer						
Broker 2	Bid						
Broker 2	Offer						
Broker 3	Bid						
Broker 3	Offer						
Average	Mid						

Broker	Bid/Offer	Dec-19	Jan-20	Feb-20	Mar-20	Apr-20	May-20
Broker 1	Bid						
Broker 1	Offer						
Broker 2	Bid						
Broker 2	Offer						
Broker 3	Bid						
Broker 3	Offer						
Average	Mid						

Table 2 - Example of Historical Energy Prices, and ratio of Off-Peak to On-Peak Broker Sheets (Quotes from Auction Closing Day)

Month	Historical Average Price	Historical Ratio of Off-Peak to On-Peak Price
January		
February		
March		
April		
May		
June		
July		
August		·
September		
October		
November		
December		

Table 3 - Data set on the Closing Day of the Auction Energy (MWh/tranche)

	On-Peak Volume ¹	Off-Peak Volume²	On-Peak Price ³	Off-Peak Price ⁴
Jun-19				
Jul-19				
Aug-19				
Sep-19				
Oct-19				
Nov-19				
Dec-19				
Jan-20				
Feb-20				
Mar-20				
Apr-20				
May-20				
Jun-20				
Jul-20				
Aug-20				
Sep-20				
Oct-20				
Nov-20				
Dec-20				
Jan-21				
Feb-21				
Mar-21				
Apr-21				
May-21				
Jun-21				
Jul-21				
Aug-21				
Sep-21				
Oct-21				
Nov-21				
Dec-21				
Jan-22				
Feb-22				
Mar-22				
Apr-22				
May-22				

On-peak and off-peak volumes will be adjusted monthly.
 On-peak and off-peak volumes will be adjusted monthly.
 Forward price (if available). If not available, adjusted historical average price will be used.
 On-peak price multiplied by the historical off-peak/on-peak ratio.

EXAMPLE

Table 4 - Data on Day 1 of the Supply Period

Broker sheets on day 1 of the Supply Period

Broker	Bid/Offer	Jun-19	Jul-19	Aug-19	Sep-19	Oct-19	Nov-19
Broker 1	Bid						
Broker 1	Offer	•					
Broker 2	Bid						
Broker 2	Offer						
Broker 3	Bid						
Broker 3	Offer						
Average	Mid						

Broker	Bid/Offer	Dec-19	Jan-20	Feb-20	Mar-20	Apr-20	May-20
Broker 1	Bid		194		•		
Broker 1	Offer						
Broker 2	Bid						
Broker 2	Offer						
Broker 3	Bid						
Broker 3	Offer						
Average	Mid						

Note: For a monthly forward price from a broker to be included, both bid and offer must be available.

EXAMPLE

Table 5 - Data set on the Closing Day of the Auction Energy (MWh/tranche)

	On-Peak Load per Tranche (MWh)	Off-Peak Load per Tranche (MWh)	Inception On-Peak Price	Current Day On- Peak Price	Change in On-Peak Price	Change in Off-Peak Price ¹	MtM
Jun-19							
Jul-19							
Aug-19							
Sep-19							
Oct-19							
Nov-19							
Dec-19							
Jan-20							
Feb-20					`		
Mar-20							
Apr-20							
May-20							
Jun-20							
Jul-20							
Aug-20							
Sep-20							
Oct-20							
Nov-20					•		
Dec-20							
Jan-21							
Feb-21							
Mar-21							
Apr-21							
May-21							
Jun-21							
Jul-21							
Aug-21							
Sep-21							
Oct-21							•
Nov-21							
Dec-21	•						
Jan-22							
Feb-22			*				
Mar-22		•					
Apr-22							
May-22							
						Total	

Note: Inception on-peak price is equal to on-peak price set on the closing day of the auction.

¹ Change in on-peak price multiplied by ratio of off-peak price to on-peak price.

APPENDIX C TO BGS-RSCP SUPPLIER MASTER AGREEMENT,

DATED _____, 2019, BY AND BETWEEN __ AND THE BGS-RSCP SUPPLIERS

Sample BGS-RSCP Letter of Credit

	(Date)
Le	tter of Credit No
At Se: (" <u>I</u>	: [One of the following: lantic City Electric Company or Jersey Central Power & Light Company or Public rvice Electric and Gas Company or Rockland Electric Company] Beneficiary") tention: [name, title, address]
۱.	We hereby establish in your favor this irrevocable transferable Letter of Credit (this
	"Letter of Credit") for the account of (the "Applicant"),
	in the amount of \$, effective immediately and available to you at
	sight upon demand at our counters at (Location). This Letter
	of Credit shall expire 364 days from date of issuance, unless terminated earlier or
	otherwise extended in accordance with the provisions hereof. It is a condition of this
	Letter of Credit that it shall be deemed automatically extended, without amendment,
	for additional period(s) of 364 days from the expiry date hereof, or any future
	expiration date unless at least ninety (90) days before its current expiration date, we
	send notice to you in writing by registered mail or overnight courier at the address
	above, ATTN .: [Title] and to the Applicant, that we do not intend to extend this
	Letter of Credit.
2.	This Letter of Credit is issued at the request of the Applicant, and we hereby
	irrevocably authorize you to draw on us, in accordance with the terms and conditions
	hereof up to the maximum amount of this I atter of Cradit subject to reduction as

	provided in Paragraph 11 hereof. This Letter of Credit may be drawn upon an Event
	of Default under the BGS-RSCP Supplier Master Agreement(s) between the
	Applicant and you, dated and the BGS-CIEP Supplier Master
	Agreement(s) between the Applicant and you, dated
3.	A partial or full drawing hereunder may be made by you on any Business Day on or
	prior to the expiration of this Letter of Credit by delivering, by no later than 11:00
	A.M. (New York, NY time ¹) on such Business Day to
	(Bank), (address), (i) a notice executed by you in
	the form of Annex 1 hereto, appropriately completed and duly signed by your
	Authorized Officer and (ii) your draft in the form of Annex 2 hereto, appropriately
	completed and duly signed by your Authorized Officer. Authorized Officer shall
	mean President, Treasurer, any Vice President or any Assistant Treasurer.

4. We hereby agree to honor a drawing hereunder made in compliance with the terms and provisions of this Letter of Credit by transferring in immediately available funds the amount specified in the draft delivered to us in connection with such drawing to such account at such bank in the United States as you may specify in your draft delivered to us pursuant to Paragraph 3 hereof, by 3:00 P.M. (New York, NY time) on the date of such drawing, if delivery of this requisite document is made prior to 11:00 AM (New York, NY time) on a Business Day pursuant to Paragraph 3 herein above, but at the opening of business on the first Business Day next succeeding the date of such drawing if delivery of the requisite document is made on or after 11:00

If the issuer of the Letter of Credit is located in an area that is not in the Eastern time zone, this time and all other times in this Letter of Credit, and the definition of a Business Day should be adjusted accordingly.

AM (New York, NY time) on any Business Day pursuant to Paragraph 3 herein above.

- 5. If a demand for payment made by you hereunder does not, in any instance, conform to the terms and conditions of this Letter of Credit, we shall give you prompt notice (not exceeding three (3) Business Days following the date of receipt of the documents) that the demand for payment was not effected in accordance with the terms and conditions of this Letter of Credit, stating the reasons that the demand for payment was not effected in accordance with such terms and conditions, and that we will upon your instructions hold any documents at your disposal or return the same to you. Upon being notified that the demand for payment was not effected in conformity with this Letter of Credit, you may attempt to correct any such nonconforming demand for payment to the extent that you are entitled to do so, provided, however, in such event a conforming demand for payment must be timely made in accordance with the terms of this Letter of Credit.
- 6. This Letter of Credit shall automatically terminate and be delivered to us for cancellation on the earliest of (i) the making by you of the drawings in an amount equal to the maximum amount available to be made hereunder, (ii) the date we receive from you a Certificate of Expiration in the form of Annex 3 hereto, or (iii) the above-stated expiration date hereof.

7. As used herein:

"Business Day" shall mean any day on which commercial banks are not authorized or required to close in New York, New York and any day on which payments can be effected on the Fedwire system.

- 8. This Letter of Credit is transferable one or more times, but in each instance only to a single transferee for the full amount available to be drawn under the Letter of Credit at the time of such transfer. Such transfer shall be effected by us upon your submission of this Letter of Credit including all amendments, if any, accompanied by the Letter of Full Transfer substantially in the form of Annex 4. This Letter of Credit may not be transferred to any person with which the United States are prohibited from doing business under the United States Foreign Assets Control Regulations or other applicable United States laws and regulations. Except for such transfer, this Letter of Credit may not otherwise be amended or modified without consent from us, you and the Applicant, and except as otherwise expressly stated herein, is subject to the Uniform Customs and Practice for Documentary Credits 2007 Revision, ICC Publication No. 600, or any successor publication thereto (the "UCP"). Any and all transfer fees, expenses and costs shall be borne by the Applicant. This Letter of Credit shall, as to matters not governed by the UCP, be governed and construed in accordance with New York law, without regard to principles of conflicts of law.
- 9. This Letter of Credit sets forth in full our undertaking, and such undertaking shall not in any way be modified, amended, changed, amplified or limited by reference to any document, instrument or agreement referred to herein, except for Annexes 1 through 4 hereto and the notices referred to herein; and any such reference shall not be deemed to incorporate herein by reference any document, instrument or agreement except as set forth above.

10.	We		(" <u>Banl</u>	<u>ς</u> ") cε	ertif	ry that	as	of _					
	(date) our	senior	unsecured	debt	is	rated	"A"	or	better	bv	S&P	Global	Ratings

("<u>S&P</u>") if rated by S&P or "A2" or better by Moody's Investors Service Inc. ("<u>Moody's</u>") if rated by Moody's. We hereby certify that our senior unsecured debt is rated by at least one of S&P and Moody's.

11. The amount which may be drawn by you under this Letter of Credit shall be automatically reduced by the amount of any drawings paid through us referencing this Letter of Credit No. _____. Partial drawings are permitted hereunder.

12. Faxed document(s) are acceptable. Presentation by fax must be made to fax number confirmed by telephone to ______.

- 13. In the event of act of God, riot, civil commotion, insurrection, war, terrorism or by any strikes or lock outs, or any cause beyond our control, that interrupts our business, and causes the place for presentation of this Letter of Credit to be closed for business on the last day of presentation, the expiration date of this Letter of Credit shall be automatically extended without amendment to a date thirty (30) calendar days after the place for presentation reopens for business.
- 14. This original Letter of Credit has been sent to [name, title, address]
 ________, the Beneficiary above (as per Applicant's instructions). The aggregate amount paid to the Beneficiary during the validity of this Letter of Credit will not exceed the amount of this Letter of Credit. Any demands or communications in the form of the attached Annexes or other communications directed to us under this Letter of Credit must be signed by an Authorized Officer of the Beneficiary. Acceptance or rejection of any amendments to this Letter of Credit must be signed by an Authorized Officer of the Beneficiary.

Very truly yours,	
(Bank)	
By:	
Name:	
Title:	
Ву:	
Name:	
Title.	

DRAWING UNDER LETTER OF CREDIT NO. , 20 To: (Bank) (Address) Attention: Standby Letter of Credit Unit Ladies and Gentlemen: The undersigned is making a drawing under the above-referenced Letter of Credit in the amount specified below and hereby certifies to you as follows: Capitalized terms used herein that are defined herein shall have the meanings ascribed thereto in the Letter of Credit. Pursuant to Paragraph 2 of the Letter of Credit No. __, 20__, the undersigned is entitled to make a drawing under the Letter dated of Credit in the amount of \$_____, inasmuch as there is an Event of Default under any BGS Supplier Master Agreement between the Applicant and us. 3. The undersigned acknowledges that, upon your honoring the drawing herein requested, the amount of the Letter of Credit available for drawing shall be automatically decreased by an amount equal to this drawing. Very truly yours, (Beneficiary) By ____ Name: Title: Date: (Applicant Name)

Annex 1 to Letter of Credit

Annex 2 to Letter of Credit DRAWING UNDER LETTER OF CREDIT NO. _____ , 20___ ON [Business Day immediately succeeding date of presentation] PAY TO: (Beneficiary) Attn: For credit to the account of . FOR VALUE RECEIVED AND CHARGE TO ACCOUNT OF LETTER OF CREDIT (Bank) (Address) (Beneficiary) By _____ Name: Title:

Annex 3 to Letter of Credit	
CERTIFICATE OF EXPIRATION OF LETTER OF CREDIT NO.	
, 20	
To: (Bank) (Address)	
Attention: Standby Letter of Cre	edit Unit
Ladies and Gentlemen:	
	you that the above referenced Letter of Credit Attached hereto is said Letter of Credit, marked
	(Beneficiary)
	Ву:
	Name:
	Title:
	Date:
cc:(Applicant Name)

LETTER OF FULL TRANSFER , 20 Bank Bank Address To Whom It May Concern: Re: Credit____ Issued by _____ Advice No For value received, the undersigned Beneficiary hereby irrevocably transfers to: (Name of Transferee) (Address) all rights of the undersigned Beneficiary to draw under the above Letter of Credit in its entirety. By this transfer, all rights of the undersigned Beneficiary in such Letter of Credit are transferred to the transferee and the transferee shall have the sole rights as beneficiary thereof, including sole rights relating to any amendments whether increases or extensions or other amendments and whether now existing or hereafter made. All amendments are to be advised direct to the transferee without necessity of any consent of or notice to the undersigned Beneficiary. The original of such Letter of Credit is returned herewith, and we ask you to endorse the transfer on the reverse thereof, and forward it direct to the transferee with your customary notice of transfer. Enclosed is remittance in the amount of \$_____ in payment of your transfer commission and in addition we agree to pay to you on demand any expenses that may be incurred by you in connection with this transfer. Transfer Commission Charges

Annex 4 to Letter of Credit

SIGNATURE AUTHENTICATED The signatory/ies of this concern is/are authorized to withdraw corporate funds.	Yours very truly,
(BANK)	Signature of Beneficiary
(Authorized Signature)	_
SIGNATURE AUTHENTICATED The signatory/ies of this concern is/are authorized to withdraw corporate funds.	
(BANK)	Signature of Transferee
(Authorized Signature)	_

APPENDIX D TO BGS-RSCP SUPPLIER MASTER AGREEMENT,

DATED _____, 2019, BY AND BETWEEN _ AND THE BGS-RSCP SUPPLIERS

Guaranty

GUARANTY (this "Guaranty"), dated as of, 2019, made by (the
"Guarantor"), a corporation organized and existing under the laws of
in favor of (the "Guaranteed Party"), a corporation organized
and existing under the laws of the State of New Jersey.
Terms not defined herein take on the meaning given to them in the BGS-RSCP Supplier
Master Agreement(s) dated and/or the BGS-CIEP Supplier Master Agreement(s)
dated (the "Agreements"). Guarantor enters into this Guaranty in consideration
of, and as an inducement for Guaranteed Party having entered into or entering into the
"Agreements" with [Name], a
[State] corporation (the "BGS Supplier"), which may involve the extension of credit by
the Guaranteed Party. Guarantor, subject to the terms and conditions hereof, hereby
unconditionally and absolutely guarantees to the Guaranteed Party the full and prompt
payment when due, subject to an applicable grace period and upon demand in writing
from the Guaranteed Party to the Guarantor's attention at the address for Guarantor set
forth in Section 11 hereof of any and all amounts payable by the BGS Supplier to the
Guaranteed Party arising out of the Agreement(s), and,

- 1. The Guarantor, as primary obligor and not merely as surety, hereby irrevocably and unconditionally guarantees the full and prompt payment when due (whether by acceleration or otherwise) of the principal and interest on any sums due and payable by the BGS Supplier as a result of an Event of Default under the Agreement(s) (including, without limitation, indemnities, damages, fees and interest thereon, pursuant to the terms of the Agreement(s)). Notwithstanding anything to the contrary herein, the maximum aggregate liability of the Guarantor under this Guaranty shall Option 1 [in no event exceed ________.] Option 2 [in no event exceed the lesser of [the credit limit amount] or the sum of the Total Exposures Amounts under the Agreement(s).] All such principal, interest, obligations and liabilities, collectively, are the "Guaranteed Obligations". This Guaranty is a guarantee of payment and not of collection.
- 2. The Guarantor hereby waives diligence, acceleration, notice of acceptance of this Guaranty and notice of any liability to which it may apply, and waives presentment and all demands whatsoever except as noted herein, notice of protest, notice of dishonor or nonpayment of any such liability, suit or taking of other action by any Guaranteed Party against, and any other notice to, any party liable thereon (including the Guarantor or any other guarantor), filing of claims with a court in the event of the

- insolvency or bankruptcy of the BGS Supplier, and any right to require a proceeding first against the BGS Supplier.
- 3. The Guaranteed Party may, at any time and from time to time, without notice to or consent of the Guarantor, without incurring responsibility to the Guarantor and without impairing or releasing the obligations of the Guarantor hereunder, upon or without any terms or conditions: (i) take or refrain from taking any and all actions with respect to the Guaranteed Obligations, any Document or any person (including the BGS Supplier) that the Guaranteed Party determines in its sole discretion to be necessary or appropriate; (ii) take or refrain from taking any action of any kind in respect of any security for any Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party; or (iii) compromise or subordinate any Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party including any security for such Guaranteed Obligation(s) or liability of the BGS Supplier to the Guaranteed Party.
- 4. Subject to the terms and conditions hereof, the obligations of the Guarantor under this Guaranty are absolute and unconditional and, without limiting the generality of the foregoing, shall not be released, discharged or otherwise affected by: extension, renewal, settlement, compromise, waiver, consent, discharge or release by the BGS Supplier concerning any provision of the Agreement(s) in respect of any Guaranteed Obligations of the BGS Supplier; (ii) the rendering of any judgment against the BGS Supplier or any action to enforce the same; (iii) the existence, or extent of, any release, exchange, surrender, non-perfection or invalidity of any direct or indirect security for any of the Guaranteed Obligations; (iv) any modification, amendment, waiver, extension of or supplement to any of the Agreement(s) or the Guaranteed Obligations agreed to from time to time by the BGS Supplier and the Guaranteed Party; (v) any change in the corporate existence (including its constitution, laws, rules, regulations or powers), structure or ownership of the BGS Supplier or the Guarantor, or any insolvency, bankruptcy, reorganization or other similar proceedings affecting the BGS Supplier or its assets, the Guarantor or any other guarantor of any of the Guaranteed Obligations; (vi) the existence of any claim, set-off or other rights which the Guarantor may have at any time against the BGS Supplier, the Guaranteed Party or any other corporation or person, whether in connection herewith or in connection with any unrelated transaction; provided that nothing herein shall prevent the assertion of any such claim by separate suit or compulsory counterclaim; (vii) the invalidity, irregularity or unenforceability in whole or in part of the Agreement(s) or any Guaranteed Obligations or any instrument evidencing any Guaranteed Obligations or the absence of any action to enforce the same, or any provision of applicable law or regulation purporting to prohibit payment by the BGS Supplier of amounts to be paid by it under the Agreement(s) or any of the Guaranteed Obligations; and (viii) except for a failure to comply with any applicable statute of limitations, any other act or omission to act or delay of any kind of the BGS Supplier, any other guarantor, the Guaranteed Party or any other corporation or person or any other event, occurrence or circumstance

- whatsoever which might, but for the provisions of this paragraph, constitute a legal or equitable discharge of the Guarantor's obligations hereunder.
- 5. The Guarantor hereby irrevocably waives (a) any right of reimbursement or contribution, and (b) any right of salvage against the BGS Supplier of any collateral security or guaranty or right of offset held by the Guaranteed Party.
- 6. The Guarantor will not exercise any rights, which it may acquire by way of subrogation until all Guaranteed Obligations to the Guaranteed Party pursuant to the Agreement(s) have been paid in full.
- 7. Subject to the terms and conditions hereof, this Guaranty is a continuing one and all liabilities to which it applies or may apply under the terms hereof shall be conclusively presumed to have been created in reliance hereon. Except for a failure to comply with any applicable statute of limitations, no failure or delay on the part of the Guaranteed Party in exercising any right, power or privilege hereunder, and no course of dealing between the Guarantor and the Guaranteed Party, shall operate as a waiver thereof; nor shall any single or partial exercise of any right, power or privilege hereunder preclude any other or further exercise thereof or the exercise of any other right, power or privilege. The rights, powers and remedies herein expressly provided are cumulative and not exclusive of any rights, powers or remedies, which the Guaranteed Party would otherwise have. No notice to or demand on the Guarantor in any case shall entitle the Guarantor to any other or further notice of demand in similar or other circumstances or constitute a waiver of the rights of the Guaranteed Party to any other or further action in any circumstances without notice or demand.
- 8. This Guaranty shall be binding upon the Guarantor and upon its successors and assigns and shall inure to the benefit of and be enforceable by the Guaranteed Party and its successors and assigns; provided, however, that the Guarantor may not assign or transfer any of its rights or obligations hereunder without the prior written consent of the Guaranteed Party. The assignment rights of the Guaranteed Party will be in accordance with the terms of the underlying Agreement(s).
- Neither this Guaranty nor any provision hereof may be changed, waived, discharged
 or terminated except upon written agreement of the Guaranteed Party and the
 Guarantor.
- 10. The Guarantor agrees that its liability as guarantor shall continue and remain in full force and effect in the event that all or any part of any payment made hereunder or any obligation or liability guaranteed hereunder is recovered (as a fraudulent conveyance, preference or otherwise) rescinded or must otherwise be reinstated or returned due to bankruptcy or insolvency laws or otherwise.
- 11. All notices and other communications hereunder shall be made at the addresses by hand delivery, by the next day delivery service effective upon receipt or by certified mail return receipt requested (effective upon scheduled weekday delivery day) or

facsimile (effective upon receipt of evidence, including facsimile evidence, that facsimile was received)

If to the Guarantor: [To be completed]

If to the Guaranteed Party: [To be completed]

- 12. If claim is ever made upon the Guaranteed Party for repayment or recovery of any amount or amounts received in payment or on account of any of the Guaranteed Obligations and the Guaranteed Party repays all or part of such amount by reason of (a) any judgment, decree or order of any court or administrative body having jurisdiction over such payee or any of its property, or (b) any settlement or compromise of any such claim effected by such payee with any such claimant (including the Guarantor), then and in such event the Guarantor agrees that any such judgment, decree, order, settlement or compromise shall be binding upon it, notwithstanding any revocation hereof or the cancellation of the Agreement(s) or other instrument evidencing any liability of the Guarantor, and the Guarantor shall be and remain liable to the Guaranteed Party hereunder for the amount so repaid or recovered to the same extent as if such amount had never originally been received by any such payee.
- 13. The Guarantor hereby certifies that it satisfies the Minimum Rating as defined in the Agreement(s).
- 14. This Guaranty shall remain in full force and effect until all Guaranteed Obligations have been fully and finally performed, at which point it will expire. The Guarantor may terminate this Guaranty upon thirty (30) days prior written notice to the Guaranteed Party which termination shall be effective only upon receipt by the Guaranteed Party of alternative means of security or credit support, as specified in the Agreement(s) and in a form reasonably acceptable to the Guaranteed Party. Upon the effectiveness of any such expiration or termination, the Guarantor shall have no further liability under this Guaranty, except with respect to the Guaranteed Obligations entered into prior to the time the expiration or termination is effective, which Guaranteed Obligations shall remain guaranteed pursuant to the terms of this Guaranty until finally and fully performed.
- 15. The Guarantor represents and warrants that: (i) it is duly organized and validly existing under the laws of the jurisdiction in which it was organized and has the power and authority to execute, deliver, and perform this Guaranty; (ii) no authorization, approval, consent or order of, or registration or filing with, any court or other governmental body having jurisdiction over the Guarantor is required on the part of the Guarantor for the execution, delivery and performance of this Guaranty except for those already made or obtained; (iii) this Guaranty constitutes a valid and legally binding agreement of the Guarantor, and is enforceable against the Guarantor,

except as the enforceability of this Guaranty may be limited by the effect of any applicable bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditor's rights generally and by general principles of equity; and (iv) the execution, delivery and performance of this Guaranty by the Guarantor have been and remain duly authorized by all necessary corporate or comparable action and do not contravene any provision of its ______ [insert appropriate corporate organizational document, such as Declaration of Trust, Limited Liability Agreement, Articles of Incorporation or by-laws] or any law, regulation or contractual restriction binding on it or its assets.

- 16. This Guaranty and the rights and obligations of the BGS Supplier and the Guarantor hereunder shall be construed in accordance with and governed by the laws of the State of New Jersey. The Guarantor and Guaranteed Party jointly and severally agree to the exclusive jurisdiction of State and federal courts located in the State of New Jersey over any disputes arising or relating to this Guaranty and waive any objections to venue or inconvenient forum. The Guarantor and Guaranteed Party each hereby irrevocably waive any and all rights to trial by jury with respect to any legal proceeding arising out of or relating to this Guaranty.
- 17. This writing is the complete and exclusive statement of the terms of this Guaranty and supersedes all prior oral or written representations, understandings, and agreements between the Guaranteed Party and the Guarantor with respect to subject matter hereof. The Guaranteed Party and the Guarantor agree that there are no conditions to the full effectiveness of this Guaranty.
- 18. Every provision of this Guaranty is intended to be severable. If any term or provision hereof is declared to be illegal or invalid for any reason whatsoever by a court of competent jurisdiction, such illegality or invalidity shall not affect the balance of the terms and provisions hereof, which terms and provisions shall remain binding and enforceable. This Guaranty may be executed in any number of counterparts, each of which shall be an original, but all of which together shall constitute one instrument.
- 19. No Trustee or shareholder of Guarantor shall be held to any liability whatsoever for any obligation under this Guaranty, and such Guaranty shall not be enforceable against any such Trustee in their or his or her individual capacities or capacity. This Guaranty shall be enforceable against the Trustees of Guarantor only as such, and every person, firm, association, trust or corporation having any claim or demand arising under this Guaranty and relating to Guarantor, its shareholders or Trustee shall look solely to the trust estate of Guarantor for the payment or satisfaction thereof.

IN WITNESS WHEI delivered as of the da of any of the Agreem	ite first above writte		
[GUARANTOR]			
By:			
Accepted and Agreed	to:		
(EDC)			
By:			